

This is a repository copy of *White paper on light sterile neutrino searches and related phenomenology*.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/219186/</u>

Version: Published Version

Article:

Acero, M.A. orcid.org/0000-0002-0835-0641, Argüelles, C.A. orcid.org/0000-0003-4186-4182, Hostert, M. orcid.org/0000-0002-9584-8877 et al. (181 more authors) (2024) White paper on light sterile neutrino searches and related phenomenology. Journal of Physics G: Nuclear and Particle Physics, 51. 120501. ISSN 0954-3899

https://doi.org/10.1088/1361-6471/ad307f

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/

OPEN ACCESS

White paper on light sterile neutrino searches and related phenomenology

To cite this article: M A Acero et al 2024 J. Phys. G: Nucl. Part. Phys. 51 120501

View the article online for updates and enhancements.



You may also like

- <u>Towards a Standard Model with six</u> fermion generations and a new breaking scale? Adrian Palcu
- <u>Particle physics using reactor antineutrinos</u> O A Akindele, N S Bowden, C Roca et al.
- <u>Non-extensive effects on the QCD</u> equation of state and fluctuations of conserved charges within Polyakov quark meson model Abdel Magied Diab

IOP Publishing

J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501 (214pp)

https://doi.org/10.1088/1361-6471/ad307f

Major Report

White paper on light sterile neutrino searches and related phenomenology

M A Acero^{1,119}[®]. C A Argüelles^{2,119}[®]. M Hostert^{3,4,5,119}[®]. D Kalra^{6,119}, G Karagiorgi^{6,119}, K J Kelly^{7,119}, B R Littlejohn^{8,119}, P Machado^{9,119}, W Pettus^{10,119}, M Toups^{9,119}, M Ross-Lonergan^{6,119}, A Sousa^{11,119} P T Surukuchi^{12,119}, Y Y Y Wong^{13,119}, W Abdallah^{14,120}, A M Abdullahi^{9,15,120}, R Akutsu^{16,120}, L Alvarez-Ruso^{17,120}, D S M Alves^{18,120}, A Aurisano^{11,120}, A B Balantekin^{19,120}, J M Berryman^{20,21,120}, T Bertólez-Martínez^{22,120}, J Brunner^{23,120}, M Blennow^{24,25,26,120}, S Bolognesi^{27,120}, M Borusinski^{28,120}, T Y Chen^{6,120}, D Cianci^{6,120}, G Collin^{29,120}, J M Conrad^{30,120}, B Crow^{28,120}, P B Denton^{31,120} M Duvall^{28,120}, E Fernández-Martinez^{25,120}, C S Fong^{32,120}, N Foppiani^{2,120}, D V Forero^{33,120}, M Friend^{34,35,120}, A García-Soto^{2,17,120}, C Giganti^{36,120}, C Giunti^{37,120}, R Gandhi^{38,120}, M Ghosh^{39,120}, J Hardin^{30,120}, K M Heeger^{12,120}, M Ishitsuka^{40,120}, A Izmaylov^{41,42,120}, B J P Jones^{43,120}, J R Jordan^{44,120}, N W Kamp^{30,120}, T Katori^{45,120}, S B Kim^{46,120}, L W Koerner^{47,120}, M H Moulai^{60,120}, H Nunokawa^{61,120}, J P Ochoa-Ricoux^{62,120} Y M Oh^{63,120}, T Ohlsson^{24,26,120}, H Päs^{64,120}, D Pershey^{65,120}, R G H Robertson^{66,120}, S Rosauro-Alcaraz^{67,120}, C Rott^{46,68,120}, S Roy^{69,120}, J Salvado^{22,120}, M Scott^{41,120}, S H Seo^{63,120}, M H Shaevitz^{6,120}, M Smiley^{20,70,120}, J Spitz^{44,120}, J Stachurska^{30,120}, M Tammaro⁷¹, T Thakore^{11,120}, C A Ternes^{37,120}, A Thompson^{72,120}, S Tseng^{73,120}, B Vegelecc^{52,120}, T Weiter ^{12,120} S Tseng^{73,120}, B Vogelaar^{52,120}, T Weiss^{12,120} R A Wendell^{74,75,120}, R J Wilson^{59,120}, T Wright^{52,120}, Z Xin^{50,51,120}, B S Yang^{76,120}, J Yoo^{76,120}, J Zennamo^{9,120}, J Zettlemoyer^{9,120}, J D Zornoza^{17,120}, J Zupan¹¹, S Ahmad⁷⁷, E Arrieta-Diaz⁷⁸, V S Basto-Gonzalez⁷⁹, N S Bowden⁸⁰, B C Cañas⁷⁹, D Caratelli⁸¹, C V Chang⁸², C Chen⁸²,

© 2024 The Author(s). Published by IOP Publishing Ltd

T Classen⁸⁰, M Convery⁸³, G S Davies⁸⁴, S R Dennis⁸⁵, Z Djurcic⁸⁶, R Dorrill⁸, Y Du⁸⁷, J J Evans⁸⁸, U Fahrendholz⁵⁷, J A Formaggio³⁰, B T Foust¹², H Frandini Gatti⁹, D Garcia-Gamez⁸⁹, S Gariazzo³⁷, J Gehrlein³¹, C Grant⁹⁰, R A Gomes⁹¹, A B Hansell⁹², F Halzen⁶⁰, S Ho⁹³, J Hoefken Zink^{94,95}, R S Jones⁹⁶, P Kunkle⁹⁰, J-Y Li⁹⁷, S C Li^{52,98}, X Luo⁸¹, Yu Malyshkin^{42,99}, C J Martoff¹⁰⁰, D Massaro^{94,95,101}, A Mastbaum¹⁰², R Mohanta³⁹, H P Mumm¹⁰³, M Nebot-Guinot⁹⁷, R Neilson¹⁰⁴, K Ni¹⁰⁵, J Nieves¹⁷, G D Orebi Gann^{20,70}, V Pandey¹⁰⁶, S Pascoli^{94,95}, G Paz¹⁰⁷, A A Petrov^{107,108}, X Qian³¹, M Rajaoalisoa¹¹, S H Razafinime¹¹, C Roca⁸⁰, G Ron¹⁰⁹, B Roskovec¹¹⁰, E Saul-Sala¹⁷, L Saldaña¹², D W Schmitz¹¹¹, K Scholberg⁶⁵, B Shakya¹¹², P L Slocum¹², E L Snider⁹, H Th J Steiger^{57,113,114}, A F Steklain¹¹⁵, M R Stock⁵⁷, F Sutanto⁸⁰, V Takhistov⁷⁵, R Tayloe¹⁰, Y-D Tsai¹¹⁶, Y-T Tsai⁸³, D Venegas-Vargas⁸⁹, M Wallbank¹¹, E Wang¹¹⁷, P Weatherly¹⁰⁴, S Westerdale⁵⁴, E Worcester³¹, W Wu⁹, G Yang²⁰ and B Zamorano⁸⁹

¹ Universidad del Atlántico, Puerto Colombia, Atlántico, Colombia

² Department of Physics & Laboratory for Particle Physics and Cosmology, Harvard University, Cambridge, MA 02138, United States of America

³ School of Physics and Astronomy, University of Minnesota, Minneapolis, MN 55455, United States of America

⁴ William I. Fine Theoretical Physics Institute, School of Physics and Astronomy, University of Minnesota, Minneapolis, MN 55455, United States of America

⁵ Perimeter Institute for Theoretical Physics, Waterloo, ON N2J 3W9, Canada

⁶ Columbia University, New York, NY, United States of America

⁷ CERN, Esplanade des Particules, 1211 Geneva 23, Switzerland

⁸ Illinois Institute of Technology, Chicago, IL, United States of America

⁹ Fermi National Accelerator Laboratory, Batavia, IL, United States of America

¹⁰ Indiana University, Bloomington, IN, United States of America

¹¹ Department of Physics, University of Cincinnati, Cincinnati, OH 45221, United States of America

¹² Wright Laboratory, Department of Physics, Yale University, New Haven, CT, United States of America

¹³ School of Physics, The University of New South Wales, Sydney NSW 2052, Australia

¹⁴ Department of Mathematics, Faculty of Science, Cairo University, Giza 12613, Egypt

¹⁵ Institute for Particle Physics Phenomenology, Durham University, Durham DH1 3LE, United Kingdom

¹⁶ TRIUMF, Vancouver, British Columbia, Canada

¹⁷ Instituto de Física Corpuscular (IFIC) and Departamento de Física Teórica, Consejo Superior de Investigaciones Científicas (CSIC) and Universidad de Valencia (UV), E-46980, Valencia, Spain

¹⁸ Los Alamos National Laboratory, Los Alamos, NM, United States of America

¹⁹ Department of Physics, University of Wisconsin, Madison, WI 53706, United States of America ²⁰ Department of Physics, University of California, Berkeley, CA 94720, United States of America

²¹ Institute for Nuclear Theory, University of Washington, Seattle, WA 98195, United States of America

²² Departament de Física Quàntica i AstrofÃsica and Institut de Ciéncies del Cosmos, Universitat de Barcelona, Diagonal 647, E-08028 Barcelona, Spain

²³ Aix Marseille Univ, CNRS/IN2P3, CPPM, Marseille, France

²⁴ Department of Physics, School of Engineering Sciences, KTH Royal Institute of Technology, AlbaNova University Center, SE-106 91 Stockholm, Sweden

²⁵ University Autonoma Madrid, Department of Theoretical Physics, Madrid, Spain

²⁶ The Oskar Klein Centre, AlbaNova University Center, SE-106 91 Stockholm,

Sweden

²⁷ Université Paris-Saclay, Gif-sur-Yvette, France

²⁸ University of Hawaii, 2505 Correa Rd, Honolulu, HI 96822, United States of America

²⁹ Department of Physics and Astronomy, University of California, Irvine, CA 92697-4575, United States of America

³⁰ Massachusetts Institute of Technology, Cambridge, MA, United States of America

³¹ Physics Department, Brookhaven National Laboratory, Upton, NY, United States of America

³² Centro de Ciências Naturais e Humanas, Universidade Federal do ABC, 09.210-170, Santo André, SP, Brazil

³ Universidad de Medellín, Carrera 87 Nº 30 - 65 Medellín, Colombia

³⁴ High Energy Accelerator Research Organization (KEK), Tsukuba, Ibaraki 305-00801, Japan

³⁵ J-PARC Center, Tokai, Japan

³⁶ Laboratoire de Physique Nucleaire et de Hautes Energies, IN2P3/CNRS, Sorbonne Universitè, Paris, France

³⁷ Istituto Nazionale di Fisica Nucleare (INFN), Sezione di Torino, Via P. Giuria 1, I–10125 Torino, Italy

³⁸ Harish-Chandra Research Institute (A CI of the Homi Bhabha National Institute), Chhatnag Road, Jhunsi, Allahabad 211019, India

³⁹ School of Physics, University of Hyderabad, Hyderabad 500046, India

⁴⁰ Tokyo University of Science, Department of Physics, Chiba, Japan

⁴¹ Imperial College London, London, United Kingdom

⁴² Institute for Nuclear Research of the Russian Academy of Sciences, Moscow, Russia

⁴³ Dept. of Physics, University of Texas at Arlington, 502 Yates St., Science Hall Rm 108, Box 19059, Arlington, TX 76019, United States of America

⁴⁴ University of Michigan, Ann Arbor, MI, 48109, United States of America

⁴⁵ King's College London, WC2R 2LS, London, United Kingdom

⁴⁶ Department of Physics, Sungkyunkwan University, Suwon 16419, Republic of Korea

⁴⁷ Department of Physics, Houston, Houston, Texas 77204, United States of America

⁴⁸ INFN Sezione di Padova, I-35131 Padova, Italy

⁴⁹ Department of Physics, Colorado School of Mines, Golden, CO, United States of America

⁵⁰ Institute of High Energy Physics, Chinese Academy of Sciences, Beijing 100049, People's Republic of China

⁵¹ School of Physical Sciences, University of Chinese Academy of Sciences, Beijing 100049, People's Republic of China

⁵² Center for Neutrino Physics, Department of Physics, Virginia Tech, Blacksburg, VA 24061, United States of America

⁵³ Michigan State University, East Lansing, MI, United States of America

⁵⁵ North Carolina Central University, 1801 Fayetteville Street, Durham, NC 27707, United States of America

⁵⁸ Department of Physics, Pennsylvania State University, University Park, PA 16802, United States of America

⁵⁹ Colorado State University, Fort Collins, CO, United States of America

⁶⁰ Dept. of Physics and Wisconsin IceCube Particle Astrophysics Center, University of Wisconsin-Madison, Madison, WI 53706, United States of America

⁶¹ Departamento de Física, Pontifícia Universidade Católica do Rio de Janeiro,

C.P. 38097, 22451-900, Rio de Janeiro, Brazil

⁶² Department of Physics and Astronomy, University of California at Irvine, Irvine, CA 92697, United States of America

⁶³ Center for Underground Physics, Institute for Basic Science (IBS), Daejeon 34126, Republic of Korea

⁶⁴ TU Dortmund, Department of Physics, D-44221, Dortmund, Germany

⁶⁵ Department of Physics, Duke University, Durham, NC 27708, United States of America

⁶⁶ University of Washington, Seattle, WA, United States of America

⁶⁷ Pôle Théorie, Laboratoire de Physique des 2 Infinis Irène Joliot Curie (UMR 9012) CNRS/IN2P3, F-91400 Orsay, France

⁶⁸ Department of physics, The University of Utah, 201 Presidents' Cir, Salt Lake City, UT 84112, United States of America

⁶⁹ Institute of Physics, Sachivalaya Marg, Sainik School Post, Bhubaneswar 751005, India

⁷⁰ Lawrence Berkeley National Laboratory, Berkeley, CA 94720, United States of America

⁷¹ Jozef Stefan Institute, Jamova 39, Ljubljana, Slovenia

⁷² Mitchell Institute for Fundamental Physics and Astronomy, Department of Physics and Astronomy, Texas A&M University, College Station, TX 77845, United States of America

⁷³ Department of Physics, Faculty of Science, The University of Tokyo, Bunkyo-ku, Tokyo 113-0033, Japan

⁴ Kyoto University, Department of Physics, Kyoto, Japan

⁷⁵ Kavli Institute for the Physics and Mathematics of the Universe (WPI), UTIAS, Japan

⁷⁶ Department of Physics and Astronomy, Seoul National University, Seoul 08826, Republic of Korea

⁷⁷ HEPG, Pakistan Institute of Nuclear Science and Technology (PINSTECH), Islamabad, Pakistan

⁷⁸ Universidad del Magdalena, Carrera 32 Nº 22 - 08 Santa Marta, Colombia

⁷⁹ Universidad de Pamplona, km 1, vía salida a Bucaramanga, Campus Universitario, 543050 Pamplona, Colombia

⁸⁰ Lawrence Livermore National Laboratory, Livermore, CA, United States of America

⁸¹ University of California Santa Barbara, Santa Barbara, CA 93106, United States of America

⁸² Department of Physics, National Taiwan Normal University, Taipei 116, Taiwan

⁸³ SLAC National Accelerator Laboratory, United States of America

⁸⁴ University of Mississippi, University, MS 38677, United States of America

⁸⁵ Cavendish Laboratory, University of Cambridge, Cambridge, United Kingdom

⁵⁴ Princeton University, Princeton, NJ, United States of America

⁵⁶ Max-Planck-Institut für Physik, München, Germany

⁵⁷ Technische Universität München, Garching, Germany

⁸⁶ Argonne National Laboratory, Lemont, IL 60439, United States of America

- ⁸⁷ CAS Key Laboratory of Theoretical Physics, Institute of Theoretical Physics,
- Chinese Academy of Sciences, Beijing 100190, People's Republic of China

⁸⁸ University of Manchester, Manchester M13 9PL, United Kingdom

⁸⁹ University of Granada & CAFPE, Granada, Spain

⁹¹ Instituto de Física, Universidade Federal de Goiás, Goiânia, GO 74690-900, Brazil

⁹² Department of Physics, Susquehanna University, Selinsgrove, PA, United States of America

⁹³ Korea Institute for Advanced Study, Seoul 02455, Republic of Korea

- ⁹⁴ Dipartimento di Fisica e Astronomia, Università di Bologna, via Irnerio 46, I-40126 Bologna, Italy
- ⁹⁵ INFN, Sezione di Bologna, viale Berti Pichat 6/2, I-40127 Bologna, Italy
- ⁹⁶ Department of Physics and Astronomy, University of Sheffield, Sheffield, United Kingdom
- ⁷ University of Edinburgh, Edinburgh EH8 9YL, United Kingdom

⁹⁸ Department of Physics and Astronomy, Purdue University, West Lafayette, IN 47907, United States of America

⁹⁹ Joint Institute for Nuclear Research, Dubna, 141980, Russia

- ¹⁰⁰ Temple University, Philadelphia, PA 19122, United States of America
- ¹⁰¹ Centre for Cosmology, Particle Physics and Phenomenology (CP3), Université Catholique de Louvain, B-1348 Louvain-la-Neuve, Belgium
- ¹⁰² Rutgers University, Piscataway, NJ, United States of America

¹⁰³ National Institute of Standards and Technology, Gaithersburg, MD 20899, United States of America

¹⁰⁴ Department of Physics, Drexel University, Philadelphia, PA 19104, United States of America

¹⁰⁵ University of California San Diego, La Jolla, CA 92093, United States of America

¹⁰⁶ Department of Physics, University of Florida, Gainesville, FL 32611, United States of America

¹⁰⁷ Department of Physics and Astronomy, Wayne State University, Detroit, Michigan 48201, United States of America

¹⁰⁸ Department of Physics and Astronomy, University of South Carolina, Columbia, SC 29205, United States of America

¹⁰⁹ Hebrew University of Jerusalem, Israel

¹¹⁰ Faculty of Mathematics and Physics, Charles University, Prague, V Holešovičkách 2, 18000 Prague, Czech Republic ¹¹¹ Enrico Fermi Institute, University of Chicago, Chicago, IL 60637, United States of

America ¹¹² Deutsches Elektronen-Synchrotron DESY, Notkestrasse 85, D-22607 Hamburg,

Germany

¹¹³ Johannes Gutenberg Universität zu Mainz, Mainz, Germany

¹¹⁴ Cluster of Excellence PRISMA+, Mainz, Germany

¹¹⁵ Department of Mathematics, Universidade Tecnológica Federal do Paraná, 80.230-901, Curitiba, PR, Brazil

¹¹⁶ University of California, Irvine, CA 92697, United States of America

¹¹⁷ School of Physics and Microelectronics, Zhengzhou University, Zhengzhou, Henan 450001, People's Republic of China

¹¹⁸ Department of Physics, Case Western Reserve University, Cleveland, OH 44106, United States of America

⁹⁰ Department of Physics, Boston University, Boston, MA 02215, United States of America

E-mail: alex.sousa@uc.edu

Received 12 December 2022 Accepted for publication 6 March 2024 Published 29 October 2024



Keywords: neutrino physics, light sterile neutrino, experimental neutrino anomalies

Executive summary

Several decades of a rich and diverse program of experimental neutrino measurements have provided an increasingly clearer picture of the elusive neutrino sector, and uncovered physics not predicted by the Standard Model (SM), such as the existence of nonzero neutrino masses implied by the surprising discovery of neutrino flavor mixing. This foundational discovery represented a welcome resolution to decades-long experimental anomalies associated with solar and atmospheric neutrino measurements.

Alongside this foundational discovery, experimental neutrino anomalies have been observed that still remain unresolved, and have served as primary drivers in the development of a vibrant short-baseline neutrino program, and in the launch of a multitude of complementary probes within a large variety of other experiments. Two of these anomalies arise from the apparent oscillatory appearance of electron (anti)neutrinos in relatively pure muon-(anti)neutrino beams originating from charged-pion decay-at-rest, specifically the LSND Anomaly, and from charged-pion decay-in-flight, the MiniBooNE Low-Energy Excess. Two other anomalies are associated with an overall normalization discrepancy of electron (anti) neutrinos expected both from conventional fission reactors, the Reactor Neutrino Anomaly, and in the radioactive decay of Gallium-71, the Gallium Anomaly. In these two latter cases, no oscillatory signature is observed, but the overall normalization deficit can be ascribed to rapid oscillations that are averaged out and appear as an overall deficit.

Historically, these anomalies were first interpreted as oscillations due to the existence of light sterile neutrinos that mix with the three SM neutrinos. This interpretation requires an oscillation frequency $\Delta m^2 \gtrsim 1 \text{ eV}^2$, implying the addition of at least one neutrino to the threeflavor mixing paradigm. This new neutrino would have to be a SM gauge singlet, thus it is referred to as *sterile*, as LEP measurements of the invisible decay width of the Z boson show only three neutrinos couple to the Z boson. However, this purely oscillatory interpretation is disfavored by several other direct and indirect experimental tests. Consequently, recent years have seen accelerating theoretical interest in more complex Beyond the Standard Model (BSM) flavor transformation and hidden-sector particle production as explanations for the anomalies. Experimental interest in testing a more diverse set of interpretations has also been growing, as well as motivation to probe deeper into potential conventional explanations. The

¹¹⁹ Editor.¹²⁰ Contributor.



Original content from this work may be used under the terms of the Creative Commons Attribution 4.0 licence. Any further distribution of this work must maintain attribution to the author(s) and the title of the work, journal citation and DOI.

discovery of new physics associated with these anomalies would be groundbreaking, and would have profound implications not only for particle physics but also for astrophysics and cosmology.

This white paper provides a comprehensive review of our present understanding of the experimental neutrino anomalies, charting the progress achieved over the last decade at the experimental and phenomenological level, and sets the stage for future programmatic prospects in addressing the anomalies. In a similar spirit to the 'Light Sterile Neutrinos: A White Paper' document from a decade ago [1], this new white paper is purposed to serve as a guiding and motivational 'encyclopedic' reference, with emphasis on needs and options for future exploration that may lead to the ultimate resolution of the anomalies.

Developments over the past decade

Following the requirements identified in [1] has led to a broader understanding of viable interpretations of the anomalies and strengthened experimental efforts—and experimental capabilities—in that direction. Notably, the requirement to probe the anomalies with multiple and orthogonal approaches (accelerator-based short/long-baseline, reactor-based short-baseline, atmospheric neutrinos, and radioactive source) in the same spirit as employed for neutrino oscillations has been realized through recent, ongoing, or impending experimental programs:

- 1. The development of new radioactive sources and detectors for improved tests of the Gallium Anomaly has been pursued and realized in the form of the BEST experiment.
- The reactor antineutrino anomaly and subsequent reactor-based activities and new results have placed a required emphasis on oscillation-testing short-baseline reactor experiments and on improved understanding of reactor neutrino fluxes.
- 3. The community has just begun a comprehensive multi-channel/multi-baseline accelerator-based short-baseline program to search for 3+N oscillations while directly addressing the MiniBooNE anomaly both in regards to oscillatory and non-oscillatory solutions.
- Recent searches for smoking-gun signatures of light sterile neutrinos with high-energy atmospheric neutrinos, such as the one performed by the IceCube Neutrino Observatory.
- 5. A direct test of the LSND anomaly using an improved decay-at-rest beam facility and experimental arrangement has just begun in the form of the JSNS² experiment.
- 6. Beyond direct anomaly tests, many alternate techniques/facilities, including direct neutrino mass measurements, long-baseline oscillation experiments, and atmospheric and astrophysical neutrino experiments, have been applied to the sterile neutrino explanation of the anomalies.

Primary focuses for the next decade

As the question of light sterile neutrino oscillations is further explored over the next several years, the community's efforts should be directed toward disentangling the plethora of possibilities that have been identified over the past ten years as viable interpretations of the experimental anomalies in the neutrino sector. The goal of these collective efforts will be to validate and solidify our understanding of the neutrino sector. Regardless of what the ongoing and upcoming experiments observe—be it a deviation from the three-neutrino picture or otherwise—the community should be prepared to address how to put these anomalies to test

or adequately distinguish between different interpretations. We see the main experimental, analysis, and theory-driven thrusts that will be essential to achieving this goal being:

- Cover all anomaly sectors: given the fundamentally unresolved nature of all four canonical anomalies, it is imperative to support all pillars of a diverse experimental portfolio—source, reactor, decay-at-rest, decay-in-flight, and other methods/sources—to provide complementary probes of and increased precision for new physics explanations.
- 2. **Pursue diverse signatures:** given the diversity of possible experimental signatures associated with allowed anomaly interpretations, it is imperative that experiments make design and analysis choices that maximize sensitivity to as broad an array of these potential signals as possible.
- 3. Deepen theoretical engagement: priority in the theory community should be placed on the development of new physics models relevant to all four canonical short-baseline anomalies and the development of tools for enabling efficient tests of these models with existing and future experimental datasets.
- 4. **Openly share data:** fluid communication between the experimental and theory communities will be required, which implies that both experimental data releases and theoretical calculations should be publicly available. In particular, as it is most likely that a combination of measurements will be needed to resolve the anomalies, global fits should be made public, as well as phenomenological fits and constraints to specific data sets.
- 5. Apply robust analysis techniques: appropriate statistical treatment is crucial to quantify the compatibility of data sets within the context of any given model, and in order to test the absolute viability of a given model. Accurate evaluation of allowed parameter space is also an important input to the design of future experiments.

The white paper is organized as follows. Section 1 provides an overall introduction and motivation for seeking resolution of the experimental neutrino anomalies, and section 2 introduces each of the anomalies in detail, placing them within historical context. Section 3 delves into the theoretical interpretation of the anomalies, detailing phenomenological consequences of various scenarios that have been or are being pursued. Section 4 goes over the broader experimental landscape, discussing the impact of null results, as well as potential conventional explanations for the anomalies, while section 5 covers results from astrophysical and cosmological indirect probes. Section 6 reviews the very diverse landscape of future experimental prospects that will be capable of addressing the anomalies. Finally, section 7 reiterates our vision for a path that will lead to the ultimate resolution of the anomalies, providing further discussion and elaboration of the focal points for the next decade listed in this executive summary.

1. Introduction and motivation

The Nobel prize-winning discovery of neutrino oscillation [2–4] has led to a three-neutrino mixing picture that is now established as a minimal extension to the Standard Model (SM), and which is only empirically motivated. This picture prescribes an 'Extended SM' (ESM), in which the neutrino sector includes three distinct neutrino mass states that are each an independent linear combination of the three neutrino weak eigenstates: ν_e , ν_{μ} , and ν_{τ} [5]. This discovery stands as one of few indisputable pieces of evidence for new physics 'beyond the SM' (BSM).

Generating neutrino masses is qualitatively different from the mass generation for any other fermions in the SM. The Higgs mechanism for neutrinos would require the existence of a right-handed neutrino, which would carry no SM gauge quantum number. This in turn would allow

for Majorana masses of these right-handed fields, opening up the possibility of a seesaw mechanism [6-10]. Several other scenarios, involving different particle content, could also explain the origin of neutrino masses, such as type-II and type-III seesaw models [11-16]. In general, the mechanism of neutrino masses would require the addition of particle content to the SM that has never been observed. The lack of experimental indication of the scale of this new physics makes the neutrino sector a promising portal to new physics. Many neutrino masses models would predict observable deviations from the ESM and could lead to a rich phenomenology. In particular, the existence of new states or gauge interactions associated with neutrinos could affect neutrino experiments in a variety of ways, for example as effects on oscillation phenomenology or new particles produced in neutrino beams or in neutrino detectors.

In particular, interest in this direction has been fanned by a series of anomalous experimental measurements, especially since the mid-1990's, which suggested the existence of new neutrino states. It is expected that these states should be 'sterile', i.e. non-weakly-interacting, in order to avoid experimental constraints from invisible $Z \rightarrow \nu \bar{\nu}$ decay measurements [17]. There is now a series of indications of neutrino phenomena deviating from the three-neutrino (ESM) paradigm, many of which have the commonality of being observed primarily in association with electron neutrino observations, from either electron neutrino or muon neutrino sources, with either Cherenkov or scintillator detectors, and at relatively 'short baselines' from the neutrino sources¹²¹. These 'short-baseline experimental anomalies', and the expansive and dedicated scientific program that has been launched over the past two decades to address them, is the focus of this paper.

One of the most widely examined theoretical frameworks considered for the interpretation of these anomalies is that of light ($\sim 1 \text{ eV}$) sterile neutrino oscillations [1]. This framework generally extends the three-neutrino paradigm of the ESM to accommodate (3+*N*) light neutrino masses and (3+*N*) neutrino flavor states, where *N* refers to additional neutrino mass states with masses of order $\sim 1 \text{ eV}$. The latter are a linear combination of primarily *N* sterile neutrino eigenstates but contain a small admixture of weak neutrino eigenstates so that they can participate in neutrino oscillations. This framework generally leads to observable neutrino oscillation amplitudes of order 1% or less, and disappearance oscillation amplitudes of up to tens of percent. The relatively small oscillation amplitudes are constrained by unitarity considerations [18], and the known oscillation amplitudes from 'medium-' and 'long-baseline' neutrino oscillation measurements.

While the light sterile neutrino oscillation framework can, in theory accommodate all shortbaseline experimental anomalies to date, it fails to accommodate the lack of corresponding oscillations in other short-baseline, long-baseline, and atmospheric neutrino measurements. One particular experimental anomaly, contributed by the MiniBooNE experiment, exacerbates this issue. The need to interpret compelling experimental results, on the other hand, has given rise to an extensive experimental neutrino program, as well as a substantial body of related phenomenological work, including many viable interpretations, from modifications of three-flavor neutrino mixing to potential couplings to hidden sectors, which we review here.

A consistent picture of short-baseline experimental anomalies has not yet formed, as will be discussed in this paper. On the other hand, new experiments launched over the past decade or about to be launched, with the goal of independently investigating either specific experimental anomalies or specific theoretical interpretations, promise to deliver new and

¹²¹ We introduce and note that the term 'short baseline' is used qualitatively, and more specifically it refers to neutrino propagation distances of O(1 km) for measurements performed with neutrino energy of O(1 GeV). More broadly, it refers to a ratio of neutrino propagation distance relative to neutrino energy of $\sim 1 \text{ km GeV}^{-1}$, corresponding to a neutrino oscillation frequency Δm^2 of $\sim 1 \text{ eV}^2$.

invaluable information that will either identify the underlying source(s) of these anomalies or guide future scientific endeavors to better understand them. The paper further discusses theoretical developments over the past decade, as well as future research programs with the ability to further elucidate this picture, with attention to synergy and complementarity of both planned and proposed programs.

Sterile neutrino states, if they exist, would reveal a new, unexpected form of a fundamental particle and possibly new types of interactions in nature. This possibility, or other BSM physics that may be the source(s) of the current experimental neutrino anomalies, has and will continue to compel particle physicists toward further experimentation in this area in the foreseeable future. The Discovery of new physics associated with these signals would be groundbreaking and would have profound implications not only for particle physics but also for astrophysics and cosmology. Additionally, in several scenarios, new physics associated with these anomalies can have a significant impact on measurements of three-neutrino oscillation parameters planned with ongoing and future long-baseline experiments, as well as on absolute measurements of neutrino mass, further necessitating their resolution. Similarly, new physics associated with these anomalies could connect to searches for neutrinoless double β decay, and direct or indirect probes for dark matter or other dark sector particle searches. Alternatively, a clear null result, or an SM explanation for the current experimental anomalies, would bring a welcome resolution to a longstanding puzzle and greatly clarify the current picture in neutrino physics.

2. Experimental anomalies

There are four long-standing anomalies in the neutrino sector that have served as primary drivers in the development of a vibrant short-baseline neutrino program over the last decade. Two come from the apparent oscillatory appearance of electron (anti)neutrinos in relatively pure muon-(anti)neutrino beams originating from charged-pion decay-at-rest, section 2.1, and charged-pion decay-in-flight, section 2.2. Two more anomalies are associated with an overall normalization discrepancy of electron (anti)neutrinos expected both from conventional fission reactors, section 2.3, and in the radioactive decay of Gallium-71, section 2.4. In these two cases, no oscillatory signature is observed, but the overall normalization deficit can be ascribed to rapid oscillations at a high Δm^2 that are averaged out and appear as an overall deficit. This section will describe all four of these anomalies in detail, presenting both their experimental arrangements as well as the experimental (anomalous) results.

Historically, the results have been discussed primarily in the context of a 3 + 1 scenario, with a single sterile neutrino. As such, the results in this section are presented in this manner; however, we emphasize that the current theoretical landscape strives to explore a much broader set of possible interpretations of the anomalies, as we describe in detail in section 3. Those include more exotic flavor conversions, section 3.1, dark sector particles produced in neutrino scattering or in the neutrino source/beam itself, section 3.2, as well as more conventional explanations due to background mismodeling or underestimation, section 3.3.

2.1. Pion decay-at-rest accelerator experiments

Pion decay-at-rest accelerator experiments provide a well-understood muon antineutrino flux of a mean energy of ~30 MeV, and negligible electron antineutrino flux contamination. As such, detectors placed at relatively short baselines (~30 m) with positron identification capability offer sensitivity to $\overline{\nu}_{\mu} \rightarrow \overline{\nu}_{e}$ oscillations. Past pion decay-at-rest experiments include the Los Alamos Neutrino Detector (LSND) [19] and the KArlsruhe Rutherford Medium

Energy Neutrino (KARMEN) [20] experiments. Among the two, LSND observed evidence for $\overline{\nu}_{\mu} \rightarrow \overline{\nu}_{e}$ oscillations that could be parametrized by two-neutrino oscillations with a Δm^{2} of ~1 eV² and an oscillation amplitude sin² $2\theta_{\mu e}$ of less than 1%. Although less sensitive, KARMEN observed no evidence of such oscillations and therefore has been historically referred to as a 'null' experiment in terms of this framework [1]. The LSND anomalous result, which motivated a number of follow-up experimental searches for short-baseline neutrino oscillations over the past nearly three decades, is described below.

The LSND detector [21] at Los Alamos National Lab consisted of a cylindrical tank, 8.3 m long with a 5.7 m in diameter, located 29.8 m from the neutrino source. LSND was designed to search for oscillations $\overline{\nu}_{\mu} \rightarrow \overline{\nu}_{e}$. Neutrinos were produced from the decay chains of charged pions to muons decaying at rest, with the charged pions produced using 798 MeV protons on a target at the Los Alamos Neutron Science Center (LANSCE). Muon antineutrinos were produced by the sequence of $\pi^+ \rightarrow \mu^+ + \nu_{\mu}$ and $\mu^+ \rightarrow e^+ + \nu_e + \overline{\nu}_{\mu}$. The related decay of π^- that would produce $\overline{\nu}_{e}$ is highly suppressed through pion capture on heavy nuclei in the vicinity of the beam target. As a result, the intrinsic $\overline{\nu}_{e}$ contamination was expected to be 7.8×10^{-4} smaller than the $\overline{\nu}_{\mu}$ flux.

The signal selection proceeded via the identification of a positron from inverse beta decay (IBD), $\overline{\nu}_e + p \rightarrow e^+ + n$, followed by detection of a 2.2 MeV photon from subsequent neutron capture that is correlated with the positron both in position and time. The interactions of ν_e inherent in the beam via $\nu_e + C_{12} \rightarrow N_{12} + e^-$ were not a major contributing background, as there was no correlated neutron capture accompanying these events, except for accidental coincidences. The target consisted of 167 metric tons of mineral oil, which was lightly doped with scintillator allowing for both the detection of Cherenkov light and isotropic scintillation light. This light was detected with 1220 8" photomultiplier tubes (PMTs) spaced uniformly around the inner surface of the tank. As the Cherenkov ring could be detected, this allowed for the determination of both energy and angle of the outgoing positron.

Cosmic rays, although abundant, were not a major source of backgrounds for LSND, being removed by timing cuts and usage of an active and optically isolated veto shield surrounding the detector. True correlated 2.2 MeV photons were separated from coincident photons from radioactivity using a likelihood ratio R_{γ} variable cut; this was defined to be the likelihood that the photon was correlated divided by the likelihood that the photon was accidental.

A series of LSND measurements were published, all in support of an excess of events observed over that expected from beam-off and beam-on neutrino background. The final results published in 2001 [19] concluded that an excess of events was observed, consistent with two-neutrino oscillations, corresponding to a background-subtracted excess of 87.9 \pm 22.4 (stat) \pm 6.0 (sys) events. Distributions of the observed excess are reproduced in figure 1 s a function of both the observed positron energy and the reconstructed L/E_{ν} , for the subset of total selected events with $R_{\gamma} > 10$ (described as a clean sample of oscillation candidate events); this selection corresponds to an excess of 49.1 \pm 9.4 events.

The 3 + 1 sterile neutrino fit, which tests an effective two-flavor $\nu_{\mu} \rightarrow \nu_{e}$ appearance probability hypothesis under the short-baseline (SBL) approximation $\Delta m_{21}^{2} \approx \Delta m_{31}^{2} \equiv 0$, was not performed simply in reconstructed neutrino energy, but as a likelihood in four dimensional E_{e} (electron energy), R_{γ} (coincidence variable), z (electron distance along tank axis) and $\cos \theta$ (electron angle w.r.t neutrino beam) space. The best-fit point was found to be at an oscillation amplitude of $\sin^{2} 2\theta_{\mu e} = 0.003$ with a mass splitting of $\Delta m^{2} = 1.2 \text{ eV}^{2}$. The resulting allowed regions are shown in figure 2 alongside then-contemporary experiments that did not see a positive signal.





Figure 1. The LSND anomalous events as a function of both L/E_{ν} (left) and observable electron energy (right), for the subset of total selected events with $R_{\gamma} > 10$ and $20 < E_e < 60$ MeV. Note the blue shaded region is for a best fit two-neutrino oscillation fit of $\sin^2 2\theta = 0.003$ and $\Delta m^2 = 1.2$ eV². Reprinted (figure) with permission from [19], Copyright (2001) by the American Physical Society.



Figure 2. The results of a two-neutrino oscillation fit performed by the LSND collaboration for all data with reconstructed electron energy $20 < E_e < 200$ MeV, showing the resulting 90 and 99% confidence level allowed regions for $\sin^2 2\theta_{\mu e}$ and Δm^2 . Shown also are the 90% CL limits from other contemporary experiments. Reprinted (figure) with permission from [19], Copyright (2001) by the American Physical Society.

2.2. Piondecay-in-flight accelerator experiments

The LSND evidence for two-neutrino oscillation beginning in the late 1990s prompted the need for an independent follow-up experiment to test the result. Such test needed to rely on different systematics and methodology, while preserving sensitivity to the same Δm^2 and $\sin^2 2\theta_{\mu e}$. A π^+ decay-in-flight accelerator beam can produce a muon-neutrino-dominated flux with higher mean energy, providing an opportunity for an independent test at a longer baseline, and through different detection methods. This independent test was realized with the Booster Neutrino Beamline (BNB) at the U.S. Fermi National Accelerator Laboratory (FNAL, or Fermilab) [22], providing a ~99.5%-pure muon neutrino beam with a mean neutrino energy of ~600 MeV, sampled by the MiniBooNE Cherenkov detector [23].

Both MiniBooNE and BNB at Fermilab were designed in such a manner so as to have comparable L/E to that of LSND ($L/E \approx 0.4-1.0 \text{ m MeV}^{-1}$) but a longer baseline (540 m relative to LSND's 30 m) and higher energy (peak energy ~700 MeV relative to LSND's ~50 MeV). Although initially envisaged as being a two-detector experimental setup, with a near detector at $L \approx 0$ m and a far detector at $L/E \approx 0.4-1.0$ m MeV⁻¹, the final MiniBooNE experiment consisted of a single detector, which was a spherical tank, 12.2 m in diameter, filled with 818 tons of mineral oil. The interior surface of the tank, including an outer veto spherical shell region, was lined with 1520 8" PMTs, including the recycled usage of all 1220 PMTs from the LSND experiment. Cherenkov and scintillation photons emitted by particles produced in neutrino interactions were used to differentiate electrons versus muons produced in ν_e versus ν_{μ} interactions, respectively.

The primary reconstruction method in MiniBooNE uses the Cherenkov rings detected on the inside surface of the detector to differentiate between electrons, muons and charged pions, and neutral pions. Protons that fall below the Cherenkov threshold in mineral oil, ~350 MeV kinetic energy, cannot be observed by their Cherenkov rings. Prompt scintillation light, however, can be used to estimate the energy of particles below the threshold. One crucial point to understand MiniBooNE's backgrounds is the fact that a single lone photon (which subsequently pair-produces a collimated e^+e^- pair) is indistinguishable from a single electron in terms of their Cherenkov ring reconstruction. The separation of neutral-current (NC) $\pi^0 \rightarrow$ $\gamma\gamma$ events thus relies entirely on reconstructing two separate Cherenkov rings. As such, the main backgrounds to searching for ν_e from $\nu_{\mu} \rightarrow \nu_e$ oscillations at MiniBooNE were:

- 1. Intrinsic ν_e in the BNB. Although an extremely pure ν_{μ} beam, the $\mathcal{O}(0.5\%)$ ν_e and $\overline{\nu}_e$ in the beam provide an irreducible background. These are constrained by the high-statistics ν_{μ} sample due to their common origin in meson decay chains.
- 2. NC π^0 events. In the scenario where one of the daughter photons from a NC π^0 decay is missed the event is indistinguishable from a single electron. This can occur due to overlapping Cherenkov cones, one photon exiting the detector before pair converting, or extremely low energy secondary photons. The NC π^0 's were constrained by a high-statistics *in situ* measurement.
- 3. NC $\Delta \rightarrow N\gamma$. Radiative decay of the Δ baryon is a predicted SM process that produces a single photon, mimicking single-electron production in MiniBooNE.
- 4. 'Dirt' events. The so-called 'Dirt' events correspond to neutrino-induced events in which the scattering takes place in the material surrounding the detector, but some particles scatter inside the detector and are reconstructed. The majority of these are photons scattering in from π^0 decays.

Although the primary observable corresponds to the reconstructed outgoing electron itself, the results are often presented and interpreted in terms of quasi-elastic reconstructed neutrino energy, defined as

$$E_{\nu}^{QE} = rac{1}{2} \Biggl(rac{2(M_n - \mathcal{B})E_e - ((M_n - B)^2 + M_e^2 - M_p^2)}{(M_n - \mathcal{B}) - E_e + \sqrt{E_e^2 - M_e^2}\cos\theta_e} \Biggr),$$

where M_n , M_p and M_e are the masses of neutron, proton and electron respectively, E_e is the reconstructed energy of the electron, $\cos \theta_e$ is the angle the reconstructed electron makes relative to the neutrino beam, and \mathcal{B} is the binding energy of the target nucleus.

The first results from MiniBooNE, published in 2007, used approximately a third of the total data set collected by MiniBooNE, corresponding to a BNB proton beam delivery of 5.58×10^{20} protons-on-target (POT). The result reported no evidence for oscillations within a two-neutrino $\nu_{\mu} \rightarrow \nu_{e}$ appearance paradigm [24], thus placing a 90% CL limit covering the majority of the allowed LSND ($\sin^{2} 2\theta_{\mu e}, \Delta m^{2}$) parameter space. Crucially, this oscillation result was performed only for the region of reconstructed neutrino energy of $E_{\nu}^{QE} > 475$ MeV (assuming quasi-elastic scattering)¹²². While this first result contained no significant excess in the $E_{\nu}^{QE} > 475$ MeV region, below this energy, an excess of events was observed. This excess, further examined by the MiniBooNE collaboration in a subsequent analysis with higher statistics, is often referred to as the MiniBooNE 'Low-Energy Excess' (or LEE), and consisted of 128.8 \pm 20.4 \pm 38.3 excess events above predicted backgrounds, corresponding to 3.0 σ , as reported in [25].

Unlike LSND, MiniBooNE was designed with the ability to switch from a neutrino- to an antineutrino-dominated beam, by switching the charged-pion focusing magnetic field polarity, preferentially focusing π^- mesons produced in proton-Be interactions toward the detector, resulting in a $\overline{\nu}_{\mu}$ -dominated neutrino flux. Although the intrinsic ν_{e} and $\overline{\nu}_{e}$ contamination remains very small in antineutrino mode (0.6%), the wrong sign contamination is not negligible (with 83.73% $\overline{\nu}_{\mu}$ and 15.71% ν_{μ} components). As such, MiniBooNE repeated its search for two-neutrino oscillations in antineutrino running mode in 2010, using data corresponding to 5.66 \times 10²⁰ POT. The antineutrino search was motivated by findings supporting large observable CP violation in short-baseline oscillations involving two additional, mostly sterile, neutrino mass states with masses of order ~1 eV [26], as well as CPTor Lorenz-violating models suggested as alternative interpretations of LSND at the time [27–29]. Given that LSND's result was obtained with antineutrinos, an independent antineutrino search-albeit less sensitive due to reduced statistics expected from a factor-of-two suppression in flux—was motivated by the need to provide an independent test of LSND regardless of CP or other symmetry violation assumption, and as a further probe of the LEE anomaly. The results from MiniBooNE's first $\overline{\nu}_{\mu} \rightarrow \overline{\nu}_{e}$ search [30] followed in 2010, and showed an excess extending both at low energy and in the oscillation signal region of $475 < E_{\nu}^{QE} < 300$ MeV. The results were found to be consistent with two-neutrino $\overline{\nu_{\mu}} \rightarrow \overline{\nu_{e}}$ oscillations with a χ^2 probability of 8.7% compared to 0.5% for background only [30], with a best fit at $\Delta m^2 = 0.064 \text{ eV}^2$ and $\sin^2 2\theta = 0.96$. When the fit was expanded to the whole energy range, $E_{\nu}^{QE} > 200$ MeV, the best fit was found to be at $\Delta m^2 = 4.42$ eV² and $\sin^2 2\theta = 0.0066$, which although the best fit itself lies outside LSND's 99% allowed contour, there was still significant overlap in the low Δm^2 allowed regions at the 90% CL.

¹²² This restriction was decided upon as part of the data unblinding process followed by the MiniBooNE collaboration, supported by the findings that spectral information in this background-dominated region did not contribute significantly to two-neutrino oscillation sensitivity, and furthermore a data to Monte Carlo prediction discrepancy was observed with both the best-fit two-neutrino oscillation hypothesis and the SM prediction.



Figure 3. The final MiniBooNE allowed regions for the full fit of all neutrino and antineutrino running mode data. Reproduced from [33]. CC BY 4.0.

Since those first results, MiniBooNE ran for approximately ten more years, collecting BNB data corresponding to a total of 18.75×10^{20} POT in neutrino running mode, and 11.27×10^{20} POT in antineutrino running mode. Major updates were published in 2013 [31], 2018 [32] and 2020 [33], and for the remainder of this summary, we will concentrate on the final 2020 results, unless noted otherwise. MiniBooNE's final results are reproduced in figure 4, showing an excess of data over background prediction in both neutrino and antineutrino data sets, as a function of the reconstructed electron candidate energy and reconstructed electron angle with respect to the beam. The excess is predominately evident below 600 MeV and has an overall significance of 4.8σ (combining neutrino and antineutrino running mode data sets). This significance is almost entirely systematics-limited in nature, and corresponds to 560.6 ± 119.6 and 77.4 ± 28.5 excess events in neutrino and antineutrino running modes, respectively. The final best-fit parameters for the full neutrino and antineutrino data sets, for a fit over the entire energy range $200 < E_{\nu}^{QE} < 3000$ MeV were found to be at

$$\Delta m^2 = 0.043 \text{ eV}^2$$
, and $\sin^2 2\theta = 0.807$,

with the best-fit χ^2/ndof for the energy range $200 < E_{\nu}^{QE} < 1250 \text{ MeV}$ being 21.7/15.5 (probability of 12.3%) compared to the background-only χ^2/ndof of 50.7/17.3 (probability of 0.01%). While this best fit is close to maximal mixing and is ruled out by a number of experiments (e.g. OPERA [34]), the 1σ allowed regions stretches to much smaller mixing angles as can be seen in figure 3 overlapping substantially with the allowed LSND regions. None of the upper portion of LSND's allowed regions, the island at higher $\Delta m^2 > 10 \text{ eV}^2$, is within the combined MiniBooNE 95% CL.



Figure 4. The final MiniBooNE results corresponding to 18.75×10^{20} POT in neutrino mode (top figures) and 11.27×10^{20} POT in antineutrino mode (bottom figures) for both the reconstructed visible energy (left) and the reconstructed angle that the Cherenkov cone makes with respect to the neutrino beam (right). Note that as the top two figures corresponding to neutrino mode are from [33], and the bottom two for antineutrino running more are from [32], the best-fit line does not correspond to the exact same point in sterile parameter space. Reproduced from [32, 33]. CC BY 4.0.

While the overall number of excess events is consistent with that expected from twoneutrino oscillations driven by LSND's best-fit parameters, a key feature of these results, highlighted in figure 4, is that the predicted signal from $\nu_{\mu} \rightarrow \nu_{e}$ oscillations corresponding to MiniBooNE's two-neutrino oscillation best-fit parameters cannot accommodate the shape of the excess. This is particularly the case in neutrino mode, especially in the most forward region of the outgoing electron $\cos \theta$ distribution (with θ representing the angle of the electron relative to the incoming neutrino beam direction). A similar feature exists as a function of the reconstructed neutrino energy (E_{ν}^{QE}) distribution in MiniBooNE. While historically the MiniBooNE excess was presented almost exclusively in terms of the reconstructed neutrino energy (E_{i}^{QE}) , this was primarily due to the most common contemporary interpretation being that of a 3 + 1 oscillatory effect. It is worth noting that if the origin of the excess is not oscillatory in nature then additional information can be gained by studying the excess as a function of other reconstructed variables. Reconstructed visible shower energy and shower angle as shown in figure 4 are two such examples, as is the reconstructed radial position of the event in the detector and its timing relative to the beam, shown for neutrino mode running in figure 5.

These observations have helped motivate and understand 'conventional' interpretations involving energy misreconstruction due to mismodeled nuclear effects [35], mis-estimated



Figure 5. The final MiniBooNE results in neutrino mode in terms of both the timing of the events relative to the beam (right) and the reconstructed radial position of the spherical detector. By studying the excess in terms of additional distributions like these, a better understanding of the excess as well as the backgrounds has begun to emerge. In this example, both the beam timing and radial distributions heavily disfavor an underestimation of the 'Dirt' component normalization being the source of the excess. Reproduced from [33]. CC BY 4.0.

backgrounds [36], as well as other beyond-SM physics [28, 29, 37–40] as the source of the MiniBooNE anomaly. Many of these interpretations are discussed in section 3. On the experimental front, the MicroBooNE experiment was proposed in 2008 to provide a direct test of the MiniBooNE anomaly; MicroBooNE recent results are discussed in section 4.3.

2.3. Reactor experiments

Even in the simplest light sterile neutrino oscillation framework (see section 3.1.1), where one additional, mostly-sterile neutrino mass state is assumed, non-zero $\nu_{\mu} \rightarrow \nu_{e}$ oscillations with 1%-level amplitudes imply that both ν_{μ} and ν_{e} disappearance must occur at short baselines, and at a level that should be observable with current and upcoming reactor experiments, atmospheric neutrino measurements, or measurements carried out with near-only or near+far detectors of long-baseline facilities at accelerators. In apparent consonance with this interpretation, measurements of $\bar{\nu}_{e}$ fluxes performed at short (of order 10–100 m) reactor-detector distances were indeed found to be anomalously low. This energy-integrated flux deficit, observed over a broad range of short baselines, is referred to as the 'reactor antineutrino anomaly' (RAA).

Reactors have played an important role in neutrino physics since their discovery because they are prodigious generators of electron-flavor antineutrinos ($\bar{\nu}_e$). Reactor $\bar{\nu}_e$ are produced from beta decays of neutron-rich fission fragments generated by the heavy fissionable isotopes ²³⁵U, ²³⁸U, ²³⁹Pu, and ²⁴¹Pu. After their production in the reactor core, these MeV-scale $\bar{\nu}_e$ are emitted isotropically. (For a broader introduction to this neutrino source, see the excellent reviews provided in [41–43]).

A typical reactor $\bar{\nu}_e$ spectrum is composed of $\bar{\nu}_e$ produced by hundreds of fission isotopes whose yields and beta decay pathways are sometimes poorly understood. Modeling of this complex neutrino source is thus extremely challenging. For this reason, before discussing anomalies at reactor neutrino experiments, we should briefly overview reactor antineutrino flux modeling methods. Modeling of reactor $\bar{\nu}_e$ spectrum uses two state-of-the-art approaches: the summation or *ab initio* method, and the beta conversion method. Both methods build $\bar{\nu}_e$ predictions for individual fissioning isotopes and aggregate them for a given reactor fuel composition. The summation method uses β -decay information from nuclear databases to first estimate $\bar{\nu}_e$ contributions of individual β -decay branches [44, 45]. Individual beta branch contributions are then summed together, with weights based on fission yields and branching fractions, to obtain a total flux and spectrum for each fissioning isotope. Since nuclear databases are incomplete and sometimes inaccurate [46], the inferred reactor antineutrino spectra have potentially large and ill-defined uncertainties. Nonetheless, significant work is being done in improving the inputs to [47–53] and accuracy of [46] these databases.

The beta conversion method instead sums together beta particle contributions from virtual decay branches that empirically add up to a measured aggregate beta spectrum for each individual fissioning isotope. After defining each virtual branch's contribution in this manner, the individual beta spectrum from each branch can be converted to an $\bar{\nu}_e$ spectrum and summed to generate a model of the aggregate $\bar{\nu}_e$ spectrum specific to the fissioning isotope. The modern converted $\bar{\nu}_e$ spectra by Mueller *et al* [54] and Huber [55] (HM model) are based on the β -decay measurements of the isotopes ²³⁵U, ²³⁹Pu, ²⁴¹Pu performed at the Institut Laue-Langevin (ILL) in the early 1980s [56–58]. An aggregate fission beta spectrum measurement of ²³⁸U was not performed at ILL¹²³, since fission of ²³⁸U required neutrons with energies higher than those available at the thermal ILL facility. Reference [43] article provides an in-depth survey of both these methods.

With modern reactor $\bar{\nu}_e$ experiments performing flux and spectrum measurements with percent-level precision, sources of $\bar{\nu}_e$ not accounted for in these two models play are also important to consider. The $\bar{\nu}_e$ arising from non-equilibrium effects [54, 60] of the beta-decaying isotopes represent one such source that is not included in the conversion method since the β -decay measurements are done on shorter timescales where the off-equilibrium effects do not have enough time to manifest. Reactor and site-specific $\bar{\nu}_e$ are additional such sources that have to be included for each individual experiment separately. These may include $\bar{\nu}_e$ originating from the neutrons interacting with non-fuel reactor elements [61] and spent nuclear fuel [62, 63] placed in close proximity to the detectors.

Reactor neutrino experiments have typically used IBD, $\bar{\nu}_e + p \rightarrow e^+ + n$, as the detection mechanism of choice due to its relatively high cross-section and the substantial background rejection capability made possible by the time-coincident signature it produces-prompt positron energy deposition followed by the delayed spatially-correlated capture of the thermalized neutron. Since neutrons are significantly heavier than e^+ , the positron energies measured by IBD detectors are used as a high-fidelity measure of the interacting neutrino energy. The $\bar{\nu}_e$ spectrum as a function of energy $E_{\bar{\nu}_e}$ measured by a detector using the IBD mechanism is:

$$\frac{\mathrm{d}N}{\mathrm{d}E_{\bar{\nu}_e}}(E_{\bar{\nu}_e}) = \frac{1}{4\pi L^2} \epsilon N_p \,\sigma_{E_{\bar{\nu}_e}} \sum_i \frac{\mathrm{d}S_i(E_{\bar{\nu}_e})}{\mathrm{d}E_{\bar{\nu}_e}},\tag{1}$$

where *L* is the mean detector baseline from the reactor, ϵ is the (typically energy-dependent) efficiency, N_p is the number of target protons, $\sigma_{E_{\bar{\nu}_e}}$ is the IBD cross-section, and $dS_i(E_{\bar{\nu}_e})$ is the antineutrino flux from isotope *i*. Whereas $dS(E_{\bar{\nu}_e})$ decreases with energy, IBD cross section increases with energy. Equation (1) has to be modified if the detector samples neutrinos from multiple reactors, the presence of non-fissioning sources of neutrinos [64], or in the presence of neutrino oscillations [65]. The uncertainties on $dS_i(E_{\bar{\nu}_e})$ and N_p are typically the dominating source of reactor-specific and detector-specific uncertainties.

¹²³ Even though a ²³⁸U β -decay measurement was performed in 2013 [59], the presented $\bar{\nu}_e$ energy had a lower limit of 2.875 MeV, and consequently *ab initio* has been the model of choice for this isotope.

The event rate measured in reactor antineutrino experiments can be expressed in a convenient way as a 'cross section per fission' $\sigma_{f,a}$, often called 'IBD yield':

$$\sigma_{f,a} = \sum_{i} f_i^a \sigma_i, \tag{2}$$

where *a* is the experiment label, σ_i is the IBD yield for the fissionable isotope *i* (with *i* = 235, 238, 239, and 241 for ²³⁵U, ²³⁸U, ²³⁹Pu, and ²⁴¹Pu, respectively), and f_i^a is the effective fission fraction of the isotope *i* for the experiment *a*. For each fissionable isotope *i*, the individual IBD yield is given by

$$\sigma_i = \int_{E_\nu^{\rm transmission}}^{E_\nu^{\rm max}} dE_\nu \, \Phi_i(E_\nu) \, \sigma_{\rm IBD}(E_\nu), \tag{3}$$

where E_{ν} is the neutrino energy, $\Phi_i(E_{\nu})$ is the neutrino flux generated by the fissionable isotope *i*, and $\sigma_{\text{IBD}}(E_{\nu})$ is the detection cross-section. The neutrino energy is integrated from the threshold energy $E_{\nu}^{\text{thr}} = 1.806 \text{ MeV}$. The numerical values of the σ_i 's predicted by a theoretical model depend on the way in which the integral in equation (3) is performed, taking into account that the neutrino fluxes are given in tabulated bins.

The IBD yields $\sigma_{f,a}$ have been measured in a broad array of reactor antineutrino experiments spanning three continents and nearly four decades. A full list of experimental measurements is provided in table 1. Some measurements were performed at compact, highly ²³⁵U-enriched research reactors, while others were performed at high-powered low-enriched commercial core reactors. Experimental reactor-detector baselines in these experiments ranged from less than 10 m to more than 1 km. Implemented IBD interaction detection technologies also varied widely between experiments. In some, IBD neutron detection was enabled using ³He counters, while in others, metal-doped liquid scintillators (⁶Li or Gd) were used. Some efforts used large-volume scintillator regions to detect the prompt IBD positron signal, while others possessed no capability to detect this signal. Despite the broad range of employed technologies, baselines, and reactor types, experiments from the 1980s to the late 2000s were generally deemed to be consistent with state-of-the-art conversion and summation predictions available at that time.

In 2011, new antineutrino flux calculations by Mueller *et al* [54] and Huber [55] using the conversion method for ²³⁵U, ²³⁹Pu, and ²⁴¹Pu and the summation method for ²³⁸U predicted detection rates substantially different than previous estimates. In conjunction with the reduction in the measured neutron lifetime, as well as the inclusion of the off-equilibrium corrections, predicted IBD yields increased, leading to a 5%–6% discrepancy between this need prediction and the average of existing measurements [54]. This discrepancy has come to be known as the 'reactor antineutrino anomaly' (RAA) [1, 66]. Subsequent flux measurements performed using blind analyses in reactor-based θ_{13} experiments following the inception of the RAA arising from historical neutrino measurements being biased toward an agreement with contemporaneous flux predictions.

Figure 6 shows the ratio of the measured to the predicted IBD yields as a function of the distance between the reactors and the detectors. A model involving sterile neutrinos that mix with active neutrinos has been invoked as a potential source for the discrepancy. Under the sterile neutrino hypothesis, a portion of the $\bar{\nu}_e$ from the nuclear reactor oscillate at frequencies of $\Delta m_{new}^2 \sim 1 \text{ eV}^2$ —much higher than the active neutrino oscillation frequencies—into sterile states that go undetected, leading to a deficit in the measurements. Since the oscillation length of eV-scale-mediated oscillations is much shorter than the baselines of most of the measurements in question, the RAA would almost entirely be reflected in a common fixed IBD

	f_{238}^a, f_{239}^a , at in units of f_{238}^a table 6; δ_a^{exp} which is con	nd f_{241}^a are 10^{-43} cm ² / is the total rrelated in 6	the effectiv fission; R_a^e l relative est each group	e fission fra $_{ABM}^{xp}$, $R_{a,EF}^{exp}$, kperimental of experim	actions of the $R_{a,\text{HKSS}}^{\exp}$ and $R_{a,\text{HKSS}}^{\exp}$ at statistical products indicated in the state of	the four ison for $R_{a,KI}^{exp}$, and $R_{a,KI}^{exp}$, applies system ated by the formula of	otopes ²³⁵ U are the rational matic uncer e braces; <i>L</i>	\sum_{a}^{238} U, \sum_{a}^{239} P os of measurtainty, δ_{a}^{cor} tainty, δ_{a}^{cor} is the source	Pu, and ²⁴¹ Pu, ured and pre- is the part of urce-detector	, respectivel dicted rates f the relativ distance. R	ly; $\sigma_{f,a}^{\exp}$ is the s for the IBD e experimenta eproduced fro	experimental yields of the l systematic u m [257]. CC	IBD yield models in ncertainty BY 4.0.
а	Experiment	f^{a}_{235}	f^{a}_{238}	f^{a}_{239}	f^{a}_{241}	$\sigma_{f,a}^{\exp}$	$R_{a,\mathrm{HM}}^{\mathrm{exp}}$	$R_{a,\mathrm{EF}}^{\mathrm{exp}}$	$R_{a, \rm HKSS}^{\rm exp}$	$R_{a,\mathrm{KI}}^{\mathrm{exp}}$	δ^{\exp}_a [%]	$\delta_a^{\rm cor}$ [%]	L_a [m]
1	Bugey-4	0.538	0.078	0.328	0.056	5.75	0.927	0.962	0.916	0.962	1.4	1.4	15
2	Rovno91	0.614	0.074	0.274	0.038	5.85	0.924	0.965	0.914	0.962	2.8		18
3	Rovno88-11	0.607	0.074	0.277	0.042	5.70	0.902	0.941	0.892	0.939	6.4	3.1 2.2	18
4	Rovno88-2I	0.603	0.076	0.276	0.045	5.89	0.931	0.971	0.920	0.969	6.4		17.96
5	Rovno88-1S	0.606	0.074	0.277	0.043	6.04	0.956	0.997	0.945	0.995	7.3		18.15
6	Rovno88-2S	0.557	0.076	0.313	0.054	5.96	0.956	0.994	0.945	0.993	7.3	3.1	25.17
7	Rovno88-3S	0.606	0.074	0.274	0.046	5.83	0.922	0.962	0.911	0.960	6.8		18.18
8	Bugey-3-15	0.538	0.078	0.328	0.056	5.77	0.930	0.966	0.920	0.966	4.2		15
9	Bugey-3-40	0.538	0.078	0.328	0.056	5.81	0.936	0.972	0.926	0.972	4.3	4.0	40
10	Bugey-3-95	0.538	0.078	0.328	0.056	5.35	0.861	0.895	0.852	0.894	15.2		95
11	Gosgen-38	0.619	0.067	0.272	0.042	5.99	0.949	0.992	0.939	0.988	5.4		37.9
12	Gosgen-46	0.584	0.068	0.298	0.050	6.09	0.975	1.016	0.964	1.014	5.4	2.0 3.8	45.9
13	Gosgen-65	0.543	0.070	0.329	0.058	5.62	0.909	0.945	0.899	0.944	6.7		64.7
14	ĬĽL	1.000	0.000	0.000	0.000	5.30	0.787	0.843	0.777	0.827	9.1		8.76
15	Krasnoyarsk87-33	1	0	0	0	6.20	0.920	0.986	0.909	0.967	5.2	4.1	32.8
16	Krasnoyarsk87-92	1	0	0	0	6.30	0.935	1.002	0.924	0.983	20.5		92.3
17	Krasnoyarsk94-57	1	0	0	0	6.26	0.929	0.995	0.918	0.977	4.2	0	57
18	Krasnoyarsk99-34	1	0	0	0	6.39	0.948	1.016	0.937	0.997	3.0	0	34
19	SRP-18	1	0	0	0	6.29	0.934	1.000	0.923	0.982	2.8	0	18.2
20	SRP-24	1	0	0	0	6.73	0.998	1.070	0.987	1.050	2.9	0	23.8
21	Nucifer	0.926	0.008	0.061	0.005	6.67	1.007	1.074	0.995	1.056	10.8	0	7.2
22	Chooz	0.496	0.087	0.351	0.066	6.12	0.990	1.025	0.979	1.027	3.2	0	≈ 1000
23	Palo Verde	0.600	0.070	0.270	0.060	6.25	0.991	1.033	0.980	1.031	5.4	0	≈ 800
24	Daya Bay	0.564	0.076	0.304	0.056	5.94	0.950	0.988	0.939	0.987	1.5	0	\approx 550
25	RENO	0.571	0.073	0.300	0.056	5.85	0.936	0.974	0.925	0.973	2.1	0	≈411
26	Double Chooz	0.520	0.087	0.333	0.060	5.71	0.918	0.952	0.907	0.953	1.1	0	≈415
27	STEREO	1	0	0	0	6.34	0.941	1.008	0.930	0.989	2.5	0	9-11

20

Major Report



Figure 6. Ratio $(R_{a,HM})$ of the measured to the predicted IBD yields as a function of baseline. HM model is used for the predicted IBD yields. Each data point corresponds to an experiment with the error bar representing the experimental uncertainty. The green line and band show the average of $R_{a,HM}$ and average uncertainty respectively. Reproduced from [257]. CC BY 4.0.

yield deficit in all relevant experiments [66]. The lack of an L/E character in this anomaly lends credence to a variety of alternate theories for its existence aside from sterile neutrino oscillations, including flux modeling inaccuracies; an in-depth discussion of possible flux model issues and recent modeling improvements is given in section 3.3.2. The most straightforward way to conclusively affirm a BSM origin for the RAA is to observe its L/Edependence; section 4.1.3 describes recent experimental efforts that probe this behavior via searches for baseline-dependent $\bar{\nu}_e$ energy spectrum distortions.

2.4. Radioactive source experiments

A complementary probe of electron neutrino disappearance to that of reactors is provided by intense radioactive sources producing copious amounts of ν_e , such as those employed by past radiochemical experiments.

Radiochemical experiments were originally designed to detect neutrinos coming from the Sun, making use of a reaction where neutrinos weakly interact with the detector chemical, converting the initial element into a radioactive isotope through the reaction

$$\nu_e + N(A, Z - 1) \to e^- + N(A, Z),$$
(4)

where Z and A are the atomic and mass numbers, respectively. This very same neutrino detection method was implemented by Ray Davis in the Homestake Gold Mine (Lead, SD), using ³⁷Cl [70], which allowed him and his collaborators to successfully detect the neutrinos predicted by the Standard Solar Model (SSM), an observation which led him to win the Nobel Prize in Physics in 2002.

Later on, two other solar neutrino experiments were constructed, GALLEX [71] and SAGE [72], this time using ⁷¹Ga as the detector medium, as suggested initially in [73]. In this case, the interaction of electron neutrinos with the gallium atoms leads to the emission of electrons and the creation of ⁷¹Ge atoms, which are then extracted and counted by means of chemical techniques, giving information about the neutrino flux.



Figure 7. Generic scheme of the radioactive source experiments. The radioactive source $({}^{37}$ Ar or 51 Cr) is located inside the tank containing liquid gallium.

Table 2. Energies, branching ratios and cross sections for the reaction in equation (6), for neutrinos produced by each radioactive source $({}^{51}$ Cr, 37 Ar) decay. Info from [78] and [261]. In particular, the cross section values are extracted by interpolating the calculations of J N Bahcall in [76]. Reprinted (table) with permission from [78], Copyright (2008) by the American Physical Society.

	⁵¹ Cr				³⁷ Ar		
E_{ν} (keV)	747.3	752.1.	427.2	432.0	811	813	
B.R.	0.8163	0.0849	0.0895	0.0093	0.902	0.098	
σ (10 ⁻⁴⁶ cm ²)	60.8	61.5	26.7	27.1	70.1	70.3	

The relevance of the physics being scrutinized (the flux of neutrinos coming from the Sun as a test of the SSM), made it necessary to ensure the detection technique was correctly understood. To do so, the GALLEX and SAGE Collaborations performed experiments under controlled conditions, exposing the detectors to specific neutrino calibration sources. These set of experiments are the so-called 'Gallium radioactive source experiments', which used artificial ⁵¹Cr and ³⁷Ar sources located inside the detectors, as schematically shown in figure 7.

In these experiments, electron neutrinos are emitted during the electron capture decay of the radioactive isotopes in the sources:

$$e^{-} + {}^{51}\mathrm{Cr} \rightarrow {}^{51}\mathrm{V} + \nu_e,$$

$$e^{-} + {}^{37}\mathrm{Ar} \rightarrow {}^{37}\mathrm{Cl} + \nu_e,$$
(5)

with neutrino energies and branching ratios as shown in table 2, and the decay nuclear levels shown in figure 8.

The electron neutrinos interact with the main component of the detectors through the process described by equation (4), which for GALLEX and SAGE becomes:

$$\nu_e + {}^{71}\text{Ga} \rightarrow {}^{71}\text{Ge} + e^-, \tag{6}$$

with the cross sections for each emitted neutrino energy as given in table 2.



Figure 8. Nuclear levels for the ⁵¹Cr and ³⁷Ar radioactive sources decay, according to equation (5). Reprinted (figure) with permission from [72, 75], Copyright (1999, 2006) by the American Physical Society.

Major Report

Table 3. Ratio of predicted and observed 71 Ge event rates as measured by GALLEX (using 51 Cr twice) and SAGE (51 Cr and 37 Ar).

GALLEX	SAGE
$\overline{ 0.953 \pm 0.11 } \\ 0.812^{+0.10}_{-0.11} \\$	$\begin{array}{c} 0.95 \pm 0.12 \\ 0.791 \substack{+0.084 \\ -0.078} \end{array}$

The experimental setup for both GALLEX and SAGE radioactive source experiments were very similar, mainly consisting of a cylindrical tank containing the chemical component acting as the detector (71 Ga), and the radioactive source located inside this tank, as schematically depicted in figure 7.

In order to determine the number of neutrinos produced by the radioactive source and interacting with the detector, the ⁷¹Ge atoms produced by reaction equation (6) are extracted from the gallium by chemical mechanisms and specific cuts are applied to select the events of interest (details of these procedures can be found in [71, 72, 74, 75]), including the relevant information about the ν_e -⁷¹Ga cross section. Uncertainties on this cross section may significantly impact the final result and its interpretation in terms of neutrino oscillations, as discussed in section 3.3.3.

After the counting procedure, the activity of the source is computed and compared to the previously directly measured activity. The ratio between the two numbers is reported in table 3 for the four performed experiments. It is important to note that the cross sections for reactions (equation (6)) used to compute these numbers were the ones calculated by Bahcall in [76] and that, as pointed out in [77], the corresponding uncertainties were not considered. Further investigation of cross-section calculations, and their prospects for providing a 'conventional' explanation for the Gallium Anomaly, is provided in section 4.

As the main purpose of these experiments was to prove the experimental techniques used for the detection of solar neutrinos, the obtained results allowed the two collaborations to conclude that their setup and procedures were well understood and that the solar neutrino measurements—a very large observed deficit on the neutrino flux when compared against the SSM—were not due to any experimental artifact and were highly reliable.

More recently, however, the difference between measured capture rates (table 3) and theoretical calculations was re-examined, especially in light of other indications of anomalous flavor transition from the LSND and MiniBooNE experiments in the 1990s and 2000s. The differences observed during this reiteration established what is known as the Gallium Anomaly. Figure 9 shows the experimental results mentioned above, together with the global average, $R_{\text{avg}} = 0.86 \pm 0.05$, which is ~3 σ less than unity.



Figure 9. Ratio of the observed and the predicted event rates as measured by the different radioactive source experiments GALLEX and SAGE. The shadowed area corresponds to the 1σ region around the weighted average, $R_{\rm avg} = 0.86 \pm 0.05$.

This discrepancy is usually interpreted as an anomalous disappearance of electron neutrinos trough short-baseline oscillations to sterile neutrinos in a framework of 3 + 1 mixing neutrinos. In the scheme in which one sterile neutrino at the eV mass scale is added to the standard three-neutrino framework, the survival probability of electron (anti) neutrinos is

$$P(\nu_e \to \nu_e) = 1 - 4 |U_{e4}|^2 (1 - |U_{e4}|^2) \sin^2\left(\frac{\Delta m_{41}^2 L}{4E}\right),\tag{7}$$

where *L* is the distance from the source to the detector, *E* is the neutrino energy, *U* is the 4 × 4 PNMS mixing matrix, and $\Delta m_{41}^2 = m_4^2 - m_1^2$, is the squared-mass difference between the heavy (mostly sterile) neutrino ν_4 and the (standard) light neutrino ν_1 (considering that $\Delta m_{41}^2 \approx \Delta m_{42}^2 \approx \Delta m_{43}^2$).

This model is implemented, for instance, in [78], where studies of the GALLEX and SAGE results (with the cross sections listed on table 2) revealed a possible indication of electron neutrino disappearance due to neutrino oscillations with $\Delta m_{41}^2 \gtrsim 0.1 \text{ eV}^2$, as the contour plots in figure 10 show.

The relevance of these results has led the neutrino community to search for alternative explanations, such as possible effects arising from cross section uncertainties, not considered in the analysis leading to the contours in figure 10 (described in section 3.3.3), and to perform new experiments (e.g. BEST, described in section 4.1.5) to test the deficit of electron neutrinos observed by the gallium radioactive experiments as described here.

The following section provides details on a diverse range of viable interpretations for these anomalies, from modifications of three-flavor neutrino mixing to potential couplings to hidden sectors.

3. Interpretations of the anomalies

In this section, we discuss the theoretical interpretations of the experimental anomalies discussed above. While seemingly compatible when presented within the empirical picture of two-neutrino oscillations, the underlying source of the anomalies may or may not be connected. In what follows, we describe three different categories of resolutions put forth in the literature, including those that can explain some—but not necessarily all—of the anomalies.



Figure 10. Allowed regions in the $\Delta m^2 - \sin^2 2\theta$ parameter space obtained from the combination of the GALLEX and SAGE radioactive source experiments. Reprinted (figure) with permission from [78], Copyright (2008) by the American Physical Society. Notice that $\sin^2 2\theta = 4 |U_{e4}|^2 (1 - |U_{e4}|^2)$.

3.1. Flavor conversion

In this subsection, we discuss flavor-conversion-based explanations to the short-baseline anomalies discussed in section 2.

We begin by discussing perhaps the simplest model that leads to flavor change in short baselines: an extension of the ESM by the inclusion of a new light sterile neutrino, referred to as the well-studied 3 + 1 sterile neutrino model. In the most simple realization of the 3 + 1 model, the sterile neutrino has no gauge interactions. It should be noted that, historically, experimental collaborations such as LSND and MiniBooNE have analyzed their data sets primarily under a two-neutrino oscillation hypothesis most closely represented by the 3 + 1 model. Additionally, in this model, provided that the two-neutrino oscillation approximation is valid, no observable CP violation effects are expected. Therefore, light sterile neutrino and antineutrino oscillation searches are effectively sensitive to the same oscillation parameters. Finally, we prepare the reader in advance, in that, the 3 + 1 model has been shown to provide an insufficient description to the globally available experimental data that have sensitivity to its observable effects. Nevertheless, it has and will likely continue to be instructively used within the community as a simple 'measure' for developing, optimizing, and comparing sensitivities of various experimental searches and comparing compatibility of different experimental results, albeit with several caveats.

After discussing this simple model in some detail, we then consider extensions and variations to this model, all of which can lead to flavor transitions at short baselines. The most straightforward extension beyond 3 + 1 is represented by the 3+N model, where N = 2, 3, ... light sterile neutrinos are introduced and associated with similarly light neutrino mass states.

Other extensions often introduce non-standard interactions (NSIs) and neutrino propagation effects.

3.1.1. 3 + 1 light sterile neutrino oscillations

3.1.1.1. Sterile neutrinos. Sterile neutrinos provide one of the simplest extensions of the SM that explain the non-zero mass of neutrinos. The right-handed, gauge-singlet fields $(N)_R$, sometimes denoted as $(\nu_s)_R$, provide the missing chiral partners for the left-handed, interacting SM neutrino fields $(\nu_\alpha)_L$. Neutrinos would then acquire a mass just like any other SM fermion via the Higgs mechanism, $m_D \overline{\nu_\alpha} N$, with $m_D = yv/\sqrt{2}$ and v the Higgs vaccum expectation value. While this observation is sufficient to resolve the puzzle of neutrino masses, it raises new questions. In particular, since N_R would carry no charges under any SM gauge symmetries, it could also have a Majorana mass, $MN_R^c N_R$. In that case, the physical spectrum could contain the light, mostly-active neutrinos, as well as potentially heavier, mostly-sterile neutrinos that interact very weakly through mixing. Note that since Majorana masses violate any one of the quantum numbers associated with N_R , it would also indicate that lepton number is violated by two units. Conversely, if lepton number is conserved, then Majorana masses are not allowed, and neutrinos are purely Dirac particles, like any other SM fermion.

Most importantly, Majorana masses need not be related to the electroweak scale. In principle, can take any value up to the Planck scale or the scale of gauge unification. If it takes values larger than the electroweak scale or, more precisely, the scale of Dirac neutrino masses, it would trigger the canonical Type-I seesaw mechanism [6, 8–10, 12, 13, 15, 79, 80]. The seesaw Lagrangian reads,

$$\mathcal{L}_{\nu-\text{mass}} \supset -Y_{\alpha i} \overline{L_{\alpha}} \widetilde{H} N_j - \frac{M_{ij}}{2} \overline{N_i^c} N_j + \text{h.c.},$$
(8)

where $\widetilde{H} = i\sigma_2 H^*$ is the conjugate of the Higgs doublet and L is the lepton doublet. After electroweak symmetry breaking,

$$M_{\nu} \simeq -\frac{\nu^2}{2} Y^T M^{-1} Y, \tag{9}$$

where M_{ν} is the mass matrix for the light, mostly-active neutrinos. In its simplest realization, the seesaw mechanism explains the smallness of the neutrino masses using a hierarchy of scales between Dirac and Majorana masses. Mixing between the heavy neutrinos and the active SM neutrinos, $|U_{\alpha i}|$, is typically of order $\mathcal{O}(M_D/M)$.

Other variations of the Type-I seesaw mechanism exist, including low-scale models where the lightness of neutrino masses is explained instead by the approximate conservation of lepton number. Seesaw models with pairs of heavy neutral leptons, N and S, with opposite lepton number, are often called extended seesaws. One of which, the inverse seesaw, has these particles form pseudo-Dirac pairs with a small mass splitting given by μ , a lepton-number-violating parameter. Light neutrino masses, in this case, are proportional to μ , which in the limit $\mu \rightarrow 0$, parametrically recovers lepton number conservation and massless neutrinos.

In extended seesaws, it is also possible that some number of the sterile neutrinos remain relatively light [81]. This may be due to cancellations, new symmetries, or because the number of fields exceeds the number of large scales in the theory. In such cases, heavy neutrinos can 'seesaw' not only the light, mostly-active neutrinos but also some of the sterile neutrinos, rendering them light as well. These models predict the existence of light sterile neutrinos that mix with active neutrinos with large mixing angles. Finally, we note that while

the seesaw mechanism provides an elegant solution to the smallness of neutrino masses, we cannot rule out the possibility that lepton number is indeed conserved and that neutrinos are Dirac A fourth neutrino, in this case, would not necessarily be related to the origin of neutrino masses but could exist based on a new pair of left and right-handed singlet fermions.

Sterile neutrinos can also solve other open problems in the SM. Heavy neutrinos above the electroweak scale can generate the observed baryon asymmetry of the Universe via their decays or oscillations. In this scenario, the lepton asymmetry generated by CP violation and out-of-equilibrium processes involving heavy neutrinos is converted into a baryon asymmetry via Sphalerons, non-perturbative SM processes. This scenario is referred to as Leptogenesis [82]. In addition, light sterile neutrinos with masses of $\mathcal{O}(1-100)$ keV can provide a warm dark matter candidate [83–87], produced out-of-equilibrium by oscillations in the early Universe. It is also possible that both of these issues are addressed by a whole new sector of sterile neutrinos, such as in the proposed ν -minimal SM [88, 89]

In this section, we discuss short-baseline oscillations generated by sterile neutrinos with masses of order $\mathcal{O}(1-10)$ eV. These sterile neutrinos could very well play the role of the seesaw partners, albeit triggering the mechanism at a relatively low scale [90, 91]. Unfortunately, they would not provide direct evidence for Leptogenesis or sterile-neutrino dark matter. Still, their discovery would be the first laboratory observation of a particle beyond the SM and strongly motivate sterile-neutrino solutions to all open problems in particle physics. Studying their properties and looking for potential lower and higher-scale partners would be of great importance in this case.

3.1.1.2. 3 + 1 oscillation probabilities. In the simplest 3 + 1 model, the standard neutrino sector is extended by an extra neutrino flavor ν_s which is a gauge singlet and does not experience weak interactions. The three neutrino flavors and the sterile neutrino are admixtures of four neutrino mass eigenstates, where m_4 is assumed to be of order ~ 1 eV, motivated by the LSND observation. Parametrically, one can extend the 3×3 leptonic mixing matrix to a 4×4 matrix $U_{\alpha i}$, with $\alpha = e, \mu, \tau, s$ and i = 1, ..., 4. Notice, however, that the last row of this extended matrix is not related to experimental observables as it pertains to the amount of sterile neutrino flavor in different beams. This matrix can be parametrized by the usual three mixing angles and CP violation phase, plus three extra mixing angles and two extra CP phases.

To avoid parametrization dependence, it is often helpful to work with the mixing matrix elements U_{e4} , $U_{\mu4}$, and $U_{\tau4}$ directly. We assume that the fourth mass eigenstate is mostly sterile and much heavier than the other ones, so that $\Delta m_{41}^2 \gg |\Delta m_{31}^2|$, Δm_{21}^2 , allowing for the approximation that Δm_{31}^2 and Δm_{21}^2 are degenerate and at zero. Furthermore, this new, large mass splitting allows for short-baseline neutrino oscillations. In the limit where oscillations due to Δm_{31}^2 and Δm_{21}^2 —the atmospheric and solar mass-squared splittings, respectively—are negligible, short-baseline oscillations can be approximated by

$$P(\nu_e \to \nu_e) \simeq 1 - 4 |U_{e4}|^2 (1 - |U_{e4}|^2) \sin^2\left(\frac{\Delta m_{41}^2 L}{4E_\nu}\right) \equiv 1 - \sin^2\left(2\theta_{ee}\right) \sin^2\left(\frac{\Delta m_{41}^2 L}{4E_\nu}\right),$$
(10)

$$P(\nu_{\mu} \to \nu_{\mu}) \simeq 1 - 4 |U_{\mu4}|^2 (1 - |U_{\mu4}|^2) \sin^2\left(\frac{\Delta m_{41}^2 L}{4E_{\nu}}\right) \equiv 1 - \sin^2\left(2\theta_{\mu\mu}\right) \sin^2\left(\frac{\Delta m_{41}^2 L}{4E_{\nu}}\right),$$
(11)

J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501

$$P(\nu_{\mu} \to \nu_{e}) \simeq 4 |U_{\mu4}|^{2} |U_{e4}|^{2} \sin^{2} \left(\frac{\Delta m_{41}^{2} L}{4E_{\nu}}\right) \equiv \sin^{2} \left(2\theta_{\mu e}\right) \sin^{2} \left(\frac{\Delta m_{41}^{2} L}{4E_{\nu}}\right).$$
(12)

Note that the above makes no explicit assumption about $U_{\tau 4}$ (or, consequently, U_{s4}); however, by unitarity considerations, $|U_{e4}|^2 + |U_{\mu 4}|^2 + |U_{\tau 4}|^2 + |U_{s4}|^2 = 1$. Note also that in the above, we have focused explicitly on electron- and muon-neutrino flavors, given that these are the transition channels that have been studied most extensively and where the shortbaseline anomalies occur.

The above three oscillation probabilities further define the effective mixing angles $4 |U_{e4}|^2 (1 - |U_{e4}|^2) = \sin^2 2\theta_{ee},$ $\sin^2(2\theta_{\alpha\beta})$ often used in the literature: $4 |U_{\mu4}|^2 (1 - |U_{\mu4}|^2) = \sin^2 2\theta_{\mu\mu}$, and $4 |U_{\mu4}|^2 |U_{e4}|^2 = \sin^2 2\theta_{\mu e}$. The two new CP violating phases from the extended 4 × 4 mixing matrix do not lead to observable effects unless effects from both Δm_{41}^2 and either Δm_{21}^2 or Δm_{31}^2 are simultaneously relevant. In scenarios with more than one light sterile neutrino (see section 3.1.3), CP-violating phases associated with the (3 + N) \times (3 + N) mixing matrix may be accessible. Returning to the 3 + 1 scenario, oscillations are relevant for $L/E_{\nu} \approx m/MeV$ or km/GeV if $\Delta m_{41}^2 \approx 1 \text{ eV}^2$. Most notably, the relationships among equations (10–12) imply that, if both $|U_{e4}|^2$ and $|U_{\mu4}|^2$ are nonzero, then electron-neutrino disappearance, muon-neutrino disappearance, and muon-to-electron-neutrino appearance must all occur at the same L/E_{ν} . More explicitly, the oscillation amplitudes for appearance and disappearance are related by $\sin^2 2\theta_{\mu e} \leq 1/4 \sin^2 2\theta_{\mu \mu} \sin^2 2\theta_{ee}$. This relation allows for combinations of experiments to over-constrain the 3 + 1 model, a feature that global fits take advantage of when performing combined analyses to experimental data sets on neutrino appearance and disappearance.

Finally, when an explicit parametrization of the 4 × 4 unitary mixing matrix U is needed for oscillations, six rotation angles and three CP phases are required. In addition to the threeneutrino mixing parameters, three new angles, θ_{14} , θ_{24} , and θ_{34} , and two new phases, δ_{24} and δ_{14} , are defined, in accordance with [92]. The full mixing matrix is then given by

$$U_{4\times4} = R_{34}S_{24}S_{14}R_{23}S_{13}R_{12}, \tag{13}$$

where R_{ij} is a real matrix of rotation by an angle θ_{ij} , and S_{ij} is a complex matrix of rotation by θ_{ij} with a CP phase of δ_{ij} . The relationship between this parametrization to the effective mixing angles as well as to the full matrix elements is shown in table 4.

3.1.1.3. Global analysis of 3 + 1 oscillations. The present status of the 3 + 1 model is best examined through the lens of a global analysis. This allows each of the myriad of shortbaseline experiments to contribute to a single statistical model according to the strength of their results. Global fits have been performed independently by several groups (see, e.g. [93–106]). While all groups find a strong preference for a 3 + 1 model compared to the SM, driven mainly by LSND and MiniBooNE, a significant tension among data sets is also consistently found. The tension lies in a simple fact: large enough mixings required to explain the LSND and MiniBooNE anomalies simultaneously lead to a large disappearance amplitude, particularly of muon neutrinos, and this is in tension with $\nu_{\mu} \rightarrow \nu_{\mu}$ data, which provide strong limits on the value of $\sin^2 2\theta_{\mu\mu}$.

Figure 11 shows the preferred region in the 3 + 1 model parameter space of several short-baseline appearance experiments, including the combination of all of them, at 99% CL for two degrees of freedom (left panel), as well as, the regions preferred by all short-baseline appearance experiments (right panel, red region) compared to the excluded region by all disappearance experiments (blue line) at 99.73% CL for two degrees of freedom [102]. Note

Table 4. Relation between the different parametrizations of neutrino mixing parameters in a 3 + 1 model. Modified from [103].

Effective angle	Full $3 + 1$ model rotation angles	Mixing elements
$\sin^2 2\theta_{ee}$	$=\sin^2 2\theta_{14}$	$= 4(1 - U_{e4} ^2) U_{e4} ^2$
$\sin^2 2 heta_{\mu\mu}$	$= 4\cos^2\theta_{14}\sin^2\theta_{24}(1 - \cos^2\theta_{14}\sin^2\theta_{24})$	$=4(1- U_{\mu4} ^2) U_{\mu4} ^2$
$\sin^2 2 heta_{ au au}$	$= 4\cos^2\theta_{14}\cos^2\theta_{24}\sin^2\theta_{34}(1 - \cos^2\theta_{14}\cos^2\theta_{24}\sin^2\theta_{34})$	$= 4(1 - U_{\tau 4} ^2) U_{\tau 4} ^2$
$\sin^2 2\theta_{e\mu}$	$=\sin^2 2\theta_{14}\sin^2\theta_{24}$	$=4 U_{\mu4} ^2 U_{e4} ^2$
$\sin^2 2\theta_{e au}$	$=\sin^2 2\theta_{14}\cos^2\theta_{24}\sin^2\theta_{34}$	$=4 U_{e4} ^2 U_{ au 4} ^2$
$\sin^2 2 heta_{\mu au}$	$=\sin^2 2\theta_{24}\cos^4\theta_{14}\sin^2\theta_{34}$	$= 4 U_{\mu 4} ^2 U_{\tau 4} ^2$

29



Figure 11. Left: preferred regions by several $\nu_{\mu} \rightarrow \nu_{e}$ appearance experiments in the 3 + 1 scenario at 99% CL for 2 degrees of freedom. Right: preferred region of short-baseline appearance experiments (red region), compared to the region excluded by disappearance experiments (blue line) at 99.73% CL for 2 degrees of freedom. Reproduced from [102]. CC BY 4.0.

that the region to the right of the blue line is excluded, and that encompasses the entirety of the appearance allowed region. While the left panel clearly indicates a strong preference for the 3 + 1 model over the usual three neutrino framework, the right panel clearly shows the tension between appearance and disappearance data: the appearance and the disappearance 99.73% CL preferred regions are disjoint.

One can be more precise and quantify the amount of tension using the parameter goodness-of-fit (PG) test [107], which compares the minimum chi-square values of the full data set to the sum of minimum values of the individual data sets, that is

$$\chi^2_{\rm PG} \equiv \chi^2_{\rm global} - \chi^2_{\rm app} - \chi^2_{\rm dis}.$$
 (14)

In [102], for example, this tension is found to yield a *p*-value of 3.7×10^{-7} when assuming that the χ^2_{PG} follows a chi-square distribution. Moreover, removing any individual null experiment from the fit does not lead to significant improvements in the *p*-value, evidencing that the tension is robust. Similar conclusions have been drawn by other global fits [103–106].

While this demonstrates the shortcomings of the 3 + 1 scenario as an explanation of the short-baseline anomalies, there are important caveats that should be highlighted. First, all global fits to date have been performed assuming the validity of Wilks' theorem [108]. While in many cases, Wilks' theorem is valid and, therefore, the test statistic follows a chi-square distribution, this is not obvious in neutrino oscillation experiments. Reference [109] has shown that these considerations are relevant for the interpretation of short-baseline reactor experiments in terms of sterile neutrinos, and the assumption of Wilks' theorem can have a significant quantitative impact on the statistical interpretation of the anomaly. Not only would this change the CL of the allowed regions, but it would also affect the outcome of the parameter goodness-of-fit test, and thus the 'amount of tension' between appearance and disappearance data. Therefore, the *p*-values quoted above should be taken with a grain of salt.

One example of the importance of the statistical treatment is shown in the left panel of figure 12 in which a Bayesian analysis is compared to the outcome of a frequentist one [103]. For the Bayesian analysis, the translucent black, red and blue regions represent the 68%, 90%, and 99% highest posterior density credible regions, respectively, while the brighter colors



Figure 12. Left: 3 + 1 global confidence regions (solid) are compared to Bayesian highest posterior density credible regions (68%, 90% and 99% credible regions in black, red and blue). Right: appearance-only (below red line) confidence regions compared to disappearance-only (above red line), with $\Delta m_{41}^2 > 1 \text{ eV}^2$ appearance-only regions shown in hatched purple. All confidence regions are shown in red, green, and blue for 90%, 95%, and 99%. Reprinted from [103], Copyright (2020), with permission from Elsevier.

refer to the frequentist analysis. The allowed regions are quite different, which shows how the statistical treatment can affect the identification of promising parameter space and the statement of the tension between data sets. Another example can be found in the right panel of figure 12, where the allowed regions for the appearance (below the red line) and disappearance (above the red line) are shown [103]. If a $\Delta m_{41}^2 > 1 \text{ eV}^2$ cut is applied to the appearance parameter space, the 99% confidence region moves to the hatched purple region. The points within this region are still preferred with respect to the null hypothesis at the 99% level and, additionally, overlap with the disappearance regions, despite being disfavored when compared to the best-fit point in the 3 + 1 scenario.

Besides all that, since reactor experiments drive the preference for nonzero U_{e4} mixing, there is an important caveat with respect to the reactor anomaly that should be discussed. The reactor anomaly has originated in a discrepancy between data and theoretical expectations based on the calculations of reactor antineutrino fluxes. Nevertheless, a large, unexpected feature in the flux around 5 MeV has been identified, the so-called 5 MeV bump. This outstanding feature lies outside the proposed theoretical uncertainties and puts in question the anomaly itself. Flux ratios can be used to mitigate the impact of the 5 MeV bump [110] but at the price of reducing the statistical power of the analysis. Therefore, more precise calculations of the reactor anomaly. A detailed discussion of this issue can be found in 4.3.2.

3.1.2. 3 + 1 light sterile neutrino oscillations and decoherence. When considering the 3 + 1 model discussed above, one has assumed that neutrinos are always coherent. However, as pointed out in [111], due to the lack of detailed calculations of the neutrino production and detection mechanism, or from additional BSM effects, this is not guaranteed. In this section, we follow the discussion given in [111] and point out that when interpreting experimental data in the context of 3 + 1 this possibility has been overlooked. This fact could partially or completely resolve the existing tension between appearance and disappearance data sets.

Currently, when deriving constraints or preferred regions on the 3 + 1 model, the experiments assume that the neutrino state is a plane wave. It is well-known that the plane-



Figure 13. Overview of the solar potential, neutrino experiments, and relevant scales. L^{osc} (dotted gray and dashed pink) and L^{coh} (dashed blue) are computed from equation (16) using $\Delta m_{41}^2 = 1 \text{ eV}^2$ and $\sigma_x = 2.1 \times 10^{-4}$ nm for $L_{\text{ste}}^{\text{coh,nuc}}$ [129], and $\sigma_x = 10^{-11}$ m for $L_{\text{ste}}^{\text{coh,\pi}}$ flight [643]. Decoherence effects are expected at $L \gtrsim L^{\text{coh}}$. Reproduced from [111]. CC BY 4.0.

wave (PW) theory of neutrino oscillations [112–114] is a simplified framework that, upon careful inspection, contains apparent paradoxes [115–117]. These can be resolved by introducing the wave packet (WP) formalism [118–123]. The applicability of the PW approximation has been studied in detail for the standard mass-squared differences [117, 121, 124–126] and has been shown to be a good approximation for current and future neutrino experiments. However, this has not been shown to be the case for mass-squared differences relevant to the LSND observation.

In the WP formalism, the oscillation probability is given by

$$P_{\alpha\beta} = \sum_{i=1}^{n} |U_{\alpha i}|^{2} |U_{\beta i}|^{2} + 2\operatorname{Re} \sum_{j>i} U_{\alpha i} U_{\alpha j}^{*} U_{\beta i}^{*} U_{\beta j} \exp\left\{-2\pi i \frac{L}{L_{\text{osc}}^{ij}} - 2\pi^{2} \left(\frac{\sigma_{x}}{L_{\text{osc}}^{ij}}\right)^{2} - \left(\frac{L}{L_{\text{coh}}^{ij}}\right)^{2}\right\},$$
(15)

where $U_{\alpha i}$ are the neutrino mixing matrix elements and *L* the experiment baseline. Here the damping of the oscillations is parametrized by a length scale σ_x that can be referred to as the wave packet size [118, 123–127] and depends on the neutrino production and detection mechanisms. These lengths are defined as

$$L_{\rm osc}^{ij} = \frac{4\pi E}{\Delta m_{ii}^2}$$
 and $L_{\rm coh}^{ij} = \frac{4\sqrt{2}E^2\sigma_x}{\Delta m_{ii}^2}$, (16)

the oscillation and coherence lengths, respectively. Here, *E* is the energy of the neutrino and Δm_{ji}^2 the mass-squared difference between the ν_j , ν_i mass eigenstates. The two last terms in the exponential of equation (15) smear the oscillation.

Most experiments fulfill $\sigma_x \ll L_{osc}^{ij}$ and thus the first dampening term can be neglected. This is not the case for the second one, which describes the decoherence arising from the



Figure 14. Impact of the smearing. On the left, the *y*-axis represents the ratio between the 3 + 1 and the 3 expected events in the NEOS experiment, for the reactor antineutrino anomaly (RAA) best-fit parameters [133]: $\Delta m_{41}^2 = 2.32 \text{ eV}^2$ and $\sin^2 2\theta_{14} = 0.14$. On the right, the solid pink and solid blue contours bound the exclusion region from Daya Bay, NEOS and PROSPECT; at two sigma for the PW approximation and WP formalism, respectively. The preferred region at two sigma for the BEST experiment is shaded for the PW approximation (pink) and the WP formalism (blue). Both figures are obtained for $\sigma_x = 2.1 \times 10^{-4}$ nm. Reproduced from [111]. CC BY 4.0.

separation of the mass eigenstates during their propagation at different velocities. Notice that under a stationary situation, all the relevant information should be in the energy spectrum. Therefore, this effect can be equivalently understood as an additional quantum uncertainty in the measurement of the true neutrino energy by the experiment [128].

Figure 13 shows several oscillation experiments compared to the sterile oscillation scale (L_{ste}^{osc}) and the decoherence scale (L_{ste}^{coh}) . For experiments with baselines smaller than L_{ste}^{coh} , decoherence can be neglected, while experiments with larger baselines will experience complete decoherence. Take into account that the effect of not resolving fast oscillations experimentally is from an observational point of view identical to a decoherence effect. Consequently, an experiment far above the L_{ste}^{osc} line would also be effectively decoherent, and no effect due to L_{ste}^{coh} would be manifest. This narrows the region of interest for the decoherence of light sterile neutrinos to the low-energy region and in particular to the reactor and radioactive sources experiments.

To show the impact of the wave packet separation [111] chooses the smallest value allowed by oscillation experiments for the wave packet size, $\sigma_x = 2.1 \times 10^{-4}$ nm [129, 130], and performs analyses searching for sterile neutrinos with and without the PW approximation. Notice that this value is far from some first-order estimations of the wave packet in various contexts [118, 120, 122, 123]; however, [111] decided to be agnostic and use an experimental result that should be robust even in more exotic scenarios. The global analysis performed in [111] considers the null results from Daya Bay [131, 132], NEOS [133], and PROSPECT [134, 135] and the anomalous results observed from radioactive sources by BEST [136].

The main result of [111] is presented in figure 14, which shows the two-sigma exclusion contours for these experiments and the positive hint regions at two sigma by BEST, in both

formalisms. The WP results become compatible not only at large values of Δm_{41}^2 but also at the region around $\Delta m_{41}^2 = 2 \text{ eV}^2$.

Reference [111] finds that the damping of the oscillations due to the wave packet size may have important consequences for low-energy light sterile neutrino searches, accommodating apparently contradictory results. The result strongly motivates further studies to improve our understanding of the physics involved in the production and detection of the nuclear reactor and radioactive source neutrino experiments.

3.1.3. 3+N light sterile neutrinos. The generic model with three active and N sterile neutrino states can be considered a viable explanation of the anomaly seen experimentally. But it also provides a sterile-sector model-independent framework for non-unitarity [137, 138] tests.

We define the unitary mixing matrix U in the whole $(3 + N) \times (3 + N)$ space, and denote its 3 × 3 active space sub-matrix as N. Then, the probability of active neutrino oscillation $P(\nu_{\beta} \rightarrow \nu_{\alpha})$ in matter can be written in the simple form as [138]

$$P(\nu_{\beta} \to \nu_{\alpha}) = C_{\alpha\beta} + \left| \sum_{j=1}^{3} N_{\alpha j} N_{\beta j}^{*} \right|^{2} - 2 \sum_{j \neq k} \operatorname{Re}[(NX)_{\alpha j} (NX)_{\beta j}^{*} (NX)_{\alpha k}^{*} (NX)_{\beta k}] \times \sin^{2} \frac{(h_{k} - h_{j})x}{2} - \sum_{j \neq k} \operatorname{Im}[(NX)_{\alpha j} (NX)_{\beta j}^{*} (NX)_{\alpha k}^{*} (NX)_{\beta k}] \sin(h_{k} - h_{j})x,$$

$$(17)$$

where α , $\beta = e$, μ , τ denote the active neutrino flavor indices, *i*, *j*, *k* = 1, 2, 3 are the indices for the light mass eigenstates, and all oscillations involving heavier mass eigenstates with $m_j^2 \gtrsim 0.1 \text{ eV}^2$, which are dominantly sterile, are averaged out.

In equation (17), $P(\nu_{\beta} \rightarrow \nu_{\alpha})$ is the leading term in a expansion by the active-sterile transition sub-matrix *W* in U [138]. The zeroth order Hamiltonian contains the 3 × 3 active space sub-matrix with the kinetic term plus matter potential given by diag(a - b, -b, -b), where a(b) denotes the Wolfenstein matter potential due to CC (NC) interactions, with a decoupled $N \times N$ sterile block. Here, h_i (i = 1, 2, 3) denote the energy eigenvalues of zeroth-order active states and *X* is the unitary matrix which diagonalizes the zeroth-order active part Hamiltonian. $C_{\alpha\beta}$ in equation (17) is a probability leaking term which takes the same form in vacuum and in matter [137, 138] as

$$C_{\alpha\beta} \equiv \sum_{J=4}^{3+N} |W_{\alpha J}|^2 |W_{\beta J}|^2.$$
(18)

The probability leaking and W^2 correction terms—in contrast to high-scale unitarity violation, observation of $C_{\alpha\beta}$ in equation (18) would testify for low-scale unitarity violation. Unfortunately, a detailed study of the sensitivity to high-scale unitarity, namely the constraints on $C_{\alpha\beta}$, has only been performed for JUNO [137]; see figure 15.

Another unique prediction of the 3+N model with low-mass sterile neutrinos is the presence of higher-order W^2 corrections. These can be in principle measured and provide a way to differentiate this scenario from high-scale unitarity violation, where the mostly-sterile mass states are assumed to be kinematically forbidden and do not participate in neutrino oscillations. The term is explicitly evaluated and plotted in figure 16 [138]. Notice the peculiar energy- and zenith-angle dependence of the term shown in figure 16. The relevant energy region of $\rho E = 50-1000$ (g cm⁻³) GeV may be explored by beam or atmospheric neutrino experiments; for example, Super-K, Hyper-K/HKK, DUNE, IceCube, or KM3NeT-ORCA, can probe this parameter space using low- to high-energy observables.


Figure 15. Constraint in $\sum_{i=1}^{3} |U_{ei}|^2 - C_{ee}$ space at 1σ , 2σ and 3σ CL expected to be obtained by JUNO like setting assuming the flux uncertainty of 3% (left panel) and 6% (right panel). Reproduced from [137]. CC BY 4.0.

Finally, another marked difference between low- and high-scale unitarity violation is that there is no deviation in the production and detection cross sections of neutrino. This is because in the low-scale scenario both CC and NC vertices are not modified since all mass eigenstates, including the mostly-sterile states, are kinematically allowed in the processes.

3.1.4. Light sterile neutrino oscillations and NSIs. While many attempted solutions to the hints of anomalous $\nu_{\mu} \rightarrow \nu_{e}$ appearance at LSND and MiniBooNE have focused on those experiments, it is also conceivable to see if the strong constraints from MINOS/MINOS+ [139] and IceCube [140] could be alleviated. As these constraints are at larger energies and over longer baselines, they would be subject to modification by non-standard neutrino interactions [141] in either the active or sterile sector. Thus a scenario with the usual sterile neutrinos explaining the short-baseline accelerator hints along with a new matter effect to modify the imprint of that sterile neutrino in the long-baseline accelerator and atmospheric constraints could be consistent with all the data. This was investigated in [142-145], which found that, while it could be possible to simultaneously explain some of the data sets in this fashion, explaining all seems to be impossible, even with both a sterile neutrino and a new interaction. In particular, [145] found that a model with a new interaction between sterile neutrinos and baryons provides an excellent fit to LSND, MiniBooNE, and IceCube data; but cannot simultaneously fit MINOS+ data due to a disagreement in the preferred values of θ_{34} . Other approaches, such as [146], used beam neutrinos forward scattering off of a locally overdense relic neutrino background to give rise to a matter effect with an energy-specific resonance that can reproduce the MiniBooNE observed excess.

More recently, [147] reiterated that these tensions with long-baseline experiments occur because new matter effects generically distort the active (anti-)neutrino mixing and mass spectrum. A dark sector model with both neutrino and vector portals was then proposed in [147] that avoids these large active spectrum distortions and is fully compatible with longbaseline experiments, including T2K, NOvA, MINOS/MINOS+, IceCube/DeepCore, and KamLAND. In this model, quasi-sterile neutrinos from a dark sector are charged under a light vector mediator with feeble couplings to SM fermions. This leads to new matter effects that generate resonant active-to-quasi-sterile neutrino oscillations within a narrow window of energy, ~250–350 MeV, to explain the MiniBooNE low energy excess. The MiniBooNE excess at mid-to-high energies, $E_{\nu} \gtrsim 400$ MeV, as well as the LSND and Gallium anomalies,



Figure 16. The order W^2 correction terms, $\delta P(\nu_{\mu} \rightarrow \nu_{\alpha}) \equiv C_{\mu\alpha} + P(\nu_{\mu} \rightarrow \nu_{\alpha})^{(2)}$, are shown as a function of the distance traveled by neutrinos in the Earth assuming a common $m_f^2 = 0.1 \text{ eV}^2$. The top, middle and bottom panels are for $\alpha = e, \tau$, and μ , respectively. In each panel the three cases are shown: N = 1 with maximal $C_{\mu\alpha}$ (solid line), the universal scaling model with N = 3 (see [138], dotted line), and the order W^2 correction terms only (dashed line). The blue (red) lines are for E = 10 (100) GeV, and the leaking term is taken as $(C_{e\mu}, C_{\tau\mu}, C_{\mu\mu}) = (20, 95, 9.6) \times 10^{-5}$ for N = 1. Reproduced from [138]. CC BY 4.0.

are explained by an additional 'vanilla' sterile neutrino which does not participate in the resonant oscillations. Besides being fully testable by the SBN program, the new matter effects in this model have interesting implications for solar neutrinos.

3.1.5. Decaying light sterile neutrinos. As discussed above, global fits to the neutrino data show that the 3 + 1 sterile neutrino model suffers from internal inconsistency amongst the datasets [103, 107]. This 'tension' in the global fits motivates considering more complicated physics scenarios. More complicated physics scenarios could involve neutrino decay. Neutrinos are not protected from decay in the SM, i.e. radiative neutrino decay of the two

heavier of the three known neutrino mass states (ν_1 , ν_2 , and ν_3) can occur, albeit, extremely slowly [148, 149]. Scenarios that include decay of the eV-scale neutrino mass state, ν_4 , are referred to as "3 + 1+decay."

The class of 3 + 1+decay models can be divided into two scenarios: visible decay and invisible decay. In the visible decay scenario, one of the decay daughters is an active neutrino, which could be detected, while the other is undetectable, Beyond the Standard Model (BSM). In the invisible decay scenarios, all decay daughters are BSM and invisible, i.e. undetectable. In either scenario, an additional degree of freedom (DOF) is introduced to those from a 3 + 1 model: strength of the coupling that mediates the decay, which determines the ν_4 lifetime. Decay of the ν_4 state causes a dampening of oscillations, resulting in different neutrino transition probabilities than in the 3 + 1 model.

Invisible decay—the 3 + 1+decay model involving *visible* and *invisible* neutrino decay was explored in the case of the IceCube Neutrino Observatory [150]. IceCube can search for anomalous muon-neutrino disappearance due to the existence of eV-scale sterile neutrinos [151, 152]. It was shown that this model can change the interpretation of the IceCube one-year null result, which had set a strong constraint on the 3 + 1 model [153].

The 3 + 1+decay model with *invisible* decay was fit to short-baseline data in [103], and subsequently, fits to the IceCube one-year dataset were combined with the short-baseline fits [154]. This model improves over the 3 + 1 model with a $\Delta \chi^2$ of 9.0, with one additional DOF. The aforementioned tension in the global fits can be quantified with a *parameter goodness-of-fit* [107]. The 3 + 1+decay model reduces the tension from a χ^2 /DOF of 28/2 to 19/3.

A search for the invisible 3 + 1+decay model using eight years of IceCube has found a preference for this model over either the three-neutrino or 3 + 1 models [155]. Under the assumption of Wilks' theorem, the three-neutrino model is disfavored with a *p*-value 3% and the 3 + 1 model is disfavored with a *p*-value of 5%. Incorporation of this result into global fits is expected to further reduce the tension from what was found in [154]. The Short-Baseline Neutrino Program at Fermi National Accelerator Laboratory offers another opportunity to search for this model of sterile neutrinos [156].

Visible decay—visible decays were discussed in [157] as an explanation of the LSND results. There, the authors proposed that a mostly-sterile neutrino ν_4 , of either Dirac of Majorana nature, could be produced in μ^+ decays and decay into ν_e and $\overline{\nu}_e$ in between the source and the detector, thereby leading to an effective flavor transition. The decay was fast due to the interactions of ν_4 with a massless scalar particle, ϕ . Subsequently, references [158–160] expanded on this scenario and argued that it could explain the MiniBooNE excess as well.

In [160], a SU (2)-invariant model is proposed where a Dirac sterile neutrino ν_s interacts with a massive scalar particle ϕ . In the mass basis, the mostly-sterile state ν_4 mixes with the electron- and muon-neutrinos, and therefore can be produced in both π^+ and μ^+ decays. The relevant interaction Lagrangian is given by

$$\mathcal{L} \supset -g_{s}U_{si}^{*}U_{s4}\phi\overline{\nu_{4}}\nu_{i} - \sum_{\alpha}\frac{g}{\sqrt{2}}U_{\alpha4}\overline{\nu_{4}}\mathcal{W}\ell_{\alpha} + \text{ h.c.}, \qquad (19)$$

where U_{si} is the mixing between the sterile state ν_s and the massive eigenstate ν_i , g_s the parityconserving, sterile neutrino coupling to the scalar ϕ , and g is the weak coupling constant.

The authors in [160] performed a fit to the MiniBooNE and LSND data. The results are shown in figure 17. While most of the signal at MiniBooNE comes from $\pi^+ \rightarrow \mu\nu_4$ production, at LSND, both $\pi^+ \rightarrow \mu^+$, $\mu^+ \rightarrow e^+\nu_4 \overline{\nu_\mu}$ as well as $\mu^+ \rightarrow e^+\nu_e \overline{\nu_4}$ contribute to the rate of inverse-beta-decays. This is because of the subsequent $\phi \rightarrow \nu \overline{\nu}$ decays, which



Figure 17. The preferred mixing of the light sterile neutrino with the muon and electron flavors, $|U_{e4}|^2$ and $|U_{\mu4}|^2$, to explain the MiniBooNE and LSND anomalies in the decaying-sterile-neutrino model of equation (19). Reproduced from [102]. CC BY 4.0.

generate an apparent $\nu \rightarrow \overline{\nu}$ transition. This effective transition is strongly constrained by solar antineutrino searches [161].

The presence of the scalar DOF in the theory also helps reconcile the model with cosmology [160, 162–172]. The secret, self-interactions in the sterile sector provide a new matter potential for sterile neutrinos in the early universe that suppresses their production. Another possibility is that the steriles interact with an ultra-light dark matter background, which also suppresses production in the early Universe [169, 170]. For more details, see section 5.1.3.

3.1.6. Lepton number violating muon decays. In this section, we focus on the possibility of using Lepton Number Violation (LNV) muon decays in addition to neutrino oscillations as an explanation for the LSND experiment. While not necessarily providing a full solution to the short-baseline puzzle, this scenario is worth considering because it is allowed by all data, it can be realized in explicit models, and is testable. Additionally, when considered in tandem with the 3 + 1 model, it opens up some parameter space in the 3 + 1 neutrino oscillations scenario by accounting for some of the LSND observation.

Lepton-flavor violating NSIs are very strongly constrained by theoretical consistency requirements and charged lepton flavor experiments. It has been pointed out in [173] that $\Delta L \neq 0$ interactions can evade these constraints. In [174] it was shown that, while most $\Delta L \neq 0$ effective operators are strongly constrained by high-precision measurements of the Michel parameters in muon decays, two such operators retain the SM prediction of $\rho = \delta = 3/4$ and are thus allowed. In addition, theoretical models that led to these two self-consistent effective operators were also developed. These effective operators are:

$$\mathcal{L}_1 \to [(\bar{\mu}_R \nu_{eL})(\nu_{aL}^T C e_L) - (\bar{\mu}_R \nu_{eL})(\ell_{aL}^T C \nu_{eL})]\langle |H^0| \rangle, \tag{20}$$

$$\mathcal{L}_2 \to (e_L^T C \nu_{eL}) (\mu_R^T C \nu_R)^* \langle |H^0| \rangle^2.$$
⁽²¹⁾

This type of NSI would lead to $\mu^{\dagger} \rightarrow e^{+} + \bar{\nu}_{\mu} + \bar{\nu}_{e}$, which would directly contribute to the muon decay-at-rest (DAR) signal in LSND. Accommodating the entire DAR signal

through such NSI would conflict with the KARMEN experiment, which also used a muon DAR beam. However, it is possible to achieve good agreement between the two experiments when one combines the LNV NSI and oscillations through a sterile neutrino, due to the differences in baselines. The LNV NSI would clearly not affect the pion decay beams in the other short-baseline accelerator experiments. The oscillation parameters obtained in [174] in the presence of the LNV NSI for LSND+KARMEN are compatible with those of global fits that include the MiniBoone and other data which rely on neutrinos from semileptonic pion decays.

It is thus interesting to consider how the presence of such LNV NSI can change the allowed sterile neutrino oscillation parameter space through the additional contribution to the LSND signal. This scenario would be testable by the different effects it produces in muon decay versus hadronic decay beams. In specific model realizations, it might also be possible to observe the effects of the new particles associated with the generation of the LNV operators in future collider experiments.

3.1.7. Large extra dimensions and altered dispersion relations (ADRs). Scenarios with sterile neutrino ADRs adopt additional terms in the standard dispersion relation $E^2 = |\vec{p}|^2 + m^2$. These terms make the oscillation amplitude energy-dependent, thus offering more freedom to accommodate various constraints and anomalies arising in short-baseline neutrino experiments. There exist various realizations of this scenario, including Lorentz violation and sterile neutrino shortcuts in warped extra dimensions [37, 175, 176]. The effect implied resembles standard matter effects but features a different energy dependence and typically applies for neutrinos and antineutrinos in the same way.

The basic idea is that active-sterile neutrino mixing is unaltered at low energies; however, a resonance conversion is present when the effect of the ADR minimizes the effective mass squared difference between active and sterile neutrinos. This effect is suppressed significantly for energies above the resonance energy. This allows to decouple sterile neutrinos at high energy and to evade stringent constraints from atmospheric and long-baseline accelerator neutrino experiments while offering the possibility to make small active-sterile neutrino mixing, observed or constrained in solar and reactor neutrino experiments, compatible with anomalies in short-baseline experiments such as LSND and MiniBooNE.

A challenge of such approaches is to find data sets compatible with all constraints in a complete framework with three active neutrinos. As has been pointed out in [177], the requirement to obtain the same flavor structure of active neutrinos below and above the resonance requires three sterile neutrinos whose mass squared differences reflect the active neutrino mass spectrum and feature democratic active-sterile flavor mixing. Moreover, different ADRs and, consequently, resonance energies are necessary for the three sterile flavors to avoid the cancellation of oscillation effects. In figure 18, the evolution of the various effective Δm^{2} 's is shown symbolically [177].

The consequent parameter space has five parameters beyond that of the SM with three massive, active neutrinos; namely a universal active-sterile Δm^2 and mixing $\sin^2 \theta$ plus three ADR parameters or resonance energies for the sterile flavors, respectively. As has been stressed in [178], where the phenomenology of two exemplary data points has been studied, it is not an easy task to find a combination of parameters that is compatible with all constraints, especially with both MiniBooNE and T2K that probe similar energy regions. Obviously, a conclusive verdict on the potential of ADR models would require a thorough scan of the complete parameter space. In [177], various promising Benchmark Mark Points (BMPs) have been analyzed. In particular BMP4 (see figure 19 for the energy dependence of Δm^2 's for this concrete data set) looks promising in this respect, as it leads to a muon-neutrino disappearance



Figure 18. Evolution of various Δm^{2*} s in ADR scenarios: symbolically. Reproduced from [177]. CC BY 4.0.



Figure 19. Evolution of various Δm^2 's for the specific example of BMP4. The vertical colored regions correspond from left to right to the energy ranges probed by reactor and Gallium experiments, LSND, MiniBooNE, and long-baseline accelerator experiments, respectively. Reproduced from [177]. CC BY 4.0.

and electron-neutrino appearance probability that is sharply peaked around the resonance energy $E_{\text{Res}} = 223.6 \text{ MeV}$ (see figure 20). Moreover, BMP4 features a rather small activesterile mixing angle $\sin^2 \theta = 10^{-4}$ that seems not to be excluded by MicroBooNE, according to the analysis in [179]. In fact, the large Δm^2 or order 30 eV² leads to fast oscillations that allow exploiting the difference in baselines between MiniBooNE and MicroBooNE (541 m versus 470 m) that amounts to roughly 8% of the oscillation length (corresponding to an oscillation probability reduced by 25% at resonance for MicroBooNE with respect to MiniBooNE) and that may be increased by finetuning the parameters.



Figure 20. $P(\nu_{\mu} \rightarrow \nu_{e})$ and $P(\nu_{\mu} \rightarrow \nu_{\mu})$ oscillation probabilities for MiniBooNE and the specific example of BMP4. Reproduced from [177]. CC BY 4.0.

In summary, ADRs represent a well-motivated and interesting tool to alter standard neutrino oscillations that may prove useful to explain the intricate framework of anomalies and constraints characterizing short-baseline neutrino oscillations.

3.1.8. Lorentz violation. The standard-model extension (SME) is an effective field theory framework to look for Lorentz symmetry violation (LV) [28]. The main interest of LV as an explanation to short-baseline anomalies is the flexibility of the SME-based Hamiltonian. One could design a suitable Hamiltonian using the SME to reproduce all neutrino and antineutrino oscillation data without the standard neutrino mass term. For example, the bicycle model [180] has the seesaw-mechanism-like texture to reproduce L/E oscillation behavior at high energy, even though Hamiltonian only has *CPT*-odd SME coefficient *a* and *CPT*-even SME coefficient *c* that make $\sim L$ or $\sim L \cdot E$ oscillation behaviors. The effective Hamiltonian of the bicycle model in 3×3 flavor basis matrix has following the texture,

$$h_{\rm eff} \sim \begin{pmatrix} aE & c & c \\ c & 0 & 0 \\ c & 0 & 0 \end{pmatrix}.$$
 (22)

This model demonstrates the possibility that LV can imitate the standard three massive neutrino oscillations. The tandem model [29] follows this, which can reproduce existing neutrino data at that time, including LSND.

The SME Lagrangian can be extended to include higher-order terms [181]. Since LV is related to a new space-time structure motivated by quantum gravity, it is natural to expect LV to show up in the non-renormalizable operators of the effective field theory. This further increases the number of model-building possibilities. The puma model [182, 183], for instance, is based on higher-order SME terms. The relevant Hamiltonian for this model is given by

$$h_{\rm eff} \sim \frac{m^2}{2E} \begin{pmatrix} 1 & 1 & 1\\ 1 & 1 & 1\\ 1 & 1 & 1 \end{pmatrix} + aE^2 \begin{pmatrix} 1 & 1 & 1\\ 1 & 0 & 0\\ 1 & 0 & 0 \end{pmatrix} + cE^5 \begin{pmatrix} 1 & 0 & 0\\ 0 & 0 & 0\\ 0 & 0 & 0 \end{pmatrix},$$
(23)

where m^2 , *a*, and *c* are tunable parameters. Suitable choices of these three parameters in this texture can reproduce all neutrino data at that time, including LSND and MiniBooNE. However, at present time this model is in tension with measurements by MINOS. The solution of this Hamiltonian provides energy-dependent oscillation lengths shown in figure 21. One line is used to reproduce solar and reactor neutrino data at low energy, and another line is used to reproduce atmospheric and long-baseline muon-neutrino disappearance



Figure 21. Solutions of the puma model [182, 183] for neutrinos (left) and antineutrinos (right) are shown by solid and dashed curves. Dotted lines are solutions from the solar and atmospheric Δm^2 . The horizontal axis is the energy and the vertical axis is the propagation length. Experimental regions are mapped by boxes or segments. Reprinted (figure) with permission from [182, 183], Copyright (2011) by the American Physical Society.

data. Around 30 to 300 MeV, these two lines drastically change shapes, and this region is used to reproduce LSND and MiniBooNE data. The model also uses a *CPT*-odd term which can make a difference between neutrino and antineutrino results. However, this model does not produce short-baseline reactor neutrino disappearance data (Daya Bay, Double Chooz, Reno) and long-baseline electron neutrino appearance data (T2K, NOvA). Thus, at present, this model is rejected as an explanation of the short-baseline puzzle.

Future of neutrino oscillation models based on Lorentz violation—it may be possible to construct an LV-based neutrino oscillation model beyond the puma model to reproduce all existing data including LSND, MiniBooNE, and other short-baseline results. However, such a model would have more fine-tuned parameters with an unusual texture or artificial cutoffs.

The difficulty to construct such a model is because LV-motivated terms have zero ($\propto E^0$) to a higher power with energy ($\propto E^1, E^2, ...$), and they dominate neutrino oscillations at high energy. So parameters introduced to explain short-baseline anomalies in general conflict with other oscillation data due to the lack of L/E oscillation behavior, which requires E^{-1} term in the Hamiltonian. One possibility is to introduce an unusual texture discussed above because they can reproduce L/E behavior with fine-tuned parameters. Another possibility is to introduce fine-tuned cutoffs in LV terms so that LV terms are limited to only certain regions to explain short-baseline anomalies. Therefore, LV-based models to explain short baseline anomalies would be unnatural, even if they exist. This is true for any other similar approach based on effective field theory, regardless of whether they are Lorentz violating or not.

Finally, the LV-based neutrino oscillation models can also be tested by studying the time dependence of the anomalies. This is the smoking-gun signature that differentiates this proposal from others. Interestingly, the MiniBooNE antineutrino data set shows a preference for a non-zero time-dependent LV component; however, this is in tension with the constraint on LV from MINOS.

3.2. Dark sectors in scattering and in the beam

The difficulty of resolving the various short-baseline anomalies by invoking solely neutrino flavor transformations, as detailed throughout section 3.1, has led to more exotic proposals, where light dark sectors can be produced alongside neutrinos in the beam or inside neutrino detectors, and mimic the experimental signatures. These model scenarios typically explain, for instance, the MiniBooNE and LSND results without violating the null results from other experiments.

In this subsection, we highlight a few related classes of models that fit this description section 3.2.1 discusses a model in which neutrinos are endowed with large transition magnetic moments, where upscattering from a light SM neutrino into a heavier neutral lepton N can be mediated through photon exchange with the nucleus. Section 3.2.2 details a model in which N interacts with a light-dark photon, where light-neutrino upscattering into this new state is mediated by the new force carrier, and N subsequently decays into e^+e^- pairs.

Models with long-lived particles produced at the neutrino source are also discussed. Section 3.2.3 discusses a model with long-lived HNLs that propagate to the detector and decay into e^+e^- and single- γ final states. In section 3.2.3, a dark matter model with light mediators is presented. The particles are produced in charged meson decays and upscatter inside the detector to produce electromagnetic showers.

3.2.1. Transition magnetic moment. Several extensions of the SM consider the existence of sterile neutrinos with Dirac or Majorana masses at or above the MeV scale. By convention, the mass eigenstates that are mostly in the direction of these sterile neutrino models are usually referred to as heavy neutral leptons (HNL), denoted here by *N*. Additional interactions notwithstanding, HNLs and light sterile neutrinos refer to the same class of particles which differ only in their mass value. Depending on the model, they can behave either as Majorana or (pseudo-)Dirac particles.

Similar to the sterile neutrino models discussed above, N can mix with SM neutrinos and interact with the Z and W bosons through mixing. However, it is also possible that this mixing is too small to be observed, and that these HNLs can be produced and decay via additional interactions. One interesting example is that of a transition magnetic moment between light SM neutrinos and N, described by the effective Lagrangian

$$\mathcal{L}_{N} = -d_{\alpha}\bar{\nu}_{\alpha L}\sigma_{\mu\nu}NF^{\mu\nu} - \frac{g}{\sqrt{2}}U^{*}_{\alpha N}\bar{\nu}_{\alpha}\gamma_{\mu}NZ^{\mu} + h.c., \qquad (24)$$

where d_{α} is the transition magnetic moment between N and SM neutrino weak eigenstate ν_{α} , $F^{\mu\nu}$ is the electromagnetic field strength tensor, $\sigma_{\mu\nu} = \frac{i}{2}(\gamma^{\mu}\gamma^{\nu} - \gamma^{\nu}\gamma^{\mu})$, and $U_{\alpha}N$ is the mixing between ν_{α} and N. Note that in an effective theory language the operator that gives rise to the first term is found at dimension six, namely, $\frac{c_{\alpha}}{\Lambda^2}L_{\alpha}\tilde{H}\sigma_{\mu\nu}NB^{\mu\nu}$, and thus $d_{\alpha} \propto c_{\alpha}\nu/\Lambda^2$, with $\nu = 174$ GeV being the Higgs vacuum expectation value. The second term above induces an effective vertex between an SM neutrino, a heavy neutrino, and the Z, which arrives due to mixing between ν_{α} and N.

The effective vertex introduced in equation (24) gives rise to new interactions relevant for existing and future neutrino experiments. First, the dipole model opens up new decay modes for mesons through off-shell virtual mediators, which can provide a source of heavy neutrinos at beam-dump experiments. For example, one introduces the weak mediated decay $\pi^+ \rightarrow \ell^+ \nu^* \rightarrow \ell^+ N\gamma$ and the Dalitz-like decay $\pi^0 \rightarrow \gamma \gamma^* \rightarrow \gamma N\nu$. Additionally, heavy neutrinos can be produced by the Primakoff up-scattering of SM neutrinos off a nuclear target A via the interaction $\nu A \rightarrow NX$. Finally, the typical observable signal in the dipole model is the decay of a heavy neutrino to an SM neutrino and a photon via the process $N \rightarrow \nu\gamma$. The relevant Feynman diagrams for these processes are shown in figure 22.

It is also possible that HNLs have non-negligible mixing with SM neutrinos, in which case the upscattering of SM neutrinos to HNLs can be mediated by both the photon, referred to as electromagnetic (EM) production, or the SM Z boson referred to as weak production. In this case, the mixing of N with muon neutrinos is the most relevant since most neutrinos in accelerator experiments are ν_{μ} and $\overline{\nu}_{\mu}$.



Figure 22. Left: dipole model interactions channels relevant for neutrino experiments which are introduced by the effective vertex in equation(24) (figure from [189]); right: preferred regions in dipole model parameter space to explain short-baseline anomalies, along with constraints from existing experiments. Reproduced from [189, 191]. CC BY 4.0.

LSND —the dipole model can explain the observed LSND excess via upscattering on carbon [40]

$$\nu_{\mu}{}^{12}C \to NnX \to \nu\gamma nX,$$
(25)

where the ν_{μ} comes from $\pi^+ \rightarrow \mu^+ \nu_{\mu}$ decay-in-flight in the LANSCE beam-stop. The Compton scattering and pair production of the photon from *N* decay mimic the signal of the prompt e^+ , which is detected in coincidence with the capture of the recoil neutron in equation (25). A heavy neutrino mass of $m_N \sim 50$ MeV can reasonably explain the LSND anomaly while avoiding constraints from the KARMEN experiment [40]. While [40] considered weak production, subsequent studies showed that EM upscattering dominates in this region of parameter space. Nevertheless, the original solution can still be accommodated in non-minimal scenarios involving more than one heavy neutrino [184].

MiniBooNE—regarding MiniBooNE, since the detection of e^{\pm} relies on reconstructing Cherenkov rings, this signature is indistinguishable from photons in the detector. Thus, the dipole model can provide an explanation of the MiniBooNE through the decay channel $N \rightarrow \nu\gamma$. In this case, the dominant source of heavy neutrinos in MiniBooNE comes from the Primakoff up-scattering of SM neutrinos produced in the BNB. Depending on the lifetime of the *N*, this can happen off of nuclei either in the dirt between the BNB and MiniBooNE or in the target material of the detector itself. We note that the decays of the heavy neutrino must be sufficiently prompt to be consistent with the timing distribution of the MiniBooNE excess [33].

The upscattering to N could proceed via EM or weak interactions with the protons of hydrogen or both with carbon nuclei in an incoherent or coherent fashion. The hadronic tensor in this case is the same as in elastic and quasi-elastic interactions of electrons and neutrinos on the corresponding targets (see chapter 4 of [185] for details). As shown in figure 16 of [186], the EM cross section on nuclei is dominated by the coherent mechanism; the incoherent one is suppressed by Pauli blocking at low four-momentum transfers, where the amplitude is enhanced by the photon propagator. On the contrary, the incoherent reaction is dominant in the weak part.

Electromagnetic upscattering only—the EM upscattering scenario with negligible N mixing with SM neutrinos has been explored in a number of studies [184, 187–190], which

J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501

Major Report



Figure 23. Photon events from radiative decay of N, \overline{N} at the MiniBooNE detector in ν -mode (top) and $\overline{\nu}$ -mode (bottom) compared to the MiniBooNE excess. For details, see [190] and [186]. Reproduced from [186]. \circ{C} DP Publishing Ltd. CC BY 3.0.

suggest that the energy distributions of the MiniBooNE excess can be accommodated by a dipole-coupled heavy neutrino with $m_N = O(100)$ MeV and $d_\mu = O(10^{-7}-10^{-6})$ GeV⁻¹. The MiniBooNE angular distribution, however, can only be explained with EM upscattering when $m_N \gtrsim 300$ MeV. In that case, the produced HNL is less boosted and decays more isotropically.

One can also consider extending the model to include an $\mathcal{O}(1)$ eV sterile neutrino in addition to the dipole-coupled HNL. Such a model can explain MiniBooNE through a combination of upscattering into N and eV-sterile-neutrino oscillations. If $m_N > \mathcal{O}(100)$ MeV, other low-energy experiments like LSND would not be sensitive to it, but they would still be sensitive to oscillations. This combination of effects has been found to decrease the tension in the sterile neutrino global picture [191].

Transition magnetic moment with weak production—we now consider weak production of HNLs in MiniBooNE. With the parameters proposed in [184], the forward-peaked dominant coherent EM contribution leads to a very narrow angular distribution not observed in the experiment (see figure 2 of [190]). The agreement can be improved by including production via the SM Z boson, which is less forward. The parameters can be constrained to the allowed range established in [40], but there are more stringent bounds for $U_{\mu N}$, in particular from radiative muon capture: $\mu^- p \rightarrow n \nu \gamma$, experimentally investigated at TRIUMF. Setting m_N to the allowed minimum of 40 MeV to have the largest possible upper bound in the mixing: $|U_{\mu N}|^2 = 8.4 \times 10^{-3}$ [192], the best fit finds $\tau_N = 9.1^{+1.1}_{-1.5} \times 10^{-10}$ seconds, $BR_{\mu} = 1.7^{+2.4}_{-1.4} \times 10^{-5}$ with a χ^2 / DoF =104/54. The different contributions to the excess are singled out in figure 23. A reasonable description of the angular distribution requires a suppression of the EM strength, as reflected by the small BR_µ compared to the original proposal of BR_µ = 10⁻², while increasing $U_{\mu N}$ as much as possible: its upper limit prevents from obtaining a more satisfactory description of the data.



Figure 24. Photon events from *N* radiative decay at MicroBooNE for 6.6×10^{20} POT in ν -mode. For details, see [190] and [186]. Reproduced with permission from [190].



Figure 25. Dark neutrino production and decay inside the MiniBooNE detector.

These results show that the heavy neutrino radiative decay hypothesis is not particularly successful in the simultaneous description of both the energy and the angular distribution of the excess, even with a degree of parameter fine-tuning. Nevertheless, based on MiniBooNE data alone, it cannot be fully excluded, at least as a partial source of the excess. Using the same number of POT as for figure 37 and the best-fit parameters, the photon events predicted at MicroBooNE are displayed in figure 24. With a total number of events of more than twice the SM ones and clearly more forward peaked, testing this possible explanation of the anomaly is within reach of the MicroBooNE experiment. It also warrants further studies for the new generation of experiments, SBND and ICARUS.

Other constraints—we now briefly discuss existing and projected constraints on the heavy neutrino transition magnetic moments. For $m_N \lesssim 1$ GeV, neutrino–electron scattering cross section measurements from Borexino, CHARM-II, DONUT, and LSND can be



Figure 26. Left: the spectrum of dark neutrino events in neutrino and antineutrino energy spectrum, as well as in the angular distribution at MiniBooNE. Right: fit to the neutrino-energy distribution at MiniBooNE in a dark neutrino model with a light dark photon ($m_{Z'} < m_4$). Reproduced from [203]. CC BY 4.0.



Figure 27. Left: Fit to the neutrino-energy distribution (left) and angular distribution (right) at MiniBooNE in a dark neutrino model with a heavy dark photon ($m_{Z'} < m_4$). Reproduced from [204]. CC BY 4.0.

translated into bounds on the transition magnetic moment [189, 193–197] (see the discussion in section 4.2.2). A single-photon search from the NOMAD experiment can be used to set a limit $\nu_{\mu}A \rightarrow (N \rightarrow \nu_{\mu}\gamma)A$ [198, 199]. Future single-photon searches at Fermilab's short baseline program may be able to probe parameter space relevant for the MiniBooNE anomaly [189]. At higher heavy neutrino masses, collider experiments can provide bounds on transition magnetic moments by looking for single-photon events with missing energy [189]. One can also constrain heavy neutrino transition magnetic moments using measurements of the relic ⁴ He abundance from Big Bang Nucleosynthesis and inferred Supernova 1987A cooling rates [189]. The proposed SHiP detector at CERN may also be able to set leading limits on heavy neutrino transition magnetic moments, especially for the $d_{e,\tau}$ couplings [189, 200]. Additionally, experiments such as Super-Kamiokande, IceCube, DUNE, and Hyper-Kamiokande may be able to place limits on the dipole model by taking advantage of the unique double-bang topology of $\nu - N$ up-scattering and subsequent N decay [193, 201, 202]. A subset of these limits, as well as the preferred regions to explain the short baseline anomalies, are shown in figure 22. We look forward to future experiments to shed light on heavy neutrino dipole portal explanations of short baseline anomalies.

3.2.2. Dark neutrinos. Many extensions of the SM to accommodate neutrino masses involve HNLs that mix with the SM neutrinos. If these HNLs interact via additional mediators, e.g. a dark photon from a secluded $U(1)_X$, and their masses are in the \sim MeV–GeV, then the upscattering of SM neutrinos into these so-called 'dark neutrinos', followed by dark neutrino decay, can explain the LEE observed by MiniBooNE [203, 204]. Similar to the dipole model (section 3.2.1), since MiniBooNE cannot distinguish an electron from a photon, it also cannot distinguish either of these from a pair of overlapping e^+ and e^- . If the e^+e^- pair are sufficiently collinear, or the energy of one of the particles falls below the detector energy threshold, the EM shower of the pair mimics that of a single electron or photon.

HNLs can be produced via rare meson decays (e.g. from kaons), and then decay inside the detector [205, 206], however, the signals predicted in this scenario can be delayed with respect to the neutrino beam, and tend to lead to forward-peaked angular distributions. However, if the new particles are produced in neutrino upscattering inside the detector, provided their decays are sufficiently short-lived ($c\tau^{\text{LAB}} < O(10)$ ns at MiniBooNE), they can be registered in coincidence with the SM neutrinos. Scenarios where e^+e^- pairs are produced in this fashion (not restricted to the dark neutrino model) have been studied in detail in [203, 204, 207–215]. We first discuss the original proposals based on a dark photon and then move on to newer proposals involving scalar particles.

Dark photon models—in addition, these models face none of the problems of the popular 3 + 1 oscillation explanation of the MiniBooNE excess, as the phenomena observed at the different SBL experiments decouple in this framework. Finally, there is the possibility of connecting these models to other prominent questions of particle physics. The discovery of neutrinos with hidden interactions would be a very strong indication of the existence of dark sectors that could contain the theorized dark matter. With the SBN program currently underway, there is the opportunity to probe large regions of the LEE model parameter space, but also more generic models of dark sector HNLs.

In [203, 204, 207, 208, 214, 216], the mediator of upscattering is a dark photon. A simplified model can be used to understand the experimental signatures. In it, an electrically-neutral fermion ν_D is charged under dark U(1)' symmetry and is assumed to mix with SM neutrinos, $\nu_{\alpha} = \sum_{i=1}^{4} U_{\alpha i} \nu_i$, ($\alpha = e, \mu, \tau, D$) where U is a 4 × 4 unitary matrix. The mediator of this new gauge group then kinetically mixes with the SM photon via $(\varepsilon/2)X'_{\mu\nu}\hat{F}^{\mu\nu}$, which leads the electrically-charged SM fermions to acquire a small coupling to the dark photon, Z'. The low-energy simplified Lagrangian reads

J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501

$$\mathcal{L}_{\text{int}} \supset -g_D U_{D4}^* U_{Di} \overline{\nu}_i \gamma_\mu N Z'^\mu - e \varepsilon Z'^\mu J_\mu^{\text{EM}}, \qquad (26)$$

where g_D is the U(1)' gauge coupling, J_{μ}^{EM} the SM electromagnetic current, and ε is the kinetic mixing parameter. We define $N \equiv \nu_4$. In a full model, N may be of Dirac or Majorana nature.

Since the dark photon interacts with the SM electric charge in an analogous way to the photon, the upscattering cross section can be calculated in an analogous way to the transition magnetic moment case, replacing the leptonic vertex and the propagator by a massive one. When the dark photon is light, coherent scattering with the nucleus is dominant and the rate is enhanced by the number of protons in the nucleus, *Z*, with respect to scattering on individual protons. For $m_{Z'} \gtrsim 1$ GeV, the incoherent contributions start to dominate and the upscattering can kick a proton out of the Carbon nucleus in MiniBooNE, for example. In models where the HNL only mixes with muon neutrinos, the upscattering cross section can be shown to scale as $\sigma_{\nu_{\mu}A \to NA} \propto |U_{\mu4}U_{D4}|^2$ by virtue of the unitarity of *U*. Therefore, $\sigma_{\nu_{\mu}A \to NA}$ can be readily constrained by existing limits on the active-heavy mixing $|U_{\mu4}|^2$ for every choice of $|U_{D4}|^2$, which usually takes values close to unity.

The decay of the HNL is prompt when $m_{Z'} < m_4$, where the dark photon is produced on shell and decays to e^+e^- . Decays to light neutrinos are suppressed by the mixing angle combination $|U_{Di}U_{Dj}|^2$, which is small for i < 4. This regime produces very forward-going HNLs, which subsequently decay to boosted e^+e^- pairs. While this produces more $e^+e^$ events that mimic single photon or single electron showers, it also leads to very forwardpeaked angular distributions, in contrast with MiniBooNE's observation [33]. The angular spectrum is less forward for $m_N \gtrsim 300$ MeV, however, at those masses constraints from highenergy experiments are severe [217].

For heavy dark photons, $m_{Z'} > m_4$, the decay is a three-body one and, therefore, the HNL is much longer-lived. Reference [204] proposed a model where $|U_{\tau 4}|^2 \gg |U_{\mu 4}|^2$, so that the decay process is effectively $N \rightarrow \nu_{\tau} e^+ e^-$. A similar proposal was made in [208, 209] where a model with two HNLs was used. In that case, the heaviest HNL decays to the intermediary state with a lifetime that is not suppressed by active-heavy mixing. Due to the heavy mediator, the upscattering happens with a larger Q^2 , and the angular distribution can be less forward. The best agreement with the angular distribution is found when the dark photon interferes with the SM Z.

Dark scalars—HNLs can also interact with additional scalars that play the role of the dark photon discussed above. In this case, the upscattering cross section lacks the *t*-channel singularity and does not asymptote to a constant, but rather falls as $1/E_{\nu}^2$ at large energies. Therefore, if the dark sector particles have masses of $\mathcal{O}(100)$ MeV, the upscattering process is largest at the energies of LSND and MiniBooNE and may avoid constraints from high-energy experiments altogether. Models of this type have been discussed in [210–213].

Dark scalars and neutrino polarizability—as pointed out in [210], the scalar mediator can also lead to the production of photons pairs via the decay chain $N \rightarrow \nu(S \rightarrow \gamma\gamma)$. At MiniBooNE, if S is lighter than the pion, it will be more boosted and its decays to overlapping photons would mimic the excess signal. It is also possible that the branching ratios for $S \rightarrow e^+e^-$ and $S \rightarrow \gamma\gamma$ are both sizeable, in which case the excess would display a non-trivial shape in energy and angle.

A scalar mediator coupling to both neutrinos and photons induces a parametrically enhanced neutrino polarizability, i.e. in low energy processes the scalar can be integrated out, resulting in dimension 7 Rayleigh operators of the form $(\bar{\nu}_i P_L \nu_j) F_{\mu\nu} F^{\mu\nu}$ (or $(\bar{\nu}_i P_L \nu_j) F_{\mu\nu} \tilde{F}^{\mu\nu}$ if the mediator is a pseudoscalar) [218]. For light mediators, with masses below ~MeV, there are stringent constraints on such neutrino polarizability models from cosmology and stellar N(k'

 $C^{Z'/b}$

 $N(k) \equiv$

Z'/H

(a)



(c)

Figure 28. Feynman diagrams for the production and the subsequent decay of the N_2 in LSND and MB.

cooling. For mediators in the MeV to few GeV regime, relevant for the MiniBooNE anomaly, these models can be probed at a multitude of neutrino facilities, from measurements of solar neutrino scattering (Borexino and Xenon-nT) to observations in high energy neutrino beams (DUNE ND), as well as using beam dump facilities and at precision e^+e^- collider experiments such as Belle II [218].

(b)

Explaining LSND—some of the models proposed in the literature above can, in principle, also explain the LSND anomaly. This sets a more stringent requirement on the theory since it requires the upscattering process to produce a neutron in the detector. The new signal should mimic the inverse-beta-decay signature with at LSND, with a prompt electromagnetic signal followed by delayed neutron capture. This already eliminates the dark photon as a potential solution, since the couplings with neutrons are much more suppressed than those with protons. In addition, depending on the mass of the heavy neutrino, the upscattering process may have too large of a threshold to be initiated by π or μ decays at rest. Instead, for masses of $\mathcal{O}(100)$ MeV, one may take advantage of the number of pions that decay in flight. This requires a large cross section, since the at these energies is much smaller than the decay-at-rest one. Models of this type were proposed in [212, 213], where the heavy neutrinos are produced via the exchange of a new scalar boson with large couplings to neutrons.

Broad requirements to explain MiniBooNE and LSND—such proposals must conform to strong demands from (a) cross section requirements in order to yield a sufficient number of total events in both LSND and MB, (b) the measured energy and angular distributions in both experiments, and finally, (c) compatibility of the new physics model and its particle content with bounds from an extensive swathe of particle physics experiments [189, 192, 193, 213, 217, 219-232].

We compare how scalar and vector mediators perform in helping to achieve a simultaneous understanding of both anomalies. Our treatment is necessarily brief, and for details and a full set of references on all that follows the reader is referred to [233].

We start by breaking up the interaction into two parts. We first consider the tree-level process leading to the up-scattering of an incoming muon neutrino, ν_{μ} , to a heavy neutral lepton (N_2) in the neutrino detector as shown in figure 28(a), with the underlying assumption that it subsequently decays promptly in the detector into either of the final states shown in figures 28(b) and (c). The mediator for the up-scattering process could be either a light neutral vector boson Z' or a light CP-even scalar H. The relevant interaction Lagrangian for the upscattering process in each case is given by

$$\mathcal{L}_{int}^{Z'} = (C_{\nu}^{Z'} \bar{\nu}_{iL} \gamma^{\mu} N_{Lj} Z'_{\mu} + h.c.) + C_{n}^{Z'} \bar{U}_{n} \gamma^{\mu} U_{n} Z'_{\mu},$$
(27)



Figure 29. The incoherent (coherent) cross section per CH_2 molecule (C atom) as a function of incoming neutrino energy. The overall constants for different kinds of mediator masses are taken from table 5.

Table 5. The overall coupling values for the vector and scalar mediators for different values of mediator masses $(m_{Z'/H})$ to produce 560 N_2 in the MB final state. The mass of N_2 is 100 MeV.

$m_{Z'/H}$	$C_{ u}^{Z'}C_n^{Z'}$	$C_{\nu}^{H}C_{n}^{H}$
50 MeV 1 GeV 300 MeV	1.04×10^{-8} 8.5 × 10 ⁻⁷	$\begin{array}{c} 9.3 \times 10^{-8} \\ 2.14 \times 10^{-6} \\ 5.35 \times 10^{-7} \end{array}$

$$\mathcal{L}_{\text{int}}^{H} = (C_{\nu}^{H}\bar{\nu}_{iL}N_{Ri}H + h.c.) + C_{n}^{H}\bar{U}_{n}U_{n}H, \qquad (28)$$

where U_n is the nucleon field and i, j = 1, 2, 3. For simplicity, we assume that for both mediators (Z' and H), the coupling to a proton is the same as its coupling to a neutron. The up-scattering cross section depends on the overall product of the coupling constants $C_{\nu}^{Z'}C_n^{Z'}$ ($C_{\nu}^H C_n^H$) for the vector (scalar) mediator.

Based on the Lagrangian above and the benchmark parameter values in figure 5, we compute the coherent and incoherent cross sections (shown in figure 29) for both mediators, required for generating the central value of the number of excess events in MB, which is 560 [33]. From figure 29, we note that the scalar and vector-mediated cross sections behave distinctly, and our representative calculations bring out the following qualitative points:

- (a) For all cases the cross section initially rises as the energy is increased from its lowest values, however, it subsequently drops for a scalar mediator whereas it remains approximately flat with increasing energy for a vector mediator. This is true for both the coherent and incoherent parts. As the neutrino energy rises, it is this relatively rapid drop in the cross section for *H* that allows solutions with a scalar mediator to comfortably skirt constraints [217] coming from CHARM II [234] and MINER ν A [235], compared to the *Z'* mediated process.
- (b) It can also be seen from figure 29 that the coherent contribution dominates over the incoherent part for lighter mediator masses, whereas the reverse is true for the higher mass choice for mediators.
- (c) For LSND, contributions to events come from the incoherent part of the cross section only, due to the presence of a neutron in the final state. In the region in the left panel of

figure 29, we note that the energy drops from MB (~800 MeV) to LSND DAR flux values (~150–200 MeV). Note that for $m_{Z'/H} = 1$ GeV (solid curves), while the incoherent cross section drops for both mediators, the vector cross section has lower values to begin with compared to the scalar. It also drops more rapidly. For example, it can be seen that the cross section for the Z' drops an order of magnitude over this energy range for $m_{Z'} = 1$ GeV. For $m_{Z'/H} = 50$ MeV (dashed curves), over this energy range, the incoherent scalar cross section is significantly higher than that for the vector. In fact, it increases as the energy is lowered, unlike its vector counterpart. This reduction in the incoherent vector to give a sufficient number of electron-like excess events at LSND, even though by using a high enough $m_{Z'}$ one may successfully evade the CHARM II and MINER ν A bounds. However, on the other hand, too low m_H gives many more events than those observed in LSND, both in the 20–60 MeV visible energy range which recorded data, and beyond 60 MeV, where only a limited number of events were seen.

(d) Finally, for scalar mediators, especially those with low masses $m_H \simeq 100$ MeV, the cross section tends to rise at low values of E_{ν} . However, in such models, if the primary decay modes of N_2 are to invisible daughters, as in [211], the incoherent interaction would mimic the neutral current interaction $\nu N \rightarrow \nu N$. This has been measured at MB [236] at these energies and found to be in agreement with the SM, providing an important restriction on such models.

Overall, the cross section and mediator mass considerations for a common solution thus appear to favor scalar mediators over vectors. Secondly, our representative calculations also point to a preference for lighter (but not ultra-light) mediators if both excesses are to have a simultaneous solution.

An examination of the angular distribution of MB is also helpful from the point of view of imposing requirements on proposed solutions. The excess in MB is distributed over all directions but is moderately forward. The cross section responsible for the production of N_2 as a function of the cosine of the angle between the momentum direction of N_2 and the beam direction has been studied in a bin-wise manner in [233] for both the coherent and incoherent contributions. It was found that when $m_{Z'/H} = 50$ MeV, almost all the produced N_2 are in the most forward bin for both mediators. However, as the mediator mass is increased to 1 GeV, there was a shift in the distribution, and the other bins were also populated for both types of mediators, even though there were qualitative differences between the two. From this, at first it appears that using a single scalar mediator and adjusting its mass to an intermediate value, as well as the mass of N_2 will allow us to find a common solution to the two anomalies as well as match the angular distribution in MB. However, further examination based on considerations related to the energy distributions in LSND and MB (for details, see [233]) reveals that this is not the case if good fits to both anomalies are desired.

Overall, as detailed in [233], energy distributions in LSND and MB, the angular distribution in MB, when combined with the stringent constraints on light singlet scalars, all suggest the use of a scalar doublet, with one light and one moderately heavy partner. We find this leads to a degree of angular isotropy while allowing a large number of events in the forward direction, consistent with observations. A combination of a moderately heavy and a light mediator complement each other well when a common solution to the two anomalies is sought. An example solution to both anomalies that incorporates all the features that have been obtained in our study has been provided in [213].

Our insistence on a solution that addresses both anomalies simultaneously is, of course, a choice. It restricts proposed solutions in ways that attempt to explain the anomalies

individually do not. However, it is noteworthy that once we demand this, and adhere to the dictates of the cross section, the observed energy and angular distributions in both experiments as well as the many constraints from various experiments [233], then we are led to a simple extension of the SM that (i) resolves both anomalies, (ii) provides a portal to the dark sector, (iii) accounts for the experimentally observed value of the muon g - 2 and ((iv) addresses the issue of neutrino mass via a Type I seesaw, in conformity with the global data on the observed values of neutrino mass-squared differences in oscillation experiments, as shown in [213].

3.2.3. Long-lived heavy neutrinos. Heavy neutrino decays to single photons—in [205], the authors propose a solution to the MiniBooNE excess with a heavy neutrino that interacts with muon-neutrinos through mixing as well as through a transition-magnetic moment. The model is the same as the one presented in figure 24, but contrary to the solution in section 3.2.1, the mixing angle $|U_{\mu N}|$ and transition magnetic moment d_{μ} are small, such that the heavy neutrino decays only in macroscopical distances. In particular, if the lifetime is larger than the distance between the target and the MiniBooNE detector, $c\tau^0 \gtrsim \mathcal{O}(500)$ m, then the heavy neutrino can be produced in pion and kaon decays via mixing, $K \to \mu N$, travel to the detector, and decay inside the active volume due to the transition magnetic moment, $N \to \nu\gamma$.

While the model was shown to successfully reproduce the energy spectrum of the MiniBooNE excess, the decay process tends to produce very forward signatures. A better agreement with the angular spectrum is achieved for larger masses of the *N*. However, if the heavy neutrino is too massive, its arrival in the MiniBooNE detector is delayed with respect to the SM neutrinos. This delay is proportional to the $(m_N/E_N)^2$, and should not exceed values much larger than several nanoseconds, as the excess events have been observed to be in the same time window as the beam neutrinos [33]. This sets a strong constraint on the mass of the heavy neutrino. For masses below 150 MeV, where agreement with the timing requirements can be achieved, production in pion decays should also be considered.

Heavy neutrino decays to axion-like particles—in order to account for the MiniBooNE excess, [206] proposes an extension of the SM with Dirac HNL that couples to a leptophilic axion-like particle (ℓ ALP). The HNL, denoted by N_D to emphasize its Dirac nature, mixes with the three SM neutrino flavors, ν_{β} . The flavor and mass eigenstates, ν_{jL} and N_D , can be transformed into each other by means of a unitary matrix U [203]:

$$\nu_{\beta} = \sum_{j=1}^{3} U_{\beta j} \nu_{jL} + U_{\beta 4} N_D.$$
(29)

The relevant Lagrangian includes the interaction of ℓ ALPs with sterile neutrinos and electrons [237]:

$$\mathcal{L}_{a\ell} = -\frac{\partial_{\mu}a}{2f_a} (c_N \bar{\nu}_D \gamma^{\mu} \gamma^5 \nu_D + c_e \bar{e} \gamma^{\mu} \gamma^5 e), \qquad (30)$$

where f_a is the ℓ ALP decay constant, c_N and c_e are the dimensionless parameters for the ℓ ALP-sterile neutrino and ℓ ALP-electron couplings, respectively.

Our ℓ ALP scenario is sketched in figure 30. In this framework, the sterile neutrino with a mass m_N of $\mathcal{O}(100)$ MeV, the ℓ ALP with a mass m_a of $\mathcal{O}(10)$ MeV and an inverse decay constant $c_e/f_a \simeq \mathcal{O}(10^{-2})$ GeV⁻¹ are considered. The Dirac-type sterile neutrino N_D , produced from charged kaon decays through its mixing with the muon neutrino, travels a distance of 500 m and decays into a ℓ ALP and a muon neutrino inside the MiniBooNE detector. The electron-positron pairs produced from the ℓ ALP decays can be interpreted as



Figure 30. ℓ ALP scenario for the explanation of the MiniBooNE excess. *L* is the travel distance of the sterile neutrino, *D* is the diameter of the MiniNooBE detector, and θ_a is the scattering angle of the ℓ ALP. Reproduced from [206]. CC BY 4.0.



Figure 31. The comparison of the numerical results of the angular and visible energy spectra in the ℓ ALP scenario and the data of the MiniBooNE experiment in the neutrino mode. Reproduced from [206]. CC BY 4.0.

electron-like events provided that their opening angle is sufficiently small. We verify that with appropriate choices of the parameters, the sterile neutrino and ℓ ALP have the proper mean decay lengths consistent with the setup of the MiniBooNE experiment. We also make sure that the values of the parameters adopted in our model are allowed by the astrophysical and experimental constraints.

Following the approach applied in [205], the angular and visible energy spectra in the neutrino mode are computed and compared with the results of the MiniBooNE experiment, as shown at the top of figure 31. At the bottom panel, the predicted total event numbers are obtained after summing over the spectra and shown as the contours consistent with the MiniBooNE excess at the 1σ to 3σ levels on the $(m_N, |U_{\mu4}|^2)$ and $(c_e/f_a, |U_{\mu4}|^2)$ planes, with the constraints obtained from other experiments. We find that the scenario with the sterile neutrino mass in the range 150 MeV $\leq m_N \leq 380$ MeV and the neutrino mixing parameter between $10^{-10} \leq |U_{\mu4}|^2 \leq 10^{-8}$ can explain the MiniBooNE excess.

3.2.4. Dark matter particles. Since the neutrinos at MiniBooNE are produced primarily from charged meson decays and the decays of daughter muons of those charged mesons, neutrinobased solutions can accommodate the absence of any excess in the dump mode, in which the charged mesons are no longer focused by magnetic horns, unlike the neutrino and



Figure 32. Best-fit regions to the MiniBooNE excess in the HNL and ℓ ALP parameter space. Limits and best-fit regions are shown on the $(m_N, |U_{\mu4}|^2)$ (left) and $(c_e/f_a, |U_{\mu4}|^2)$ (right) planes. Reproduced from [206]. CC BY 4.0.

antineutrino modes. Essentially, the neutrino-based explanations work well because a key feature of the excess seems to be correlated to the focusing or suppression of charged mesons.

This poses a challenge to dark sector interpretations of the excess (e.g. using π^0 or dark bremsstrahlung production channels [223, 224]), which have been more constrained and less holistic in their explanation of the excess with respect to their counterparts in neutrino BSM physics thus far. One solution proposed in [215] opens up the possibility for dark sector explanations by means of connecting the dark sector to the physics of charged meson decays, something that had previously been overlooked. The authors in [215] considered various DM scenarios involving couplings to muons, namely those in equations (31), (32), and (33):

$$\mathcal{L}_{S} \supset g_{\mu}\phi\bar{\mu}\mu + g_{n}Z_{\alpha}'\bar{u}\gamma^{\alpha}u + \frac{\lambda}{4}\phi F_{\mu\nu}'F^{\mu\nu} + \text{h.c.}, \qquad (31)$$

$$\mathcal{L}_P \supset ig_{\mu}a\bar{\mu}\gamma^5\mu + g_n Z'_{\alpha}\bar{u}\gamma^{\alpha}u + \frac{\lambda}{4}aF'_{\mu\nu}\tilde{F}^{\mu\nu} + \text{h.c.}, \qquad (32)$$

$$\mathcal{L}_{V} \supset e(\epsilon_{1}V_{1,\mu} + \epsilon_{2}V_{2,\mu})J_{\text{EM}}^{\mu} + (g_{1}V_{1,\mu} + g_{2}V_{2,\mu})J_{D}^{\mu} + (g_{1}'V_{1,\mu} + g_{2}'V_{2,\mu})J_{D}^{\prime\mu}.$$
(33)

Here, three massive bosons have been introduced; a long-lived scalar ϕ and pseudoscalar a, and a short-lived vector A'_{α} decaying to DM fermions χ , χ' with $F'_{\mu\nu} \equiv \partial_{\mu}A'_{\nu} - \partial_{\nu}A'_{\mu}$. The muonic couplings allow for the 3-body decays of the form $M \to \mu\nu X$ ($M = \pi^{\pm}, K^{\pm}$) became possible, as shown in figure 33. The 3-body nature of this decay mechanism is not phase-space suppressed in the contraction of $\nu^{\dagger}\mu_{\uparrow}$, as opposed to the ordinary 2-body decay which selects out only the combination $\nu_{\downarrow}\mu_{\uparrow}$. The branching ratios for scalar, pseudoscalar, and vector DM production are shown in figure 34. In the \mathcal{L}_V scenario, the detector signature would then take place through DM upscattering, $\chi N \to \chi' N (\chi' \to \chi e^+ e^-)$, while scenarios \mathcal{L}_S and \mathcal{L}_P consider long-lived ϕ/a scattering in the detector through a Primakoff-like process $\phi N \to \gamma N$ via a heavy mediator Z'. The parameter space which fits the MiniBooNE excess for \mathcal{L}_S , \mathcal{L}_P , and \mathcal{L}_V is shown in figure 35.

However, the muonic portal is not the only possibility, and DM production in the 3-body π^{\pm} and K^{\pm} decays from DM-quark couplings can also be treated. In this sense, the scope of



Figure 33. 3-body charged meson decay into a scalar, pseudoscalar, or vector. Analogous processes exist for π^- and K^- decay. Reproduced from [215]. CC BY 4.0.



Figure 34. Branching ratios for the 3-body production of scalars, pseudoscalars, and vectors via a charged meson with mass $M = m_{\pi^+}$, m_{K^+} . Reproduced from [215]. CC BY 4.0.



Figure 35. The credible regions for fits to the MiniBooNE cosine spectrum with \mathcal{L}_V (left) and $\mathcal{L}_{S,P}$ (right) are shown at 68% (dark-shaded) and 95% (light-shaded). Reproduced from [215]. CC BY 4.0.

the 3-body decay solution can cover several coupling schemes, and should be testable at other experiments with similar meson production capabilities.

This has broad implications for accelerator facilities, such as LBNF (DUNE), that also have magnetic focusing horns that should be sensitive to forward-produced DM from the meson decays. Constraints on the parameter space from accelerator-based searches at CHARM, MINER ν A, and T2K can also be considered, but their smaller POT and exposures do not give them sensitivity to the MiniBooNE excess. Other neutrino experiments such as CCM, JSNS, and COHERENT that produce stopped mesons and lack magnetic focusing horns can also probe the parameter space relevant for the excess, since while the DM signal from meson decays will be isotropic, their detectors are situated much closer to the beam targets to be sensitive to the DM flux.

3.3. Conventional explanations

While the majority of the explanations explored above rely on some new or BSM physics to give rise to the various anomalies, the possibility that the origin lies in more conventional explanations, such as an underestimated background, mis-modelling in simulation, or overconstrained cross-section uncertainties, must still be considered. While these explanations are generally difficult to test directly without access to (often) collaboration-internal experimental tools and data sets, there have been several attempts in recent years to test individual anomalies in this direction. In one such example [36], it was shown that allowing a combination of theoretical uncertainties in different background channels to fluctuate in unison is not sufficient to resolve the MiniBooNE anomaly; however, it can reduce the significance of the MiniBooNE excess. In this section, we discuss several such possible 'conventional' interpretations for the anomalies.

3.3.1. Single-photon production. The solution to the MiniBooNE puzzle may have important implications for our understanding of neutrinos and their interactions. In addition to the interpretation in terms of new physics, the MiniBooNE anomaly could be a manifestation of new forces of nature, while unaccounted or poorly modeled SM backgrounds cannot be entirely discarded. Once Cherenkov detectors like MiniBooNE misidentify single photon tracks as electrons, the excess of events could be due to their products through both SM and BSM mechanisms.

In the SM, single photons can be emitted in NC interactions, NC1 γ , on nucleons, $\nu(\bar{\nu})N \rightarrow \nu(\bar{\nu})\gamma N$, or on heavy nuclei, via incoherent or coherent (where the nucleus remains in its ground state) scattering. Theoretical models for the elementary NC1 γ [238–240] take into account s- and u-channel amplitudes with nucleons and Δ (1232), but also heavier baryon resonances, in the intermediate state, figure 36. The structure of nucleon pole terms at threshold is determined by symmetries. The extension towards higher energy and momentum transfers, required to predict cross sections at MiniBooNE, is performed by the introduction of phenomenologically parametrized weak and electromagnetic form factors. The same strategy has been adopted for resonance terms The $\Delta(1232)$ excitation followed by radiative decay is the dominant mechanism, as correctly assumed by MiniBooNE, but the contribution of non-resonant terms is also sizable. The uncertainty in the elementary NC1 γ cross section is dominated by the leading $N - \Delta$ axial transition coupling, $C_5^A(q^2 = 0)$, which is related to the $\Delta N\pi$ coupling (known from πN scattering) by a Goldberger-Treiman relation, but has also been found to be $C_5^A(0) = 1.18 \pm 0.07$ [241] from data on π production induced by neutrino scattering on hydrogen and deuterium. It is worth stressing that the $\Delta N\gamma$ couplings responsible for the resonance radiative decay are directly related to helicity



Figure 36. Feynman diagrams for NC single photon emission considered in the literature. The first two diagrams stand for direct and crossed baryon pole terms with nucleons and baryon resonances in the intermediate state. The third diagram represents *t*-channel meson exchange contributions. Reproduced from [186]. © IOP Publishing Ltd. CC BY 3.0.

amplitudes $A_{1/2,3/2}$ known with few-percent accuracy from photo-nucleon interactions [5]. Furthermore, owing to isospin symmetry, these quantities also constrain the vector part of the weak Δ production. This implies that large uncertainties in the Δ radiative decay couplings are at odds with hadron phenomenology but would also have an observable impact in weak pion production. Among the non-resonant contributions, *t*-channel ω -meson exchange, was proposed as a solution for the MiniBooNE anomaly [38] because of the rather large (although uncertain) couplings and the ω isoscalar nature, which enhances its impact on the coherent NC1 γ reaction. However, actual calculations found this contribution small compared to Δ (1232) excitation [238, 242]. Nuclear effects, in particular the broadening of the Δ resonance in the nucleus, are important for single photon emission [240, 243]. On ¹²C they cause a reduction of about 30% in the cross section (see figure 9 of [240]).

With the ingredients outlined above, the SM single- γ contribution to the number events in the MiniBooNE detector and their distributions have been calculated [244, 245] using the available information about the detector mass and composition (CH₂), neutrino flux and, quite significantly, photon detection efficiency. Results are to a large extent consistent with the (data based but relying on an improvable reaction model) MiniBooNE estimate (figures 4, 6 and 8 of [245]) and, therefore, insufficient to explain the excess. The impact of two-nucleon meson-exchange reaction mechanisms has been recently investigated [246] and found to be small (a factor of around 9 at $E_{\nu} = 500$ MeV compared to single-nucleon mechanisms). In [247] it was estimated that the NC1 γ background should be enhanced by a factor between 1.52 and 1.62 over the MiniBooNE estimate, depending on the energy range and mode. Such an enhancement, shrinks the excess and significantly reduces the appearance-disappearance tension in global fits but is at odds with the earlier described theoretical calculations. An upper limit for the NC1 γ cross section on liquid argon has been recently obtained by the MicroBooNE experiment [248], as discussed in the following section. It disfavors that the excess could be solely attributed to this reaction channel but new results with higher statistics are required for a firm conclusion. Assuming 6.6×10^{20} POT from the experiment's run plan, the distributions of the NC 1γ events calculated with the model of [240] are given in figure 37 (adapted from [249]). Comparison to future data shall offer valuable information about this process.

3.3.2. Reactor flux modeling. One conventional explanation for the Reactor Antineutrino Anomaly (RAA) has been gaining momentum thanks to the significant experimental and theoretical progress made in the past decade. Reactor neutrino experiments that can measure oscillations without any reliance on reactor neutrino models (discussed further in



Figure 37. SM prediction for NC1 γ events at MicroBooNE for 6.6 \times 10²⁰ POT in ν -mode. Reproduced from [186]. © IOP Publishing Ltd. CC BY 3.0.

section 4.1.3) have been chipping away at RAA-suggested sterile neutrino oscillation parameter space. Additionally, neutrino, nuclear physics, and nuclear theory evidence have recently emerged suggesting that RAA may, at least in part, be caused by problems with reactor flux models. While experimental developments are discussed in section 4.3.2, recent theoretical advancements in the flux prediction landscape that lend support to non-BSM origin to the RAA are discussed here.

The statistical significance of the RAA depends not only on the magnitude of offset between reactor $\bar{\nu}_e$ data and the Huber-Mueller prediction but also on the size of the error bands applied to those predictions. In the years following the inception of the RAA in 2011, a variety of reactor modeling studies have argued that the 2%–3% error budget assigned to this prediction is likely underestimated. In particular, the role of forbidden beta transitions in altering the $\bar{\nu}_e$ flux and spectrum reported by conversion predictions is not considered in the formulation of Huber-Mueller model error bands [55]. When naive treatments of forbidden decay contributions are included in the prediction, variations in the flux of 4% or more are observed [250] and are reflected in the community-driven report in [251]. On the other hand, a more recent conversion calculation (termed the HKSS model) that attempts to account for forbidden transition contribution using nuclear shell model-based calculations shows strong deviations from the Huber-Mueller model in spectral shape. However, it finds no major discrepancy in reported IBD yields [252].

The past ten years have also brought about substantial development of state-of-the-art summation calculations, thanks to improved nuclear data evaluations and new nuclear structure measurements for a range of high-Q, high-yield fission daughters [47–53]. The improved summations can be compared to conversion calculations to provide an assessment of the latter's robustness. As early as 2012, improved summation calculations were shown to generate reduced flux predictions with respect to earlier iterations [253], indicating modest over-prediction of ²³⁸U fluxes in the Huber–Mueller prediction. When comparing to conversion predictions, modern summation calculations were shown to predict different fuel-dependent $\bar{\nu}_e$ flux variations [254]. Most recently in 2019, Estienne, Fallot *et al* [44] used comprehensive improvements and updates in nuclear databases to generate a summation model (referred to as the EF Model) with a total predicted flux of a few percent lower than the measured global IBD yield average, but with a ²³⁵U prediction ~6% smaller than that predicted by the HM conversion model. These comparisons are suggestive of a possible issue with conversion-predicted fluxes for individual isotopes.

Table 6. Theoretical IBD yields of the four fissionable isotopes in units of 10^{-43} cm²/fission predicted by different models [257]. Reproduced from [257]. CC BY 4.0.

Model	σ_{235}	σ_{238}	σ_{239}	σ_{241}
HM EF	6.74 ± 0.17 6.29 ± 0.31	10.19 ± 0.83 10.16 + 1.02	4.40 ± 0.13 4.42 ± 0.22	6.10 ± 0.16 6.23 ± 0.31
HKSS KI	6.82 ± 0.18 6.41 ± 0.14	$\begin{array}{c} 10.10 \pm 1.02 \\ 10.28 \pm 0.84 \\ 9.53 \pm 0.48 \end{array}$	$\begin{array}{c} 4.42 \pm 0.22 \\ 4.45 \pm 0.13 \\ 4.40 \pm 0.13 \end{array}$	6.17 ± 0.16 6.10 ± 0.16

The IBD yields predicted by these various models can be compared to the global IBD yields listed in table 9 and to specialized Daya Bay [255] and RENO [256] dataset reporting yields as a function of varying contents of the reactor cores; the latter datasets are described further in section 4.3.2. In particular, if comparisons are made under the hypothesis that the RAA is generated by sterile neutrino oscillations, the different reactor flux models generate substantially differing interpretations. To illustrate, figure 38 shows the contours of the 2σ allowed regions in the $(\sin^2 2\vartheta_{ee}, \Delta m_{41}^2)$ plane of 3 + 1 active-sterile neutrino mixing parameters [257]. One can see that an indication in favor of neutrino oscillations only for the HM and HKSS conversion models, which exhibit a significant reactor rate anomaly above 2σ (see table 7). Considering the EF model, for which the reactor rate anomaly is less than a few percent, the 2σ exclusion curves in figure 38 allow only small values of $\sin^2 2\vartheta_{ee}$, including $\sin^2 2\vartheta_{ee} = 0$, which corresponds to a lack of any statistically significant indication of sterile neutrino oscillations. It should be noted here that error estimates for summation calculations are ill-defined, but are generally expected to be similar in magnitude to those provided by conversion predictions; efforts to provide more robust error envelopes are underway [258].

For all the reactor flux models, it is also worth noting that upper bounds exist for the value of the mixing parameter $\sin^2 2\vartheta_{ee}$, with exact limits dependent on the value of Δm_{41}^2 . For $\Delta m_{41}^2 \gtrsim 2 \text{ eV}^2$ the upper bounds for $\sin^2 2\vartheta_{ee}$ are between 0.14 and 0.25. Figure 38 shows that these bounds and the solar bound [259] are in agreement, but in tension with the large mixing [136] required to explain the anomaly of the GALLEX [260], SAGE [75], and BEST [136] gallium experiments with short baseline neutrino oscillations. This is a puzzling recent development in the phenomenology of short-baseline neutrino share that may require an explanation extending beyond both conventional explanations and the simplest possible model of 3 + 1 active-sterile neutrino mixing.

3.3.3. The gallium anomaly and interaction cross-section uncertainties. The first analysis establishing the existence of the Gallium Anomaly [78] did not consider the uncertainties on the cross section of the detection process in equation (6) and, as mentioned in previous sections, this has been a source of subsequent investigation as a possible avenue for resolving this anomaly.

It is now clear that this quantity is of paramount relevance for the possible interpretation of the Gallium Anomaly. Different calculations of the cross section have been published after the seminal work by Bahcall [76], and the values are summarized in table 8.

Also, it has been pointed [77] that the rather large uncertainties come from the fact that only the cross section of the transition from the ground state of 71 Ga to the ground state of 71 Ge is known with precision from the measured rate of electron capture decay of 71 Ge to 71 Ga. In fact, recent improvements on measurements of this transition [261–263] indicate that the Gallium Anomaly persists. However, electron neutrinos produced by processes in



Figure 38. Contours of the 2σ allowed regions in the $(\sin^2 2\vartheta_{ee}, \Delta m_{41}^2)$ plane obtained from the combined neutrino oscillation fit of the reactor rates in table 7 and the Daya Bay [255] and RENO [256] evolution data. The blue, red, green, and magenta curves correspond, respectively, to the HM, EF, HKSS, and KI models in table 6. Also shown are the contour of the 2σ allowed regions of the Gallium anomaly obtained in [136] from the combined analysis of the GALLEX, SAGE and BEST data (orange curve), and the 2σ bound obtained from the analysis of solar neutrino data in [259] (dark red vertical line). Reproduced from [257]. CC BY 4.0.

Table 7. Average ratio \overline{R}_{mod} obtained in [257] from the least-squares analysis of the reactor rates in table 1 and of the Daya Bay [255] and RENO [256] evolution data for the IBD yields of the models in table 6. The RAA columns give the corresponding statistical significance of the reactor antineutrino anomaly. The descriptions of all the models are given in text. Reproduced from [257]. CC BY 4.0.

Model	Rates		Evoluti	on	Rates + Evolution		
	\overline{R}_{mod}	RAA	\overline{R}_{mod}	RAA	\overline{R}_{mod}	RAA	
HM	$0.936\substack{+0.024\\-0.023}$	2.5σ	$0.933^{+0.025}_{-0.024}$	2.6σ	$0.930\substack{+0.024\\-0.023}$	2.8σ	
EF	$0.960^{+0.033}_{-0.031}$	1.2σ	$0.975\substack{+0.032\\-0.030}$	0.8σ	$0.975\substack{+0.032\\-0.030}$	0.8σ	
HKSS	$0.925^{+0.025}_{-0.023}$	2.9σ	$0.925\substack{+0.026\\-0.024}$	2.8σ	$0.922\substack{+0.024\\-0.023}$	3.0σ	
KI	$0.975\substack{+0.022\\-0.021}$	1.1σ	$0.973\substack{+0.023\\-0.022}$	1.2 σ	0.970 ± 0.021	1.4σ	

equation (5), can also be absorbed through transitions from the ground state of 71 Ga to two excited states of 71 Ge at 175 keV and 500 keV.

When the aforementioned uncertainties are taken into account, different results for the total cross section for the radioactive sources are obtained, resulting in changes to the measured and expected 71 Ge event rates. On the other hand, complete calculations of the



Figure 39. Nuclear levels for the 71 Ga transitions to 71 Ge. Reprinted (figure) with permission from [264], Copyright (2012) by the American Physical Society.

Table 8. Summary of the cross section values recently published.

$\sigma (10^{-46} \text{ cm}^2)$	References
$58.1^{+0.21}_{-0.16}$	[76]
59.3 ± 0.14	[263]
59.10 ± 0.114	[548]
56.7 ± 0.06	[268]
59.38 ± 0.116	[674]

cross sections of the interaction process, equation (6), for neutrinos produced by ⁵¹Cr and ³⁷Ar sources are given in [264] as

$$\sigma = \sigma_{gs} \left(1 + \xi_{175} \frac{BGT_{175}}{BGT_{gs}} + \xi_{500} \frac{BGT_{500}}{BGT_{gs}} \right).$$
(34)

Here, σ_{gs} is the cross section of the transitions from the ground state of ⁷¹Ga to the ground state of ⁷¹Ge, BGT_{gs} is the corresponding Gamow–Teller strength, and BGT₁₇₅ and BGT₅₀₀ are the Gamow–Teller strengths of the transitions from the ground state of ⁷¹Ga to the two excited states of ⁷¹Ge at about 175 and 500 keV as shown in figure 39. The coefficients of BGT₁₇₅/BGT_{gs} and BGT₅₀₀/BGT_{gs} are determined by phase space: $\xi_{175}({}^{51}Cr) = 0.669$, $\xi_{500}({}^{51}Cr) = 0.220$, $\xi_{175}({}^{37}Ar) = 0.695$, $\xi_{500}({}^{37}Ar) = 0.263$ [76].

Table 9 shows the different values of the rate when four different approaches (other than the one by Baxton, R_B) are used to compute the cross sections [264]: R_{HK} uses information about the Gamow–Teller strengths from Haxton [265] and Krofcheck *et al* [266]; for R_{FF} , the corresponding numbers are taken from Frekers *et al* [261]; R_{HF} uses BGT₁₇₅ from Haxton and BGT₅₀₀ from Frekers *et al*; and R_{JUN45} uses calculations using nuclear shell-model wave functions obtained by exploiting recently developed two- nucleon interactions [267] (see also [268] for additional details).

The last column of table 9 shows the corresponding weighted average for each case showing the Gallium Anomaly with a statistical significance of 3σ , 2.9σ , 3.1σ , and 2.3σ , respectively for the last four cases. This confirms the Gallium Anomaly and retains the indication in favor of a short-baseline disappearance of electron neutrinos, possibly due to neutrino oscillations. Figure 40 shows the 90% contours in the $|U_{e4}|^2 - \Delta m_{41}^2$ plane obtained from the analysis of the measured and expected ⁷¹Ge event rates, considering the neutrino survival probability equation (7), where one can see that the squared-mass difference Δm_{41}^2



Figure 40. The 90% allowed regions in the $|U_{e4}|^2 - \Delta m_{41}^2$ plane obtained from the analysis of the measured and expected ⁷¹Ge event rates. Reproduced from [268]. CC BY 4.0.

Table 9. Ratios of measured and expected 71 Ge event rates in the GALLEX and SAGE experiments. The last column corresponds to the weighted average [264]. Information in the last row is from [268]. Reprinted (table) with permission from [264], Copyright (2012) by the American Physical Society. Reproduced from [268]. CC BY 4.0.

	GALLEX1	GALLEX ₁	SAGE _{Cr}	SAGE _{Ar}	Avg.
R _B	0.95 ± 0.11	$0.81\substack{+0.10 \\ -0.11}$	0.95 ± 0.12	0.79 ± 0.08	0.86 ± 0.05
$R_{\rm HK}$	0.85 ± 0.12	0.71 ± 0.11	$0.84\substack{+0.13\\-0.12}$	0.71 ± 0.09	0.77 ± 0.08
$R_{\rm FF}$	0.93 ± 0.11	$0.79\substack{+0.10\\-0.11}$	$0.93\substack{+0.11\\-0.12}$	$0.77\substack{+0.09 \\ -0.07}$	0.84 ± 0.05
$R_{\rm HF}$	$0.83\substack{+0.13 \\ -0.11}$	0.71 ± 0.11	$0.83\substack{+0.13 \\ -0.12}$	$0.69\substack{+0.10\\-0.09}$	$0.75\substack{+0.09 \\ -0.07}$
R _{JUN45}	0.97 ± 0.11	0.83 ± 0.11	0.97 ± 0.12	0.81 ± 0.08	0.88 ± 0.05

allowed values are $\sim 1 \text{ eV}^2$ or larger, for the different approaches to compute the $\nu_e - {}^{71}\text{Ga}$ cross section.

A revision of the statistical significance of the Gallium Anomaly considering different calculations of the neutrino detection cross section, has been performed to compare the 3 + 1 neutrino oscillation hypothesis with the Reactor Antineutrino Anomaly and with the inclusion of data from tritium experiments and from experiments measuring solar neutrinos (Giunti et al 2022b). Remarkably, it was found that the Gallium Anomaly is in strong tension with bounds obtained from the other data sets. In addition, when all data are combined, the corresponding parameter goodness of fit is below 0.042%, implying a tension of $\sim 5\sigma$, leading to the conclusion that it should be necessary to seek for alternative solutions to the shortbaseline oscillations for this anomaly.

3.4. Summary of interpretations

A summary of the interpretations detailed in this section is provided in table 10. The columns, from left to right, are the following: broad classes of models; specific models that fall in each class; the experimental signature of each model; which anomalies each model can address (LSND, MiniBooNE, Reactor and Gallium anomalies, respectively), and the corresponding references. For convenience, in the first column, we also indicate the sections of this document that are related to each class of models.

Finally, in table 11 we summarize which experimental efforts can probe which signatures of new physics that can address the anomalies. For concreteness, we focus on experiments that are either recent, are expected to be upgraded, or are still under proposal. The leftmost column shows broad classes of experiments grouped arbitrarily by their source and type of experiment. The other columns present which specific experiments, in each of these classes, can probe a given experimental signature resulting from different interpretations of the anomalies. For clarity, the 'decays in flight' column title refers to particles produced by the decay of mesons, muons, or taus in flight.

4. Broader experimental landscape

In this section, we review the broader landscape of existing experimental results with relevant sensitivity to interpretations of the LSND, MiniBooNE, Reactor, and Gallium anomalies.

4.1. Flavor conversion

Other short-baseline, as well as long-baseline, neutrino experiments can look for anomalous flavor conversions. In the following sections, we discuss direct tests of MiniBooNE and LSND in $\nu_{\mu} \rightarrow \nu_{e}$ and $\overline{\nu}_{\mu} \rightarrow \overline{\nu}_{e}$ searches, as well as in ν_{μ} and ν_{e} disappearance. We also discuss the ν_{e} and $\overline{\nu}_{e}$ disappearance in the context of the Gallium and reactor anomalies, where both null results and hints are observed.

4.1.1. Pion decay-at-rest accelerator experiments. The KARMEN (KArlsruhe Rutherford Medium Energy Neutrino) experiment was located at the highly pulsed spallation neutron source ISIS of the Rutherford Laboratory (UK). ISIS protons had an energy of 800 MeV and were delivered to the water-cooled Ta-D₂O target with a repetition rate of 50 Hz. The time structure of the ISIS protons (double pulses with a width of 100 ns separated by 325 ns) allowed a clear separation of ν_{μ} events due to π^+ decay from $\bar{\nu}_{\mu}$ and ν_e events due to μ^+ decay.

The KARMEN detector [269] was a segmented liquid scintillator calorimeter, located 17.7m from the ISIS target at an angle 100 degrees relative to the proton beam. The active target consisted of 65 m³ of liquid scintillator segmented into 608 modules with gadolinium-coated paper placed between modules for efficient detection of thermal neutrons. KARMEN performed a search for $\bar{\nu}_{\mu} \rightarrow \bar{\nu}_{e}$ oscillations, analogous to the LSND search, using $p(\bar{\nu}_{e}, e^{+})n$ and found measured rates agreed with background expectations [20]. The sterile neutrino 90% confidence interval (C.I.) obtained by the KARMEN measurement is shown in figure 41 in relation to the LSND allowed regions. Although KARMEN did not see the LSND-like signal, it did not exclude the entirety of the LSND 99% allowed regions although it did strongly disfavour the larger $\Delta m^2 > 10 \text{ eV}^2$ solutions.

Category	Model	Signature	Anomalies				Pafarancas
	Widden		LSND	MiniBooNE	Reactors	Sources	References
Flavor transitions sections 3.1.1– 3.1.3, 3.1.5	(3+1) oscillations	oscillations	1	1	1	1	Reviews and global fits [102, 103, 105, 106]
,	(3+1) w/invisible sterile decay	oscillations w/ ν_4 invisible decay	\checkmark	1	\checkmark	\checkmark	[150, 154]
	(3+1) w/sterile decay	$ u_4 \rightarrow \phi \nu_e $	\checkmark	1	\checkmark	\checkmark	[158, 161–675]
Matter effects sections 3.1.4, 3.1.7	(3+1) w/anom- alous matter effects	$ \nu_{\mu} \rightarrow \nu_{e} \text{ via matter} $ effects	1	1	X	X	[142, 146, 676–678]
	(3+1) w/quasi- sterile neutrinos	$ \nu_{\mu} \rightarrow \nu_{e} \text{ w/resonant } u_{s} \text{ matter effects} $	1	1	1	1	[147]
Flavor violation section 3.1.6	Lepton-flavor-vio- lating μ decays	$\mu^+ ightarrow e^+ u_lpha \overline{ u_e}$	\checkmark	x	X	x	[173, 174, 679]
	neutrino-flavor- changing bremsstrahlung	$ u_\mu A ightarrow e \phi A$	1	1	X	x	[680]
Decaysin flight section 3.2.3	Transition magn- etic mom., heavy ν decay	$N ightarrow u \gamma$	X	1	X	X	[205]
	Dark sector heavy neutrino decay	$N \rightarrow u(X \rightarrow e^+e^-)$ or $N \rightarrow u(X \rightarrow \gamma\gamma)$	x	\checkmark	X	X	[206]
Neutrino scattering sections 3.2.1, 3.2.2	neutrino-induced upscattering	$\nu A \rightarrow NA, N \rightarrow$ $\nu e^+ e^- \text{ or } N \rightarrow \nu \gamma \gamma$	1	1	X	X	[203, 204, 207–214]

65

Table 10. New physics explanations of the short-baseline anomalies categorized by their signature. Notation: \checkmark —the model can naturally explain the anomaly, \checkmark —the model can partially explain the anomaly, \checkmark —the model cannot explain the anomaly.

Major Report

Tab	le 10. (Continued.)							
Category	Model Transition magn- etic mom. or polarizability photons	Signature $\nu A \rightarrow NA, N \rightarrow \nu \gamma$ or $\nu A \rightarrow \nu \gamma A$	Anomalies				References	
			LSND	MiniBooNE	Reactors	Sources	Telefonces	
			1	J	X	x	[40, 184, 186, 187, 189, 191, 218, 232, 681]	
Dark matter Scat- tering section 3.2.4	dark particle- induced upscattering	γ or e^+e^-	x	1	X	X	[215]	
Section 5.2.4	dark particle- induced inverse Primakoff	γ	1	1	x	x	[215]	

66

Table 11. Summary of future experimental prospects to probe new physics explanations of the anomalies. We emphasize that all experiments can constrain the new physics models discussed in this paper in one way or another, but we list those that can provide a direct test of the respective model.

Source	3 + 1 oscillations	Anomalous matter effects	Lepton flavor violation	Decays in flight	Neutrino- induced upscattering	Dark-particle-induced upscattering
Reactor	DANSS upgrade, JUNO- TAO, NEOS II, Neutrino-4 upgrade, PROSPECT-II					
Radioactive source	BEST-2, IsoDAR, THEIA, Jinping					
Atmospheric	IceCube upgrade, KM3NET, ORCA and ARCA, DUNE, Hyper-K, THEIA				IceCube upgrad ARCA, DU	de, KM3NET, ORCA and NE, Hyper-K, THEIA
Pion/kaon decay- at-rest	JSNS ² , COHERENT, CAP- TAIN-Mills, IsoDAR, KPIPE		JSNS ² , COHERENT, CAPTAIN-Mills, Iso- DAR, KPIPE, PIP2-BD			COHERENT, CAP- TAIN-Mills, KPIPE, PIP2-BD
Beam Short Baseline	SBN				SBN	۷
Beam long baseline	DUN	NE, Hyper-K, ESSn	uSB	DUNE	, Hyper-K, ESSnu	SB, FASER ν , FLArE
Muon decay-in- flight		<i>v</i> STORM			ν STORM	
Beta decay and electron capture	KATRIN/TRISTAN, Project- 8, HUNTER, BeEST, DUNE- ³⁹ Ar, PTOL- EMY, $2\nu\beta\beta$					

67



Figure 41. The KARMEN sterile neutrino 90% C.I. compared to other then contemporary experiments. Reprinted (figure) with permission from [20], Copyright (2002) by the American Physical Society.

4.1.2. Pion decay-in-flight accelerator experiments

4.1.2.1. Short-baseline experiments. MiniBooNE. While MiniBooNE's ν_e and $\bar{\nu}_e$ appearance searches in the Fermilab BNB have resulted in the observation of anomalous excesses, MiniBooNE BNB ν_{μ} and $\bar{\nu}_{\mu}$ CC measurements have been relatively well understood, including a disagreement between data and the Monte Carlo prediction that has been attributed to cross-section effects and uncertainties [270]. As a result, MiniBooNE has been able to perform searches for ν_{μ} and $\bar{\nu}_{\mu}$ disappearance, both exclusively, and inclusively, resulting in limits (using approximately the first half of data collected by MiniBooNE) as shown in figure 42 [271]. The results complement those from prior short-baseline ν_{μ} and $\bar{\nu}_{\mu}$ disappearance searches, from the CCFR [272] and CDHS [273] experiments.

It should be noted that, in these searches, due to the lack of a near detector (ND), large flux and cross-section uncertainties limited MiniBooNE's sensitivity particularly in the high Δm^2 range, where (fast) oscillations are expected to lead to an overall normalization deficit, and are thus masked by systematic uncertainties on the overall ν_{μ} and $\bar{\nu}_{\mu}$ CC rate normalization. More powerful searches were performed by MiniBooNE in combination with measurements by the SciBooNE experiment, as described next.

MiniBooNE/SciBooNE. The SciBooNE experiment was located 100 m downstream from the BNB target and ran simultaneously with MiniBooNE from 2007 to 2008. A simultaneous ν_{μ} disappearance search under a (3 + 1) scenario was performed in the SciBooNE and MiniBooNE detectors with the BNB operating in forward horn current mode [274]. A separate $\bar{\nu}_{\mu}$ disappearance search was performed in the SciBooNE and MiniBooNE detectors with the BNB operating in reverse horn current mode [275]. Exclusion (at 90% CL)



Figure 42. MiniBooNE's 90% CL sensitivity (dashed line) and limit (solid line) for ν_{μ} (top) and $\bar{\nu}_{\mu}$ (bottom) disappearance under a 3 + 1 scenario. Previous limits by CCFR (dark grey) and CDHS (light grey) are also shown. Reprinted (figure) with permission from [271], Copyright (2009) by the American Physical Society.

regions for these two searches are shown in figure 43, and are consistent with exclusion limits from other accelerator-based neutrino experiments, including the short-baseline CCFR and CDHS experiments, and the long-baseline MINOS experiment [276], as well as a MiniBooNE-only search limit.

MiniBooNE-NuMI. In addition to neutrinos from the BNB, MiniBooNE has also been able to study neutrinos from the Fermilab-based NuMI beam, viewed by MiniBooNE at an off-axis angle of 6.3° [277]. MiniBooNE measured both ν_e and ν_{μ} CCQE events from the NuMI off-axis beam (with no wrong-sign discrimination), as shown in figure 44 (top). While an appearance or disappearance search with these data sets was not performed by MiniBooNE¹²⁴, the data was found to agree with expectation. The NuMI off-axis beam at the MiniBooNE location is much higher in ν_e content than the BNB, as well as in neutrinos contributed from kaon decays in the beamline. Because of a higher intrinsic ν_e background, this data set was particularly limited in sensitivity to light sterile neutrino oscillations.

¹²⁴ The MiniBooNE-NuMI ν_e data set has been analyzed under a 3 + 1 appearance scenario in [94], where it was shown that, despite the small observed data excess, the data showed no significant preference for oscillations over the null hypothesis.



Figure 43. MiniBooNE-SciBooNE 90% CL limits from a joint ν_{μ} disappearance search (left) [274] and a joint $\bar{\nu}_{\mu}$ disappearance search (right) [275]. Reprinted (figure) with permission from [274, 275], Copyright (2012) by the American Physical Society.

MicroBooNE. Recently MicroBooNE has published its first results on direct oscillation searches due to light sterile neutrinos [278]. The analysis uses high-statistics CC ν_{μ} and CC ν_{e} selections, developed as part of the inclusive ν_{e} search [279], with data collected over a three year period with a total exposure of 6.369 × 10²⁰ protons-on-target, representing approximately half of the total MicroBooNE data-set. A full 3 + 1 neutrino model was studied, capitalizing on the seven channels of ν_{e} and ν_{μ} selections. Although a small deficit of ν_{e} was observed, the data was found to agree with the 3ν (no-sterile) hypothesis within the 1σ level. Since the data was found to be consistent with the no-sterile hypothesis, exclusion contours were calculated and can be found in figure 45, where regions of the LSND anomaly at both high Δm^{2} and low Δm^{2} were excluded. As the analysis makes use of a a very pure ν_{e} selection, exclusions are shown both for ν_{e} appearance (sin² 2 $\theta \mu e$) and ν_{e} disappearance (sin² 2 θee) profiling over the other parameters of the full 3 + 1 fit.

This work highlighted an important aspect of studying full 3 + 1 neutrino oscillations in the BNB. As the BNB is a ν_{μ} dominated beam with a non-negligible intrinsic ν_e component, cancellation can occur in any $\nu_{\mu} \rightarrow \nu_e$ appearance signal due to $\nu_e \rightarrow \nu_e$ disappearance occurring in parallel. This cancellation can lead to a diminished oscillation effect in comparison to when one studies the (un-physical) 2ν approximation. The use of the NuMI beam, which has a different ratio of intrinsic ν_e to ν_{μ} could help break this cancelletion effect and includion of the NuMI beam data in the 3 + 1 result is an ongoing effort for the MicroBooNE collaboration.

Prior to this collaboration result, the publicly available MicroBooNE ν_e CC data sets [279–282] have also been studied by the phenomenology community [179, 283] and in combination with the MiniBooNE data by the MiniBooNE collaboration [284]. Although these studies involve several assumptions and approximations that are not inherent in the official results, the qualitative conclusions are largely the same, with the exception of the analysis [283] which founds a preference for ν_e disappearance at the $\sim 2\sigma$ level but notably did not use the full systematic uncertainty accounting of the collaboration's data release. However, a more recent analysis [179], as well as the official collaboration release, incorporated systematic uncertainties from the collaboration's data release and accounting for


Figure 44. Top: reconstructed energy distributions of ν_{μ} (left) and ν_{e} (right) events in MiniBooNE from the NuMI off-axis beam. The figure is from [277]. Bottom: limit extracted from a fit to the ν_{e} distribution under a 3 + 1 hypothesis. Reprinted (figure) with permission from [94, 277], Copyright (2009) by the American Physical Society.



Figure 45. The MicroBooNE 3 + 1 CLs exclusion contours at the 95% CL in the plane of Δm_{41}^2 versus (left) $\sin^2 2\theta_{\mu e}$ and (right) $\sin^2 2\theta_{ee}$. A full 3 + 1 treatment was implemented, with parameters not shown profiled over for the sensitivity and resulting data exclusions. Shown also is the (unphysical) ν_e appearance-only and ν_e disappearance-only sensitivities for comparison. Reproduced from [278]. CC BY 4.0.



Figure 46. Left: preferred regions in the simplified two-flavor model for MiniBooNE (gray), MicroBooNE inclusive analysis (blue), MicroBooNE CCQE analysis (red), KARMEN (green), and OPERA (brown) as indicated. For reference, the predicted sensitivity of SBN is also shown. Reproduced from [179]. CC BY 4.0. Right: preferred regions in the 3 + 1 model parameter space for a MiniBooNE-only fit and a joint fit including the MicroBooNE ν_e CCQE result. Reproduced from [284]. CC BY 4.0.

neutrino energy reconstruction smearing found this preference for ν_e disappearance to be statistically insignificant.

Reference [179] has also provided the first analysis of the effect of MicroBooNE's ν_{e} results in the context of a light sterile neutrino explanation of MiniBooNE. This includes both an analysis in the 'simplified two-flavor' model, in which only $\nu_{\mu} \rightarrow \nu_{e}$ oscillations are taken into account, without significant $\nu_{\mu} \rightarrow \nu_{\mu}$ or $\nu_e \rightarrow \nu_e$ effects, consistent with MiniBooNE analyses prior to 2022 as well as the complete 3 + 1 model. It is found that, while MicroBooNE prefers the null hypothesis, i.e. no new sterile neutrino, the allowed regions from MiniBooNE at $\sim 3\sigma$ are still allowed at the same confidence level by the MicroBooNE results, see left panel of figure 46. Given its large dataset, the results of MicroBooNE's inclusive analysis [279] are found to be the most statistically powerful, while the CCQE sample provides a considerably weaker constraint. Besides that, the MiniBooNE collaboration has performed a fit of both MiniBooNE and MicroBooNE CCQE data [281] to a 3 + 1sterile neutrino model [284], properly accounting for oscillations in the backgrounds. As expected, the impact of adding MicroBooNE's CCQE sample to the fit has a marginal effect on the preferred regions for the 3 + 1 model, see right panel of figure 46. While the two experiments' results are combined into a joint likelihood, the systematic uncertainties that the two experiments have in common, particularly those associated with their common neutrino flux from the booster neutrino beam and the neutrino-nucleus cross section model, have not been combined.

NOMAD. The NOMAD experiment [285], which ran at CERN using protons from the 450 GeV SPS accelerator, employed a conventional neutrino beamline to create a wideband 2.5–40 GeV neutrino energy source. These neutrinos were created with a carbon-based, low-mass, tracking detector located 600m downstream of the target. This detector had fine spatial resolution and could search for muon-to-electron and muon-to-tau oscillations. No signal was observed in either mode, and this experiment set a limit to $\nu_{\mu} \rightarrow \nu_{e}$ appearance that excluded

the majority of the 99% CL-allowed LSND region above 10 eV^2 at 90% CL, but had a significantly worse limit than KARMEN below 10 eV^2 .

CCFR. The CCFR experiment was carried out at Fermilab in 1984 [286]. The experiment made use of a narrow band beamline, with meson energies set to 100, 140, 165, 200, and 250 GeV, yielding ν_{μ} and $\bar{\nu}_{\mu}$ beams that ranged from 40 to 230 GeV in energy. A two-detector experimental setup carried out a disappearance search, with the ND at 715 m and the far detector (FD) at 1116 m from the center of the 352 m long decay pipe. The calorimetric detectors were constructed of segmented iron with a scintillator and spark chambers, and each had a downstream toroid to measure the muon momentum. The data showed no evidence for a distance-dependent modification of the neutrino flux and ruled out oscillations of ν_{μ} into any other single type of neutrino for $30 < \Delta m^2 < 1000 \text{ eV}^2$ and $\sin^2(2\theta) > 0.02 - 0.20$.

CDHS. The CDHS experiment [287] at CERN searched for ν_{μ} disappearance with a twodetector design of segmented calorimeters with iron and scintillator. The experiment used 19.2 GeV protons on a beryllium target to produce mesons that were subsequently focused into a 52 m decay channel. The detectors were located 130 m and 885 m downstream of the target. The experiment set a limit at 95% CL and set constraints that are comparable to the MiniBooNE ν_{μ} disappearance limit described above, but extending to slightly lower Δm^2 .

4.1.2.2. Long-baseline experiments. The phenomenology of sterile neutrino-driven oscillations at long baselines consists of interference phenomena arising from at least two distinct oscillation frequencies and several scale-determining mixing angles and phases. In the context of a 3 + 1 model with the addition of one new flavor state ν_s and a new mass state ν_4 extending the three-flavor PMNS matrix to a 4 × 4 unitary matrix, and assuming the parametrization $U = R_{34}R_{24}R_{14}R_{23}R_{13}R_{12}$, the extended matrix is written as:

$$U = \begin{bmatrix} U_{e1} & U_{e2} & e^{-i\delta_{13}}s_{13}c_{14} & e^{-\delta_{14}}s_{14} \\ U_{\mu 1} & U_{\mu 2} & -e^{-i(\delta_{13}-\delta_{14}+\delta_{24})}s_{13}s_{14}s_{24} + c_{13}s_{23}c_{24} & e^{-i\delta_{24}}c_{14}s_{24} \\ U_{\tau 1} & U_{\tau 2} & -e^{i\delta_{24}}c_{13}s_{23}s_{24}s_{34} + c_{13}c_{23}c_{34} - e^{-i(\delta_{13}-\delta_{14})}s_{13}s_{14}c_{24}s_{34} & c_{14}c_{24}s_{34} \\ U_{s1} & U_{s2} & -e^{i\delta_{24}}c_{13}s_{23}s_{24}c_{34} - c_{13}c_{23}s_{34} - e^{-i(\delta_{13}-\delta_{14})}s_{13}s_{14}c_{24}c_{34} & c_{14}c_{24}c_{34} \end{bmatrix},$$
(35)

which introduces three new mixing angles, θ_{14} , θ_{24} , and θ_{34} , two new CP violating phases, δ_{14} and δ_{24} , and one new linearly independent mass-splitting that is chosen to be Δm_{41}^2 in this case. Neglecting Δ_{21} and using unitarity to rewrite any terms containing $U_{\alpha 1}$ or $U_{\alpha 2}$, the probability of ν_{μ} survival in a 3 + 1 model can be written as

$$P(\nu_{\mu} \to \nu_{\mu}) \approx 1 - 4 |U_{\mu3}|^2 (1 - |U_{\mu3}|^2 - |U_{\mu4}|^2) \sin^2 \Delta_{31} -4 |U_{\mu3}|^2 |U_{\mu4}|^2 \sin^2 \Delta_{43} - 4 |U_{\mu4}|^2 (1 - |U_{\mu3}|^2 - |U_{\mu4}|^2) \sin^2 \Delta_{41}.$$
(36)

Of particular interest are NC neutrino interactions, since as the three active flavors participate in the NC interaction at the same rate, the NC sample is insensitive to three-flavor oscillations between ND and FD. However, if a sterile neutrino exists, sterile-neutrino appearance would cause a depletion in the NC channel. The NC survival probability is thus defined as $1 - P(\nu_{\mu} \rightarrow \nu_s)$. Similar to equation (36), this can be written as

$$1 - P(\nu_{\mu} \to \nu_{s}) \approx 1 - 4 |U_{\mu3}|^{2} |U_{s3}|^{2} \sin^{2} \Delta_{31} -4 |U_{\mu4}|^{2} |U_{s4}|^{2} \sin^{2} \Delta_{41} -4 \operatorname{Re}(Z)(\sin^{2} \Delta_{31} - \sin^{2} \Delta_{43} + \sin^{2} \Delta_{41}) -2 \operatorname{Im}(Z)(\sin 2 \Delta_{31} + \sin 2 \Delta_{43} - \sin 2 \Delta_{41}),$$
(37)

where $Z = U_{\mu4}^* U_{s4} U_{\mu3} U_{s3}^*$. The phenomenology of ν_{μ} -CC and NC disappearance driven by sterile neutrino oscillations is complicated at long baselines due to the interference of three-flavor oscillations and sterile oscillations, which do not occur at short baselines. To perform two-detector analyses typical of long-baseline accelerator neutrino experiments, effects at both short and long baselines must thus be understood.

For instance, in the case of the short-baseline oscillation, only probed by the ND, the oscillation probability for NC disappearance is approximately given by

$$1 - P(\nu_{\mu} \rightarrow \nu_{s}) \approx 1 - \cos^{4}\theta_{14}\cos^{2}\theta_{34}\sin^{2}2\theta_{24}\sin^{2}\Delta_{41}, \tag{38}$$

where $\Delta_{ji} = \frac{\Delta m_{ji}^2 L}{4E}$. For the typical beam neutrino energies and ND baselines, when $\Delta m_{41}^2 < 0.05$, oscillations are not visible in the ND. Starting at $\Delta m_{41}^2 \sim 0.5 \text{ eV}^2$, oscillations begin to be visible at low energies in the ND, and as Δm_{41}^2 increases, the first oscillation maximum moves to higher energies. At sufficiently high Δm_{41}^2 values, the entire ND sees rapid oscillations that can no longer be resolved and are seen as a constant normalization shift described by

$$1 - P(\nu_{\mu} \to \nu_{s}) \approx 1 - \frac{1}{2}\cos^{4}\theta_{14}\cos^{2}\theta_{34}\sin^{2}2\theta_{24}.$$
 (39)

For ν_{μ} -CC at the ND, the oscillation probability can be approximated as

$$P(\nu_{\mu} \to \nu_{\mu}) \approx 1 - \sin^2 2\theta_{24} \sin^2 \Delta_{41}, \tag{40}$$

which behaves similarly to NC disappearance except it depends only on θ_{24} , and in the rapid oscillation case the normalization shift is given by $(1/2)\sin^2 2\theta_{24}$.

However, when considering long baselines, terms oscillating at the atmospheric frequency cannot be neglected. Approximating the NC disappearance probability to first order in small mixing angles gives

$$1 - P(\nu_{\mu} \to \nu_{s}) \approx 1 - \cos^{4}\theta_{14}\cos^{2}\theta_{34}\sin^{2}2\theta_{24}\sin^{2}\Delta_{41} - \sin^{2}\theta_{34}\sin^{2}2\theta_{23}\sin^{2}\Delta_{31} + \frac{1}{2}\sin\delta_{24}\sin\theta_{24}\sin2\theta_{23}\sin\Delta_{31}.$$
(41)

In this expression, the first term is identical to the short-baseline approximation. The second and third terms both oscillate at the atmospheric frequency. If $\theta_{34} > 0$, the second term is nonzero, and if $\sin \delta_{24}$ and θ_{24} are non-zero, the third term will not be zero. In either case, this creates an oscillation dip visible at the FD regardless of the value of Δm_{41}^2 . It is notable that this will happen even though NC disappearance cannot occur in the standard three-flavor paradigm. It should also be noted that the third term is CP-odd, since NC disappearance is effectively sterile neutrino appearance, so $\bar{\nu}$ data will also add to the sensitivity of longbaseline experiments to sterile neutrinos. For ν_{μ} -CC disappearance, expanding to second order in small mixing angles one finds

$$P(\nu_{\mu} \to \nu_{\mu}) \approx 1 - \sin^2 2\theta_{23} \cos 2\theta_{24} \sin^2 \Delta_{31} - \sin^2 2\theta_{24} \sin^2 \Delta_{41},$$
(42)

which can be rewritten as

$$P(\nu_{\mu} \to \nu_{\mu}) \approx 1 - \sin^{2} 2\theta_{23} \sin^{2} \Delta_{31} + 2 \sin^{2} 2\theta_{23} \sin^{2} \theta_{24} \sin^{2} \Delta_{31} - \sin^{2} 2\theta_{24} \sin^{2} \Delta_{41}.$$
(43)

The first term is the standard approximation for three-flavor ν_{μ} -CC disappearance. The second term also oscillates as a function of Δm_{31}^2 , the atmospheric frequency, but it is driven by sterile mixing. Even at large Δm_{41}^2 values, this term does not enter into rapid oscillations. Thus, even for large mass splittings where the ND is well inside the rapid oscillation regime, the FD will still show shape variations in addition to normalization changes due to the terms oscillating at the atmospheric frequency with a magnitude that scales with $\sin^2 \theta_{24}$. Furthermore, when considering ν_{μ} disappearance alone, one could define an effective atmospheric mixing angle to account for both sterile and standard oscillations by combining the first two terms in equation (43):

$$\sin^2 2\theta_{23}^{\text{eff}} = \sin^2 2\theta_{23} \cos 2\theta_{24}. \tag{44}$$

While it may seem this would imply that the depth of the atmospheric dip would be insensitive to the large Δm_{41}^2 regime, it actually provides a constraint due to θ_{23}^{eff} having been measured to be close to maximal. Due to the cos $2\theta_{24}$ factor, a non-zero θ_{24} can only drive θ_{23}^{eff} away from maximal.

In addition, experiments that can make precise measurements of ν_e CC interactions can look for $\stackrel{(-)}{\nu}_e$ -CC disappearance at the far detector following

$$P(\stackrel{(-)}{\nu_{e}} \to \stackrel{(-)}{\nu_{e}}) \approx 1 - \sin^{2}2\theta_{13}\sin^{2}\Delta_{31} - \sin^{2}2\theta_{14}\sin^{2}\Delta_{41},$$
(45)

valid only if one assumes sterile-driven muon neutrino disappearance and electron neutrino appearance do not occur.

Finally, it is worth noting the near detectors of long-baseline experiments can be used to conduct short-baseline searches following the same methodologies described in section 4.1.2.1

MINOS/MINOS+. The main injector neutrino oscillation search (MINOS) experiment was a long-baseline neutrino oscillation experiment using the neutrinos at the main injector (NuMI) neutrino beam and two detectors placed within a 735 km baseline. The NuMI beam is produced by collisions with a graphite target of protons accelerated to 120 GeV at Fermilab's Main Injector. The secondary products of these collisions, pions, and kaons, are focused by two parabolic magnetic horns and eventually decay into muons and neutrinos inside a 675 m-long decay pipe filled with helium. The muons are absorbed in the rock and neutrinos continue towards the 1 kton ND, 1 km downstream of the target, and beyond, towards the 5.4 kton FD. The relative positions of target and horns can be changed to tune the beam spectrum to lower or higher neutrino energies. The MINOS run concluded in 2012, with a total exposure of over 15×10^{20} protons-on-target (POT) in neutrino and antineutrino mode since the start of data taking in 2005. The MINOS+ experiment operated the MINOS detectors using the NuMI beam upgraded from 320 to 700 kW of beam power, part of the



Figure 47. The NuMI neutrino energy spectrum for the MINOS+ medium-energy tune, shown as the red solid line. The NO ν A spectrum shown in blue is obtained with the same tune at a 14 mrad offset from the beam axis (MINOS+ is on-axis). For comparison, the spectrum corresponding to the NuMI low-energy tune used by MINOS is shown as the gold histogram. Reprinted from [644], with the permission of AIP Publishing.

NOvA experimental setup. Instead of the low-energy configuration used for the MINOS run, MINOS+ used the NOvA medium-energy configuration, which for MINOS+ corresponds to a neutrino energy spectrum peaked around 7 GeV, as shown in figure 47. Exposure of the MINOS+ detectors to a beam peaked above the three-flavor oscillation maximum provided excellent sensitivity to new physics through precise measurements of muon neutrino disappearance between the ND and FD. MINOS+ operated from 2014 through 2016 and accumulated only neutrino mode data. A combined analysis of the 10.6 $\times 10^{20}$ POT of MINOS neutrino data and 5.8 $\times 10^{20}$ POT of MINOS+ neutrino data using a two-detector fitting technique placed stringent limits on sterile driven muon neutrino disappearance within a 3 + 1 model, as shown in figure 48 [139]. The null results from MINOS/MINOS+ are one of the primary drivers, along with IceCube results, of the large tension between appearance and disappearance data when attempting to explain observations purely through sterile neutrino mixing. This is further evidenced in combinations with reactor data looking for electron (anti)neutrino disappearance, as described in section 4.1.3.4

NOvA. The NuMI off-axis ν_e appearance (NOvA) experiment is a long-baseline accelerator neutrino experiment based at Fermilab and the Far Detector Laboratory at Ash River, Minnesota. NOvA has as its primary goal to measure three-neutrino mixing parameters, including the determination of the neutrino mass ordering, by looking for the appearance of electron neutrinos or antineutrinos, and the disappearance of muon neutrinos or antineutrinos, using the NuMI neutrino beam produced at Fermilab. This is accomplished by using two detectors separated by 810 km, placed 14 mrad off the NuMI beam axis. Due to the off-axis placement, the detectors sample a narrow range of neutrino energies between 1 and 4 GeV, peaking at 2 GeV as shown in figure 49. This configuration is chosen to drastically reduce the feed-down of NC interactions of higher-energy neutrinos, which typically represent the dominant background to the measurement of ν_e CC interactions in on-axis experiments. The 0.33 kton ND is located underground next to the MINOS ND hall at Fermilab, while the 14 kton FD is positioned at the surface in Ash River, Minnesota. Both detectors are composed of extruded 32-cell PVC modules filled with liquid scintillators. The



Figure 48. The MINOS and MINOS+ 90% Feldman–Cousins exclusion limit compared to the previous MINOS result [645] and results from other experiments. The Gariazzo *et al* region is the result of a global fit to neutrino oscillation data [646]. Reproduced from [139]. CC BY 4.0.

cells are read out by 32-pixel avalanche photodiodes (APDs). NOvA began collecting data in 2014 and has so far accumulated large samples in both neutrino-dominated and antineutrino-dominated modes.

NOvA placed constraints on sterile neutrinos via searches for differences in the rate of NC neutrino interactions between the Near and Far detectors. The analysis was based on 6.05×10^{20} protons-on-target taken in neutrino-dominated mode, and 95 NC candidates were selected at the Far Detector compared with $83.5 \pm 9.7(\text{stat.}) \pm 9.4(\text{syst.})$ events predicted assuming mixing only occurs between active neutrino species. Therefore, NOvA found no evidence of active-sterile neutrino mixing. Interpreting these results within a 3 + 1 model results in constraints on the sterile mixing angles of $\theta_{24} < 20.8^{\circ}$ and $\theta_{34} < 31.2^{\circ}$ at the 90% C. L. for 0.05 eV² $\leq \Delta m_{41}^2 \leq 0.5$ eV², the range of mass splittings for which no significant oscillations over the ND baseline are expected [288]. The energy spectrum of the NC selected candidates at the Far detector is shown along with a comparison of the allowed regions for the sterile matrix elements obtained by NOvA with similar constraints by SuperK and IceCube-DeepCore in figure 50.

NOvA has also reported results on the first search for sterile antineutrino mixing in an antineutrino beam, using an exposure of 12.51×10^{20} protons-on-target from the NuMI beam at Fermilab running in antineutrino-dominated mode. NOvA observed 121 NC antineutrino candidates at the FD, compared to a prediction of $122 \pm 11(\text{stat.}) \pm 15(\text{syst.})$ assuming mixing between only three active flavors. Therefore, no evidence for $\bar{\nu}_{\mu} \rightarrow \bar{\nu}_{s}$ oscillations is observed. In this case the 3 + 1 model constraints on the mixing angles are found to be $\theta_{24} < 25^{\circ}$ and $\theta_{34} < 32^{\circ}$ at the 90% C.L. for $0.05 \text{ eV}^2 \leq \Delta m_{41}^2 \leq 0.5 \text{ eV}^2$ [289]. The



Figure 49. The plot on the left displays the predicted neutrino energy spectra of the NuMI beam during the NO ν A run, including the on-axis spectrum sampled by MINOS + as the dotted line, as well as the spectrum at different off-axis positions. The 14 mrad off-axis position of the NOvA detectors is identified by the red line. The plot on the right shows neutrino energy as a function of the parent pion energy and the angle between the neutrino and the decaying pion. Reproduced with permission from [647].

antineutrino energy spectrum at the FD and obtained allowed regions are shown in figure 51. Finally, NOvA has recently presented results from a two-detector fit at Neutrino 2022, using similar techniques to the MINOS/MINOS+ experiment, extending limits on the θ_{24} and θ_{34} angles over a large range of Δm_{41}^2 values. These results are presented in figure 52.

T2K. The Tokai-to-Kamioka experiment [290] is a neutrino oscillation experiment based in Japan. T2K's primary goal is to measure three flavor neutrino oscillation parameters. An intense source of (anti)neutrinos is produced at J-PARC and is directed toward a series of detectors placed 280 m from the target, and a massive detector 295 km away (Super-Kamiokande, SK [291]). The neutrino beam composition is predominantly muon neutrino flavor, with a small (0.5%) fraction of electron neutrinos [292]; the beamline elements can be configured to produce a predominantly antineutrino beam as well. The 'near detectors' include the INGRID detector [293], WAGASCI detectors [294, 295], and ND280 detector suite. These detectors observe interactions from slightly different neutrino energy spectrums, with INGRID's peak neutrino energy approximately at 1 GeV (on-axis), ND280 at 0.6 GeV (2.5 degrees off-axis), and WAGASCI at 0.8 GeV (1.5 degrees off-axis). T2K has operated since 2009 and has produced a series of results on light sterile neutrinos using the ND280 detectors and SK with subsets of the data taken. T2K plans to install improvements to the ND280 detector [296], and, with improvements to the beamline, T2K will have further opportunities for updated or expanded analyses between detectors.

CC electron neutrino interactions were selected in ND280 to test for $3 + 1 \nu_e$ disappearance, motivated by radiochemical experiments [71, 72, 297] and discrepancies in reactor experiments energy spectra [66]. Isolating CC ν_e candidates was challenging, with significant backgrounds to the ν_e selection from photons (from, for example, the inactive material surrounding the active scintillator target). These backgrounds were controlled by dedicated photon selections and with additional systematic uncertainty estimation on the production of π^0 from neutrino interactions on non-scintillator materials. The analysis also



Figure 50. Left: calorimetric energy spectrum for NC neutrino candidates in the NOvA FD data compared with three-flavor prediction; right: the NOvA 68% (dashed) and 90% (solid) Feldman–Cousins non-excluded regions (shaded) in terms of $|U_{\mu4}|^2$ and $|U_{\tau4}|^2$, where it is assumed that $\cos^2 \theta_{14} = 1$ in both cases, compared to SuperK and IceCube-DeepCore constraints. Reprinted (figure) with permission from [288], Copyright (2017) by the American Physical Society.



Figure 51. Left: calorimetric energy spectrum for NC antineutrino candidates in the NOvA FD data compared with three-flavor prediction; Right: the NOvA 68% (dashed) and 90% (solid) Feldman–Cousins non-excluded regions (shaded) in terms of $|U_{\mu4}|^2$ and $|U_{\tau4}|^2$, where it is assumed that $\cos^2 \theta_{14} = 1$ in both cases, compared to SuperK and IceCube-DeepCore constraints. Reproduced from [289]. CC BY 4.0.

assumes no ν_{μ} disappearance, and systematic uncertainty from the flux and cross section models were reduced using dedicated ν_{μ} selections as is done for T2K three flavor oscillation analyses. T2K excludes regions of interest at 95% CL with 5.9×10^{20} POT between approximately $\sin^2 2\theta_{ee} > 0.3$ and $\Delta m_{eff}^2 > 7 \text{ eV}^2$ [298]. An updated analysis of large mixing regions would be useful in light of recent MicroBooNE results [280].

A search for light sterile neutrinos using SK has been performed in 2019 [299], using $14.7 \times 10^{20} (7.6 \times 10^{20})$ POT in neutrino (antineutrino) mode. It is focusing on a 3 + 1 model where a single sterile neutrino is added and assumed to be mixing with ν_2 and ν_3 mass eigenstates through new parameters θ_{24} and θ_{34} (T2K has limited sensitivity to other parameters, that are thus neglected $\theta_{14} = \delta_{14} = \delta_{24} = 0^{\circ}$).

The analysis consists of a simultaneous fit of the CC muon, electron and NC neutrino samples. While CC channels are mainly sensitive to the new mass splitting Δm_{41}^2 and to θ_{24} , the NC channel measures the active neutrino survival probability and is also sensitive to θ_{34} . The five CC analysis samples are the same as the one used in standard oscillation analyses [300], but it has been the first time NC π^0 (single π^0 production where the pion decays and produces two Cherenkov rings) and NC γ -deexcitation samples are used in oscillation measurements.



Figure 52. Left: NOvA 90% Feldman–Cousins excluded region in $\sin^2 \theta_{24}$ versus Δm_{41}^2 , obtained from a two-detector fitting method, compared to allowed and exclude regions from other experiments. Right: NOvA 90% Feldman–Cousins excluded region in $\sin^2 \theta_{34}$ versus Δm_{41}^2 , compared to limits reported by other experiments. Reproduced from [648]. CC BY 4.0.

Most of the systematic parameters are constrained in the same way as in the 3 flavor analysis (using e.g. ND constraints), but additional 30% normalization uncertainties are added for the new NC samples. The joint maximum-likelihood fits allows to draw exclusion limits in the $(\sin^2 \theta_{24}, \Delta m_{41}^2)$ plane: the most stringent limits on θ_{24} are obtained for $\Delta m_{41}^2 < 3 \times 10^{-3} \text{ eV}^2$. Similarly, limits in the $(\sin^2 \theta_{24}, \cos^2 \theta_{24} \sin^2 \theta_{34})$ plane have been obtained although more statistics in the NC samples and additional systematic studies are needed to further improve the measurement. A comparison of the exclusion regions obtained with other existing constraints is shown in figure 53.

One limitation of work to date on T2K is the completeness of the assessment of interaction model uncertainties as applied to short-baseline analyses. T2K analyses so far assume no ν_{μ} disappearance, however the interaction model systematic uncertainties are assessed based on external and ND280 measurements. Those measurements are placed close to production and therefore could be sensitive to a ν_{μ} disappearance signal, potentially biasing a dedicated ν_{μ} disappearance search. T2K studied the possible impact of a subset of interaction model uncertainties on a ND280 ν_{μ} disappearance result [301] and found it to be robust, but this does not consider a full re-assessment of where external data is used to inform the model. Current efforts in T2K cross-section measurements and the implementation of *ab initio* computations in the context of three-flavor analysis would greatly benefit such studies as well. Certain event selections, like coherent neutrino interactions or low ν selections [302] may have better theoretical understanding for a single detector analysis, but can be challenging to use due to statistics or acceptance for T2K.

In the future, joint analyses, including ν_{μ} disappearance, ν_{e} appearance/disappearance may also be performed either between several near detectors (with different technologies and/



Figure 53. The T2K 90% CL. exclusion limits on $\sin^2 \theta_{24}$ as a function of Δm_{41}^2 compared to other experiments. Areas on the right are excluded. Reproduced from [299]. CC BY 4.0.

or neutrino energy spectra) or benefiting from the complementary coverage of near versus far facilities. The upgraded near detector, for example, will have improved detection thresholds and sensitivity to neutrons, possibly enabling the selection of NC events at the near site.

OPERA and ICARUS. The Oscillation Project with Emulsion-tRacking Apparatus (OPERA) was a long-baseline accelerator neutrino experiment that sampled the CERN Neutrinos to Gran Sasso (CNGS) beam with a detector placed at the Gran Sasso laboratory (LNGS) 730 km from the neutrino production source, having as primary physics goal to observe $\nu_{\mu} \rightarrow \nu_{\tau}$ appearance. This goal required the use of emulsion detection technology, such that the detector was a hybrid apparatus made of a nuclear emulsion/lead target complemented by electronic detectors. The detector 'target' region had a total mass of about 1.25 kt and was composed of two identical sections, each with 31 walls of emulsion cloud chamber bricks, interleaved by planes of horizontal and vertical scintillator strips used to select bricks in which a neutrino interaction had occurred. The exquisite spatial resolution afforded by the emulsion layers enabled the identification of the characteristic 'kink' due to the decay of the final-state τ particle from a ν_{τ} interaction. The target region was complemented by a magnetized muon spectrometer.

Reconstruction of neutrinos interacting within the emulsion layers was conducted by automated scanning robots. Due to the high-energy tau production threshold ($E_{\nu_{\tau}} \gtrsim 3.5$ GeV), the CNGS neutrino flux was distributed at higher energies than other LBL experiments, as shown in figure 54 [303]. OPERA collected CNGS beam data from 2008 to 2012, with an integrated exposure of 17.97×10^{19} protons-on-target. A total of 19 505 neutrino interaction events in the detector target were recorded by the electronic detectors, of which 5603 were fully reconstructed in the emulsion layers.



Figure 54. Fluxes of the different CNGS beam neutrino components at LNGS. Reproduced from [303]. © 2009 IOP Publishing Ltd and SISSA. All rights reserved.

OPERA used its full data sample to conduct a search for sterile neutrino mixing within the context of a 4-flavor (3 active + 1 sterile) model based on both the ν_{τ} and ν_{e} appearance channels, with the ν_{μ} CC disappearance channel not being considered given the low sensitivity to sterile mixing for that channel due to ambiguities with potential NC disappearance [34]. Defining $\sin^{2}2\theta_{\mu\tau} = 4|U_{\tau4}|^{2}|U_{\mu4}|^{2}$, $\sin^{2}2\theta_{\mu e} = 4|U_{e4}|^{2}|U_{\mu4}|^{2}$, exclusion regions of Δm_{41}^{2} versus $\sin^{2}2\theta_{\mu\tau}$ and $\sin^{2}2\theta_{\mu e}$ were computed and are shown in figure 55. The result is restricted to positive Δm_{41}^{2} values since negative values are disfavored by results on the sum of neutrino masses from cosmological surveys [304]. The results are consistent with no active-sterile neutrino mixing. For $\Delta m_{41}^{2} > 0.1 \text{eV}^{2}$, the upper limits on $\sin^{2}2\theta_{\mu\tau}$ and $\sin^{2}2\theta_{\mu e}$ are set to 0.10 and 0.019 both for the case of Normal Ordering and Inverted Ordering. The values of the oscillation parameters ($\Delta m_{41}^{2} = 0.041 \text{eV}^{2}$, $\sin^{2}2\theta_{\mu e} = 0.92$) corresponding to the MiniBooNE combined neutrino and antineutrino best-fit [32] are excluded with a *p*-value of 8.9×10^{-4} , corresponding to a significance of 3.3σ .

The ICARUS (imaging cosmic and rare underground signals) experiment operated the T600 liquid argon time projection chamber detector, with a fiducial mass of 447 tons, exposed to the same CNGS beam as OPERA, and located at the same 730 km from the neutrino production target in the LNGS. The baseline and CNGS beam typical energy determine that ICARUS is primarily sensitive to sterile mixing in the region where $L/E \sim 36.5$ km GeV⁻¹. ICARUS conducted a search for sterile neutrino mixing by looking for excess ν_e appearance in the data sample collected in 2010 and 2011. This search assumed a simplified two-flavor model (one active, one sterile) and found no evidence of sterile neutrino oscillations, as detailed in [305]. However, it was pointed out soon after the publication of the ICARUS results that the limits obtained on $\sin^2 2\theta_{\mu e}$ by using a two-flavor approximation (which works well for a short-baseline experiment) in a long-baseline setup neglects sizable four-flavor effects, induced by the interference of the new large squared-mass splitting Δm_{41}^2 with the atmospheric one. The analysis also neglected contributions to the four-flavor oscillation probabilities arising from the intrinsic ν_e component of the CNGS beam. It is estimated that these four-flavor effects weaken the reported ICARUS constraints by up to a factor of 3 [306]. The ICARUS T600 detector has since been moved to Fermilab, where it will be operated as the 'Far' detector for Fermilab's Short-Baseline Neutrino program. ICARUS completed



Figure 55. Left: OPERA 90% C.L. exclusion region in the Δm_{41}^2 and $\sin^2 2\theta_{\mu\tau}$ parameter space for the normal (NH, solid line) and inverted (IH, dashed line) hierarchy of the three standard neutrino masses. The exclusion regions by NOMAD [649] and CHORUS [650] are also shown. Right: OPERA 90% C.L. exclusion region in the Δm_{41}^2 and $\sin^2 2\theta_{\mu e}$ plane is shown for the normal (NH, solid line) and inverted (IH, dashed line) hierarchy of the three standard neutrino masses. The plot also regions in the Δm_{41}^2 and $\sin^2 2\theta_{\mu e}$ plane is shown for the normal (NH, solid line) and inverted (IH, dashed line) hierarchy of the three standard neutrino masses. The plot also reports the 90% C.L. allowed region obtained by LSND [19] (cyan) and MiniBooNE combining ν and $\bar{\nu}$ mode [32] (yellow). The blue and red lines represent the 90% C.L. exclusion regions obtained in appearance mode by NOMAD [285] and KARMEN2 [20], respectively. The 90% C.L. exclusion region obtained in disappearance mode by the MINOS and DayaBay/Bugey-3 joint analysis [324] is shown as green line. The black star (*) corresponds to the MiniBooNE 2018 best-fit values for the combined analysis of ν and $\bar{\nu}$ data. Reproduced from [34]. CC BY 4.0.

installation at Fermilab in Spring 2022 (the last section of the cosmic ray tagger and overburden) and is beginning its first physics run. The prospective ICARUS contributions to the Fermilab SBN program are detailed in section 6.2.1.1.

4.1.3. Reactor neutrino experiments. A conclusive way to test whether RAA is due to mixing between active and sterile neutrinos is by searching for baseline-dependent sterile neutrino-induced spectral variations. In this section, we summarize the results from the experiments that performed a relative spectral search for $3 + 1 \bar{\nu}_e$ oscillations induced by active to sterile mixing. Relative oscillation searches curtail the dependence on the reactor $\bar{\nu}_e$ spectrum model and detection efficiency by comparing the ratio of energy spectrum at different baselines to the corresponding predicted ratio under the oscillation hypothesis. Baseline-dependent spectral measurement is done either by using a segmented detector, a single-volume movable detector, or by the use of multiple detectors placed at different baselines. At reactor $\bar{\nu}_e$ energies, since eV-scale neutrinos induce oscillations at meter scales,

purpose-built experiments place their detectors as close to the reactor as possible, ideally <10 m. Furthermore, experiments built to search for θ_{13} are also sensitive to sterile neutrino oscillations but at lower oscillation frequencies, typically for $\Delta m_{13}^2 < 0.1 \text{eV}^2$.

All the experiments discussed in this section use scintillator detectors and the IBD interaction process, where a characteristic signature is provided by the timing and spatial coincidence between prompt positron annihilation signal and delayed neutron capture signal producing two separate flashes ($\Delta t \sim \mu s$). Refer to section 2.3 for more details on the IBD mechanism. Experiments also typically use an external neutron capture agent (other than hydrogen), either Gadolinium or ⁶Li, to increase IBD efficiency by increasing the neutron capture cross-section. ⁶Li has the added benefit of constraining the spatial extent of the delayed signal since the capture products have high dE/dx in scintillators. In addition to the delayed coincidence, experiments also use various combinations of active and passive shielding, detector segmentation, and pulse shape discrimination (PSD) for background reduction. PSD-capable scintillators generate pulse shapes that are particle-dependent and could in principle be used for both prompt and delayed selection. Whenever possible, experiments also measure reactor-off data to constrain reactor-uncorrelated IBD-mimicking backgrounds. Following is a discussion of the individual reactor experiments that searched for sterile neutrino-induced oscillations. A list of purpose-built experiments and their features pertinent to the eV-scale oscillation search can be found in table 12.

4.1.3.1. Short-baseline experiments. **DANSS**. The Detector of AntiNeutrinos based on Solid State Scintillator (DANSS) experiment [307] consists of 1 m³ highly segmented, plastic scintillator detector with each scintillator bar wrapped in Gd-loaded reflector. Light and signal readout is performed using wavelength-shifting fibers and a combination of pixel photon detectors and PMTs operated at room temperature. The experiment samples $\bar{\nu}_e$ from a 3 GW_{th} low enriched uranium (LEU) reactor at Kalinin Nuclear Power Plant in Russia. As opposed to most other reactor $\bar{\nu}_e$ experiments, the DANSS detector is located below the reactor core, benefiting from its 50-meter water equivalent (m.w.e) overburden. The detector is placed on a lifting platform which enables measurements at the baselines of 10.9 m, 11.9 m, and 12.9 m. Due to its close proximity to the reactor core, the detector measures a high $\bar{\nu}_e$ flux of around 4000 events per day.

The DANSS detector was commissioned in 2016 and started taking data in October 2016. The experiment has collected around 4 million signal events over three years at three different positions. The biggest source of background that constitutes 13.8% of the IBD signal events is the accidental coincidence background, along with non-negligible cosmogenic backgrounds and IBD backgrounds from neighboring reactors. The data excludes a large area in sterile neutrino parameter space (Δm^2 , $\sin^2 2\theta_{ee}$) and most interestingly excludes the best-fit point of RAA and Ga experiments at more than 5σ level as shown in figure 56. Additionally, the IBD positron spectrum was measured and compared to the simulated Huber-Mueller spectrum. Although the measured spectrum disagrees with DANSS's Huber-Mueller-derived prediction, final oscillation measurements are largely independent of the choice of models due to the ratios taken at different baselines [307].

NEOS. The Neutrino Experiment for Oscillation at Short baseline (NEOS) aims to search for light sterile neutrinos by detecting electron antineutrinos (v_e) from a reactor at a very short baseline. The NEOS detector consists of a liquid scintillator (LS) target (~1000 liter with ~0.5% Gd loading by weight), two buffer zones filled with mineral oil where 19 8 inch PMTs per a zone are attached, and a muon veto system. NEOS had been installed in the tendon gallery of the fifth reactor of the Hanbit Nuclear Power Plant in Korea for two periods, from Aug. 2015 to May 2016 (NEOS-I) and from September 2018 to October 2020 (NEOS-

Table 12. Details of the short-baseline experiments that have been specifically built to search for eV-scale sterile neutrinos. Short baselines are preferable for oscillation searches involving eV-scale sterile neutrinos. HEU reactors are typically $\sim 10x$ smaller than commercial LEU power reactors and are preferred for eV-scale oscillation searches since they have smaller source positions oscillation washout. Detector segmentation, capture agent, and pulse shape discrimination capabilities are highly beneficial for background reduction.

8	Experiment	Baseline (m)	Reactor type	Reactor power (MW <i>th</i>)	Detector Size	Target	Sterile ν Search Strategy
•	DANSS [682]	11–13	LEU	3000	1 m ³	Segmented PS with Gd coating	Multi-site
	NEOS [133]	24	LEU	2800	1 m ³	Single-volume GdLS + PSD	Single-site
	Neutrino-4 [498]	6-12	HEU	90	2 m^3	Segmented GdLS	Multi-site/zone
	PROSPECT [134]	7–9	HEU	85	4 m^3	Segmented ⁶ LiLS + PSD	Multi-zone
	STEREO [656]	9–11	HEU	57	2 m ³	Segmented GdLS + PSD	Multi-zone



Figure 56. Left panel: exclusion curves at 90% C.L. (filled area) and 90% C.L. sensitivity contours (dashed line) are shown. Expected regions from RAA and GA are also shown. Reproduced from [651]. © IOP Publishing Ltd. All rights reserved. Right panel: exclusion curves for 3 + 1 neutrino oscillations in the $\sin^2 2\theta_{14}$, Δm_{14}^2 parameter space obtained by NEOS. Reprinted (figure) with permission from [133], Copyright (2017) by the American Physical Society.

II). For the NEOS-II, the modifications include newly produced target GdLS and a minor modification in the muon veto system. With a 24 m baseline from the core of the reactor (2.8 GW_{th} in 100%) about 2000 IBD, $v_e + p \rightarrow e^+ + n$, events per day are observed. The signal-to-background ratio is 22 thanks to relatively good overburden (~20 m.w.e.). Background contribution from the nearest neighboring reactor (d = 256 m) is found to be less than 1% of the total v_e flux from the fifth reactor. Calibration data using radioactive sources, ²²Na, ¹³⁷Cs, ⁶⁰Co, PoBe, and ²⁵²Cf, had been taken regularly.

NEOS-I [133] using 180 (46) live days of reactor-ON (-OFF) data excluded the RAA best fit at 90% C.L. by comparing the prompt energy spectrum of Daya Bay where the sterile neutrino oscillation effect is averaged out. The best-fit was found at $(\sin^2 2\theta, \Delta m^2) = (0.05, 1.70 \text{ eV}^2)$ with χ^2/NDF for 3ν and 4ν are 64.0/61, and 57.5/59, respectively. The corresponding p-value is estimated to be 22%. The well-known '5 MeV excess' was clearly observed as well in NEOS-I. Recently a joint analysis [308] between NEOS-I and RENO was performed, yielding a slightly improved result beyond its previous result using early Daya Bay data. More details on the NEOS-I and RENO joint analysis is discussed in the following sub-section (section 4.1.3.3).

Neutrino-4. Neutrino-4 is the only reactor neutrino experiment performing relative spectral comparison that has reported evidence of oscillations. The detector is a 1.8 m³ Gd-doped liquid scintillator detector divided into 10 rows each row consisting of 5 sections each of size 0.225 m × 0.225 m × 0.85 m. The detector samples $\bar{\nu}_e$ from the 57 MW_{th} SM-3 reactor, and HEU in Dimitrovgrad, Russia. A movable platform enables the detector to sample baselines from 6 m–12 m. The short reactor on (off) cycles of 8–10 (2–5) days enable the experiment to perform rapid signal and background measurements.

The detector collected data for five years from 2016 to 2021 with \sim 300 events/day. The latest analyzed dataset consists of 720 (860) calendar days of reactor on (off) data with a



Figure 57. Left: 1σ (blue), 2σ (green), and 3σ (yellow) suggested regions from the Neutrino-4 sterile neutrino oscillation search. Right: L/E distribution of background-subtracted IBD rates reported by the Neutrino-4 Experiment. Reproduced from [498]. CC BY 4.0.

signal-to-background ratio of ~0.5. The experiment performed a sterile neutrino search in the *L/E* space and observed evidence for 2.9σ neutrino oscillation effect with the best-fit at $\Delta m_{14}^2 = 7.3 \pm 0.13$ (stat) ± 1.16 (syst) and $\sin^2 \theta_{14}$ =0.36 ± 0.12 (stat) (figure 57). It is important to point out that these results are widely debated by several groups [109, 309–311]. Additionally, PROSPECT and STEREO experiments [312, 313] also individually disfavor the best-fit point and a significant portion of the Neutrino-4 suggested parameter space.

PROSPECT. The Precision Reactor Oscillation and SPECTrum (PROSPECT) experiment [314, 315] is a U.S.-based reactor neutrino experiment installed at short baselines of 6.7–9.2m from the 85 MW High Flux Isotope Reactor (HFIR) at Oak Ridge National Laboratory. The 4-ton ⁶Li-loaded PSD-capable liquid scintillator [316] detector is composed of a two-dimensional grid of optically isolated segments [312, 317] as shown in figure 58. Low mass, highly reflective, rigid separators [318] were used to achieve segmentation and PMTs enclosed in mineral oil-filled acrylic housings were installed at either ends of the segments for signal readout.

The detector shielding was optimized based on the neutron and γ ray background measurements performed at HFIR [319] and consisted of top-heavy hydrogenous shielding to reduce cosmogenic backgrounds and a fixed local lead shield to mitigate reactor-specific backgrounds. PROSPECT was the first on-surface reactor neutrino experiment to achieve a signal-to-background (S:B) > 1, thanks to the high background suppression enabled by the detector design.

PROSPECT was commissioned in February 2018 and started collecting data in March. The first oscillation search result was published in 2018 [134] with a relatively small dataset of 33 (28) reactors on (off) live days which was followed by a result with a longer dataset consisting of 96 (73) reactor on (off) live days composed of >50k signal events in 2020. The oscillation search was done by performing a relative comparison of baseline-dependent spectra minimizing the dependence on the reactor neutrino model [135].

A combination of the compact HFIR reactor core (a cylinder of diameter 0.435 m and height of 0.508 m) with the fine detector segmentation (14.5 cm × 14.5 cm cross-section) enabled high sensitivity to oscillation frequencies of $\Delta m^2 > 1 \text{ eV}^2$. PROSPECT observed no statistically significant indication of $\bar{\nu}_e$ to sterile neutrino oscillations. Using Feldman–Cousins



Figure 58. Left: schematic of the PROSPECT detector, shielding, and electronics in the HFIR Experiment Room. The center of the detector is located at a distance of 7.84 m from the center of the reactor core (red). Right: Cutout of the PROSPECT active segmented detector enclosed in the containment and shielding. Also shown is the vertical local lead shield adjoining the reactor pool wall that was installed to mitigate the reactor-related backgrounds. The figure was a modified version of the figure. Reproduced from [533]. © IOP Publishing Ltd. All rights reserved.

technique [320], PROSPECT excluded a significant portion of the RAA suggested parameter space at 95% CL and the best-fit point at 2.5σ as shown in figure 59. A complementary technique using the Gaussian CLs method [321] also provides similar exclusion.

The PROSPECT detector was decommissioned in 2020 after an unexpected HFIR downtime. During the course of data taking, it was also observed that a number of segments have slowly lost functionality due to failures of PMT electronics induced by the ingress of the liquid scintillator into the PMT housings. Due to these reasons, the results shown above–corresponding to the full PROSPECT dataset–are primarily dominated by statistics. However, the collaboration is carrying out two major analysis modifications by leveraging distinctive detector features. (a) Splitting the data-taking period into multiple time-frames and (b) allowing the use of segments with single PMTs. The former would allow for an increase in statistics by allowing segments with non-functioning PMTs to be used for part of the dataset and the latter increases the S:B by improving particle identification and consequently reduces the backgrounds. Each of the analysis improvements is individually expected to increase the effective statistics by \sim 50% and enable considerable improvement in the statistical power of the PROSPECT's oscillation search.

STEREO. The STEREO experiment [322] is a Gd-doped liquid scintillator detector located at the ILL in Grenoble, France. The detector uses a high-flux 58 MW_{th} research reactor that consists of highly enriched Uranium. The detector is located 9 m away from the reactor core and is a segmented detector where the position of different segments of the detector serves as different baselines varied between 9 and 11 m as shown in figure 60 (left).

The IBD signal events are selected using a cut-based approach where selection cuts on energy and time variables are optimized by trading off between detection efficiency and background rejection. Furthermore, antineutrino signal rates separated from the remaining background using pulse shape discrimination (PSD) where pulses generated from neutrons have longer tails compared to that of gamma. Hence, a ratio of the pulse tail to total charge is used to mitigate neutron-related background.

Major Report



Figure 59. Results from PROSPECT's search for $\bar{\nu}_e$ to sterile neutrino oscillations. Exclusion contours were drawn using Feldman–Cousins (black) and Gaussian CLs (red) methods. Green and yellow bands show the 1σ and 2σ PROSPECT sensitivities to the sterile neutrino oscillations. Also shown for comparison are the RAA-suggested parameter space and the best-fit point. Reproduced from [135]. CC BY 4.0.



Figure 60. (Left) STEREO setup. 1–6: target cells (baselines from core: 9.4–11.1 m). Reproduced from [322]. CC BY 4.0. (Right) Exclusion contour (red) and exclusion sensitivity (blue) at 95% C.L. Overlaid are the allowed regions of the RAA (grey) and its best fit point (star). Reproduced from [313]. CC BY 4.0.

The test of sterile neutrino oscillations is performed using ratios of energy spectrum at six different segments to that of the first segment and therefore making the measurement independent of absolute normalization and of the prediction of the reactor spectrum. With 273 (520) days of reactor-on (-off) data, STEREO found no evidence for sterile neutrino oscillations, and the results are compatible with the null oscillation hypothesis. The data excluded RAA best-fit point at *p*-value $<10^{-4}$ and exclusion curve with 3 + 1 neutrino oscillations scenario is shown in figure 60 (Right).



Figure 61. Left: Feldman–Cousins (FC) 90% C.L. and 90% CL_s exclusion regions from an oscillation analysis of 1230 d of Daya Bay data. The dashed red line shows the 90% C.L. median sensitivity along 1 σ and 2 σ bands. The excluded region for the original Bugey-3 limit [652] is shown in green, while the resulting CL_s contour from Daya Bay and its combination with the reproduced Bugey-3 results are shown in grey and black, respectively. Reproduced from [325]. CC BY 4.0. Right: RENO's 95% C.L. exclusion contour for the oscillation parameters $\sin^2 2\theta_{14}$ and $|\Delta m_{41}^2|$. The black solid contour represents an excluded region obtained from spectral distortion between near and far detectors. The green shaded band represents expected 1 σ exclusion contours due to a statistical fluctuation. The blue dotted contour represents its median. Reproduced from [653]. CC BY 4.0.

4.1.3.2. Medium-baseline experiments. Daya Bay. Daya Bay's unique configuration makes it an excellent experiment to search for sterile-active neutrino mixing [132, 323–325]. In this experiment, electron antineutrinos emitted from six 2.9 GW_{th} nuclear reactors are detected in eight identically-designed antineutrino detectors (ADs) placed underground in two near experimental halls (EHs) and one far hall. The two near halls, EH1, and EH2, are located \sim 350–600 m away from the reactors, whereas the far hall, EH3, is located \sim 1500–1950 m from the reactors.

Daya Bay's latest constraints in the $(\sin^2 2\theta_{14}, \Delta m_{41}^2)$ parameter space, obtained with a 1230 d data set, are shown in figure 61 [325]. Two complementary analysis methods are used to set the exclusion contours, one relying on the Feldman–Cousins (FC) frequentist approach and the other on the CL_s approach. Daya Bay is most sensitive to $\sin^2 2\theta_{14}$ in the 10^{-3} eV² $\leq |\Delta m_{41}^2| \leq 0.3$ eV² region, where a distortion from the standard three-neutrino oscillation framework would be visible through a relative comparison of the rate and energy spectrum of reactor antineutrinos in the different EHs. For $|\Delta m_{41}^2| \geq 0.3$ eV², the oscillations are too fast to be resolved, and the sensitivity arises primarily from comparing the measured rate with the expectation, resulting in less stringent limits.

Daya Bay ceased operations in December 2020 after collecting data for over 3000 d. Throughout this time it accumulated the largest sample of reactor antineutrinos to date, consisting of more than six million events. This sample is still being analyzed and final results are expected to be released by early 2023. The sizable increase in statistics, combined with potential reductions in systematic uncertainties, implies that significant improvements over

the existing limits are expected. The new constraints will likely remain the best in the world for the foreseeable future in the $|\Delta m_{41}^2| \lesssim 0.3 \text{ eV}^2$ region, which no experiment in the horizon is expected to cover at the time of writing.

Double chooz. Double Chooz [326] consists of two nearly identical gadolinium-doped liquid scintillator detectors located close to the nuclear power plant comprising two 4.25 GW nuclear reactors. The near (far) detectors are located underground at an overburden of 120 m (300 m) at a distance of 469 m and 355 m (1115 m and 998 m) from the two reactor cores. The detector-reactor locations are such that the relative contributions to both the detectors from the reactors are very similar which helps reduce reactor-related uncertainties.

This analysis includes three datasets amounting to a total 5 year long dataset. The first (FD-I) dataset consists of 455.21 d of livetime collected with the far detector before the commissioning of the near detector. The second (FD-II) and third (ND) datasets are collected during the same period of time and consists of 362.97 d and 257.96 d of livetime respectively. The livetime for ND is lower than FD-II because of the higher muon rate causing larger deadtime in the near detector. In order to obtain a measurement independent of absolute flux predictions, the experiment directly compares the event rates measured in the two identical detectors which helps in canceling most of the reactor flux and detection efficiency-related uncertainties. The experiment does not see any indications of sterile neutrino oscillations and set exclusion limits in similar regions of Δm_{14}^2 as the other θ_{13} experiments.

RENO. The RENO collaboration has reported a search result for light sterile neutrino oscillations. The search is performed using six 2.8 GW_{th} reactors and two identical detectors located at 294 m (near) and 1383 m (far), respectively, from the center of six reactor cores at the Hanbit Nuclear Power Plant Complex in Yonggwang. The reactor flux-weighted baseline is 410.6 m for the near detector and 1445.7 m for the far detector, respectively. The near (far) underground detector has 120 m (450 m) of water equivalent overburden. A spectral comparison between near and far detectors was performed to search for reactor 3 + 1 light sterile neutrino oscillations [327–331].

The RENO sterile analysis uses roughly 2200 live days of data taken in the period between August 2011 and February 2018 amounting to 850 666 (103 212) $\overline{\nu}_e$ candidate events in the near (far) detector. The details of pull terms and systematic uncertainties are described in [331]. Exclusion regions at 95% confidence level are set for $\Delta \chi^2 > 5.99$ and are shown in figure 61. Exclusion contours obtained using the Gaussian CL_s method [321, 332] show negligible difference with the pictured $\Delta \chi^2$ method. Figure 61 shows the 95% C.L. exclusion contour and median sensitivity including 1 σ band due to statistical fluctuations.

The limit of $\sin^2 2\theta_{14}$ is mostly determined by a statistical uncertainty, while the systematic uncertainties dominate in the $|\Delta m_{41}^2| \lesssim 0.06 \text{ eV}^2$. The uncertainty of background is a dominant systematic source in the $0.003 \lesssim |\Delta m_{41}^2| \lesssim 0.06 \text{ eV}^2$ region, and the energy-scale uncertainty is a major limiting factor in the $|\Delta m_{41}^2| \lesssim 0.008 \text{ eV}^2$ region. The RENO result provides the most stringent limits on sterile neutrino mixing at $|\Delta m_{41}^2| \lesssim 0.002 \text{ eV}^2$ using the $\overline{\nu}_e$ disappearance channel. Adding data taken since 2018 and reducing the above systematic uncertainties will improve the results significantly.

4.1.3.3. Joint fits of reactor neutrino experiments. As discussed above, the conclusive way to test whether RAA is due to active-sterile mixing is by searching for sterile neutrino-induced spectral variations as a function of the baseline. With the exception of the Neutrino-4 experiment, no experiment has claimed to observe statistically significant hints of oscillations. Nonetheless, modest hints of oscillation in the other experiments have been reported, and it is worth considering the global fits of these datasets to determine and build a broader context



Figure 62. Left: 2σ CL Feldman–Cousins (FC) exclusion curve of the combined reactor neutrino data. Also overlayed are the 2σ exclusion FC curves from solar data, $\nu_e - {}^{12}$ C scattering data from LSND and KARMEN, 95% C.L. exclusion from the KATRIN experiment, and 2σ gallium suggested region. Right: 1σ , 2σ , and 3σ CL FC suggested regions for a combination of gallium anomaly and relative reactor measurements. The suggested regions are at higher Δm^2 mainly driven by the BEST and Neutrino-4 experiments. Reproduced from [333]. CC BY 4.0.

from their individual results. A joint analysis [333] using Monte Carlo simulations with data from a combination of the short-baseline reactor neutrino experiments DANSS, NEOS, Neutrino-4, PROSPECT, and STEREO shows that the combination of these datasets is statistically compatible with the three neutrino model. The exclusion curve from this joint fit is shown in figure 62.

Although a combination of relative reactor neutrino datasets is incompatible with sterile neutrino-induced oscillations, they are in good agreement with the Gallium Anomaly. This is primarily driven by the data from BEST and Neutrino-4 experiments which prefer oscillations with $\Delta m_{14}^2 > 5 \text{ eV}^2$ —a region where other reactor experiments have minimal sensitivity. Experiments with sensitivity in the $\Delta m_{14}^2 > 5 \text{ eV}^2$ —a region where other reactor experiments have minimal sensitivity. Experiments with sensitivity in the $\Delta m_{14}^2 > 5 \text{ eV}^2$ are needed to fully address these suggested regions. This could be achieved by a combination of upcoming reactor experiments (see sections 6.3) and β -decay experiments (see sections 4.1.6, 6.6).

4.1.3.4. Joint analysis with accelerator experiments. It is impossible for a single experiment to cover all the parameter space of interest to experimentalists and phenomenologists, which motivates the undertaking of joint analyses carried out by the members of the experimental collaborations that properly treat systematic uncertainties and their correlations. Of particular interest is the combination of Daya Bay's data with those of other reactor experiments operating at shorter baselines to cover a wide range of $|\Delta m_{41}^2|$ values. A case in point is the joint fit of the Daya Bay and Bugey-3 data presented in [325] that results in the black contour of figure 61. Furthermore, powerful constraints on sterile-driven neutrino oscillations can also



Figure 63. Comparison of the MINOS, MINOS+, Daya Bay, and Bugey-3 combined 90% CL_s (left) and 99% CL_s (right) limits on $\sin^2 2\theta_{\mu e}$ to the LSND and MiniBooNE 90% and 99% C.L. allowed regions, respectively. The limit also excludes the 90% and 99% C.L. regions allowed by a fit to global data by Gariazzo *et al* where MINOS, MINOS+, Daya Bay, and Bugey-3 are not included [101, 654], and the 90% and 99% C.L. regions allowed by a fit to all available appearance data by Dentler *et al* [102] updated with the 2018 MiniBooNE appearance results [32]. Reproduced from [325]. CC BY 4.0.

be extracted from combining data from reactor experiments with data from long-baseline accelerator experiments. In a 3 + 1 scenario, reactor experiments primarily measure $|U_{e4}|^2 = \sin^2 \theta_{14}$ through electron antineutrino disappearance, while long-baseline accelerator experiments are typically most sensitive to $|U_{\mu4}|^2 = \sin^2 \theta_{24} \cos^2 \theta_{14}$ through measurements of muon (anti)neutrino disappearance. The product of the two matrix elements provides the amplitude of short-baseline electron neutrino appearance in a primary muon neutrino source:

$$P_{\nu_{\mu} \to \nu_{e}}^{SBL} = 4|U_{e4}|^{2}|U_{\mu4}|^{2}\sin^{2}\left(\frac{\Delta m_{41}^{2}L}{4E}\right),$$
(46)

where $4|U_{e4}|^2|U_{\mu4}|^2 = \sin^2 2\theta_{14} \sin^2 \theta_{24} \equiv \sin^2 2\theta_{\mu e}$. Possibly the most representative examples of this type of combination are found in Refs. [324, 325], where constraints on θ_{14} from the Daya Bay and Bugey-3 experiments are combined with constraints on θ_{24} from the MINOS/MINOS+ experiments to constrain the effective mixing parameter $\sin^2 2\theta_{\mu e}$. This work culminated in the most stringent constraints to date from disappearance searches on active to sterile neutrino oscillations and probed the parameter space allowed by the LSND and MiniBooNE anomalies, as shown in figure 63.

As highlighted in [334], there are attractive opportunities in combining Daya Bay's data with other current and future experiments, such as PROSPECT, STEREO, NEOS, and JUNO-TAO. It is worth noting that the Daya Bay collaboration plans to publicly release its full data set once all final results have been released [335], allowing such combinations to occur even well after the collaboration has dissolved. Similarly, following end of data taking in 2016, the MINOS/MINOS+ CL_s surfaces remain available for use in future combinations.

4.1.4. Atmospheric neutrino experiments. IceCube—neutrino telescopes, such as the IceCube Neutrino Observatory in the South Pole, play a unique role in searches for new

physics associated with the short-baseline anomalies [140, 151, 152, 336]. For atmospheric neutrino oscillation baselines, the L/E range of the short-baseline anomalies corresponds to neutrino energies of order 100 GeV-1 TeV. Because this closely matches the peak of the detected neutrino flux at IceCube, it is natural to expect any Lorentz invariant (LI, implying scaling as L/E) phenomenon connected with the short-baseline anomalies [1] may be meaningfully testable at IceCube.

Sensitivity to certain models is even further enhanced by fortuitous features of this energy range, notably: (1) For sterile neutrinos with 0.1-10 eV^2 mass splittings, a resonant matter effect leads to dramatic enhancement of oscillations at ~ 1 TeV, leading to sensitivity far exceeding that for vacuum-like oscillations [337–340]; (2) Any new physics model that invokes non-LI effects scaling positively with E will be enhanced at IceCube relative to all other experiments at lower L/E—these include anomalous decoherence [341–343] and Lorentz violation [344–346] models; (3) this energy range offers unique access to the ν_{τ} appearance sector [337, 347]. Other notable features of the IceCube event sample are its 'broadband' nature, covering five decades of energy of atmospheric neutrinos (10 GeV-1 PeV) and 3.5 decades of baselines (20-12 750 km); high statistical precision, owing to the high total exposure of a billion-ton detector operating stably for ten years collecting 70,000 atmospheric neutrino events per year, and well-controlled cross section uncertainties due to the predominance of deep-inelastic scattering interactions. These features have enabled world-leading sensitivity to the parameters governing three-flavor oscillations as well as nonstandard oscillation models including sterile neutrinos [140, 151–153], tests of low-energy manifestations of quantum gravity [345, 348], neutrino decay [150], and NSIs [349-351].

Present generation light sterile neutrino searches at IceCube. IceCube has made powerful sterile neutrino searches in both high (≥ 400 GeV) and low (≤ 60 GeV) energy ranges. The former targets the matter resonance [337, 338] expected for $\Delta m^2 \sim \mathcal{O}(1 \text{ eV}^2)$ splittings, and is one of the world's most sensitive in the ν_{μ} disappearance channel at eV²-scale mass splittings. The latest generation analysis [151, 152] uses a sample of 305735 reconstructed ν_{μ} events and excludes mixing angles down to $\sin^2 2\theta_{24} \sim 0.02$ at $\Delta m^2 \sim 0.2$ at 99% CL. At 90% CL the analysis yields a closed contour that may be interpreted as a statically weak hint of a signal, with a best-fit point at $\sin^2 2\theta_{24} = 0.10$ and $\Delta m_{41}^2 = 4.5$ eV². This result is consistent with the no sterile neutrino hypothesis with a *p*-value of 8.0%. The 90% CL contour is shown in figure 64, left.

At low energy, sterile neutrino mixing within an extended neutral lepton mixing matrix enhances the standard oscillation probability proportionally to the matter column density traversed [352]. The effect is approximately independent of Δm^2 , as oscillation cycles are irresolvable within detector energy resolution. IceCube has tested for this effect using a multiflavor sample over an energy range of 10–60 GeV [140], with the strongest effect expected at an energy of ~20 GeV for upgoing muons. The analysis yielded no evidence of anomalous oscillations, setting a limit on the extended PMNS matrix elements $|U_{\mu4}|^2 = \sin^2 \theta_{24}$ and $|U_{r4}|^2 = \sin^2 \theta_{34} \cos^2 \theta_{34}$, marked IC2017 (NO) in figure 64, center.

Next generation sterile neutrino searches at IceCube. The sterile neutrino sensitivity at IceCube has yet to be exhausted, with near-term improvements expected from event samples already in hand. At low energies, a sterile neutrino search using the full ten year dataset of $\geq 300\ 000$ events with $E_{\nu} \leq 100\ \text{GeV}$ spanning all flavors is underway [353], promising unique sensitivity in the $|U_{\mu4}|^2$, $|U_{\tau4}|^2$ plane (figure 64, center). At higher energies ($\geq 400\ \text{GeV}$), attention to date has been focused on searches for ν_{μ} disappearance through non-zero θ_{24} , motivated by the necessity of its finite value if sterile neutrino osculations were responsible for the short-baseline ($\nu_{\mu} \rightarrow \nu_e$)/($\bar{\nu}_{\mu} \rightarrow \bar{\nu}_e$) anomalies ($\theta_{14} = \theta_{34} = 0$ leads to the



Figure 64. Left: results from IceCube's high-energy muon-neutrino disappearance search. Reproduced from [151]. CC BY 4.0. Middle: results from IceCube low- and high-energy analyses. Reproduced from [353]. © 2021 IOP Publishing Ltd and Sissa Medialab. All rights reserved. Right: expected sensitivity with cascades. Reproduced from [347]. CC BY 4.0.

most conservative limits on θ_{24} [354]). Efforts are now underway to incorporate the high energy cascade event sample into these analyses, which includes topologies associated with ν_e and ν_{τ} CC and all flavor NC interactions. This extension promises sensitivity to both ν_e and ν_{τ} appearance signatures associated with non-zero θ_{14} and θ_{34} , respectively. Preliminary studies [347] of the sensitivity in this channel suggest that ν_{τ} appearance signatures are discoverable for values of θ_{34} consistent with world data and IceCube's existing θ_{24} limits; and that values of θ_{14} consistent with reactor [66] and gallium/BEST [136, 268, 355] anomalies may yield observable ν_e appearance signatures. The expected sensitivity of the combined high energy ν_{μ} disappearance and cascade appearance signatures is shown in figure 64, right. Augmentations of the sterile neutrino searches using machine learning techniques, starting-event topologies, and improved reconstruction methods are also underway, expected to provide continuing improvements over the coming Snowmass period.

Super-K. Super-Kamiokande (Super-K, SK) has performed a search for light sterile neutrinos using approximately 4000 live-days of atmospheric neutrino data [356]. The SK analysis focuses on light sterile neutrinos with mass-squared differences greater than 0.1 eV². In the energy range of the Super-K analysis, predominantly below 10 GeV, such a large masssquare difference implies that the oscillation effects due to light sterile neutrinos are averaged out. Thus, Super-K analysis is insensitive to the mass-square difference, but only to the mixing elements. In the Super-K analysis, they choose to constraint the mixing elements: $|U_{\tau 4}|$ and $|U_{\mu 4}|$. In principle, Super-K has also sensitivity to $|U_{e4}|$, but this is a subleading effect. Additionally, the Super-K also is sensitive to the additional CP-violating phases that appear in the presence of a sterile neutrino. The effect of these CP-violating phases is subleading but does not always yield conservative results on the mixings and thus the results should be taken with this caveat [337].

The Super-K analysis found no significant evidence of a light sterile neutrino and reported constraints on $|U_{\tau4}|$ and $|U_{\mu4}|$ for $\Delta m_{41}^2 > 0.1 \text{ eV}^2$; concretely they limit $|U_{\mu4}|$ to less than 0.041 and $|U_{\tau4}|$ to be less than 0.18 at 90% C.L. These results are shown in figure 65 and have been superseded by constraints from ANTARES and IceCube; compare to figure 64.

ANTARES. The ANTARES neutrino telescope [357], a sub-gigaton-scale neutrino telescope, had been designed and optimized for the exploration of the high-energy universe by using neutrinos as cosmic probes. However, its energy threshold of about 20 GeV was sufficiently low to be sensitive to the first atmospheric oscillation minimum. The majority of



Figure 65. Constraint on $|U_{\tau 4}|$ and $|U_{\mu 4}|$ for $\Delta m_{41}^2 > 0.1 \text{ eV}^2$ obtained by Super-K. Reprinted (figure) with permission from [356], Copyright (2015) by the American Physical Society.

the neutrino events have been recorded with energies between 100 GeV and few TeV, an energy range rich in signatures of eV-scale sterile neutrinos. This is illustrated in figure 66 which shows the ν_{μ} survival probability for maximal mixing of θ_{23} and different combinations of the mixing parameters θ_{24} , θ_{34} and δ_{24} with

$$U_{\mu4} = e^{-i\delta_{24}}\sin\theta_{24},\tag{47}$$

$$U_{\tau 4} = \sin \theta_{34} \cos \theta_{24}.$$
 (48)

The fast oscillations due to $\Delta m_{41}^2 \gtrsim 0.5 \text{ eV}^2$ are unobservable due to the limited energy resolution of the detector, making Δm_{41}^2 not measurable.

The ANTARES neutrino telescope was located in the Mediterranean Sea, 40 km off the coast of Toulon, France, at a mooring depth of about 2475 m. The detector was completed in 2008 and took data until February 2022. ANTARES was composed of 12 detection lines, instrumenting a water volume of about 15 Mtons. ANTARES data collected from 2007 to 2016 with a total detector lifetime of 2830 d have been used to constrain $U_{\mu4}$ and $U_{\tau4}$ [358]. A total of 7710 low-energetic atmospheric neutrino candidate events have been selected, largely dominated by muon-(anti)-neutrino charge-current events, identified thanks to a long-range up-going muon track. Particular attention was paid to consistent handling of the complex phase δ_{24} in conjunction with the neutrino mass ordering. The limits from the ANTARES analysis are shown in figure 67. As expected from figure 66 ANTARES is particularly sensitive if both $U_{\mu4}$ and $U_{\tau4}$ are non-zero and improves existing limits from [140, 359] substantially.

After profiling over the other variable, the following limits on the two matrix elements can be derived:

$$|U_{\mu4}|^2 < 0.007 \ (0.13) \text{ at } 90\% \ (99\%) \text{ CL},$$
(49)

$$|U_{\tau4}|^2 < 0.40 \ (0.68) \text{ at } 90\% \ (99\%) \text{ CL.}$$
 (50)



Figure 66. Survival probability of vertically up-going ν_{μ} at ANTARES as a function of neutrino energy for different values of mixing angles θ_{24} , θ_{34} and δ_{24} with $\Delta m_{41}^2 = 0.5 \text{ eV}^2$, $\Delta m_{31}^2 = 2.5 \cdot 10^{-3} \text{ eV}^2$ and $\sin^2 2\theta_{23} = 1$. Reproduced from [358]. CC BY 4.0.

4.1.5. Radioactive source experiments. **BEST**. The baksan experiment on sterile transitions (BEST) [355] was proposed to probe the possibility of the short-baseline electron neutrino disappearance using the same process studied by GALLEX and SAGE, equation (6). The disappearance of neutrinos was suggested as an explanation of the deficit in observed events measured by the previous radioactive source experiments (see section 3.3.3).

Although the main experimental idea is the same, the BEST configuration provides for simultaneous measurements at two different baselines. In this case, a spherical vessel (*inner target*) is located inside a cylindrical container (*outer target*), both filled with liquid gallium (the detector). The radioactive source (51 Cr) is placed inside the sphere, as shown in figure 68.

Here again, the emitted neutrinos interact with the detector through reaction equation (6), and the produced ⁷¹Ge atoms are extracted and counted for each vessel separately [360]. The ratio of the measured rate of ⁷¹Ge production at each distance to the expected one considering the cross section and experimental efficiencies are presented in table 13 [361]. Recall that the cross section of the reaction in equation (6) is used in the calculation of these ratios, and BEST used the Bahcall results [76], including conservative uncertainties [136]. The ratios are 4.2σ and 4.8σ less than unity, respectively, supporting the gallium anomaly observed by other experiments.

BEST performed an analysis of these results in the framework of short-baseline electron neutrinos disappearance due to neutrino oscillation governed by the survival probability in equation (7), using $\sin^2 2\theta = 4 |U_{e4}|^2 (1 - |U_{e4}|^2)$. The study leads to the allowed regions shown in figure 69, corresponding to 1σ , 2σ , and 3σ confidence levels (left plot), with the best-fit at $(\sin^2 2\theta, \Delta m^2) = (0.42, 3.3 \text{ eV}^2)$. When combined with the results from GALLEX and SAGE (table 3), the allowed regions are the ones illustrated in the right plot of figure 69, where correlated cross section uncertainties were considered. In this case, the best fit is located at $(\sin^2 2\theta, \Delta m^2) = (0.34, 1.25 \text{ eV}^2)$ [361].

Some of the calculations of the cross section of the process equation (6) appear to decrease the significance of the gallium anomaly related to the GALLEX and SAGE experiments (section 3.3.3); nonetheless, the results are still consistent with a possible short-



Figure 67. 90% (left) and 99% (right) CL limits for the 3 + 1 neutrino model in the parameter plane of $|U_{\mu4}|^2 = \sin^2\theta_{24}$ and $|U_{\tau4}|^2 = \sin^2\theta_{34}\cos^2\theta_{24}$ obtained by ANTARES (black lines), and compared to the ones published by IceCube/DeepCore [140] (red) and Super-Kamiokande [359] (blue). The dashed lines are obtained for NH and $\delta_{24} = 0^{\circ}$ while the solid lines are for an unconstrained δ_{24} (ANTARES) or for IH and $\delta_{24} = 0^{\circ}$ (IceCube/DeepCore) respectively. The colored markers indicate the best-fit values for each experiment. The 1D projections after profiling over the other variable are also shown for the result of this work. Reproduced from [358]. CC BY 4.0.

baseline electron neutrino disappearance produced by active to sterile neutrino oscillations. On the other hand, the measurements by BEST are in agreement with the former source experiments and confirm the gallium anomaly with larger significance.

4.1.6. Beta spectrum searches. High precision beta spectroscopy enables searches for sterile neutrinos from the sub-eV- up to the MeV-scale. Exploiting the kinematics of the weak process, these experiments offer a clean probe of the coupling of the electron flavor neutrino to the different mass states. The electron flavor neutrino emitted in β -decay does not have a well-defined mass but is rather an admixture of the neutrino mass eigenstates. The existence of a hypothetical sterile neutrino(s) implies the electron flavor neutrino mayalso contain a small admixture of (at least) a new fourth neutrino mass eigenstate, m_4 .

The β -decay spectrum, $R_{\beta}(E)$ will be altered:

$$R_{\beta}(E) = \cos^2 \theta_s R_{\beta}(E, m_{\beta}^2) + \sin^2 \theta_s R_{\beta}(E, m_4^2)$$
(51)

composed of both the spectrum corresponding to the traditional electron-weighted neutrino mass m_{β} (from the three active neutrinos) and a spectrum associated with m_4 . The maximal energy of each spectrum contribution is $E_0 - m_i$, with the largest neutrino mass resulting in the lowest endpoint energy. The amplitudes of the two decay branches are given by $\cos^2 \theta_s$ and $\sin^2 \theta_s$, respectively, where θ_s is the active-to-sterile mixing angle. The resulting signature of a sterile neutrino is thus a kink-like distortion of the measured spectrum at an energy of $E_0 - m_4$.

Spectral measurements across a broad energy range therefore enable searches for kink features and thus sterile neutrinos. Although not an L/E oscillation signature, detection would



Figure 68. BEST detector configuration. Two Ga target volumes detect neutrino interactions from a 51 Cr source. Reproduced from [136]. CC BY 4.0.



Figure 69. Allowed regions obtained in the $\sin^2 2\theta - \Delta m_{41}^2$ parameter space from the analysis of the BEST results only (left) and from the BEST results combined with results from GALLEX and SAGE (right). Reproduced from [361]. CC BY 4.0. Note that here $\sin^2 2\theta = 4 |U_{e4}|^2 (1 - |U_{e4}|^2)$.

Table 13. Ratio of observed to predicted ⁷¹Ge event rates as measured by BEST using ⁵¹Cr with its inner and outer targets.

Inner	Outer		
0.79 ± 0.05	0.77 ± 0.05		



Figure 70. Landscape of historical limits on sterile neutrino from beta spectrum searches. Results from the last decade are highlighted in color and labeled.

be an unambiguous detection of sterile neutrinos subject to drastically different systematics. As a complementary probe, these searches exclusively probe the sterile neutrino hypothesis, and not other physics explanations of the existing anomalies. For eV-scale sterile neutrinos, this signature naturally appears in the region of interest for direct neutrino mass measurements studying the beta endpoint.

4.1.6.1. Historical context. The flexibility of the beta spectrum method has led to its use in placing strong constraints on sterile neutrinos. These limits span more than six decades in m_4 —from 1.5 eV up to 2.5 MeV—and down to $\sin^2\theta \sim 10^{-4}$ (see figure 70). Isotopes placing the strongest constraints across this energy range include ³H [362–365], ¹⁸⁷Re [366], ⁶³Ni [367], ³⁵S [368], ⁶⁴Cu [369], ¹⁴⁴Ce-¹⁴⁴Pr [370], ⁷Be [371], and ²⁰F [372].

Tritium endpoint searches for direct neutrino mass study have enjoyed great success, leveraging the sophisticated detector and source development for that science. Both KATRIN [364, 365] and Troitsk [362, 363] have derived eV-scale limits from the primary neutrino mass physics data. Additionally, dedicated searches away from the endpoint open up a wider range to keV masses. The progress of the ongoing KATRIN experiment for eV steriles is described in detail below (section 4.1.6.2), with discussion of sensitivity to keV steriles with an upgraded detector later (section 6.6.1). At higher mass, the BeEST experiment has obtained the strongest limits to steriles up to 0.85 MeV based on their Phase II prototype [371], heralding the advent of new technologies with significant improvements in sensitivity; BeEST is discussed later in the context of their full sensitivity (section 6.6.3).

4.1.6.2. KATRIN. The Karlsruhe Tritium Neutrino experiment (KATRIN) [373–375] provides a high-precision electron spectrum measurement of tritium β -decay, ${}^{3}\text{H} \rightarrow {}^{3}\text{He}^{+} + e^{-} + \bar{\nu}_{e}$ (endpoint $E_{0} = 18.57$ keV, half-life $t_{1/2} = 12.32$ yr). KATRIN is designed to improve the sensitivity on the effective neutrino mass, m_{β} , to 0.2–0.3 eV (90% CL). Based on the science measurement campaigns taken in 2019 [376, 377], KATRIN can constrain the mass and mixture of a sterile neutrino that would manifest itself as a distortion of the β -electron spectrum. The signature is a kink-like feature, as shown in a simulation presented in figure 71. The first light sterile neutrino result is based on data from KATRIN's



Figure 71. (a) Electron spectrum of KATRIN data $R(\langle qU \rangle)$ over the interval $[E_0 - 40 \text{ eV}, E_0 + 50 \text{ eV}]$ from all 274 campaign 1 tritium scans and the three-neutrino mixing best-fit model $R_{\text{cale}}(\langle qU \rangle)$ (line). The integral β -decay spectrum extends to E_0 on top of a flat background R_{bg} . 1- σ statistical errors are enlarged by a factor 50. (b) Simulation of an arbitrary sterile neutrino imprint on the electron spectrum. The ratio of the simulated data without fluctuation, including a fourth neutrino of mass $m_4 = 10 \text{ eV}$ and mixing $|U_{e4}|^2 = 0.01$, to the three-neutrino mixing model is shown (red solid line). (c) Integral measurement time distribution. Reproduced from [378]. CC BY 4.0.

first high-purity tritium campaign, which ran from April to May, 2019, at an average source activity of 2.45×10^{10} Bq [378]. An updated result is based on KATRIN's second campaign, which ran from October to November, 2019, achieving a source activity of 9.5×10^{10} Bq [364].

The integral β -electron spectrum is scanned repeatedly in the range of $[E_0 - 90 \text{ eV}, E_0 + 135 \text{ eV}]$ by applying non-equidistant HV settings to the spectrometer electrode system. Each scan lasted 2 h. At each HV set point, the transmitted electrons are counted. Figure 71 shows the measurement time distribution. Stable scans are selected with an overall scanning time of 522 h (campaign one) and 744 h (campaign two). Detector variations are minimized by gathering events from the 117 most similar pixels and combining them into a single effective pixel analysis. The resulting spectra, $R(\langle qU \rangle)$, include a combined 5.2×10^6 expected tritium events on a flat background. Even for the campaign one search, which exhibited lower source rates and higher backgrounds, a high signal-to-background ratio was achieved, rapidly increasing from 1 at $\langle qU \rangle = E_0 - 12$ eV to >70 at $\langle qU \rangle = E_0 - 40$ eV. In the campaign two search, the highest signal-to-background ratio improves to 235 at $\langle qU \rangle = E_0 - 40$ eV. The modeled experimental

spectrum $R_{\text{calc}}(\langle qU \rangle)$ is the convolution of the differential β -spectrum $R_{\beta}(E)$ with the response function $f(E - \langle qU \rangle)$, and an energy-independent background rate R_{bg} :

$$R_{\text{calc}}(\langle qU \rangle) = A_{\text{s}} \cdot N_{\text{T}} \int R_{\beta}(E) \cdot f(E - \langle qU \rangle) \, \mathrm{d}E + R_{\text{bg}}, \tag{52}$$

where A_s is the tritium signal amplitude. N_T denotes the number of tritium atoms in the source multiplied with the accepted solid angle of the setup $\Delta\Omega/4\pi = (1 - \cos\theta_{\text{max}})/2$, with $\theta_{\text{max}} = 50.4^\circ$, and the detector efficiency (0.95). The function $f(E - \langle qU \rangle)$ describes the transmission probability of an electron as a function of its surplus energy $E - \langle qU \rangle$. KATRIN extends the experimental modeling and statistical analysis to constrain both the sterile neutrino mass squared m_4^2 and its mixing amplitude $|U_{e4}|^2$, following the same strategy as for the m_β analysis [379]. In the 3 + 1 active-sterile neutrino model extension the electron spectrum, R_β , is replaced by $R_\beta(E, m_\beta, m_4) = (1 - |U_{e4}|^2)R_\beta(E, m_\beta^2) + |U_{e4}|^2R_\beta(E, m_4^2)$, where U is the extended 4×4 unitary matrix, $R_\beta(E, m_\beta^2)$ is the differential electron spectrum associated with decays the include active neutrinos in the final state, and $R_\beta(E, m_4^2)$ describes the additional spectrum associated to decays involving a sterile neutrino of mass m_4 .

The observable integral spectrum R_{calc} is henceforth modeled with six free parameters: the four original parameters $(A_s, E_0, R_{\text{bg}}, m_{\nu}^2)$ [376], m_4^2 and $|U_{e4}|^2$. This extended model $R_{\text{calc}}(\langle qU \rangle)$ is then fitted to the experimental data $R(\langle qU \rangle)$.

In order to mitigate bias, the full analysis is first conducted on a Monte Carlo (MC) data set before turning to the actual data. The fit of $R(\langle qU \rangle)$ with $R_{\text{calc}}(\langle qU \rangle)$ is performed by minimizing the standard χ^2 -estimator. In a 'shape-only' fit, both E_0 and A_s are left unconstrained. To propagate systematic uncertainties, a covariance matrix is computed after performing $\mathcal{O}(10^4)$ simulations. The sum of all matrices encodes the total uncertainties of $R_{\text{calc}}(\langle qU \rangle)$, including HV set-point-dependent correlations. The χ^2 -estimator is then minimized to determine the best-fit parameters, and the shape of the χ^2 -function is used to infer the uncertainties. The fit range $[E_0 - 40 \text{ eV}, E_0 + 50 \text{ eV}]$ is chosen such that statistical uncertainties on $|U_{e4}|^2$ dominate over systematic uncertainties in the whole range of m_4^2 considered [376, 379]. The experimental result agrees well with the sensitivity estimates and are displayed in figure 72. They are showing no evidence for a sterile neutrino signal and are compared with short-baseline neutrino oscillation experiments measuring the electron (anti-) neutrino survival probability $P(\Delta m_{41}^2, \sin^2(2\theta_{ee}))$. An estimation of KATRIN's five-year (1000 live-day) sensitivity is also presented.

4.2. Dark sectors in scattering and in the beam

The MiniBooNE low energy excess has traditionally been interpreted as a potential hint of neutrino oscillations at short baseline driven by a new sterile state. However, simple interpretations (e.g. 3 + 1 models) are in tension with other oscillation measurements (see the discussion in the sections above), in particular searches for muon flavor disappearance driven by the same Δm^2 . Given this tension, it is natural to consider other possible new physics explanations for the LEE. These new physics explanations must have detector signatures and production mechanisms which are consistent with the existing MiniBooNE measurements.

The measured angular distribution of the events in the LEE constrains the allowed interaction channels in the detector. For concreteness, let us suppose that the LEE is due to some new particle X. Simple signatures involving elastic scattering on electrons (e.g. $Xe^- \rightarrow Xe^-$) or fully visible decays (e.g. $X \rightarrow e^+e^-$) produce events which are very forward, inconsistent with the measurement which features a broad angular distribution. While there is a highly significant forward component to the excess, especially apparent in the 2021



Figure 72. 95% confidence level exclusion curves in the $(\sin^2(2\theta_{ee}), \Delta m_{41}^2)$ plane obtained from the analysis of KATRIN data with fixed $m_\nu \langle m_\beta \rangle = 0$. The green contours denote the 3 + 1 neutrino oscillations allowed at 95% confidence level by the reactor and gallium anomalies [66]. KATRIN data improve the exclusion of the high Δm_{41}^2 values with respect to DANSS, PROSPECT, and STEREO reactor measurements [134, 655, 656]. Mainz [657] and Troitsk [658] exclusion curves [659] are also displayed for comparison. An estimation of KATRIN's final sensitivity is represented by the dotted line. Reproduced from [364]. CC BY 4.0.

MiniBooNE LEE result, the excess remains significant across a broad range of angles, and one cannot simply presume that the forward excess is due to new physics while the rest of the excess is due to background underestimation. Considering semi-visible decays (e.g. $X \rightarrow X' + \gamma$) lessens the disagreement somewhat, but such models are still strongly disfavored, with either too many forward events or too many backward events depending on the X mass. In light of these constraints, the only class of models which can adequately reproduce the LEE angular distribution are those involving inelastic scattering, similar to the neutrinonucleus scattering in the standard sterile neutrino interpretation.

4.2.1. Beam dump searches. Neutrino experiments as well as dedicated electron or proton beam dump experiments can search for dark sector particles by looking for the decays or scatterings of states produced at the target station. This technique has been extensively explored in the literature in the search for light dark matter and mediators [380–383]. Measurements at LSND and MiniBooNE provide competitive limits in several light dark sectors.

Of particular relevance are the results of the MiniBooNE-DM run. MiniBooNE ran in a beam dump configuration where the proton beam was aimed directly at the downstream beam dump instead of onto the neutrino production target. In this mode, charged meson decay in flight was suppressed, reducing the 'background' neutrino flux and enhancing sensitivity to new physics. No excess of events was observed in beam dump mode, implying that any new physics production modes that have a simple scaling with the number of protons on target like neutral meson decays or continuum processes (i.e. bremsstrahlung) are ruled out because they should produce a signature in beam dump mode. This leaves production from charged meson decay in flight as the only viable production mode for new particles that can explain the LEE.



Figure 73. Left: the neutrino flux at MiniBooNE, MINER ν A, and CHARM-II are shown as shaded regions. In lines the upscattering cross section for producing heavy neutrinos through coherent and incoherent Z' scattering is shown. Right: The single shower event spectra as a function of dE/dx at MINER ν A (top) and of $E\theta^2$ at CHARM-II (bottom). Dashed lines indicate analysis cuts. Reproduced from [217]. CC BY 4.0.

New physics explanations of the MiniBooNE LEE based on inelastic scattering signatures in the detector are favored due to the compatibility with the measured angular distribution of the excess. The MiniBooNE-DM results imply that such processes should be initiated by either neutrinos or new particles produced in charged meson decays at the target [224].

4.2.2. Neutrino-electron scattering measurement. Models that can explain the LSND or MiniBooNE anomaly through the production of new particles that decay to electromagnetic showers can be constrained by measurements of neutrino-electron ($\nu - e$) scattering. Since single photons and collimated e^+e^- pairs appear as single showers, they can be searched for by looking for in the photon-like sidebands of this measurement. In [217], the authors propose a technique to constrain e^+e^- explanations of LSND and MiniBooNE using data from MINER ν A and CHARM-II. These experiments were located in the NuMI and CERN SPS beams, respectively, and therefore cover a much broader and higher-energy neutrino flux than LSND and MiniBooNE (see left panel of figure 73).

In the dark photon model discussed in section 3.2, the HNL decays to overlapping $e^+e^$ predict new signals in the large-dE/dx sideband of $\nu - e$ scattering, with moderate values of $E\theta^2$. The energy deposition dE/dx, defined as the deposited energy in the first centimeters of a electromagnetic shower, helps discriminate between electron-like showers from photon-like showers, which have typically twice the energy deposited per cm than the former. The variable $E\theta^2$, defined as the shower energy times the square of the shower angle with respect to the neutrino beam, is also used to reduce backgrounds since in the boosted electron in $\nu - e$ scattering obeys $E\theta^2 < 2m_e$, while backgrounds, mostly from π^0 decay and ν_e CCQE, can have much broader angular distributions. For the dark sector signatures, the resulting overlapping e^+e^- pairs can be quite forward, especially when the dark photon is light. Examples of the dark sector predictions at MINER ν A as a function of dE/dx and at CHARM-II as a function of $E\theta^2$ are shown in figure 73.

Since the backgrounds have large uncertainties and the search is not tailored to the dark sector signals, the sensitivity is not enough to rule out all explanations of the MiniBooNE excess. However, due to the higher-energies at MINER ν A and CHARM-II, explanations with large HNL masses, often preferred due to the broader angular distributions at MiniBooNE,

can be robustly excluded. In figure 73, the limits are shown in the mixing of the HNL with muon neutrinos, $|U_{\mu4}|^2$, and their mass m_4 . Other parameters like kinetic mixing ϵ and the dark sector coupling α_D , as well as the dark photon mass m'_Z are fixed. Two curves are shown for each experiment: a solid curve for the nominal choice of background uncertainty, and a dashed curve corresponding to the case where uncertainties were inflated by a factor of a few (see caption). For MINER ν A LE (ME), this corresponds to 30% (40%) background normalization uncertainty in the nominal case and 100% uncertainty in the conservative case. For CHARM-II these correspond to 3% and 10%, respectively. The uncertainties at CHARM-II are constrained by the sideband at large $E\theta^2$.

Future measurements, including antineutrino-electron scattering at MINER ν A can improve on the limits discussed above. The limits can also be recasted onto models with photon final states, like those discussed in figure 3.2.1. Finally, HNLs produced via scalar mediators are less constrained by this technique since the upscattering cross section decreases at higher energies.

4.2.3. Searches for long-lived particles. The possibility that meson decays produce longlived HNLs in neutrino beams is strongly constrained by direct searches at T2K [384], MicroBooNE [385], and ArgoNeuT [386], as well as by other past-generation experiments such as PS-191 [387] (recently re-evaluated in [388] and [389]), CHARM [390], CHARM-II [391], and NuTeV [392]. The constraints are placed on minimal models with a single HNL Nthat interacts with the weak bosons only, with interaction strength suppressed by small mixing angles $U_{\alpha N}$. At first sight, this model differs from the light sterile neutrino only for the mass of the additional neutrino, which is supposed to be in the 10-500 MeV range. However, this difference results in a radically different phenomenology. Because these new particles are so much heavier than the standard neutrinos, no oscillation is possible: the HNL mixes with the other neutrinos but loses coherence immediately with the rest of the wave packet. As a result, in the minimal scenario, in which no other particle or interaction is present aside from one or more HNLs, the HNL is produced in the beam through mixing and decays in the detector. The available decay channels depend on the mass and include $e^+e^-\nu$, $e\mu\nu$, $e\pi$, $\mu^+\mu^-\nu$, $\mu\pi$. However, this minimal model is ruled out by a combination of Big Bang Nucleosynthesis, which provides a lower bound on the mixing parameters [393], and experiments with neutrino beams, which lead to upper bounds, with no available parameter space in between these bounds (left plot in figure 75).

In particular, the T2K Near Detector ND280, a modular detector able to resolve details of interactions track and identify individual final state particles, sets the best limits on these models. For this analysis, the three TPCs filled with gaseous argon provide a low-density decay volume for the HNLs, with zero background from neutrino interactions, as illustrated in the left plot in figure 76. The limits have been derived through a search with little expected background and zero observed events in every analysis channel [394], later extended and combined with BBN constraints [388].

This model becomes more interesting if new interactions are present. These new interactions allow new decay modes, hence shortening the lifetime and relaxing BBN limits.

For example, if the HNL possesses a magnetic moment, it could decay electromagnetically into $\nu\gamma$. For magnetic moments of the order of PeV, this model could explain the MiniBooNE anomaly [205]. However, this explanation is constrained by short-baseline experiments, like MicroBooNE and ND280 (right plot in figure 75). Thanks to the high density of the liquid argon, MicroBooNE identifies single photons through conversion to e^+e^- pairs. ND280 is instead sensitive to the branching ratio into off-shell photons, which



Figure 74. The MINER ν A and CHARM-II constraints in the parameter space of the HNL mixing with muons as a function of its mass, at 90% C.L. Solid lines show the nominal constraints, while dashed ones show constraints with inflated systematic errors on the background. The arrows on the vertical lines indicate where more than a given percentage of the total excess events at MiniBooNE are predicted in the forward-most angular bin. Reproduced from [217]. CC BY 4.0.



Figure 75. Left: exclusion plot for the minimal scenario, showing the mixing with the muon flavour versus the HNL mass. Right: exclusion plot for the non-minimal scenario where HNLs possess a magnetic moment. It shows the mixing with the muon flavour versus the magnetic dipole moment for an HNL mass of 250 MeV. The region of interest to explain the MiniBooNE excess is shown in green [205]. The dark grey region is taken from or extrapolated by [660]. Reproduced from [388]. CC BY 4.0.

results in a genuine e^+e^- pair. In this case, the rate is lower by a factor of 100 but benefits from a zero background search.

Upscattering recasts. As discussed in sections 3.2, HNLs can be short-lived if they interact via additional forces via a dark photon or scalar, for instance. In these models, the lifetime of the HNL can range from tens of meters to sub-mm, depending on the choice of parameters. In this regime, standard neutrinos scattering on nuclei can produce a HNL, as illustrated in the right panel of figure 76. This particle propagates inside the detector and


Figure 76. Left: schematic representations of an HNL decaying in flight in one of the ND280 GArTPCs, as considered in [388, 394]. Right: the HNL is produced in the dense layers of lead through upscattering and subsequently decays in one of the GArTPCs as considered in [395].

subsequently decays into an e^+e^- pair plus another neutrino. Searches for e^+e^- final states at neutrino detectors can be used to constraint this possibility. In particular, the near detector of T2K, ND280, and the MicroBooNE detector have both been used to search for the decay of long-lived particles into e^+e^-

In the scenario of a dark photon heavier than the HNL, lifetimes range between a few cm to several meters. In this case, a dark neutrino would be produced in the Pi0 detector (POD), a very dense detector composed of high-Z materials such as lead, which lies upstream with respect to the TPCs. The production is particularly copious as the process is coherent, thus scaling with Z^2 and benefiting the high Z materials. In the scenario of a dark photon lighter than the HNL, the decay proceeds through an on-shell dark photon, resulting in a larger decay width and shorter lifetimes, sub-millimetre. In this case, the HNL is produced and decays at the same point in the detector. The fine-grained detectors (FGD) of ND280 come into play for this study. This plastic scintillator is dense enough to make the rate for production through upscattering significant while allowing precise tracking and identification of the e^+e^- pair. Because of the larger density, backgrounds from the beam or photons that convert inside the FGD are present. On the other hand, the invariant mass of the e^+e^- pair is an excellent quantity to discriminate this background as it peaks around the dark photon mass because of the on-shell decay - making this analysis a peak search. Explanations of MiniBooNE under this model are not entirely ruled out yet [395], but the next generation of analyses, together with an upgrade to the detector [296] and a much larger dataset [396] that will be collected by T2K, are expected to probe the entire parameter space of interest.

4.3. Conventional explanations and other searches

Partly motivated by the lack of a single interpretation that can simultaneously explain all observed experimental anomalies, the possibility that the anomalies represent a collection of conventional explanations (or that in combination with new physics) has also been discussed; see, e.g. [36], in which the possibility that the MiniBooNE anomaly is a combination of underestimated backgrounds is explored. In this case it was shown that no combination of varying the known backgrounds can completely explain the MiniBooNE LEE, it can potentially reduce the significance to 3σ if several less well-known backgrounds saturate their prescribed errors.

Key among the anomalies is the MiniBooNE low-energy excess, where it has been shown repeatedly that the excess shape at low energy is incompatible with 3+N light sterile neutrino oscillations, raising the question whether this excess is due to another effect. Furthermore, as

this energy region is dominated by non- ν_e backgrounds, conventional interpretations for the excess that do not rely on new physics or introduce electrons into the MiniBooNE detector have been raised. The majority of these rest on the known fact that MiniBooNE—as a Cherenkov detector—had no ability to distinguish a single photon from a single electron, and in particular target the various photon backgrounds irreducibly contributing to and dominating the MiniBooNE observed ν_e CCQE rate at low energy.

In the reactor sector, the experimental search for sterile neutrinos were primarily motivated by the anomalously low rates of fluxes measured by the past reactor neutrino experiments. Overestimated $\bar{\nu}_e$ rates based on mismodeled reactor models, against which the fluxes were compared, could also explain the observed discrepancy without invoking new BSM particle physics.

4.3.1. Constraints on single-photon production. The majority of conventional interpretations to the MiniBooNE excess propose that the excess is due to mis-identified photons from various neutrino interactions that produce photons and no charged lepton in the final state. In MiniBooNE, these mis-identified backgrounds are primarily contributed by π^0 production, due to reconstruction inefficiencies (where one photon from the π^0 decay may be missed) or more rare processes such as NC single-photon production through $\Delta(1232)$ resonance production and subsequent radiative decay. Many of these processes can and have been constrained in situ at MiniBooNE. This was the case for the rate and momentum-dependence of misidentified NC π^0 decays, which were constrained by a high-statistics measurement of events with two reconstructed electromagnetic shower Cerenkov rings, representing the two photons from NC π^0 decay. Similarly, the 'dirt' background component in MiniBooNE, which was NC π^0 -dominated, was directly constrained by a high-statistics measurements of events close to the detector boundary. These processes were further studied and disfavored as the source of the MiniBooNE anomaly by studying both the radial distribution of the excess as well as the timing of the events relative to the known beam bunch timing [33]. On the other hand, photons from the radiative decay of the $\Delta(1232)$ baryon produced in neutrino NC interactions (NC $\Delta \rightarrow N\gamma$, where N = p, n) were an irreducible background in MiniBooNE. This background was not constrained in situ, but rather the rate was constrained indirectly through its correlation with the NC π^0 measured rate (which proceeds predominantly via $\Delta \rightarrow$ $N\pi^0$ decay).

The MicroBooNE collaboration has recently performed a direct search for single-photon events coming from neutrino-induced NC production of the $\Delta(1232)$ baryon resonance with subsequent Δ radiative decay [248]. As discussed in section 3.3.1, Δ decay is expected to be the dominant source of single-photon events in neutrino-argon interactions below 1 GeV. Although Δ radiative decay is predicted in the SM, and measurements of photoproduction [397] and virtual Compton scattering [398] are well described by theory, this process has never been directly observed in neutrino scattering. In a fit to the radial distribution of the MiniBooNE data with statistical errors only, an enhancement of NC $\Delta \rightarrow N\gamma$ by a normalization factor of 3.18 (quoted with no uncertainty) was found to provide the best fit for the observed excess, and in good spectral agreement with the observed excess across a number of reconstructed kinematic variables [33]. The MicroBooNE collaboration searched directly for an excess of this magnitude, interpreted as an overall enhancement to the theoretically-predicted NC $\Delta \rightarrow N\gamma$ rate.

MicroBooNE utilized the strength of LArTPC neutrino detectors to search for this excess of single-photons from the Δ decay, with and without a proton track present in the interaction (referred to as $1\gamma 1p$ and $1\gamma 0p$ final states). The presence of the reconstructed track in the $1\gamma 1p$ selection allowed for the tagging of the neutrino interaction vertex, and subsequent



Figure 77. Summary of MicroBooNE's NC $\Delta \rightarrow N\gamma$ single-photon searches for both with (left) and without (right) an associated proton. Reproduced from [248]. CC BY 4.0. No evidence of an enhanced NC $\Delta \rightarrow N\gamma$ rate is observed.

reconstruction of the photon conversion distance in argon, which led to improved background rejection compared to the $1\gamma 0p$ selection, where the lack of associated hadronic activity prevented neutrino vertex tagging.

The analysis used approximately half of the total collected MicroBooNE data to date (6.9 × 10²⁰ POT). It selected and simultaneously fitted the $1\gamma 1p$ and $1\gamma 0p$ samples together with two additional, mutually exclusive but highly correlated samples with high NC π^0 purity. These high-statistics NC π^0 samples effectively constrained the rate and systematic uncertainty of NC π^0 production in argon, which was the dominant background to the $1\gamma 1p$ and $1\gamma 0p$ selections.

MicroBooNE observed no evidence for an excess of NC Δ radiative decay, as shown in figure 77. The measurement ruled out the normalization enhancement factor of 3.18 as an explanation to the MiniBooNE low-energy excess at 94.8% CL (1.9 σ), in favor of the nominal prediction for NC $\Delta \rightarrow N\gamma$. Note that this was a model-dependent test of the MiniBooNE excess, and therefore does not necessarily apply universally to any photon-like interpretation. Those include, e.g. coherent single-photon production, which is expected to be a rarer process in MicroBooNE than NC $\Delta \rightarrow N\gamma$, or BSM processes that manifest as single-photon events, such as co-linear e^+e^- from the decay of new particles. Those will be the target of dedicated follow-up MicroBooNE analyses, as well as model-independent single-photon searches.

4.3.2. Reactor flux models. As discussed in section 2.3, modeling the reactor $\bar{\nu}_e$ flux is quite challenging. State-of-the art models used to compare against measured IBD yields employed the conversion approach and relied on β -decay measurements performed at ILL in the 1980s. The presence of mistakes in these beta-decay measurements, as well as incorrect assumptions present in the conversion process, could lead to mis-modeled reactor $\bar{\nu}_e$ fluxes. Recent $\bar{\nu}_e$ and β -decay measurements have played a crucial role in further exploring flux prediction issues as a possible source of the RAA.

Daya Bay [255] and RENO [256] experiments—leveraging their IBD yield measurements as a function of evolving fission fractions in the reactors—were able to perform simultaneous IBD yield measurements of ²³⁵U and ²³⁹Pu. While continuing to observe the same time-integrated IBD yield deficit that defines of the RAA, these results also showed that while ²³⁹Pu yields are in good agreement with the models, ²³⁵U is at 0.925 \pm 0.015 (>3 σ) of



Figure 78. IBD yields of ²³⁵U and ²³⁹Pu from Daya Bay (blue), RENO (red), and a combined fit to Daya Bay and RENO. Horizontal and vertical black lines show the predicted IBD yields for ²³⁹Pu and ²³⁹Pu respectively based on the Huber-Mueller (HM) model. While ²³⁹Pu yields agree with the model within 1 σ , a >3 σ discrepancy can be noticed in ²³⁵U yields. Reproduced from [404]. CC BY 4.0.

the modeled yield as shown in figure 78. While these results are in slight tension ($\sim 1\sigma$) with the average ²³⁵U IBD yields from the pure ²³⁵U measurements done using HEU reactors [399], they are in approximate agreement ($\sim 0.5\sigma$) with the modern ²³⁵U yield measured by the STEREO experiment [400]. It is also worth noting that summation-predicted IBD yields are in good agreement with the Daya Bay's flux evolution measurements [44, 254].

To test the possibility of mis-modeled IBD yields in the conversion model, Kopeikin *et al* [401] performed simultaneous β -decay spectrum measurements of ²³⁵U and ²³⁹Pu at the Kurchatov Institute, generating a conversion prediction, referred to as the KI Model that uses completely different primary inputs than those used by the Huber model. While the measurement was designed to achieve an extremely high degree of correlation between ²³⁵U and ²³⁹Pu measurements, it exhibited lower statistical precision than the ILL measurements of the 1980s. The KI model found a consistently higher ratio of ²³⁵U to ²³⁹Pu β -decay rates over the full energy range compared to the ILL measurements as shown in figure 79. Potential bias in β decay measurements have also been pointed out in a separate study [402].

These new conversion prediction results, together with the $\bar{\nu}_e$ flux evolution measurements from Daya Bay and RENO and the summation-conversion flux evolution mis-matches mentioned in section 3.3.2 collectively suggest mis-modeled ²³⁵U IBD yields as a major contributor to the RAA. This collective picture is well-illustrated in figure 80, which shows predictions and $\bar{\nu}_e$ -based measurements of IBD yields for the dominant fission isotopes ²³⁵U and ²³⁹Pu, given in terms of a ratio with respect to the Huber–Mueller conversion prediction. Conversion predictions based on the ILL beta measurements (Huber–Mueller and the HKSS model discussed in section 3.3.2) appear to deviate from all other measurements and predictions. Huber-Mueller models show substantial deficits with respect to direct IBD



Figure 79. Ratio of the cumulative β spectra for ²³⁵U and ²³⁹Pu as measured at ILL [661] (blue) and at KI [662] (red). KI sees consistently lower ratios than ILL. Reprinted (figure) with permission from [401], Copyright (2021) by the American Physical Society.

yield measurements, whereas discrepancies between EF and KI models with respect to the data are not large enough to claim the existence of any anomaly whatsoever.

It is important to stress that while these flux-based indications are fairly clear in their suggestion of flux prediction issues in the Huber-Mueller model and cast doubt on a pure sterile neutrino interpretation of the RAA, this global flux picture still leaves substantial room for short-baseline sterile oscillation phenomena at reactors. As previously mentioned in section 3.3.2, it is likely that the error bars assigned to most, if not all, predictions in figure 80 are under-estimated, leaving ample room for sterile oscillation amplitudes around the 20% level or lower. Moreover, multiple studies show that hybrid models containing both incorrect flux predictions and sterile neutrinos also provide a good fit to global IBD yield datasets [403, 404]; some of these scenarios produce best-fit oscillation-induced deficits well in excess of 10%. In light of this degeneracy between incorrect flux predictions and constant oscillation-induced deficits, short-baseline reactor measurements capable of directly probing the deficit's L/E character are the better bet for cleanly elucidating the the role played by short-baseline sterile oscillations in the reactor sector.

4.3.3. MicroBooNE ν_e CC measurement. MicroBooNE has additionally performed a search that explicitly tests the nature of the MiniBooNE excess in a physics-model-agnostic way. Specifically, MicroBooNE has carried out three independent analyses to investigate whether the MiniBooNE observed low-energy excess can be described by an effective enhancement of ν_e CC scattering at low energy given by unfolding the MiniBooNE observed excess distribution. This unfolding predicts a factor of 5–7 enhancement of ν_e CC interactions below 500 MeV in true neutrino energy.

The data used in this search correspond to approximately half the data set collected by MicroBooNE during its entire operational run time in the Fermilab BNB. The search was carried out by three separate analyses, each targeting different exclusive and inclusive ν_e CC final states, and using separate reconstruction paradigms and signal selections. The ν_e final states explicitly targeted by MicroBooNE include a pion-less final state topology with one electron and with 0 or N > 1 protons as part of the visible hadronic final state; a CCQE-like



Figure 80. Best-fit points and 95 % CL, and 99 % CL contours of ²³⁵U and ²³⁹Pu IBD yields obtained using integrated rates (red), evolution measurements by Daya Bay and RENO (purple), and combined integrated rates and evolution measurements (gray). The axes correspond to r_i (\overline{R}_{HKSS} as described above) for ²³⁵U and ²³⁹Pu along x and y direction respectively. Predictions from HM, EF, and HKSS models are shown in cyan, orange, and blue contours respectively. Black dashed line represents $r_{235} = r_{239}$. It is clear from the plot that while the measured IBD yields agree with the predictions from the EF (*ab inito*) model for both isotopes, only ²³⁹Pu agrees with the HM and HKSS models—both of which are conversion models relying on ILL β spectra measurements —pointing towards an issue with the conversion approach-based ²³⁵U predictions. Reproduced from [110]. CC BY 4.0.

final state topology with one electron and with only 1 proton as part of the visible hadronic final state; and a CC-inclusive final state topology with one electron and with any number of charged pions or protons as part of the visible hadronic final state. While probing different event topologies with distinct event reconstruction and selection methods, the three independent analyses still share several common aspects, including the signal model, a Geant4-based simulation of the neutrino beam, the detector response model, and a tuned variation of the GENIE v3 event generator incorporating the most up to date knowledge of neutrino scattering in the <1 GeV energy range. The results from each of the three analyses are presented in detail in [279, 281, 282], and are summarized in [280]. The final observed and predicted distributions from each search are reproduced in figure 81, with the observed data to Monte Carlo ratios shown in figure 82.

While statistics-limited, the three mutually-compatible analyses collectively reported no excess of low-energy ν_e candidates, and were found to be either consistent with or modestly lower than the predictions for all ν_e event classes, including inclusive and exclusive hadronic final-states, and across all energies. With the exception of the pion-less, zero-proton selection, which was the least sensitive to a simple model of the MiniBooNE low-energy excess, MicroBooNE rejected the hypothesis that an enhancement of ν_e CC interactions at low energy



Figure 81. MicroBooNE's four targeted ν_e CC spectra. Reproduced from [280]. CC BY 4.0.

is fully responsible for the MiniBooNE low-energy excess at >97% CL for both exclusive and inclusive event classes. Additionally, MicroBooNE disfavored generic ν_e interactions as the primary contributor to the excess, with a 1σ (2σ) upper limit on the inclusive ν_e CC contribution to the excess of 22% (51%).

It should further be noted that while these searches do not explicitly test the possibility of light sterile neutrino oscillations, a recent MicroBooNE publication [278], as well as independent phenomenological studies, have further analyzed the observed data within that context, including ν_e appearance and/or ν_e disappearance. The results of those are summarized in section 4.1.2.1.

5. Astrophysical and cosmological indirect probes

It has been known for a long time that astrophysical observations can provide powerful constraints on BSM physics. Stellar evolution arguments have long been invoked to constrain couplings between light BSM particles and SM particles [405]. In particular, the success of a core-collapse supernova and the synthesis of heavy elements via the *r*-process therein maybe turned into an argument for or against sterile neutrino states (see, e.g. [1]). In the past two decades, however, the strongest astrophysical statements on light sterile neutrinos have come from cosmological measurements, i.e. those observations that probe the Universe on the largest length scales. We discuss in this section the relevant theoretical arguments and the most recent observational constraints.



Figure 82. Summary of MicroBooNE ν_e CC measurements. Reproduced from [280]. CC BY 4.0.

5.1. Cosmology

The standard hot big bang model predicts that the three generations of SM neutrinos are held in a state of thermodynamic equilibrium with other SM particles via the weak interaction in the first second post-big bang. At temperatures below about 20 MeV, the dominant equilibrating interactions are

$$\begin{array}{l}
\nu e \to \nu e, \\
\nu \bar{\nu} \to e^+ e^-, \\
\nu \nu \to \nu \nu.
\end{array}$$
(53)

As the Universe expands and cools, these interactions become less frequent. When the Universe cools to a temperature of $\mathcal{O}(1)$ MeV, the interaction rate per neutrino Γ drops below the Hubble expansion rate H. From this point onwards, the neutrinos are said to be 'decoupled' from the thermal plasma. The typical energy of the neutrino ensemble at decoupling is $E \sim 3T \gg m_{\nu}$, i.e. the ensemble is largely ultra-relativistic at decoupling. Because of this, the neutrinos retain to a high degree of accuracy their relativistic Fermi–Dirac phase space distribution parametrized by a temperature and possibly a nonzero chemical potential.

Shortly after neutrino decoupling, the e^+e^- plasma becomes non-relativistic at $T \sim 0.5$ MeV. Here, kinematics favor the annihilation of e^+e^- pairs. The energy released in this process 'reheats' the photon population. The neutrino population, however, does not feel this reheating, because they have already decoupled: in other words, the weak interaction processes are no longer efficient at transporting the energy released from the annihilation to the neutrino sector. The net effect is that the neutrinos emerge from the annihilation event a little colder than the photons. Assuming (i) ideal gases, (ii) instantaneous neutrino decoupling, and (iii) ultra-relativistic electrons/positrons at neutrino decoupling, one can show using entropy conservation arguments

Table 14. Leading-digit contributions from various SM corrections, in order of importance, that make up the final $N_{\rm eff}^{\rm SM} - 3$ (adapted from [407]). The largest, m_e/T_d correction results from dropping the assumption of an ultra-relativistic electron-positron population; finite-temperature quantum electrodynamics corrections (FTQED) to the QED equation of state (EoS) enter at $\mathcal{O}(e^2)$ and $\mathcal{O}(e^3)$, where *e* is the elementary electric charge; the non-instantaneous decoupling+spectral distortion correction is defined relative to an estimate of $N_{\rm eff}^{\rm SM}$ in the limit of instantaneous decoupling assuming $T_d = 1.3453$ MeV [683]; and Type (a) FTQED corrections to the weak rates refer to neutrino–electron scattering rates corrected with thermal masses. Reproduced from [407]. © 2021 IOP Publishing Ltd and Sissa Medialab. All rights reserved.

Standard-model corrections to $N_{\rm eff}^{\rm SM}$	Leading-digit contribution
m_e/T_d correction	+0.04
$\mathcal{O}(e^2)$ FTQED correction to the QED EoS	+0.01
Non-instantaneous decoupling+spectral distortion	-0.005
$\mathcal{O}(e^3)$ FTQED correction to the QED EoS	-0.001
Neutrino flavor oscillations	+0.0005
Type (a) FTQED corrections to the weak rates	$\lesssim \! 10^{-4}$

$$T_{\nu} = \left(\frac{4}{11}\right)^{1/3} T_{\gamma}, \tag{54}$$

where T_{ν} and T_{γ} are the neutrino and the photon temperatures, respectively, after e^+e^- annihilation.

equation (54) is generally taken to define the standard neutrino temperature after $e^+e^$ annihilation. In reality, however, dropping any one of the three aforementioned assumptions can result in percent-level corrections to the relation. These corrections are commonly absorbed into the definition of the effective number of neutrinos $N_{\rm eff}$ via

$$\rho_{\nu} = N_{\rm eff} \, \frac{7\pi^2}{120} \, T_{\nu}^4 = N_{\rm eff} \, \frac{7}{8} \left(\frac{4}{11}\right)^{4/3} \rho_{\gamma}.$$
(55)

Here, ρ_{ν} is the actual total neutrino energy density after e^+e^- annihilation (but still deep in the radiation domination era when the neutrinos are ultra-relativistic), the quantity $(7\pi^2/120)T_{\nu}^4$ denotes the energy density in one family of thermally-distributed relativistic neutrinos with a temperature T_{ν} defined in equation (54), and ρ_{γ} is the actual energy density in the photon population, i.e. what eventually becomes the cosmic microwave background (CMB) radiation. Precision calculations of the SM prediction of $N_{\rm eff}$ find $N_{\rm eff}^{\rm SM} = 3.0440 \pm 0.0002$ [406, 407], including the effects summarized in table 14.¹²⁵

The effective number of neutrinos N_{eff} is of interest to cosmology primarily because energy density in relativistic particles affect directly the Hubble expansion expansion rate during the radiation domination era. In the epoch after e^+e^- annihilation, the expansion rate is given by

¹²⁵ Note that because the parameter N_{eff} is defined in relation to the neutrino energy density in an epoch when the neutrinos are ultra-relativistic, there is no ambiguity in the definition (equation (55)) even if the neutrinos should become non-relativistic at a later time because of their nonzero masses, provided the masses do not exceed the eV scale.

J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501

$$H^{2}(t) = \frac{8\pi G}{3}(\rho_{\gamma} + \rho_{\nu}),$$
(56)

where G is the gravitational constant. With the photon energy density ρ_{γ} having been measured to better than 0.1% accuracy by the FIRAS instrument on board COBE [408, 409], constraints on H(t) in the early universe can be interpreted as bounds on the ratio ρ_{ν}/ρ_{γ} and hence N_{eff} .

From a particle physics standpoint, a thermal population of light sterile neutrinos is one possible cause of a N_{eff} that differs from the standard $N_{\text{eff}}^{\text{SM}}$ value. However, it is important to emphasize that, as far as the Hubble expansion rate H(t) is concerned, any thermal background or non-thermal population (e.g. from decays) of non-photon light particles such as axions, majorons, or even gravitons will contribute to N_{eff} . Such scenarios have been considered by many authors, including Refs. [410–416]. Likewise, any process that alters the thermal abundance of neutrinos (e.g. a low reheating temperature) or affects directly the expansion rate itself (e.g. a time-dependent gravitational constant *G*) can mimic a non-standard N_{eff} value. Yet another way to change N_{eff} is to tinker with the photon energy density itself (via, e.g. interaction with millicharged particles [417] or late kinetic decoupling of the dark matter [418]), while preserving the neutrino energy density.

In the case of a non-standard N_{eff} due to a BSM light particle, the mass of the new particle can also impact on the evolution of cosmological density perturbations via its role as a 'hot dark matter'; the mathematical description of this effect goes beyond the Hubble expansion rate (equation (56)).

In the following, we describe first how a thermal population of light sterile neutrinos can arise in the early universe through a combination of neutrino flavor oscillations and scattering with other SM particles. We then discuss the various signatures of light sterile neutrinos in cosmological observables such as the light elemental abundances from big bang nucleosynthesis (BBN), the CMB anisotropies and the large-scale structure (LSS) distribution, constraints from current observations, as well as various proposals on how to get around them.

5.1.1. Light sterile neutrino thermalization. If a sterile neutrino state mixes sufficiently strongly with any of the active neutrino states, a thermal population of light sterile neutrinos that adds to $N_{\rm eff}$ can be produced prior to neutrino decoupling via a combination of active-sterile neutrino oscillations and collisions (i.e. elastic and inelastic scattering) with the primordial plasma of SM particles. Roughly speaking, as the Universe cools, an initial population comprising only active neutrinos can begin to oscillate into sterile neutrinos once the oscillation frequency, given by $\Delta m^2/(2E)$, becomes larger than the Hubble expansion rate H(t). The role of collisions is then to force a neutrino into a flavor eigenstate and hence 'measure' the flavour content of the ensemble. Since the probability of measuring a sterile flavor is nonzero, signifying that an active neutrino has turned into a sterile state, collisions also play the role of refilling any gap in the active neutrino distribution vacated by the oscillation process. This effect and the region of parameter space leading to thermalization of the sterile neutrino was found by the early works of, e.g. [419–422] If sterile neutrinos do become thermalized, then we expect them to have the same temperature as the active neutrinos.

Nowadays, light sterile neutrino thermalization in the early universe can be computed precisely, using a generalized Boltzmann formalism developed in [423, 424] which tracks the flavor evolution of a neutrino ensemble under the influence of neutrino oscillations and scattering. Schematically, the generalized Boltzmann equation for the one-particle reduced density matrix of the neutrino ensemble, $\rho(t, p)$, in the Friedmann-Lemaître–Robertson–

Walker (FLRW) universe of standard hot big bang cosmology is given by

$$\partial_t \rho - p H \partial_p \rho = -\mathbf{i}[\mathbb{H}, \rho] + \mathcal{I}[\rho], \tag{57}$$

where ∂_t and ∂_p are partial derivatives with respect to the cosmic time t and physical momentum p respectively, H is the Hubble expansion rate, $[\mathbb{H}, \rho] \equiv \mathbb{H}\rho - \rho\mathbb{H}$ denotes a commutator between the flavor oscillations Hamiltonian \mathbb{H} and ρ , and the collision integrals $\mathcal{I}[\rho]$ encapsulate all non-unitary (scattering) effects on ρ . In the fully CP symmetric case, one set of density matrix $\rho(t, p)$ suffices to describe the evolution of the whole neutrino ensemble including antineutrinos. If however the system is CP asymmetric, we will need to introduce a separate one-particle reduced density matrix $\bar{\rho}(t, p)$ for the antineutrino ensemble. Here, we follow the convention of [423], and define the density matrices using the "transposed notation, e.g. for a 3 active+1 sterile system one would have

$$\varrho(t,p) = \begin{pmatrix} \varrho_{ee} & \varrho_{e\mu} & \varrho_{e\tau} & \varrho_{es} \\ \varrho_{\mu e} & \varrho_{\mu\mu} & \varrho_{\mu\tau} & \varrho_{\mus} \\ \varrho_{\tau e} & \varrho_{\tau\mu} & \varrho_{\tau\tau} & \varrho_{\taus} \\ \varrho_{se} & \varrho_{s\mu} & \varrho_{s\tau} & \varrho_{ss} \end{pmatrix}, \qquad \bar{\varrho}(t,p) = \begin{pmatrix} \bar{\varrho}_{ee} & \bar{\varrho}_{\mu e} & \bar{\varrho}_{\tau e} & \bar{\varrho}_{se} \\ \bar{\varrho}_{e\mu} & \bar{\varrho}_{\mu\mu} & \bar{\varrho}_{\tau\mu} & \bar{\varrho}_{s\mu} \\ \bar{\varrho}_{e\tau} & \bar{\varrho}_{\mu\tau} & \bar{\varrho}_{\tau\tau} & \bar{\varrho}_{s\tau} \\ \bar{\varrho}_{es} & \bar{\varrho}_{\mu s} & \bar{\varrho}_{\tau s} & \bar{\varrho}_{ss} \end{pmatrix}.$$
(58)

This convention enables the equations of motion to be expressed in a more compact form.

Working in the flavor basis, the oscillations Hamiltonian under this convention is

$$\mathbb{H}(p, t, T(t)) = \pm \frac{U\mathbb{M}U^{\dagger}}{2p} + \sqrt{2} G_F[\mathbb{N}_{\ell}(T(t)) - \mathbb{N}_{\bar{\ell}}(T(t)) + \mathbb{N}_{\nu}(t) - \mathbb{N}_{\bar{\nu}}(t)]$$

$$\mp 2\sqrt{2} G_F p \left[\frac{\mathbb{E}_{\ell}(T(t)) + \mathbb{E}_{\bar{\ell}}(T(t)) + \mathbb{P}_{\ell}(T(t)) + \mathbb{P}_{\bar{\ell}}(T(t))}{m_W^2} + \frac{4}{3} \frac{\mathbb{E}_{\nu}(t) + \mathbb{E}_{\bar{\nu}}(t)}{m_Z^2} \right],$$
(59)

which contains a vacuum and an in-medium part. The vacuum part consists of the neutrino squared-mass difference matrix \mathbb{M} and the vacuum mixing matrix U, with the '+' sign pertaining to neutrinos and the '-' sign to the antineutrinos. The in-medium part proportional to the Fermi constant G_F , i.e. the 'matter potential', contains a CP symmetric and a CP asymmetric correction to the neutrino dispersion relation. The CP asymmetric part is similar to the usual matter potential found in, e.g. the Sun, with the matrix $\mathbb{N}_{\ell} - \mathbb{N}_{\bar{\ell}}$ denoting the asymmetry in the number density of charged leptons; in a 3 + 1 system \mathbb{N}_{ℓ} would take the form

$$\mathbb{N}_{\ell} \equiv \frac{1}{2\pi^2} \int dp \ p^2 \ \text{diag}(f_e, f_{\mu}, f_{\tau}, 0) = \text{diag}(n_e, n_{\mu}, n_{\tau}, 0), \tag{60}$$

where $f_{\ell}(p, T)$ is the (equilibrium) occupation number of the charged lepton ℓ . A similar expression exists also for \mathbb{N}_{ℓ} . Typically, the charged-lepton asymmetries are of order 10^{-10} , so this $\mathbb{N}_{\ell} - \mathbb{N}_{\bar{\ell}}$ term is not strictly a necessary ingredient. Observe however the additional term proportional to $\mathbb{N}_{\nu} - \mathbb{N}_{\bar{\nu}}$, where

$$\mathbb{N}_{\nu} \equiv \frac{1}{2\pi^2} \int \mathrm{d}p \; p^3 \; S_a \varrho S_a,\tag{61}$$

and similarly for $\mathbb{N}_{\bar{\nu}}$. This CP asymmetric term describes neutrino self-interaction in the presence of a large $(>10^{-5})$ neutrino asymmetry. Current cosmological data constrain neutrino asymmetries only to $\mathcal{O}(10^{-2})$ [425], so the presence of a sizeable $\mathbb{N}_{\nu} - \mathbb{N}_{\bar{\nu}}$ term is an interesting possibility. In standard calculations of light sterile neutrino thermalization, however, this asymmetry is set to zero. The quantity S_a is a diagonal matrix that projects out

only the active neutrino states, since sterile states are by definition interaction-less. In a 3 + 1 system it takes the form $S_a = \text{diag}(1, 1, 1, 0)$.

In the CP symmetric portion of the in-medium terms, m_W and m_Z denote respectively the W and Z boson mass, while the terms \mathbb{E}_{ℓ} and \mathbb{P}_{ℓ} are momentum-integrals of some combinations of the charged-lepton energy $E_{\ell}(p) \equiv (p^2 + m_{\ell}^2)^{1/2}$ and occupation number $f_{\ell}(p, T)$, and \mathbb{E}_{ℓ} , \mathbb{P}_{ℓ} are their antiparticle counterparts. For a 3 + 1 system, we have

$$\mathbb{E}_{\ell} \equiv \frac{1}{2\pi^2} \int \mathrm{d}p \; p^2 \operatorname{diag}(E_e f_e, E_{\mu} f_{\mu}, E_{\tau} f_{\tau}, 0) = \operatorname{diag}(\rho_e, \rho_{\mu}, \rho_{\tau}, 0), \qquad (62)$$

$$\mathbb{P}_{\ell} \equiv \frac{1}{6\pi^2} \int dp \ p^2 \ diag \left(\frac{p^2}{E_e} f_e, \ \frac{p^2}{E_{\mu}} f_{\mu}, \ \frac{p^2}{E_{\tau}} f_{\tau}, \ 0 \right) = diag(P_e, \ P_{\mu}, \ P_{\tau}, \ 0), \quad (63)$$

where ρ_{ℓ} and P_{ℓ} are, respectively, the energy density and pressure of the charged lepton ℓ . Lastly, \mathbb{E}_{ν} is the equivalent of equation (62) for an ultra-relativistic neutrino gas,

$$\mathbb{E}_{\nu} \equiv \frac{1}{2\pi^2} \int \mathrm{d}p \; p^3 \; S_a \varrho S_a,\tag{64}$$

where S_a is again a diagonal matrix that projects out only the active neutrino states.

Note that the $\mathbb{E}_{\ell} + \mathbb{P}_{\ell}$ term in equation (59) differs from its usual presentation found in, e.g. equation (2.2) of [426], which has $\mathbb{E}_{\ell} + \mathbb{P}_{\ell}$ replaced with $(4/3)\mathbb{E}_{\ell}$. First reported in [427], the former is in fact the more general result, while $(4/3)\mathbb{E}_{\ell}$ applies strictly only when the charged leptons are ultra-relativistic.

The collision integral $\mathcal{I}[\varrho]$ incorporates in principle all weak scattering processes wherein at least one neutrino appears in either the initial or final state. All published calculations to date, however, account only for $2 \rightarrow 2$ processes involving (i) two neutrinos and two charged leptons anyway distributed in the initial and final states, and (ii) neutrino– neutrino scattering. Then, schematically, $\mathcal{I}[\varrho]$ comprises 9D momentum-integrals that, at tree level, can be systematically reduced to 2D integrals of the form

$$\mathcal{I}[\varrho(p,t)] \propto G_F^2 \int \mathrm{d}p_2 \,\mathrm{d}p_3 \,\Pi(p,p_2,p_3;t) \,F(\varrho),\tag{65}$$

where Π is a scalar function representing the scattering kernel, and *F* is a phase space matrix including quantum statistics.

The vast majority of existing works on light sterile neutrino thermalization solve the above set of equations of motion for a 1 active+1 sterile system. Because of the complexity of the collision integrals $\mathcal{I}[\varrho]$, a variety of approximations have been introduced to simplify the integrals and hence speed up the computation [424, 428–432], although of course it is also possible to solve the collision integrals exactly if percent-level precision is required [432]. The publicly available code LASAGNA [433] solves a 1 + 1 system and also allows for the possibility of a CP asymmetry. On the other hand, multi-flavor (i.e. 3 + 1) effects have been considered in [434] using the momentum-averaged approximation and in [166] using a multi-momentum approach. Most recently, a publicly available multi-flavor code FortEPiaNO [426] has also become available, which is capable of solving a system with up to 3 active+3 sterile species, although it does not provide the CP asymmetric case.

Irrespective of how exactly the system is solved $(3 + 1, 1 + 1, approximate or exact collision integrals, etc), however, the general conclusion is that the active-sterile neutrino squared mass difference and mixing required to explain the short-baseline anomalies will inevitably lead to a fully thermalized light sterile neutrino population with a temperature similar to that of the active neutrinos. In other words, in terms of the <math>N_{\text{eff}}$ parameter, we expect



Figure 83. The change in the effective number of neutrinos, $\Delta N_{\text{eff}} \equiv N_{\text{eff}} - N_{\text{eff}}^{\text{SM}}$, at a temperature T = 0.1 MeV from active-sterile neutrino oscillations in a 1 + 1 system, where δm^2 is the squared mass difference and θ is the effective mixing angle between the active and the sterile state. Reproduced from [432]. CC BY 3.0.

the canonical light sterile neutrino solution to the short-baseline anomalies to lead to $N_{\rm eff} \sim 4$. Figure 83 and 84 show the prediction for $N_{\rm eff}$ as a function of the mixing parameters in a 1 + 1 and 3 + 1 scheme respectively. We discuss in the next section the observable consequences of this light sterile neutrino population.

5.1.2. Observable consequences. Three standard cosmological probes are sensitive to the presence of a light sterile neutrino population primarily through its contribution to increasing $N_{\rm eff}$ and hence the Hubble expansion rate, and/or through its nonzero mass and hence its role as a 'hot dark matter'. We describe these probes below.

5.1.2.1. Big bang nucleosynthesis. The discussion of light element formation from protons and neutrons via the process of big bang nucleosynthesis (BBN) (see [435] for a recent review) usually begins at a temperature around $T \sim 0.7$ MeV, when the weak processes

$$\nu_e + n \leftrightarrow e^- + p,
\bar{\nu}_e + p \leftrightarrow e^+ + n$$
(66)

become inefficient compared with the Hubble expansion rate and equilibrium can no longer be maintained. When these process go out of equilibrium, the ratio of neutron-to-proton number density also freezes out, to a value given by

$$\frac{n_n}{n_p}\Big|_{\rm fr} = \exp\left(-\frac{m_n - m_p}{T_{\rm fr}}\right),\tag{67}$$

where $T_{\rm fr}$ is the freeze-out temperature, and $m_{n,p}$ are the neutron and proton masses. For $T_{\rm fr} \simeq 0.7$ MeV, this ratio evaluates approximately to 1/6. Free neutron decay over a lifetime of about 880 s, however, will reduce it to a smaller number by the end of BBN.



Figure 84. The final effective number of neutrinos from active-sterile neutrino oscillations in a 3 + 1 scheme, under the constraint $\sum \log_{10} |U_{i4}|^2 = -13$. Reproduced from [426]. © 2019 IOP Publishing Ltd and Sissa Medialab. All rights reserved.

In standard BBN, the formation of elements commences at a temperature controlled by the baryon-to-photon ratio η , when the energy in the photon bath per baryon has become sufficiently low such that newly formed nuclei are no longer immediately broken apart. For η

 $\sim 6 \times 10^{-10}$, this temperature is around $T \sim 0.1$ MeV. The first element to be formed is Deuterium, followed by the production of heavier nuclei. Of particular note is Helium-4, whose mass fraction is defined as

$$Y_p \equiv \frac{4n_{\text{He 4}}}{n_n + n_p},\tag{68}$$

where n_{He4} is the number density of ⁴He. Because ⁴He has the largest binding energy amongst the light elements, the bulk of all initially available neutrons will eventually end up bound in Helium-4 nuclei, i.e. we expect $n_{\text{He}} \ _4 \simeq n_n/2$. Then, to estimate Y_p from equation (68) we simply need to note that the neutron-to-proton ratio typically drops to about 1/7 at the end of BBN via neutron decay. From this we find $Y_p \simeq 0.25$.

Besides Helium-4 mass fraction, small amounts of Deuterium and ³He (D/H \sim ³He/H $\sim O(10^{-5})$), as well as traces of ⁶Li and ⁷Li, are expected to remain. Unlike for ⁴He, however, there are no simple ways to estimate their abundances, and we must rely on solving a set of Boltzmann equations to track their number densities. Several publicly available codes can perform this task, including AlterBBN [436, 437], PArthENOPE [438, 439], and PRIMAT [435]. In standard BBN, barring experimental uncertainties in the nuclear reaction rates and the free neutron lifetime, the baryon-to-photon ratio η alone enters these Boltzmann equations and is hence the sole free parameter in the determination of the elemental abundances.

A non-standard neutrino sector can alter this picture in two different ways. Firstly, as can be seen in equation (66), electron neutrinos participate directly in the CC weak interactions that determine the neutron-to-proton ratio. If because of non-standard physics these neutrinos should end up at $T \leq 1$ MeV with an energy spectrum that departs strongly from an equilibrium relativistic Fermi–Dirac distribution with zero chemical potential, then the equilibrium of the processes (equation (66)) could shift to a different point and in so doing alter the neutron-to-proton ratio. A particularly well studied example in this regard is the case of a nonzero electron neutrino chemical potential μ_e , which shifts the neutron-to-proton ratio at weak freeze-out in a manner well described by

$$\left. \frac{n_n}{n_p} \right|_{\rm fr} = \exp\left(-\frac{m_n - m_p}{T_{\rm fr}} - \frac{\mu_e}{T_{\rm fr}}\right). \tag{69}$$

The effects of more general distortions to the ν_e and/or $\bar{\nu}_e$ energy spectra on n_n/n_p need to be computed numerically using a Boltzmann code.

Secondly, neutrinos of all flavors influence the expansion rate of the Universe prior to and during BBN through their energy densities via equation (56). Therefore, if because of new physics the total neutrino energy should be larger than the standard $N_{\rm eff}^{\rm SM}$, the freeze-out of the processes (equation (66)) would occur at a higher temperature. This in turn pushes up the neutron-to-proton ratio via equation (67) and hence the Helium-4 mass fraction as well via equation (68). Figure 85 shows the Deuterium, Helium-4 and Lithium-7 abundances computed using PArthENOPE [438, 439], as a function of the baryon density ω_b , the excess number of relativistic degrees of freedom $N_s \equiv \Delta N_{\rm eff} \equiv N_{\rm eff} - N_{\rm eff}^{\rm SM}$, and the electron neutrino degeneracy parameter $\xi \equiv \mu_e/T$.

For the specific problem of a light sterile neutrino with mixing parameters compatible with hints from terrestrial experiments, only the second effect is relevant. This is because the relatively large mass-squared difference and mixing between the active and the sterile neutrino states essentially guarantee full thermalization of the sterile species prior to the decoupling of the active neutrinos (see figures 83 and 84). In other words, $\Delta N_{\text{eff}} \simeq 1$ and the phase space distribution of the sterile states follows closely the equilibrium relativistic Fermi–

J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501



Figure 85. Deuterium, Helium-4 and Lithium-7 abundances, as computed using PArthENOPE [438, 439], as a function of cosmological parameters: ω_b is the baryon density, N_s is our $\Delta N_{\rm eff} \equiv N_{\rm eff} - N_{\rm eff}^{\rm SM}$, and $\xi \equiv \mu_e/T$ is the electron neutrino degeneracy parameter. Reproduced from [454]. © 2011 IOP Publishing Ltd and SISSA. All rights reserved.

Dirac distribution of the active neutrinos. This also implies equipartition amongst the four neutrino flavors, such that any further active-sterile flavor oscillations after neutrino decoupling will not cause the ν_e phase space distribution to deviate from a thermal distribution. Thus, to constrain such light sterile neutrino scenarios, we only need to extend standard BBN with one extra free parameter, namely, N_{eff} . We emphasize, however, that more general cases of active-sterile neutrino mixing would require that we determine the ν_e energy spectrum as well, in order to determine the full effect of light sterile states on the light elemental abundances.

Current BBN constraints. Current astrophysical observations put the Helium-4 mass fraction at $Y_p = 0.2449 \pm 0.0040$ [440], the Deuterium abundance at D/H = $(2.527 \pm 0.030) \times 10^{-5}$ [441], and the Lithium-7 abundance at ⁷Li/H = $(1.58 \pm 0.3) \times 10^{-10}$ [442], while the Helium-3 abundance is constrained to ³He/H < $(1.1 \pm 0.2) \times 10^{-5}$ [443]. Of these, only the D/H and Y_p measurements have sufficient precision to probe N_{eff} during the BBN epoch. How D/H and Y_p probe N_{eff} can be seen in figure 86, which shows in the top panel the 68.27% and 95.45% contours in the $(\Omega_b h^2, N_{\text{eff}})$ -plane, where $\Omega_b h^2 \equiv \omega_b$ is the baryon density, obtained from various data combinations, and in the bottom panel the corresponding 1D marginalized posterior for N_{eff} .

As can be seen, the Helium-4 mass fraction alone is already quite sensitive to N_{eff} , although this measurement is not particularly useful for pinning down the baryon-to-photon ratio η and hence the baryon density $\Omega_b h^2$. This is because η has no strong influence on the rate at which ⁴He is formed, other than setting the initial time of BBN. In contrast, the Deuterium abundance is strongly sensitive to both N_{eff} and η , which do directly affect the



Figure 86. Top: 68.27% and 95.45% contours in the $(\Omega_b h^2, N_{\text{eff}})$ -plane obtained from various data combinations. Bottom: 1D marginalized posterior for N_{eff} . Reprinted from [435], Copyright (2018), with permission from Elsevier.

formation rate. However, because these two measurements have opposite degeneracy directions on the $(\Omega_b h^2, N_{\text{eff}})$ -plane, in combination they provide a good measurement of both $\Omega_b h^2$ and N_{eff} . For the latter, [435] finds

$$N_{\rm eff} = 2.88 \pm 0.27, \quad (68\% \,{\rm CL})$$
 (70)

using the code PRIMAT [435]. Using PArthENOPE [438, 439] instead would have yielded a central value about 2% smaller [435]. Thus, the conclusion here is that current measurements of primordial elemental abundances are completely consistent with the $N_{\rm eff}^{\rm SM} = 3.0440$, and shows no evidence of any extra relativistic degrees of freedom.

5.1.2.2. Cosmic microwave background and large-scale structure. Unlike BBN, probes of the Universe's inhomogeneities such as the CMB temperature and polarization anisotropies and the large-scale matter distribution are not sensitive to the flavor content of the neutrino sector, only to its contribution to the stress-energy tensor. If neutrinos are massless, then the

 N_{eff} parameter as defined in equation (55) alone characterizes their effects on the Universe's evolution. If neutrinos are massive, then, in addition to N_{eff} and the neutrino masses, in principle it is also necessary to know the exact form of the neutrino momentum distribution in order to solve the evolution equations for the inhomogeneities exactly. However, unless the deviations from a thermal relativistic Fermi–Dirac spectrum is of order unity, it suffices to specify only the temperature of the distribution, as late-time cosmological probes are currently not very sensitive to spectral distortions in the neutrino sector. For this reason, most existing analyses characterize the neutrino sector, including light sterile neutrinos produced as described in section 5.1.1, only in terms of the neutrino mass spectrum and the N_{eff} parameter. The light sterile states are assumed to share the same temperature as the standard active neutrinos.

One interesting variation to the above is the case in which an excess $N_{\text{eff}} > N_{\text{eff}}^{\text{SM}}$ is due to a thermalized particle species that has temperature different from the standard neutrino temperature and/or has a different spin statistics, e.g. a thermalized bosonic particle species such as an axion that follows the Bose–Einstein distribution. If this new particle is massless, then again it suffices to describe its phenomenology in terms of its contribution to N_{eff} alone. If however the particle species is massive, then in addition to its mass, its temperature also plays a role in determining the 'hotness' of the resulting hot dark matter, where the relation between the hot dark matter's temperature and abundance is fixed by the particle's spin statistics.

Cosmic microwave background primary anisotropies. The CMB primary anisotropies refer to the temperature and polarization fluctuations imprinted on the last scattering surface. These are sensitive to the physics of the early universe up to the time of photon decoupling ($T \sim 0.1 \text{ eV}$ or redshift $z \sim 1000$), and differ from the secondary anisotropies which are additional spatial fluctuations gathered by the CMB photons as they free-stream from the last scattering surface to the observer and hence sensitive to late-time/low-redshift physics.

The effects of a non-standard $N_{\rm eff}$ value on the CMB primary anisotropies and its associated parameter degeneracies with the present-day Hubble expansion rate H_0 and physical matter density $\omega_{\rm m}$ in the context of Λ CDM cosmology have been discussed extensively in, e.g. [1, 444]. Broadly speaking, if the particles that make up $N_{\rm eff}$ are ultrarelativistic at the time of CMB formation, then a non-standard $N_{\rm eff}$ can manifest itself in the following ways:

1. The redshift of matter-radiation equality z_{eq} controls the ratio of radiation to matter at the time of photon decoupling and hence the evolution of the potential wells. This in turn affects the peak height ratios of the CMB temperature anisotropy spectrum. With seven acoustic peaks measured by the Planck mission [445], the equality redshift z_{eq} has been measured to percent level precision in Λ CDM-type cosmologies. As a probe of N_{eff} , however, we note that

$$z_{\rm eq} = \frac{\omega_{\rm m}}{\omega_{\gamma} + 0.227 N_{\rm eff}} - 1,$$
 (71)

where ω_{γ} is the present-day photon energy density. In other words, $N_{\rm eff}$ is exactly degenerate with the physical matter density $\omega_{\rm m}$, and measuring $z_{\rm eq}$ alone does not determine $N_{\rm eff}$.

2. The angular sound horizon θ_s , which determines the CMB acoustic peak positions, is another quantity sensitive to N_{eff} . Defined as $\theta_s \equiv r_s/D_A$, where r_s is the sound horizon at photon decoupling and D_A is the angular diameter distance to the last scattering surface, the parameter dependence of θ_s in Λ CDM cosmologies is as follows, J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501

$$\theta_s \propto \frac{\Omega_{\rm m}^{-1/2}}{\int_{a^*}^{1} \frac{{\rm d}a}{a^2 \sqrt{\Omega_{\rm m} a^{-3} + (1 - \Omega_{\rm m})}}},$$
(72)

where $\Omega_{\rm m} \equiv \omega_{\rm m}/h^2$, *h* is the reduced Hubble expansion rate defined via $H_0 = 100h$ km s⁻¹ Mpc⁻¹, a^* is the scale factor at photon decoupling, and we have held $z_{\rm eq}$ and the baryon density $\omega_{\rm b}$ fixed. This relation implies that while θ_s constrains the parameter combination $\omega_{\rm m}/h^2$, it does not constrain $\omega_{\rm m}$ and *h* individually. Since there already exists an exact degeneracy between $N_{\rm eff}$ and $\omega_{\rm m}$ through $z_{\rm eq}$ (see equation (71)), this additional ($\omega_{\rm m}$, *h*)-degeneracy through θ_s immediately sets up a three-way degeneracy between $N_{\rm eff}$, $\omega_{\rm m}$, and *h*, which needs to be broken by some other means. In more complex models, degeneracies between $N_{\rm eff}$ and a nonzero spatial curvature Ω_k or a non-canonical dark energy equation of state are also possible.

3. The angular diffusion scale $\theta_d \equiv r_d/D_A$, where r_d is the diffusion scale at photon decoupling, characterizes the scale at which the CMB temperature anisotropy power spectrum becomes suppressed due to diffusion damping (or Silk damping). The phenomenon of diffusion damping occurs at the CMB damping tail, i.e. at multipoles $l \gtrsim 1000$, and was first measured by the Atacama Cosmology Telescope [446] and the South Pole Telescope [447], and now by the Planck CMB mission [445].

For fixed z_{eq} , ω_b and a^* , the angular diffusion scale has a parameter dependence

$$\theta_d \propto (\Omega_{\rm m} H_0^2)^{1/4} \,\theta_s,\tag{73}$$

where θ_s is the angular sound horizon of equation (72). Thus, a simultaneous measurement of θ_d , θ_s , and z_{eq} by a CMB mission such as Planck immediately constitutes a measurement of $\omega_m \propto \Omega_m H_0^2$ and hence N_{eff} . Figure 87 shows the signature of N_{eff} in the CMB damping tail.

Large-scale matter distribution. There are many different ways to quantify and probe the large-scale matter distribution. The most basic quantity, however, is the present-day matter power spectrum P(k).

The broad-band shape of the large-scale matter power spectrum in Λ CDM-type cosmologies by two quantities: the comoving wavenumber at matter-radiation equality

$$k_{\rm eq} \equiv a_{\rm eq} H(a_{\rm eq})$$

\$\approx 4.7 \times 10^{-4} \sqrt{\Omega_m(1 + z_{\rm eq})} h \text{ Mpc}^{-1}, (74)

which fixes the location of the 'turning point' of P(k), and the baryon-to-matter density fraction

$$f_b \equiv \frac{\omega_b}{\omega_m},\tag{75}$$

which determines the suppression in power at $k > k_{eq}$ due to baryon acoustic oscillations (BAO). In addition, the matter power spectrum has small-amplitude oscillatory features which are the manifestation of the BAO themselves. These oscillatory features originate in the same early universe physics as the acoustic peaks in the CMB anisotropy power spectra, and when analyzed together with CMB data, can act as a powerful standard ruler for distance measurements.

If some of the neutrinos are massive and become non-relativistic at late times, then they can constitute a fraction of the present-day dark matter content. However, this neutrino dark matter is 'hot', in the sense that the neutrinos, although non-relativistic, come with a significant thermal velocity dispersion, which tends to hinder its gravitational clustering on



Figure 87. Signature of $N_{\rm eff}$ in the damping tail of the CMB TT power spectrum. Top: here, the cosmological parameters have been adjusted such that the baryon density $\Omega_b h^2$, the redshift of matter-radiation equality $z_{\rm eq}$, the angular sound horizon θ_s , and the normalisation at $\ell = 200$ are fixed for all $N_{\rm eff}$ cases shown. Middle: like the top panel, but with the spectrum normalisation fixed at $\ell = 400$ instead. Bottom: like the middle panel, but here the Helium-4 fraction Y_p is also allowed to vary such that all $N_{\rm eff}$ cases end up with roughly the same angular diffusion scale θ_d . Reprinted (figure) with permission from [444], Copyright (2013) by the American Physical Society.

small scales. In terms of the present-day large-scale matter power spectrum, keeping the total matter density fixed but replacing some of the cold dark matter with neutrino hot dark matter suppresses P(k) at k-values larger than the free-streaming wave number by an amount dependent on the neutrino fraction

$$f_{\nu} \equiv \frac{\omega_{\nu}}{\omega_{\rm m}} = \frac{\sum m_{\nu}/94 \,\,\mathrm{eV}}{\omega_{\rm m}}.$$
(76)

Thus, neutrino masses also influence the overall shape of the matter power spectrum.

Since it is already possible to pin down z_{eq} , Ω_m , and ω_m using the CMB primary anisotropies, measurements of the matter power spectrum P(k) generally do not improve the constraint on N_{eff} in the simplest $\Lambda CDM + N_{eff}$ fit. However, it must be noted that after the formation of the CMB primary anisotropies, these fluctuations are gravitationally lensed by the intervening matter distribution as the CMB photons propagate from the last scattering surface to the observer, contributing to the so-called CMB secondary anisotropies at multipoles $\ell \gtrsim 500$. In other words, any CMB anisotropy signal at $\ell \gtrsim 500$ will always include some information about P(k), and this is particularly useful for the purpose of constraining the neutrino mass sum $\sum m_{\nu}$.

Current CMB and LSS constraints. In a standard Λ CDM parameter inference, estimating cosmological parameter values from the CMB and related observations involve varying six free parameters related to cosmology: the baryon density ω_b , the cold dark matter density ω_c , the Hubble parameter *h*, the spectral index n_s and amplitude A_s of the primordial curvature perturbation, and the optical depth to reionization τ . Analyses of the Planck CMB data also require that we vary of order 20 nuisance parameters to model the foregrounds and instrumental systematics. These are later marginalized.

To constrain radiation excess, at minimum we need to add N_{eff} as a free parameter to this list. Doing so the Planck collaboration finds [445]

 $N_{\rm eff} = 3.00^{+0.57}_{-0.53}$ (95% CL, Planck TT + lowE), (77)

$$N_{\rm eff} = 2.92^{+0.36}_{-0.37}$$
 (95% CL, Planck TTTEEE + lowE), (78)

$$N_{\rm eff} = 3.11^{+0.44}_{-0.43}$$
 (95% CL, Planck TT + lowE + lensing + BAO), (79)

$$N_{\rm eff} = 2.99^{+0.34}_{-0.33}$$
 (95% CL, Planck TTTEEE + lowE + lensing + BAO), (80)

using various combinations of the Planck temperature and *E*-polarization measurements (TTTEEE and lowE), the lensing potential extracted from the Planck temperature maps, as well as the BAO measurements from 6dFGS [448], SDSS-MGS [449], and BOSS DR12 [450]. As can be seen, in all cases, the inference returns an estimate of $N_{\rm eff}$ that is remarkably consistent with the SM prediction of $N_{\rm eff} = 3.0440$.

Since the CMB anisotropies are also sensitive to the large-scale matter distribution at low redshifts because of the weak gravitational lensing signal inherent in all CMB power spectrum, one can also derive a constraint on the neutrino mass sum $\sum m_{\nu}$ from the Planck CMB data at the same time as we constrain N_{eff} . In a 8-parameter $\Lambda \text{CDM} + N_{\text{eff}} + \sum m_{\nu}$ fit, the Planck collaboration finds [445]

$$N_{\rm eff} = 2.96^{+0.34}_{-0.33}$$
 (95% CL, Planck TTTEEE + lowE + lensing + BAO),
 $\sum m_{\nu} < 0.12 \text{ eV}.$ (81)

Note that the fit assumes three degenerate neutrino mass eigenstates of equal abundances. The role of $N_{\rm eff}$ is merely to dial up or down the abundances, which is why $N_{\rm eff}$ can go below the standard value of $N_{\rm eff}^{\rm SM} = 3.0440$. This combined fit is to be compared with the $N_{\rm eff}$ constraint quoted in equation (80) in a 7-parameter fit of the same data combination, which has the same error bars (about 11%) and a central value off only by 2%. It is also interesting to compare it with the constraint obtained on the neutrino mass sum $\sum m_{\nu}$ from 7-parameter $\Lambda \text{CDM} + \sum m_{\nu}$ fit [445]

$$\sum m_{\nu} < 0.13 \text{ eV}$$
 (95% CL, Planck TT + lowE + lensing + BAO), (82)

$$\sum m_{\nu} < 0.12 \text{ eV}$$
 (95% CL, Planck TTTEEE + lowE + lensing + BAO), (83)

which is identical to the upper limit obtained from the 8-parameter fit (equation (81)). Thus, one can conclude from this comparison that there is no strong degeneracy between the N_{eff} parameter and the neutrino mass sum $\sum m_{\nu}$ in the current generation of precision cosmological data.

What about constraints on the sterile neutrino mass m_s ? The Planck collaboration [445] also reports a 'massive sterile neutrino' fit to their data in a scenario in which the three active neutrinos are assumed to have a fixed minimum mass sum of $\sum m_{\nu} = 0.06$ eV and the effective sterile neutrino mass is defined as $m_{\nu,\text{sterile}}^{\text{eff}} \equiv \Omega_{\nu,\text{sterile}} h^2$ (94.1 eV). This effective mass is related to the physical sterile neutrino mass via

$$m_s = (\Delta N_{\rm eff})^{-1} m_{\nu \, \rm sterile}^{\rm eff},\tag{84}$$

assuming the sterile states have the same temperature as the SM neutrinos. Imposing the priors $\Delta N_{\text{eff}} \ge 0$ and $m_s < 10$ eV, they find the constraints

$$N_{\rm eff} < 3.29 \quad (95\% \text{ CL}, \text{ Planck TTTEEE} + \text{lowE} + \text{lensing} + \text{BAO}),$$

 $m_{\nu,\text{sterile}}^{\rm eff} < 0.65 \text{ eV}.$ (85)

Taking $\Delta N_{\rm eff}$ to be a maximum allowed 0.29, the mass bound corresponds to an upper limit of $m_s < 2.24$ eV on the physical sterile neutrino mass. Thus, while cosmological measurements do constrain m_s in any interesting way, it cannot completely rule out a 1 eV light sterile provided thermalization is kept at below the 30% level. In other words, the extent of thermalization as quantified by the $N_{\rm eff}$ parameter remains the limiting factor for the light sterile neutrino scenario in standard cosmology.

Lastly, we note that while the cosmological bound on $N_{\rm eff}$ given in equation (81) is already strongly indicative that the short-baseline light sterile neutrino is in serious tension with precision cosmological measurements, it is nonetheless possible to analyse cosmological and oscillation data together in a consistent way. This has been done most recently in [451], which considers a 3 + 1 scenario, computes the corresponding light sterile neutrino thermalization using the thermalization code FortEPiaNO [426] and the associated $N_{\rm eff}$ with mixing parameters consistent with laboratory measurements, and feeds the output into a CMB analysis. Figure 88 shows the constraints on the $(\Delta m_{41}^2, |U_{e4}|^2)$ - and $(\Delta m_{41}^2, |U_{\mu4}|^2)$ -planes.

5.1.3. Can we evade cosmological constraints? We have seen that the canonical 1 eV-mass light sterile neutrino motivated by the short baseline anomalies is in strong tension with cosmological measurements primarily because of the non-detection of a non-standard $N_{\rm eff} \sim$ 4. However, a criticism often levelled at cosmological constraints is that all parameter estimation via statistical inference are inherently dependent on the cosmological framework assumed in the inference exercise. Given the large number of unknowns in cosmology, e.g. the nature of dark energy, inflation, etc., critics argue that there may exist a corner of this vast unknown parameter space in which a completely thermalized light sterile neutrino state with a mass close to 1 eV might be permitted to live.

To investigate this possibility, the Planck collaboration has provided a large number of analyses of expanded cosmological parameter spaces, often in combination with external, non-CMB data sets. These are available at the Planck Legacy Archive (https://pla.esac.esa. int/home). Of particular interest to light sterile neutrinos is the 10-parameter $\Lambda \text{CDM} + N_{\text{eff}} + \sum m_{\nu} + w_0 + n_{\text{run}}$ fit, where w_0 is the equation of state parameter of the dark energy ($w_0 = 1$ for a cosmological constant), and n_{run} is the running of the scalar spectral index, a parameter related to the initial conditions of the Universe ($n_{\text{run}} = 0$ in single-field inflation). Using the usual Planck TTTEEE+lowE+lensing+BAO data combination together



Figure 88. Left: 2D marginalized 68% and 95% constraints on the mass splitting Δm_{41}^2 and mixing matrix element $|U_{e4}|^2$ from cosmology (blue), from the tritium β -decay end-point measurements by KATRIN (green), and from neutrinoless double- β -decay experiments (red). The preferred regions of reactor experiments are also indicated. Right: cosmological 68% and 95% marginalized constraints on the $|U_{\mu4}|^2$ (blue) versus constraints from the ν_{μ} disappearance measurements of IceCube and MINOS+ (grey). Reproduced from [451]. CC BY 4.0.

with the Hubble parameter measurement of [452] and Supernova Ia data from the Pantheon sample [453], the constraints on N_{eff} and $\sum m_{\nu}$ in this extended parameter fit are

$$N_{\rm eff} = 3.11^{+0.37}_{-0.36} \quad (95\% \text{ CL}),$$

$$\sum m_{\nu} < 0.16 \text{ eV}. \tag{86}$$

Clearly, while the bound on the neutrino mass sum $\sum m_{\nu}$ has relaxed somewhat and the central N_{eff} has shifted a little up, relative to the more limited bounds in equations (80) and (81) the error bars on N_{eff} have not weaken significantly, and the canonical light sterile neutrino is still in tension with precision cosmological measurements in this expanded parameter space. The upward shift in N_{eff} can be attributed to the discrepancy between the Planck inference of the Hubble expansion rate and the local measurement of [452]. Because of the degeneracy between N_{eff} and H_0 in the CMB primary anisotropies, combining Planck data with local measurements—the latter of which prefer a higher value of H_0 —tends to drag up the inferred N_{eff} as well. The neutrino mass sum $\sum m_{\nu}$, on the other hand, has long been known to be somewhat degenerate with the dark energy equation of state parameter w_0 . However, the combination of BAO and Supernova Ia data can lift this degeneracy very effectively.

Thus, expanding the cosmological parameter space no longer appears to do much for the light sterile neutrino case (in the sense of allowing a larger N_{eff}) the way it once did [454]. In order to get around cosmological constraints, we need to introduce new physics that directly affects the cosmological phenomenology of the light sterile states. Since the main problem of the canonical light sterile neutrino is that its thermalization in the early universe raises N_{eff} to an unacceptably large level for BBN and CMB/LSS, all known new physics solutions so far involve tampering with the thermalization process, in order to maintain N_{eff} at as close to the SM value as possible. A number of ideas have been proposed and explored throughout the years (though not all are guaranteed to work as desired), including

- Large chemical potentials or, equivalently, number density asymmetries for the active neutrinos [166–455, 457],
- 2. Secret interactions of the sterile neutrinos [162–164, 169, 170, 458–463], and
- 3. Low reheating temperature of the Universe [165, 464–467].

Large chemical potentials for the active neutrinos. As discussed earlier in section 5.1.1, in a standard calculation of light sterile neutrino thermalization, the active neutrino asymmetries, defined as $L_{\alpha} = (n_{\nu_{\alpha}} - n_{\bar{\nu}_{\alpha}})/n_{\gamma}$, are assumed to be zero. However, if for some reason some of these asymmetries are large—usually taken to mean $L_{\alpha} > 10^{-5}$ —then the CP asymmetric term $\sqrt{2} G_F(\mathbb{N}_{\nu} - \mathbb{N}_{\bar{\nu}})$ in the Hamiltonian (equation (59)) can act as a large matter effect to suppress oscillations between the active and sterile states. If this suppression is effective before neutrino decoupling, then it is possible to maintain N_{eff} at close to the SM prediction [455]. For active-sterile neutrino mass splittings in the range $\Delta m^2 \simeq 0.2 \rightarrow 10 \text{ eV}^2$, the minimum neutrino asymmetries required to effect some degree of suppression are L > $10^{-4} \rightarrow 5 \times 10^{-3}$ [456]. To significantly suppress thermalization, however, asymmetries as large as $L \sim 10^{-2}$ are required [166].

Unfortunately, aside from the difficulty in explaining how such large neutrino asymmetries could have arisen in the first place, this solution also suffers from other undesirable effects, namely, significant distortion to the ν_e and $\bar{\nu}_e$ energy spectra. While it is possible to suppress active-sterile oscillations with the choice of $L \sim 10^{-2}$ before neutrino decoupling, beyond this critical point vacuum oscillations will inevitably take over and distort the active neutrino energy spectra as a result [456]. [166] has computed this distortion and its effect on the light elemental abundances. They find that the larger the neutrino asymmetry employed to suppress light sterile neutrino thermalization, the larger the spectral distortion and the resulting Helium-4 mass fraction Y_p . Thus, while large neutrino asymmetries do improve the outcome for $N_{\rm eff}$, at the same time they also affect at least one important observable in an undesirable way. The solution is therefore far from fool-proofed.

Self-interaction or NSI for the sterile neutrino. These solutions also work on the principle of suppressing sterile neutrino thermalization through the introduction of a non-standard matter potential for the sterile state in the oscillation Hamiltonian (equation (59)). They differ primarily in their coupling structures.

References [162–164, 458, 461, 462] consider an interaction of the form

$$\mathcal{L}_{\text{int}} = g_X \bar{\nu}_s \gamma^\mu P_L \nu_s X_\mu, \tag{87}$$

where the sterile neutrino self-interaction is mediated by a MeV-mass vector boson X. This leads to the addition of a matter potential to the Hamiltonian (equation (59)) of the form

$$\mp 2\sqrt{2} G_X p \left[\frac{4}{3} \frac{\mathbb{E}_s + \mathbb{E}_{\bar{s}}}{m_X^2} \right],\tag{88}$$

where $G_X \equiv (\sqrt{2}/8)g_X^2/m_X^2$, m_X is the mass of the X boson, and \mathbb{E}_s and $\mathbb{E}_{\bar{s}}$ are defined like \mathbb{E}_{ν} and $\mathbb{E}_{\bar{\nu}}$ in equation (64), but with a projection matrix $S_a = \text{diag}(0, 0, 0, 1)$ that singles out the sterile state for coupling. The collision integrals $\mathcal{I}[\rho(p, t)]$ also need to be modified appropriately. For $g_X \gtrsim 10^{-2}$ and $g_X \lesssim 10$ MeV, the new interaction (equation (87)) is able to suppress sterile neutrino thermalization and hence preserve N_{eff} at close to the SM value without altering BBN predictions. However, at times after neutrino decoupling, the same interaction also leads to equipartition amongst the active and sterile states. That is, if at neutrino decoupling the neutrino number densities are $(n_e, n_\mu, n_\tau, n_s) = (1, 1, 1, 0)$, the secret interaction will redistribute it to (3/4, 3/4, 3/4, 3/4). Thus, the mass of sterile state m_s will nonetheless contribute to the hot dark matter energy density probed by the CMB anisotropies On the other hand, [459, 460] consider a self-interaction of the mass eigenstate ν_4 mediated by a massless pseudoscalar ϕ :

$$\mathcal{L}_{\text{int}} = g_{\phi} \phi \bar{\nu}_4 \gamma_5 \nu_4. \tag{89}$$

Major Report

As with the massive vector boson case above, the secret interaction engenders a matter potential, which in turn suppresses the production of sterile states. The transition from no to full thermalization happens in the range of coupling values $10^{-6} < g_{\phi} < 10^{-5}$ [459]. The authors further argue that because the secret interaction happens exclusively for the mass eigenstate ν_4 , the interaction cannot equilibrate the active and sterile states and whatever is the $\Delta N_{\rm eff}$ produced at neutrino decoupling is also the only component of the neutrino population that carries a mass of $m_s \simeq 1$ eV. Thus, the scenario can easily evade both limits on $N_{\rm eff}$ and m_s .

Lastly, the solution of [169, 170] invokes a coupling of the sterile state to an ultra-light real scalar field of mass $m_{\phi} < 5 \times 10^{-17}$ eV that also contributes to the cold dark matter. At early times, the coupling induces an effective mass for the sterile state, which suppresses active-sterile oscillations and hence thermalization of the sterile state in much the same way as the two scenarios discussed above. After neutrino decoupling, the ϕ field starts to oscillate coherently. Since unlike the MeV vector boson case the model does not lead to the equilibration of the active and sterile states, both limits on N_{eff} and m_s can be easily avoided.

Low reheating temperature. Low reheating temperature scenarios [165, 464–467] refer to those cases in which the Universe transitions to radiation domination at temperatures below $T \sim 10$ MeV. This might happen because of a very low inflation energy scale, or because some non-standard physics causes the Universe to enter a period of matter domination immediately prior to the most recent phase of radiation domination, and the transition back to radiation domination takes place at $T \sim O(1)$ MeV.

A low reheating temperature appears to be a viable way to evade cosmological constraints on light sterile neutrino states. If reheating occurs at $T \sim O(1)$ MeV, even the SM active neutrinos have barely enough time to interact before neutrino decoupling happens. Depending on how exactly reheating happens, some of the active neutrino species may not even reach equilibrium number or energy densities. If light sterile neutrino thermalization was to happen at the same time, the shortage of active neutrinos in the plasma would also slow down the production rate. For these reasons, it is possible to engineer a scenario in which the final N_{eff} is close to the SM value, while the ratio of sterile to active states remains smaller than 1 to 3. Naively, this makes it possible to satisfy N_{eff} as well as m_s bounds from CMB and/or BBN.

In practice, however, whether or not the solution works depends on the details of the reheating model. Reference [467], for example, finds that if the parent particle responsible for reheating decays exclusively into electromagnetically interacting radiation, then a low temperature reheating can indeed render light sterile neutrinos consistent with measurements of the primordial elemental abundances. If however the parent particle decays mainly into hadrons, then together with the presence of active-sterile neutrino mixing, the primordial synthesis of light elements can proceed in a way incompatible with observations for a wide range of the mass and the hadronic branching ratio of the parent particle.

In addition to low reheating scenarios, cosmological scenarios where entropy is conserved and the expansion rate is modified can significantly affect light sterile neutrino



Figure 89. The experimental setups of the JSNS² and JSNS²-II and their sensitivities. Reproduced with permission from [663]. CC BY-NC-ND 4.0.

constraints. This includes, for example, scalar-tensor theories, and is discussed in detail in [465, 466], and also in [468] (for resonance sterile neutrino production).

6. Future experimental prospects

6.1. Decay-at-rest accelerator experiments

6.1.1. JSNS² and JSNS²-II. The JSNS² (J-PARC Sterile Neutrino Search at the J-PARC Spallation Neutron Source) [469–471] and its second phase JSNS²-II [472], aim to search for neutrino oscillations with Δm^2 near 1 eV² at the J-PARC Materials and Life Science Experimental Facility (MLF). Figure 89 shows the experimental setup and search sensitivities. An intense neutrino beam from muon decay at rest is produced by a spallation neutron target with the 1 MW beam of 3 GeV protons created by a Rapid Cycling Synchrotron (RCS). Neutrinos come predominantly from μ^+ decay : $\mu^+ \rightarrow e^+ + \bar{\nu}_{\mu} + \nu_e$. An oscillation of $\bar{\nu}_{\mu}$ to $\bar{\nu}_e$ through the fourth mass eigen-state is searched for by detecting the Inverse-Beta-Decay (IBD) interaction $\bar{\nu}_e + p \rightarrow e^+ + n$, followed by gammas from neutron capture on Gd. The JSNS² detector, as the near detector in the JSNS²-II setup, contains 17 tonnes of Gd-loaded liquid scintillator and is located 24 meters away from the Mercury target. The new far detector of JSNS²-II, currently under construction, is located outside the MLF building with a baseline of 48 meters. The far detector contains 32 tonnes of Gd loaded liquid



Figure 90. The JSNS² detector during the installation (left) and the construction status of the new far detector of the JSNS²-II (right).

scintillator as a neutrino target. Both JSNS² and JSNS²-II employ a Hydrogen target for the neutrino source (μ decay-at-rest), and a neutrino detection channel (IBD) identical to LSND. With improvements made by the short pulsed beam and the neutron capture signal, JSNS² and JSNS²-II will provide clean and direct tests of the LSND anomaly. JSNS² started data taking in 2020, and accumulated 1.45×10^{22} Proton-On-Target (POT) by 2021, 13% of the approved POT by J-PARC, which corresponds to 1 MW beam power for 3 years. An extensive analysis is ongoing (e.g. [473]). The search sensitivity of JSNS² with the full design POT is shown in the bottom-middle plot of figure 89. The construction of the far detector of JSNS²-II started in September 2021 and the aim is to start data taking in 2023. It will provide additional sensitivity, especially in the low Δm^2 region with 1 MW beam power for 5 years.

Figure 90 shows a picture of the JSNS² detector during the installation in June 2020 and the construction status of the new far detector of the JSNS²-II.

6.1.2. COHERENT at the SNS. The COHERENT collaboration can perform a powerful test of oscillations of sterile neutrinos by considering NC disappearance. Parameter space favored by a global fit of oscillation data to a 3 + 1 scenario is accessible to COHERENT in the near future with later data giving a much stronger constraint.

COHERENT measures coherent, elastic neutrino-nucleus scattering (CEvNS) and other low-energy neutrino scattering processes at the Spallation Neutron Source (SNS) at Oak Ridge National Lab (ORNL). CEvNS is a neutral current process whose only signature is a low-energy nuclear recoil which was first measured by COHERENT on CsI in 2017 [474]. The cross section is very large compared to other neutrino scattering cross sections below 50 MeV and is precisely predicted. The SNS is an intense source of π decay-at-rest neutrinos with energies 0-53 MeV, ideal for measuring CEvNS. The width of the SNS beam, 360 ns FWHM, is small compared to the muon lifetime so that the neutrino flux separates in time to a prompt ν_{μ} flux from $\pi^+ \to \mu^+ \nu_{\mu}$ and a delayed flux of $\bar{\nu}_{\mu}/\nu_e$ from $\mu^+ \to e^+ \bar{\nu}_{\mu} \nu_e$. The ν_{μ} flux is monoenergetic with $E_{\nu} = 29.8$ MeV. COHERENT builds and commissions several CEvNS detectors for operation at the SNS at baselines of 19.3 to 28 m. These baselines place COHERENT detectors at the first oscillation maximum for the ν_{μ} flux assuming the global best fit of Δm_{41}^2 . Since the neutrino flux at the SNS includes both $\nu_{\mu}/\bar{\nu}_{\mu}$ and ν_{e} , CEvNS searches can simultaneously search for $\nu_{\mu} \rightarrow \nu_{s}$ and $\nu_{e} \rightarrow \nu_{s}$ disappearance with favorable sensitivity to both θ_{14} and θ_{24} with the same experiment. Additionally, the largest systematic uncertainty, the neutrino flux normalization, is correlated between all detectors which mitigates its effect on a joint fit.

There are plans for three future detectors suitable for searching for sterile neutrinos through CEvNS disappearance in the near future. The first is an upgrade of the CENNS10



Figure 91. Sensitivity of COHERENT CEvNS detectors to constrain sterile neutrino parameter space assuming a 3 + 1 model compared to the LSND and MiniBooNE allowed regions. A global fit to all short-baseline oscillation data is also shown. Reproduced with permission from [664].

detector which made the first CEvNS measurement on argon [475]. This will be a liquid argon calorimeter with 610 kg of fiducial mass with a baseline of 28 m. Detector performance is well understood from experience with CENNS10 operations and data. A 10 kg CsI scintillation detector at a 19.3 m baseline is also planned. This detector will be undoped and cooled to 77 K which can dramatically increase light yield while reducing background scintillation within the crystal [476]. This ensures a low threshold, allowing tests of CEvNS disappearance with low-energy recoils. Finally, a 50 kg germanium PPC detector at 22 m is planned as an upgrade to the 17 kg array currently being commissioned at the SNS. The sensitivity of a joint fit using three years of data from all three detectors to search for a sterile neutrino through NC disappearance with CEvNS is shown in figure 91. This would test the parameter space preferred by a global fit at 90% confidence [101].

Into the next decade, ORNL is investing in the SNS, doubling its power and constructing a new second target station (STS) with one in every four beam spills delivered to the STS to supplement work at the first target station (FTS). Though the upgrade will not be completed until the 2030s, it will facilitate a strong search for sterile neutrinos. The two targets would only be 140 m apart, meaning a large flux of neutrinos from both sources would pass through each CEvNS detector at the SNS. Similar to two-detector oscillation experiments, this mitigates systematic uncertainties from neutrino interaction modeling and detector response by observing neutrinos from a near and far flux source. The dominant remaining uncertainty comes from π^+ production in each beam target which is small. With this control of uncertainties, a test of NC disappearance is possible at the 1%-level. Sensitivity to sterile neutrino oscillations in a 3 + 1 framework for a 10 t fiducial argon calorimeter running for five years, when placed 20 m from the STS and 120 m from the FTS, is shown in figure 91. Exploiting flux from both targets, this large detector could test $\sin^2 2\theta_{\mu e}$ values of 10^{-5} at the global best fit Δm_{41}^2 and could test the LSND and MiniBooNE preferred regions for $\Delta m_{41}^2 > 0.04 \text{ eV}^2$ at high confidence.



Figure 92. The interior of the CCM200 detector. Of the 200 PMTs, 80% are coated with wavelength shifter (TBP) to shift 128 nm scintillation light to the visible, leaving 20% uncoated (darker, more reflective PMTs in image), aiding discrimination of Cherenkov light. TPB foils cover the walls. The light-tight interior is surrounded by a veto region instrumented with PMTs. CCM is now running.

6.1.3. Coherent CAPTAIN-Mills. The physics program of the Coherent Captain Mills (CCM) Experiment comprises searches for new particles in the weak sector, including Dark Photons, Axion-like Particles (ALPs), and heavy neutral leptons in the keV to MeV mass range, extending the coverage of open parameter space for these searches at the order of magnitude level. Many of these particles are invoked as alternative or additional explanations to oscillations involving sterile neutrinos as the source of MiniBooNE anomaly. Thus, the results of CCM from the ongoing run at Los Alamos National Laboratory (LANL) have direct bearing on phenomenology presented in this white-paper. Here, we describe the CCM detector, present a relevant CCM search for production of new bosons by charged meson decays [215] as an example of the impact of the results, and summarize other searches that can be performed.

The CCM experiment is relatively new to the scene of experiments to understand the phenomenology of short-baseline anomalies. The experiment was conceived in 2017 and prototyped using 'CCM120,' which tested 120 PMTs for the SBND liquid argon (LAr) experiment. First physics results from CCM120 were recently published [477, 478]. In 2019, the LANL LDRD office and DOE Dark Matter New Initiative program recognized the relevance of the CCM rare-particle searches to dark matter studies and provided funding for an upgrade to 200 PMTs for 'CCM200.' This 5t fiducial-volume (10t total) LAr detector with 50% PMT coverage, seen in figure 92, was completed in Autumn 2021. The detector is unusual for accelerator-based liquid argon experiments, in that it utilizes only light collection -no time projection chamber. The detector is being commissioned now, and data totaling 2.25×10^{22} POT will be collected in three runs between 2022-24, at the Lujan spallation neutron center. This facility targets 100 microamps on tungsten of 800 MeV protons with 275 ns spills at 20 Hz. This is a prolific source of neutrinos from stopped pion and muon decay, and, potentially, a source for production of new particles, such as ALPs [479], that can be observed in CCM200 through interactions or decays. CCM200 is located 90° off-axis and 20 m from the target.

The CCM200 design has a combination of features related to light collection that makes it powerful and unique. The first is its PMT coverage (8" Hammamatsu R5912-MOD), which is orders of magnitude higher per unit volume than any LAr TPC experiment. Furthermore, the large charge dynamic range of the PMT's and electronics enables the detector to have reconstructed energy sensitivity from 10 keV to over 200 MeV. The second is the rate of PMT readout, which provides information at 500 MHz and is synced to the accelerator



Figure 93. Left: example diagrams for three-body meson decay producing scalar or pseudoscalar(axion-like) particles that interact in the detector. Right: allowed regions for scalers (green) and pseudoscalers (blue) that account for the MiniBooNE low energy excess, presented as a function of coupling versus new particle mass [215]. Lines: predictions of 10 000, 1000, and 100 signal events in solid, dashed and dotted, respectively. Reproduced from [215]. CC BY 4.0.

providing 2-ns absolute timing relative to the 275 ns beam pulse. This is key for separating out early speed of light particles from the prolific beam related neutrons. Third, as can be seen in the figure 92, 80% of the PMTs are coated with 1,1,4,4-Tetraphenyl-1,3-butadiene (TPB), while 20% are uncoated. The TPB allows observation of the scintillation light from LAr, which emits 40 000 photons MeV^{-1} in zero electric field–×4 brighter than typical oil-based liquid scintillator—at 128 nm wavelength, by shifting to the visible to penetrate the PMT glass. The uncoated tubes, which is unique to the CCM design, allows clean observation of Cherenkov light. An R&D goal of CCM is to make the first use of observed Cherenkov light on an event-by-event basis in an analysis to reject backgrounds, since, for a signal, the direction of the Cherenkov ring is opposite the Lujan target.

These design features make CCM particularly ideal for searches for electromagnetic signatures of new physics produced in the target, which is a signature of popular explanations for the short-baseline anomalies. The most popular new-physics explanation has been oscillations involving sterile neutrinos. CCM can explore the recent large-mixing angle result from the BEST experiment [136] using ν_e disappearance for the pion decay-at-rest neutrino beam, since the threshold for ν_e -argon scatters is 1.5 MeV. In the longer term, an upgraded CCM complex can be modified to perform a two-detector search for $\nu_{\mu} \rightarrow \nu_{e}$ in the LSND range that may be motivated by JSNS² results [480]. However, recently, community interest has turned to new particles to explain the observed anomalies. Motivated by this, for this short review, we are featuring an example of CCM's new-particle-discovery capability.

As an example of an interesting new model that CCM can address, consider the proposed explanation of the MiniBooNE Low Energy Excess (LEE) [33] from three-body meson decay [215]. The diagrams for production of a new scalar or pseudoscalar particles that will interact in the detector to produce a single photon exchanging a light vector boson (Z') with the nucleus are shown in figure 93, left. The allowed region for the LEE is shown in figure 93, right, and this model (model 1) finds a good fit to both the angular and energy dependence of the LEE [33]. Another version of this model (model 2) which also can fit the data involves the emission a light vector boson Z' from the pion decay (just like the scalar/pseudoscalar). V will then produce a photon by exchanging a scalar with the nucleus at the detector. This model explains the LEE. All these new mediators emerging from the charged pion decays, so far we have discussed, can be coupled to only quarks.

CCM will be able to probe both models. The main production channel will be π^0 decay into gamma and Z'. In model 1, Z' can be assumed to dominantly decay into a pair of scalars, which will subsequently produce a photon from the scattering at the detector as needed to resolve LEE. Z' also can decay into a photon and the scalar. Here, our assumption is that Z' does not decay into a pair of visible particles promptly. Under this assumption, figure 93 shows the predicted number of signal events at CCM in the allowed parameter space to explain the LEE. Therefore, CCM200 has the capability for discovery, if this new physics is the source of the LEE. Model 2 probes the MiniBooNE anomaly more directly since the same Z' that emerges from the charged pion decay to address the anomaly also can be produced from the π^0 decay; therefore no assumption is needed to correlate the LEE and a possible signal at the CCM. This possibility is under investigation at present.

Outside of models explaining the anomalies, CCM engages in a broad range of newphysics searches. Limits on leptophobic dark photons from CCM120 [477, 478], the prototype run, will be extended by two orders of magnitude in CCM200. Searches for the QCD axion can close the last remaining open-window at masses >0.1 MeV [479]. Searches for "neutrissimos"–not so heavy neutral heavy leptons–that have focused on >100 MeV masses to address the LEE are being extended to lower masses in CCM. Although these searches are not directly tied to explanations of the anomalies, a discovery would inevitably demand investigation on whether the observed new-physics is related.

In summary, CCM is a small, fast-timescale experiment that is already taking data at LANL. Its results have the potential to change our thinking about the anomalies.

6.1.4. PIP2-BD: GeV proton beam dump at fermilab's PIP-II linac. The completion of the PIP-II superconducting LINAC at Fermilab as a proton driver for DUNE/LBNF in the late 2020s creates an attractive opportunity to build a GeV proton beam dump facility at Fermilab dedicated to and designed from the ground up for HEP with excellent sensitivity to eV-scale sterile neutrinos via neutral current disappearance using the CEvNS reaction (see [481]). Thus, relative to spallation neutron facilities tailored to neutron physics and optimized for neutron production operating at a similar proton beam power, a HEP-dedicated beam dump facility would be designed to suppress rather than maximize neutron production and implement a beam dump made from a lighter target such as carbon, which can have a pion-to-proton production ratio up to \sim 2 times larger than heavier Hg or W targets. The facility could also accommodate multiple, 100-ton-scale high energy physics experiments located at different distances from the beam dump and at different angles with respect to the incoming proton beam. This flexibility further improves the sensitivity of dark sector and sterile neutrino searches, by allowing relative measurements at different distances and angles to constrain uncertainties in expected signal and background rates.

The continuous wave capable PIP-II LINAC at Fermilab can simultaneously provide sufficient protons to drive megawatt-class $\mathcal{O}(\text{GeV})$ proton beams as well as the multi-megawatt LBNF/DUNE beamline. By coupling the PIP-II LINAC to a new Booster-sized, permanent magnet or DC-powered accumulator ring, the protons can be compressed into pulses suitable for a proton beam dump facility with a rich physics program. The accumulator ring could be located in a new or existing beam enclosure and be designed to operate at 800 MeV but with an upgrade path allowing for future operation in the GeV range. The accumulator ring would initially provide 100 kW of beam power, limited by stripping foil heating, and have a $\mathcal{O}(10^{-4})$ duty factor. One variant of this accumulator ring would be a ~100 m circumference ring operating at 1.2 GeV with a pulse width of 20 ns and a duty factor of $\mathcal{O}(10^{-6})$, which would greatly reduce steady-state backgrounds. Another is an



Figure 94. PIP2-BD 90% confidence limits on active-to-sterile neutrino mixing compared to existing ν_{μ} disappearance limits from IceCube [151] and a recent global fit [103], assuming a 5 year run (left). Also shown are the 90% confidence limits for ν_{μ} disappearance (left), ν_e disappearance (middle), and ν_e appearance (right), assuming the $\bar{\nu}_{\mu}$ and ν_e can be detected with similar assumptions as for the ν_{μ} . Reproduced with permission from [481].

accumulator ring coupled to a new rapid cycling synchrotron replacing the Fermilab Booster with an increased proton energy of 2 GeV and an increased beam power of 1.3 MW [482].

Decay-at-rest neutrinos from a stopped pion beam dump provide an excellent source of ν_{μ} , $\bar{\nu}_{\mu}$, and ν_{e} with a time structure that can separate ν_{μ} from $\bar{\nu}_{\mu}$ and ν_{e} . CEvNS provides a unique tool to definitively establish the existence of sterile neutrinos through active-to-sterile neutrino oscillations [483]. Using CEvNS, we can explore both mono-energetic u_{μ} disappearance with $E_{\nu} = 30$ MeV and the summed disappearance of ν_{μ} , $\bar{\nu}_{\mu}$, and ν_{e} to ν_{s} , which can also put constraints on $\nu_{\mu} \rightarrow \nu_{e}$ oscillation parameters in a 3 + 1 sterile neutrino model. We consider here a setup consisting of identical 100-ton LAr scintillation detectors, located 15 m and 30 m away from a carbon proton beam dump with a 20 keV recoil energy threshold and an efficiency of 70%. The 100-ton scale scintillation-only detector assumes a cylindrical volume with a 5 m height and 2.5 m radius, and we perform a full Geant4-based [484] scintillation photon simulation with wavelength shifting and propagation to PMTs along the endcaps and side walls of the detector. Based on simulation studies, we assume the detectors have a position resolution given by $\sigma = \frac{40 \text{ cm}}{\sqrt{T/20 \text{ keV}}}$ in each spatial dimension, where T is the nuclear recoil produced by the neutrino interaction within the detector. This information allows the possibility for a 'rate+shape' fit using five bins in the reconstructed neutrino propagation distance with a bin width of 1 m matched to the expected resolution of the reconstructed neutrino propagation distance. If the prompt ν_{μ} can be separated from the delayed $\bar{\nu}_{\mu}$ and ν_{e} , one can exploit the mono-energetic feature of the ν_{μ} flux and perform a joint rate + shape disappearance fit of CEvNS events in the near and far detectors as a function of reconstructed position.

In calculating the sensitivity, we assume the neutron background in this dedicated facility could be suppressed to a negligible level for this experiment and that the signal-to-noise ratio for the remaining steady-state backgrounds is 1:1. In figure 94, we compute the 90% confidence limits on the $\nu_{\mu} \rightarrow \nu_{S}$ mixing parameter $\sin^{2} 2\theta_{\mu S}$ for a 5 year run of an upgraded 1.2 GeV proton accumulator ring operating with a pulse width of 20 ns, a duty factor of $\mathcal{O}(10^{-6})$, and a 75% uptime, assuming a 9% normalization systematic uncertainty correlated between the two detectors and Also shown are the 90% confidence limitsa 36 cm path length smearing. Also shown are the 90% confidence limits for ν_{μ} disappearance, ν_{e} disappearance, and ν_e appearance, assuming the $\bar{\nu}_{\mu}$ and ν_e can be detected with similar assumptions as for the ν_{μ} .

6.1.5. KPIPE at fermilab. The KPIPE experimental concept, outlined in [485], calls for a very long (120 m) and thin (1.5 m radius) cylindrical detector close to and oriented radially outward from an intense beam-dump source of monoenergetic 236 Mev ν_{μ} from charged-kaon decay-at-rest ($K^+ \rightarrow \mu^+ \nu_{\mu}$, with branching ratio of 64%) to achieve sensitivity to short-baseline muon-neutrino disappearance. The idea is to search for an L/E-dependent oscillation wave using fixed-*E* neutrinos with minimal background and only modest detector requirements.

The KPIPE detector, relying on liquid scintillator and silicon photomultipliers (or PMTs), is designed to look for 236 MeV $\nu_{\mu}n \rightarrow \mu^{-}p$ interactions, which provide a unique double-flash coincidence due to the muon decay following the initial prompt event. Mapping these interactions as a function of distance along the detector pipe, with a nominal, no-oscillation expectation of a $1/r^2$ rate dependence, provides sensitivity to muon-flavor disappearance. Given a beam dump, decay-at-rest neutrino source, the beam-based ν_{μ} background (from decay-in-flight mesons) to these signal events is expected to be completely sub-dominant, at the 1-2% level. While cosmics can be considered a concern for such a surface or near-surface detector, this background can be mitigated by typical accelerator duty factors of $\sim 10^{-6}-10^{-5}$ combined with the short charged kaon lifetime (13 ns). The monoenergetic neutrino source, combined with low decay-in-flight background and small beam duty factor, means that the signal-to-background ratio is expected to be well over 50:1 in the scenarios considered. This large ratio means that the detector requirements, in particular the photocoverage, can be quite modest. In fact, a preliminary estimate at [486] predicts that the entire KPIPE detector would cost \$5M.

The KPIPE detector was originally envisioned to be paired with the 3 GeV, 730 kW (currently, with 1 MW planned) J-PARC Spallation Neutron Source. Aside from the primary proton energy, which is above the kaon production threshold, and the high power, this source is particularly attractive because the beam timing structure, two ~80 ns pulses separated by 540 ns at 25 Hz, provides an extremely low duty factor (4×10^{-6}), essential for cosmic background rejection. The drawback of this source, however, is that the 3 GeV primary proton energy, while above threshold, is somewhat lower than optimal for charged kaon production per unit power: at 3 GeV, the MARS15 software package [487] predicts 0.007 KDAR ν_{μ} /POT. With an increase in proton energy to 8 GeV, for example, the production rate increases by a factor of 10–0.07 KDAR ν_{μ} /POT. Spatial and facility issues, especially in consideration of the existing materials-science-focused beamlines and experiments, also means that optimal detector placement, with KPIPE calling for a 120 m long detector with closest distance of 32 m from the neutrino source, is challenging.

The future Fermilab particle accelerator complex [482], including PIP-II [488] and the RCS upgrade [489], can provide an optimal beam-dump/stopped-kaon neutrino source for KPIPE, in terms of beam energy (8 GeV), beam timing ($\sim 10^{-5}$ duty factor), and spatial considerations (see [490]). Using the detector and Fermilab-accelerator assumptions shown in table 15, and scaling based on the detailed study in [485], we expect KPIPE could achieve the sensitivity to short-baseline ν_{μ} disappearance shown in figure 95. As can be seen, this sensitivity surpasses, and is highly complementary to, SBN (6years) at $\Delta m^2 > 10 \text{ eV}^2$ for both scenarios considered and $\Delta m^2 > 1 \text{ eV}^2$ for the RCS upgrade era case.

6.1.6. IsoDAR. Through tracing $\bar{\nu}_e$ disappearance continuously across L/E from 1 to $\sim 10 \text{ m MeV}^{-1}$, the IsoDAR (Isotope Decay At Rest) experiment uniquely addresses the fundamental question raised by this white paper: What, if any, new physics phenomenology

Experimental assumptions	
Detector length	120 m
Active detector radius	1.45 m
Closest distance to source	32 m
Liquid scintillator density	0.863 g cm^{-3}
Active detector mass	684 tons
Primary proton energy	8 GeV
Target material	Hg or W
KDAR ν_{μ} yield (MARS15)	0.07 ν_{μ} /POT
ν_{μ} CC σ @ 236 MeV (NuWro)	$1.3 \times 10^{-39} \text{ cm}^2/\text{neutron}$
KDAR signal efficiency	77%
Vertex resolution	80 cm
Light yield	4500 photons MeV^{-1}
Uptime (5 years)	5000 h/year
ν_{μ} creation point uncertainty	25 cm
PIP-II era assumptions	
Proton rate (0.08 MW)	$1.0 \times 10^{21} \text{ POT/year}$
Beam duty factor	1.6×10^{-5}
Cosmic ray background rate	110 Hz
Raw KDAR CC event rate	2.7×10^4 events/year
RCS upgrade era assumptions	
Proton rate (1.2 MW)	$1.5 \times 10^{22} \text{ POT/year}$
Beam duty factor	5.3×10^{-5}
Cosmic ray background rate	360 Hz
Raw KDAR CC event rate	4.0×10^5 events/year

Table 15. Summary of the relevant KPIPE experimental parameter assumptions. Reproduced with permission from [490].

underlies the short-baseline anomalies? Despite the enormous consequences if new physics is the cause, the question has been unanswered for more than 20 years. Incremental improvements on our present approaches are likely to yield more of the same confusing results. IsoDAR represents an entirely new approach-the experiment makes use of a flux from ⁸Li β decay, produced through a 60 MeV proton beam that is targeted on ⁹Be to yield neutrons that enter a surrounding isotopically-pure ⁷Li sleeve and capture. When this source is paired with the 2.3 kton Yemilab liquid scintillator detector (LSC), approximately 1.6 million IBD events can be reconstructed in 5 years of running. The high statistics, relatively high energy, E, of $\bar{\nu}_e$ from ⁸Li decay, and the ideal matching baseline, L, due to the size of the LSC, gives unprecedented capability to study the L/E dependence of short-baseline disappearance in an agnostic manner, determining its cause without design assumptions that bias toward specific underlying physics models. Figure 96 illustrates the power of IsoDAR to resolve various popular proposals for the source of the effect, with a 3 + 1 model at top left; 3 + 1with nonzero quantum mechanical wave packet effects at top right; introduction of additional sterile neutrinos, in this case 3 + 2 at bottom left; and introduction of new interactions, in this case 3 + 1 + decay, at bottom right. These examples show that IsoDAR can clearly elucidate the underlying oscillation-related phenomenology of the electron-flavor short-baseline anomalies, even in the case of physics that produces very complex waveforms.

IsoDAR has received preliminary approval to run at Yemilab in the configuration shown in figure 97. [491] provides an overview of the technology and installation-plan. In figure 97,



Figure 95. The 90% CL sensitivities of the KPIPE at Fermilab scenarios considered here, in both the PIP-II and RCS upgrade eras. For reference, we also show the expected 90% C.L. SBN sensitivity (6 years) [500], existing 90% C.L. MiniBooNE +SciBooNE limit [274], and 99% allowed region from the Collin *et al* global fit [100]. Reproduced with permission from [490].

the cyclotron that drives the flux production is located at the far right. This novel 5mA H_2^+ ion accelerator, producing 10 mA of 60 MeV protons, yields an order of magnitude higher proton beam current than on-market cyclotrons at similar energies. Since the 2013 Snowmass Study, cyclotron development has culminated in the design described in [492], which presents start-to-end simulations and prototypes of components now under test [493, 494]. As seen in figure 97, the proton beam is transported from right to left and then bent through 180° to the target surrounded by the sleeve, hence fast neutrons are directed away from the LSC detector, shown in green. Substantial R&D and engineering have established successful target, sleeve and shielding designs [495, 496].

IsoDAR@Yemilab is designed to address issues that have arisen during studies of reactor and MegaCurie source experiments. The $\bar{\nu}_e$ flux is generated from a single, well understood isotope, avoiding issues faced by reactor flux modeling. The $\bar{\nu}_e$ energy range is from about 3–13 MeV, with peak at ~6 MeV, well beyond environmental backgrounds and backgrounds from neutron capture. The source creation region is compact (~41 cm at 1 σ) and isotropic. The experiment has the capability of event-by-event reconstruction, with prompt (e⁺) energy resolution of 2.3% and vertex resolution of 4 cm at 8 MeV [497], in contrast to MegaCurie (MCi) source experiments that do not reconstruct event kinematics. Use of the well-known IBD cross section is also an advantage over the gallium experiments. Lastly, and importantly, the size of the LSC detector, when combined with the energy range of the source, leads to the wide L/E range, allowing precision reconstruction of the oscillation wave across many cycles.

A simple 3 + 1 model is traditionally used for cross-comparison of experimental reach. The reach of IsoDAR in $\sin^2 2\theta_{ee}$ at 95% CL is presented in table 16, column 2 for a range of Δm^2 . For comparison, column 3 shows the combined limits at 95% CL from Prospect, RENO and Daya Bay [111]. The IsoDAR mixing angle reach is $\times 4$ ($\times 35$) that of the reactor limits at 1 (8) eV². For allowed region comparisons for Neutrino-4 [498] and the Gallium experiments J. Phys. G: Nucl. Part. Phys. 51 (2024) 120501



Figure 96. The IsoDAR@Yemilab capability to trace $\bar{\nu}_e$ disappearance versus L/E for IBD interactions. Top left and right present a 3 + 1 example without and with wavepacket effects described in [111]. Bottom left and right are 3 + 2 and 3 + 1+decay models for the global best fit points in [103]. Orange is the true underlying model. Points represent the measurement capability. See text for further discussion. Reproduced from [491]. © 2022 IOP Publishing Ltd and Sissa Medialab. All rights reserved.



Figure 97. Layout of IsoDAR@Yemilab in the Yemilab caverns. The excavation of the IsoDAR cavern complex is complete. Reproduced from [491]. © 2022 IOP Publishing Ltd and Sissa Medialab. All rights reserved.

[136], we use the 2σ lower edge in $\sin^2 2\theta_{ee}$. In the Neutrino-4 case, the allowed region is narrow in Δm^2 and does not coincide with 8 eV², so we present the mixing angle reach for the best fit mass splitting of 7.3 eV².
Table 16. Quantitative comparison of the low $\sin^2 2\theta_{ee} 2\sigma$ reach of electron-flavor experiments in the Δm^2 range of interest. IsoDAR sensitivity is based on assumptions in [497]. Combined reactor limits are from PROSPECT, NEOS, and Daya Bay. N/A indicated Δm^2 is not within 95% CL allowed region. The Neutrino-4 2σ reach is quoted at 7.3 eV².

Δm^2	IsoDAR@Yemilab Sensitivity	Combined Reac- tor [111] Limits	Neutrino- 4 [498] Allowed	Gallium [136] Allowed
1 eV^2	0.004	0.016	N/A	0.28
2 eV^2	0.004	0.07	N/A	N/A
4 eV^2	0.005	0.13	N/A	0.27
8 (7.3) eV^2	0.008	0.28	(0.12)	0.28

As a result of the novel design, IsoDAR@Yemilab design is able to elucidate $\bar{\nu}_e$ disappearance across L/E of 1 to ~10 m MeV⁻¹ without guidance from any phenomenological model. Figure 96 illustrates the complex oscillation waves that are able to be differentiated in 5 years of running, with efficiency included. The upper plots illustrate the oscillation wave without (left) and with (right) wavepacket effects as discussed in [111], at a point where the combined reactor limit and gallium allowed region overlap assuming the wavepacket model. Comparison of the two plots shows that the distinctive damping due to wavepacket effects can be observed given IsoDAR's high statistics. The lower plots present 3 + 2 and 3 + 1+decay models evaluated at the best fit points from [103]. The value of the high statistics and excellent reconstruction of IsoDAR is particularly emphasized by the 3 + 2 case, where the second predicted modulation is clear due to the capability of using very fine binning. The orange line indicates the true underlying distribution, while the points with error bars present the expected measurements, illustrating the loss of information from finite statistics and bin sizes. The 3 + 1 (top) and 3 + 2 points (bottom right) also include detector energy and position smearing.

The experiment will also collect ×4 the world's sample $\bar{\nu}_e$ -electron elastic scattering events in 5 years which may help further decipher new physics, depending on the source. In fact, IsoDAR@Yemilab has an extensive discovery-level physics program beyond searching for the short-baseline anomalies, including an order of magnitude improvement in NSI searches through elastic scattering from electrons [497], unique neutrino-based searches for Z' signatures [497], and exotic non-neutrino searches, such as for neutrons shining through walls [499]. As such, IsoDAR represents a leap forward for electron-flavor neutrino experiments.

6.2. Decay-in-flight accelerator experiments

6.2.1. Short-baseline experiments

6.2.1.1. The fermilab SBN program. The Short-Baseline Neutrino (SBN) program consists of three LArTPC detectors located along the BNB at Fermilab: the MicroBooNE detector, which completed operations in 2021; the ICARUS detector, which began operations in the BNB in 2021; and the upcoming short-baseline near detector (SBND), which is expected to begin operations in 2023. This program represents an exciting opportunity for a multibaseline search for light sterile neutrino oscillations in multiple exclusive or inclusive oscillation channels, and a test of the 3 + 1 light sterile neutrino oscillation interpretation of past experimental anomalies at $\geq 5\sigma$ [500]. In particular, ν_{μ} CC measurements across the three detectors will probe $\nu_{\mu} \rightarrow \nu_{\mu}$ oscillations with world-leading sensitivity as shown in figure 98



Figure 98. SBN light sterile neutrino sensitivities in the $\nu_{\mu} \rightarrow \nu_{e}$ appearance channel (left) and $\nu_{\mu} \rightarrow \nu_{\mu}$ disappearance channel (right) according to the SBN proposal [500]. The 3σ (5σ) sensitivities are given by the solid (dotted) red curves. The LSND 90% C. L. (99% C.L.) allowed region is shown as shaded blue (grey) [19]. The global $3\sigma \nu_{e}$ (ν_{mu}) appearance (disappearance) regions from [102] are shown by the shaded red region (black line), and the global best fit regions from [101] are shown in green. Reproduced from [156]. CC BY 4.0.

(right); ν_e CC measurements will probe $\nu_{\mu} \rightarrow \nu_e$ and/or $\nu_e \rightarrow \nu_e$ oscillations with sensitivity as shown in figure 98 (left) and 99. Additionally, NC-based oscillation searches have been proposed, e.g. [501], with unique sensitivity to U_{si} under a 3+N model, as well as potentially $U_{\tau i}$ (for i > 3) when combined with ν_e and ν_{μ} CC-based appearance and disappearance searches.

The power of a multi-baseline and multi-channel search has been shown to be advantageous not only for 3 + 1 searches, but for 3 + N searches more generally. For example, [502] has found that SBN is capable of ruling out 85%, 95% and 55% of the 99%-globallyallowed parameter space region 126 of 3 + 1, 3 + 2, and 3 + 3 light sterile neutrino oscillation parameters at 5σ CL, assuming a null observation, particularly when appearance and disappearance effects are studied simultaneously (including correlations). This is illustrated in figure 100, for the 3 + 3 scenario. Additionally, it has been pointed out that within the context of 3+N oscillations with N > 1, SBN offers an opportunity for measuring potential CP violation in the leptonic sector, particularly if future antineutrino beam running is possible with SBN. In particular, if antineutrino exposure is considered, for maximal values of the (3 + 2) CP violating phase ϕ_{54} , SBN could be the first experiment to directly observe $\sim 2\sigma$ hints of CP violation associated with an extended leptonic sector. This is illustrated in figure 101, for the 3 + 2 scenario. Furthermore, a planned analysis using the ICARUS detector can probe meter-scale oscillations within the detector volume, consistent with sterile mass splittings $\approx 7 \text{ eV}^2$ and $\sin^2 2\theta_{14} \approx 0.4$, providing a test of the allowed region claimed by the Neutrino-4 experiment.

Beyond searches for physics associated with eV-scale sterile neutrinos, SBN's main physics goals include detailed studies of neutrino-argon interactions at the GeV energy scale, enabled by millions of neutrino interactions that will be recorded on argon in its high precision detectors. SBND's anticipated high statistics, in particular, provide a unique opportunity for first-ever measurements of rare SM-predicted neutrino interaction processes at

¹²⁶ Global allowed regions as of 2018.



Figure 99. Due to the large intrinsic ν_e statistics at SBND, SBN is also sensitive to ν_e disappearance, probing $\sin^2 2\theta_{ee}$ at high $\Delta m^2 \ge 0.2 \text{ eV}^2$ (assumes $\theta_{24} = 0$). This provides a complementary probe of oscillations traditionally probed using reactor antineutrinos at a much lower (MeV) energy scale. Reprinted (figure) with permission from [502], Copyright (2017) by the American Physical Society.



Figure 100. SBN's coverage of globally-allowed 3 + 3 light sterile neutrino oscillation parameters, defined as the fraction of 99%-CL-globally-allowed parameter space that can be ruled out by SBN at a given CL indicated by the *x*-axis, assuming a null observation. Coverage of 3 + 1 and 3 + 2 globally-allowed parameter space is provided in [502]. Reprinted (figure) with permission from [502], Copyright (2017) by the American Physical Society.

0.1–1 GeV, including rare photon production processes such as coherent NC single-photon production, NC $\Delta \rightarrow N\gamma$ radiative decay, or production and radiative decay of heavier resonances. Additionally, the high statistics coupled with the unprecedented event



Figure 101. The significance at which SBN can observe CP violation in the (3+2) sterile neutrino scenario, as a function of true CP violating phase ϕ_{54} , for two injected signals corresponding to the global (3+2) best fit point (red lines) as well as the point with largest total allowed mixings (blue lines), for a variety of POT in neutrino and antineutrino running modes at SBN. Reprinted (figure) with permission from [502], Copyright (2017) by the American Physical Society.

reconstruction, excellent particle identification, and fine-sampling calorimetry of the SBN detectors' LArTPC technology opens up invaluable opportunities for new physics searches.

In particular, the capabilities of LAr detectors will allow for greater discrimination between e^- , e^+e^- , γ , and $\gamma\gamma$ final states, as well as to identify final state hadron multiplicities. For models in which new particles are produced in neutrino-nucleus scattering, it will be possible to search for a hadronic vertex associated with the displaced decay position. The EM showers in this case may not point back to the original vertex due to missing energy. Furthermore, for models that explain MiniBooNE with new heavy particles produced in meson decays, the SBN detectors can also leverage the decays-at-rest of kaons produced in the NuMI absorber [503].

Reference [156] provides a broad overview of such new physics, including their signatures in SBN. Here, we limit the discussion to a summary of ones suggested as interpretations to short-baseline anomalies, extending beyond eV-scale sterile neutrinos:

- 1. SBN can probe eV-scale sterile neutrinos decaying to active neutrinos and a Majoron or gauge boson, which would lead to new features in the active neutrino energy spectrum with respect to 3+N scenarios.
- 2. Large extra-dimension models, such as ones proposed as an explanation of the reactor anomaly, would affect both appearance and disappearance channels at SBN.
- 3. Resonant $\nu_{\mu} \rightarrow \nu_{e}$ oscillations that arise in the presence of a light scalar boson that couples only to neutrinos and could induce a MSW effect sourced by the cosmic neutrino background could also be probed with SBN, through the search of $\nu_{\mu} \rightarrow \nu_{e}$ transitions

and lack of ν_{μ} disappearance (as the latter would be suppressed compared to a vanilla 3 +1 scenario).

- 4. Violation of Lorentz and CPT symmetry would lead to modifications in the oscillation probability measurable at SBN, such as direction-dependent effects, neutrinoantineutrino mixing, annual modulations, and energy dependent effects on observable mass splittings and mixing angles.
- 5. Sterile neutrinos and ADRs, also proposed as an explanation of the short-baseline anomalies, would have a similar phenomenology in SBN to that of the usual 3 + 3 sterile scenario (while evading the constraints from long-baseline and atmospheric neutrino experiments).
- 6. Charged current non-standard interactions (CCNSI) in the lepton sector could lead to a number of observable effects, such as (1) deviations of the SM CC quasi-elastic cross section, (2) modification of angular and energy distributions due to the presence of new Lorentz structures, and (3) flavor violation such as $\nu_{\mu}n \rightarrow e^-p$; At SBN, CCNSIs can lead to an apparent baseline-independent $\nu_{\mu} \rightarrow \nu_e$ conversion.
- 7. Dark neutrino sectors connected to the standard neutrino sector, allowing for neutrino upscattering into a heavy state which could then decay to a light neutrino and a gauge boson within a detector, followed by the gauge boson decay to visible particles such as e^+e^- could also be measurable at SBN. As shown in figure 102, an e^+e^- pair can give rise to four distinct topologies in LAr, depending on the lifetime of the parent particle and on the angle between the charged leptons. A variety of signatures could be probed, including pair production of e^+e^- , $\mu^+\mu^-$ or $\pi^+\pi^-$ induced by neutrino interactions, with little to no hadronic activity and with the same signal strength at all three detectors, since there is no L/E dependence.
- 8. Heavy neutrinos and transition magnetic moment proposed as an explanation of the LSND/MiniBooNE anomalies would be observable at SBN as anomalously large single-photon production with small hadronic activity. LArTPC $e \gamma$ discrimination capability places SBN in a special position to probe these scenarios. Again, the signal strength would be the same at all three detectors, since there is no L/E dependence in these models either.

6.2.1.2. nuSTORM. The 2020 Update of the European Strategy for Particle Physics (ESPP) [504] recommended that muon beam R&D should be considered a high-priority future initiative and that a programme of experimentation be developed to determine the neutrino cross-sections required to extract the most physics from the DUNE and Hyper-K long-baseline experiments. The ENUBET [505–507] and nuSTORM [508, 509] collaborations have begun to work within and alongside the CERN Physics Beyond Colliders study group [510] and the international Muon Collider Collaboration [511] to carry out a joint, five-year design study and R&D programme to deliver a concrete proposal for the implementation of an infrastructure in which:

- ENUBET and nuSTORM deliver the neutrino cross-section measurement programme identified in the ESPP and allow sensitive searches for physics beyond the SM to be carried out; and in which
- 2. A 6D muon ionisation cooling experiment is delivered as part of the technology development programme defined by the international Muon Collider Collaboration.

With their existing proton-beam infrastructure, CERN and Fermilab are both uniquely well-placed to implement ENUBET, nuSTORM, and the 6D-cooling experiment as part of



Figure 102. The four main topologies of e^+e^- LEE models at MicroBooNE: coherent and incoherent scattering with well-separated or overlapping e^+e^- pairs.

the required muon collider demonstrator. The design of ENUBET, carried out within the framework of a European Research Council funded design study, includes the precise layout of the kaon/pion focusing beamline, photon veto and timing system as well as the development and test of a positron tagger together with the required electronics and readout. The feasibility of implementing nuSTORM at CERN has been studied by the CERN Physics Beyond Colliders study group while a proposal to site nuSTORM at FNAL was developed for the last Snowmass study in 2013. The FNAL study focused on the optimisation of the muon storage ring to provide exquisite sensitivity in the search for sterile neutrinos. In the Physics Beyond Colliders study, the muon storage ring was optimised to carry out a definitive neutrino-nucleus scattering programme using stored muon beams with momentum in the range 1–6 GeV while maintaining its sensitivity to physics beyond the SM.

The study of nuSTORM is now being taken forward in the context of the demonstrator facility required by the international Muon Collider Collaboration that includes the 6D muon ionization cooling experiment. The muon-beam development activity is being carried out in close partnership with the ENUBET collaboration and the Physics Beyond Colliders Study Group. In consequence we now have the outstanding opportunity to forge an internationally collaborative activity by which to deliver a concrete proposal for the implementation of the nuSTORM infrastructure.

On top of the program outlined above, nuSTORM can still provide unprecedented sensitivity to light sterile neutrinos. In particular, it allows to search for short-baseline oscillations in $\nu_e \rightarrow \nu_{\mu}$ appearance, the CPT-conjugate channel of the appearance hypothesis at LSND. This would be possible due to charge selection of muons in a magnetic field, which can discriminate the $\overline{\nu}_{\mu}$ produced in μ^+ decays from the $\nu_e \rightarrow \nu_{\mu}$ oscillations. A detailed study of nuSTORM's sensitivity to sterile neutrinos was performed in [512], which focused on a 3.8 GeV muon ring design for siting at FNAL. The study considered a 1.3 kt magnetized ironscintillator detector at 2 km from the ring with an exposure of 10^{21} POT, corresponding to $\approx 2 \times 10^{18}$ useful muon decays. The sensitivity curves covered the entire LSND and MiniBooNE regions of preference at more than 5σ . This impressive sensitivity was achieved thanks to the muon signature, which is subject to low levels of backgrounds, and the low systematic uncertainties on the neutrino flux.



Figure 103. LBNF neutrino beam fluxes at the DUNE Far detector for neutrinoenhanced forward horn current (FHC) beam running (left), and antineutrino-enhanced reverse horn current (RHC) beam running (right). Reproduced from [665]. CC BY 4.0.

Thanks to its unique neutrino beam, nuSTORM is also sensitive to other explanations to short-baseline anomalies. It would stand out as a unique test of lepton-flavor-violation in muon decays. Because the beam is derived from μ^+ decays, any exotic branching ratio of the muon, such as $\mu^+ \rightarrow e^+\nu_\alpha \overline{\nu_e}$, would be a striking signature in a near detector with electron-positron discrimination capabilities, such as in a magnetized, low-density detector. This type of near detector would also benefit the sensitivity to models with neutrino upscattering to new dark particles with decays to e^+e^- .

6.2.2. Long-baseline experiments

6.2.2.1. DUNE. The DUNE experiment is a next-generation, long-baseline neutrino oscillation experiment, designed to be sensitive to ν_{μ} to ν_{e} oscillations. The experiment consists of a high-power, broadband neutrino beam, a powerful precision multi-instrument Near detector complex located at Fermi National Accelerator Laboratory in Batavia, Illinois, 574 m away from the neutrino production target, and a massive Liquid Argon Time Projection Chamber Far detector located at the 4850 ft level of the Sanford Underground Research Facility (SURF), 1300 km away in Lead, South Dakota, USA. The anticipated total fiducial mass of the Far detector is 40 kton. The long baseline of 1300 km provides sensitivity, in a single experiment, to all parameters governing neutrino oscillations. Due to the high-power proton beam facility, the Near detector consisting of precision detectors capable of off-axis data taking for improved constraining of systematics, and the massive Far detector, DUNE provides enormous opportunities to probe BSM phenomena in both new particle production and interactions, and in neutrino propagation effects.

DUNE expects to begin data taking operations in 2029 with half of the full Far detector, and start beam data taking operations in 2031 with the 1.2 MW long-baseline neutrino facility (LBNF) beam, upgradable to 2.4 MW. The LBNF neutrino beam flux sampled on-axis by the Far detector is shown in figure 103. The wide-band range of energies provided by the LBNF beam afford DUNE significant sensitivity to probe sterile mixing, which would typically cause distortions of standard oscillations in energy regions away from the three-flavor $\nu_{\mu} \rightarrow$



Figure 104. Regions of L/E probed by the DUNE Near and Far detectors compared to 3-flavor and 3 + 1-flavor neutrino disappearance and appearance probabilities. The gray-shaded areas show the range of true neutrino energies probed by the ND and FD. The top axis shows true neutrino energy, increasing from right to left. The top plot shows the probabilities assuming mixing with one sterile neutrino with $\Delta m_{41}^2 = 0.05$ eV², corresponding to the slow oscillations regime. The middle plot assumes mixing with one sterile neutrino with $\Delta m_{41}^2 = 0.5$ eV², corresponding to the bottom plot includes mixing with one sterile neutrino with $\Delta m_{41}^2 = 50$ eV², corresponding to the rapid oscillations regime. As an example, the slow sterile oscillations cause visible distortions in the three-flavor ν_{μ} survival probability (blue curve) for neutrino energies. Reproduced from [514]. CC BY 4.0.

 ν_{μ} disappearance maximum. Therefore, DUNE sterile mixing probes reach a broad range of potential sterile neutrino mass splittings by looking for disappearance of CC and NC interactions over the long distance separating the ND and FD, as well as over the short baseline of the ND. The DUNE sterile neutrino mixing studies shown below assume a minimal 3 + 1 oscillation scenario with three active neutrinos and one sterile neutrino, with a new independent neutrino mass-squared difference, Δm_{41}^2 , and for which the mixing matrix is extended with three new mixing angles, θ_{14} , θ_{24} , θ_{34} , and two additional phases δ_{14} and δ_{24} .

Figure 104 shows how the standard three-flavor oscillation probability is distorted at neutrino energies above the standard oscillation peak when oscillations into sterile neutrinos are included and the energy ranges DUNE ND and FD are sensitive to those distortions.

The sterile neutrino effects have been implemented in GLoBES via the existing plug-in for sterile neutrinos and NSI [513]. The DUNE ND plays a very important role in the sensitivity to sterile neutrinos both directly, for rapid oscillations with $\Delta m_{41}^2 > 1 \text{ eV}^2$ where the sterile oscillation matches the ND baseline, and indirectly, at smaller values of Δm_{41}^2 where the ND is crucial to reduce the systematic uncertainties affecting the FD to increase its sensitivity. For these studies, the DUNE ND is assumed to be an identical scaled-down



Figure 105. The left plot shows the DUNE sensitivities to θ_{14} from the ν_e CC samples at the ND and FD, assuming $\theta_{14} = 0$, along with a comparison with the combined reactor result from Daya Bay and Bugey-3. The right plot is adapted from [666] and displays sensitivities to θ_{24} using the ν_{μ} CC and NC samples at both detectors, along with a comparison with previous and existing experiments. In both cases, regions to the right of the contours are excluded. Reproduced from [514]. CC BY 4.0.

version of the FD, with identical efficiencies, backgrounds and energy reconstruction. The full set of systematic uncertainties employed in the sterile neutrino studies, as well as the methodology accounting for non-negligible beam-induced baseline spreads between production target and ND, are described in [514].

By default, GLoBES treats all systematic uncertainties included in the fit as normalization shifts. However, depending on the value of Δm_{41}^2 , sterile mixing will induce shape distortions in the measured energy spectrum beyond simple normalization shifts. As a consequence, shape uncertainties are very relevant for sterile neutrino searches, particularly in regions of parameter space where the ND, with virtually infinite statistics, has a dominant contribution. The correct inclusion of systematic uncertainties affecting the shape of the energy spectrum in the two-detector fit GLoBES framework used for this analysis posed technical and computational challenges beyond the scope of the study. Therefore, for each limit plot, we present two limits bracketing the expected DUNE sensitivity limit, namely: the black limit line, a best-case scenario, where only normalization shifts are considered in a ND +FD fit, where the ND statistics and shape have the strongest impact; and the grey limit line, corresponding to a worst-case scenario where only the FD is considered in the fit, together with a rate constraint from the ND.

For sensitivity to θ_{14} , the dominant channels are those regarding ν_e disappearance. For simplicity, only the ν_e CC sample is analyzed and the NC and ν_{μ} CC disappearance channels are not taken into account. This is expected to be improved by using more complex multichannel fits in future studies, as highlighted and recommended by [179]. The sensitivity at the 90% C.L., taking into account the systematic uncertainties mentioned above, is shown in figure 105, along with a comparison to current constraints. For the θ_{24} mixing angle, the ν_{μ} CC and NC disappearance samples are analyzed jointly. Results are shown in figure 105, along with present constraints.



Figure 106. *Left*: comparison of the DUNE sensitivity to θ_{34} using the NC samples at the ND and FD with previous and existing experiments. Regions to the right of the contour are excluded. *Right*: DUNE sensitivities to $\theta_{\mu e}$ from the appearance and disappearance samples at the ND and FD are shown on the top plot, along with a comparison with previous existing experiments and the sensitivity from the future SBN program. Regions to the right of the DUNE contours are excluded. Reproduced from [514]. CC BY 4.0.

In the case of the θ_{34} mixing angle, disappearance in the NC sample, the only contributor to this sensitivity, is probed. The results are shown in figure 106. Further, a comparison with previous experiments sensitive to ν_{μ} , ν_{τ} mixing with large mass-squared splitting is possible by considering an effective mixing angle $\theta_{\mu\tau}$, such that $\sin^2 2\theta_{\mu\tau} \equiv 4|U_{\tau4}|^2|U_{\mu4}|^2 = \cos^4 \theta_{14} \sin^2 2\theta_{24} \sin^2 \theta_{34}$, and assuming conservatively that $\cos^4 \theta_{14} = 1$, and $\sin^2 2\theta_{24} = 1$. This comparison with previous experiments is also shown in figure 106. The sensitivity to θ_{34} is largely independent of Δm_{41}^2 , since the term with $\sin^2 \theta_{34}$ in the expression describing $P(\nu_{\mu} \rightarrow \nu_s)$, depends solely on the Δm_{31}^2 mass splitting.

Finally, sensitivity to the $\theta_{\mu e}$ effective mixing angle, defined as $\sin^2 2\theta_{\mu e} \equiv 4|U_{e4}|^2|U_{\mu 4}|^2 = \sin^2 2\theta_{14} \sin^2 \theta_{24}$, is shown in figure 106, which also displays a comparison with the allowed regions from LSND and MiniBooNE, as well as with present constraints and projected constraints from Fermilab's SBN program.

DUNE will also have the ability to conduct short-baseline sterile probes, for instance, by searching for anomalous sterile-driven ν_{τ} appearance in the Near detector. The τ lepton is not directly observable in the DUNE detectors due to its short 2.9 × 10⁻¹³ s lifetime, and it is only produced for interactions where the incoming ν_{τ} has an energy of ~3.5 GeV due to the relatively large τ mass of 1776.82 MeV. However, the final states of τ decays, ~65% into hadrons, ~18% into $\nu_{\tau} + e^- + \bar{\nu}_e$, and ~17% into $\nu_{\tau} + \mu^- + \bar{\nu}_{\mu}$, are readily identifiable in the DUNE ND, given the excellent spatial and energy resolution of the ND instruments, namely, ND-LAr, ND-GAr, and SAND, which are complementary in providing sensitivity to different decays channels. While within a three-flavor scenario, the DUNE ND baseline is far too short for $\nu_{\mu} \rightarrow \nu_{\tau}$ oscillations to occur, ν_{τ} originating in sterile-neutrino driven fast oscillations could be detected. In particular, probing the $\tau \rightarrow \mu$ detection channel with highenergy muons in the final state, which is challenging due to muon containment and backgrounds, becomes very accessible through the use of the ND-GAr magnetic field and the SAND detector further downstream, with studies indicating that ND-GAr's reconstructible muon momentum via curvature extends beyond 15 GeV/c. This sensitivity will be strongly enhanced when operating LBNF in the high-energy tune, aimed at enriching the available sample of ν_{τ} at the Far detector, while extending sensitivity to anomalous ν_{τ} appearance at the Near detector [515]. Preliminary studies using LBNF's nominal flux, and including ND-LAr and ND-GAr, estimate that DUNE's sensitivities to anomalous ν_{τ} appearance may extend beyond those of previous searches carried out by NOMAD and CHORUS.

6.2.2.2. Hyper-Kamiokande. Hyper-Kamiokande (HK) is a large-scale water Cherenkov detector with a fiducial volume of about 188 kton which is approximately 8.4 times larger than Super-Kamiokande. HK is currently under construction in Japan and operations are scheduled to begin in 2027 together with the upgraded J-PARC neutrino beam. The physics capabilities of HK cover a broad range of topics including a search for sterile neutrino mixing [516]. There are various major approaches currently being considered.

While we focus in the following on the sensitivity to sterile neutrino searches, HK can also investigate other exotic scenarios like the breaking of Lorenz and CPT invariance, as demonstrated in T2K in [517], and non-standard neutrino interactions, as studied with Super-K's atmospheric neutrino observations [518].

Mixing of light sterile neutrino will be investigated with the Hyper-Kamiokande data at a baseline of 295 km baseline. T2K reported a limit on $\sin^2 \theta_{24}$ for $10^{-4} \text{ eV}^2 < \Delta m_{41}^2 < 3 \times 10^{-4} \text{ eV}^2$ using both CC and NC samples at Super-Kamiokande [299]. More stringent limit will be set by Hyper-Kamiokade with more than 20 times higher statistics by the combination of a larger fiducial volume and an upgraded J-PARC neutrino beam.

Hyper-Kamiokande's near detectors will measure the neutrino beam flux and crosssection at different baselines. Each detector has the capability to test the existence of sterile mixing at certain values of L/E but it should be noted that the sensitivity will be further enhanced by combined measurements among the detectors where the ND280 works as a near detector and give constraint to the IWCD measurement.

The IWCD (Intermediate Water Cherenkov Detector) instrument has sensitivity to sterile neutrino mixing in the $\nu_{\mu} \rightarrow \nu_{e}$ channel with a baseline of ~1 km and energy of 0.5–1 GeV, which matches the L/E at LSND and MinoBooNE. As a remarkable advantage, IWCD can measure the neutrino flux at different off-axis angles by moving the detector along its vertical pit. As the energy spectrum changes with the off-axis angle, IWCD can rule out some potential interpretations by the combination of the measurement, such as feed-down from high energy due to nuclear effects or unexpected background. The design of the IWCD detector is still under investigation, but it has potential to test the allowed region given by LSND, as indicated by the studies from the NuPRISM collaboration [519].

The off-axis ND280 has sensitivity to few eV^2 sterile neutrinos. A first search was published in [298]. It is anticipated that the upgrades currently being done for T2K, consisting in one fully active target (Super-FGD), two High-Angle TPCs, and a Time Of Flight system, and possible further upgrades under study for Hyper-K, will boost the sensitivity of ND280. Searches for sterile neutrinos with the upgraded ND280 will have several advantages with respect to [298], including a larger target mass, a lower threshold to reconstruct leptons, better performances in distinguishing electrons from gammas, and the larger exposure that will be collected in HK. ND280 has also sensitivity to search for relatively heavy sterile neutrinos produced by the decay of Kaons produced by the beamline, as demonstrated in [384].

6.2.2.3. ESSnuSB. The ESSnuSB (European Spallation Source Neutrino Super Beam) experiment [520, 521] is a proposed long-baseline neutrino oscillation experiment to use a neutrino superbeam produced using 2.0 GeV protons from the ESS Linac in Lund, Sweden,



Figure 107. Bounds (95 % CL) on sterile mixing parameters in the $\sin^2 \theta_{14}$ — $\sin^2 \theta_{24}$ plane for ESSnuSB (left and middle panels). The panel on the right shows a comparison of ESSnuSB with other long-baseline neutrino oscillation experiments. Reproduced from [667]. CC BY 4.0.

resulting in a 5 MW beam peaked at $E_{nu} = 0.4$ GeV. By sampling this beam at a distance of 540 km from Lund, using a large underground Water Cherenkov detector with a 500 kton fiducial mass, ESSnuSB will make measurements of the three-flavor oscillation second maximum, which would enable discovery of leptonic CP violation for 56% of δ_{CP} values for 10 years of data taking, and 65% of δ_{CP} values if an upgrade of the beam power to 10 MW and of the proton energy to 2.5 GeV is realized. With this experimental setup, ESSnuSB is also able to place bounds on the sterile mixing parameters, but the CP sensitivity of ESSnuSB may also be affected by the existence of light sterile neutrinos. The studies below assume two possible detector configurations for ESSnuSB and compare them: (1) combined far (FD) and near (ND) detectors with correlated systematics; and (2) an FD only with an overall systematic uncertainty.

The analysis uses the GLoBES [522, 523] software to simulate ESSnuSB. An ND and an FD are explicitly simulated to reduce systematic uncertainties [520]. The FD is a 1 Mt MEMPHYS-like water-Cherenkov detector [524] located at a distance of 540 km from the source, while the ND is assumed to have the same efficiency and background rejection capabilities as the FD [524] with a fiducial volume of 0.1 kt and placed at a distance of 0.5 km from the source. A beam power of 5 MW with 2.5 GeV protons capable of producing 2.7×10^{23} POT/year is assumed. The results are presented for a total data exposure 10 years, with 5 (5) years running in neutrino (antineutrino) mode. Throughout the simulations of the combined near and far detectors, the same treatment of systematics as in [525] was employed. In all cases, the best-fit values of the parameters from [526] assuming Normal Ordering were used.

The upper panels of figure 107 display the sensitivity of ESSnuSB to the sterile mixing angles θ_{14} and θ_{24} for particular values of Δm_{41}^2 at 95 % CL. The left panel shows the results for combined FD+ND, while the middle panel presents the results for the FD using the ND as a counting experiment. From the left panel, it is apparent that for $\Delta m_{41}^2 = 0.01 \text{ eV}^2$ the bounds are weak, as for this value of Δm_{41}^2 the oscillations have not yet developed for the ND, and therefore, the existing sensitivity comes from the FD, where the oscillations are averaged out. Increasing the value of $\Delta m_{41}^2 = 1 \text{ eV}^2$. Increasing the value of Δm_{41}^2 further, the oscillations tend to become averaged out and again the sensitivity decreases. From the middle panel, where there is no spectral information on the ND, the sensitivity for $\Delta m_{41}^2 \sim 1 \text{ eV}^2$ is lost and similar sensitivities are obtained for all values of Δm_{41}^2 . The right panel of figure 107



Figure 108. Leptonic CP violation sensitivity of ESSnuSB for four different values of the sterile mixing phase δ_{24} as a function of true δ_{13} . Reproduced from [667]. CC BY 4.0.

compares the sensitivity of ESSnuSB to the sterile mixing parameters with other future longbaseline neutrino oscillation experiments, specifically T2HK [527], T2HKK [527], and DUNE [528]. The comparison shows that the sensitivity of ESSnuSB is slightly worse than the other experiments in most of the region of the parameter space. However, as can be seen from the left panel in figure 107, it is expected that the ESSnuSB sensitivity would be considerably improved in the presence of an ND.

Figure 108 presents the CP violation sensitivity of ESSnuSB due to δ_{13} for four different values of the sterile mixing phase δ_{24} , illustrating how the presence of sterile neutrino oscillations would modify the interpretation of leptonic CP violation measurements. The sterile mixing parameters assumed are $\sin^2 \theta_{14} = \sin^2 \theta_{24} = 0.025$ [103], $\theta_{34} = \delta_{34} = 0^\circ$, and $\Delta m_{41}^2 = 1 \text{ eV}^2$. The left panel shows the sensitivity for FD+ND, and in the right one, the sensitivities considering only the FD with 8 % (10 %) overall systematics in signal (background). Again, when including the ND, the sterile mixing parameters are better constrained and the sensitivity to CP violation improves.

The ESS Linac proton beam is expected to turn on in 2025. Construction of the neutrino beam facility is under study and may begin with a low-energy nuSTORM-like ND complex starting in 2024 with operations starting in 2027.

6.3. Reactor neutrino experiments

In the period following Snowmass 2021, emphasis in the reactor sector will be placed on 50 km-scale baseline experiments probing the SM neutrino mass hierarchy and the solar mixing angle, such as JUNO, and on and very short-baseline experiments aiming to probe sterile oscillations and perform high-statistics measurements of reactor $\bar{\nu}_e$ fluxes and spectra. Short-baseline reactor experiments will be particularly important in furthering understanding of the source of enduring reactor-sector anomalies, in addressing existing claims of non-standard oscillation observation in recent reactor experiments, and in providing orthogonal experimental datasets valuable in global BSM physics fits.

Data from multiple channels, energies, and sources will be crucial for disentangling possible BSM effects manifested in the anomalies, since these effects may manifest

differently in different experimental regimes. In the stable of global measurements, shortbaseline reactor measurements are unique in their capability to very purely probe sterile oscillation effects. This is due to the lower energies involved in interactions and decays in the reactor, which prohibits production and decay of heavier hidden-sector particles, and their very short baselines, which minimize the impact of NSIs. Even in the case of purely oscillation-driven BSM explanations, reactor experiments offer unique benefits due to their unambiguously pure flavor content: this is in contrast to interpretations of short-baseline decay-in-flight and decay-at-rest experiments, which will be complicated by competition between appearance and disappearance effects in individual flavor channels [179]. These points serve to emphasize the substantial value added by acquiring datasets from *all* of experiment types: short-baseline reactor datasets will find greater utility and application in the coming decade if they are accompanied by short-baseline accelerator datasets, and vice versa.

A summary of new short-baseline reactor oscillation measurements currently underway or planned for the next five years is provided in table 17. For future experiments, some specific reactor experiment parameters are particularly valuable in achieving the future short-baseline reactor physics goals mentioned above. While reactor experiments in the past decade have provided excellent limits in the oscillation phase space region below a few eV^2 , limits above this mass splitting are substantially weaker; to address this, very short (<10 m) baseline experiments using compact cores and segmented detectors are particularly valuable. To provide more robust probes of isotopic IBD yields and spectra and thus better understanding of the fidelity of existing flux predictions, high-statistics dataset are needed from reactors of widely varying fuel content. In this context, experiments running at HEU reactors and over a full cycle at a single LEU reactor core would be particularly valuable. As demonstrated in table 17, most of these expected future experiments fulfill one or more of these key experimental requirements.

The sterile neutrino oscillation probing powers of these experiments, expressed in terms of sensitivity to $\sin^2 2\theta_{14}$ in a 3 + 1 oscillation framework, are shown on the right plot in figure 109, with the left plot also indicating some phase space parameter ranges of interest. In addition to showing individual experiments, the right hand plot indicates the style of analysis assumed in each curve: one using only relative measured spectral shapes, one using knowledge of the absolute measured spectrum, and one using only rate information. Ratebased analyses provided by reactor flux measurements clearly lack the sensitivity of shapebased ones; combined with limitations in knowledge of absolute reactor fluxes (section 4.1.3), rate-based analyses appear, at present, unlikely to yield much fruit in the pursuit of sterile neutrinos. Of the shape-based measurements, DANSS and TAO are likely to dominate sensitivities in the lower- Δm^2 regime, while PROSPECT-II is likely to dominate future sensitivity in the less-explored region above $\sim 3 \text{ eV}^2$. DANSS and TAO may achieve some power in probing the oscillation phase space region suggested by Neutrino-4, while PRO-SPECT-II should address this region at high confidence level while also covering phase space suggested by the Gallium anomaly that is currently unaddressed by spectrum-based reactor analyses. Finally, the combination of DANSS, TAO, and PROSPECT-II datasets should generate percent-level sensitivity to active-sterile couplings in the electron disappearance channel for all Δm^2 space below roughly 10 eV², enabling greater clarity in interpretation of CP-violation results from DUNE [529].

6.3.1. DANSS upgrade. One of the limitations of DANSS experiment was that the oscillation spectrum was smoothed out by the finite energy resolution. Resolution is limited mainly by the light collection system which makes further progress in DANSS physics program very challenging. Because of which, it has become necessary to upgrade the detector

Experiment	Baseline (m)	Reactor	Reactor	Detector	Target	Sterile ν
DANSS [530] JUNO-TAO [531] NEOS-II Neutrino-4 upgrade PROSPECT II [533]	11-13 30 24 6-12 7.9	Type LEU HEU LEU HEU HEU	Power (MW <i>th</i>) 3000 4600 2800 90 85	Size $1 m^3$ 2.8 ton $1 m^3$ $2 m^3$ 4 ton	Segmented PS with Gd coating Single GdLS Single-volume GdLS + PSD Segmented GdLS Segmented ⁶ Lit S + PSD	Search strategy Multi-site Single-site Single-Site Multi-site/zone

Table 17. List of the future short-baseline reactor neutrino experiments that will be able to search for sterile neutrino oscillations.



Figure 109. Left: claimed sensitivities (90% CL) to sterile neutrinos of the upcoming reactor neutrino experiments listed in table 17. Also shown is the unexplored parameter space obtained by comparing reactor rates + evolution datasets to the reevaluated HM model based on the KI suggested normalization correction as discussed in section 3.3.2. NEOS and JUNO-TAO are shown in dashed lines to indicate that the experiments make use of the absolute spectral shapes. Also shown (solid) is a version of JUNO-TAO's curve that is produced by making use of the experiment's position reconstruction to generate virtual segments that provides the ability to perform a completely absolute shape independent analysis. The plot was generated using the same approach as described in [668]. Right: Sensitivities of the future reactor neutrino experiments in comparison with the Gallium Anomaly and Neutrino-4 suggested parameter space. Also included is the line at $\sin^2 2\theta = 0.3$ (dashed black) corresponding to the limit which has to be excluded to avoid ambiguities in the future LBL CP-violation measurements.

design in order to improve energy resolution. The upgrade plan is to replace current strips with new strips with larger cross section [530]. In the current design, the strips are coated with titanium oxide for light reflection and gadolinium oxide for neutron capture which produces relatively thick dead layer with titanium and gadolinium. The non-uniformities in the light collection system is the another limiting factor for resolution. The light collection system consists of three wavelength shifting fibers which are readout by SiPMs and PMTs are on the same side of the strip which introduces non-uniformities. To tackle this issue, the upgrade plan is to have SiPM only readout from both sides of the strips. Removing the PMTs from light collection system will allow for an in increase in the sensitive volume of 70%. The sensitivity to sterile neutrinos with the upgraded detector after 1.5 years of running is shown in figure.

6.3.2. JUNO-TAO. The taishan antineutrino observatory (TAO) will be a satellite detector of the JUNO experiment [531]. Located at a baseline of approximately 30 m from one of the Taishan 4.6 GW_{th} cores in a basement 9.6 m below ground level, TAO's neutrino target will consist of 2.8 tons of gadolinium-doped liquid scintillator contained in a 1.8 m diameter acrylic sphere. This sphere will be surrounded by a copper spherical shell supporting an array of Silicon Photomultipliers (SiPMs) with 94% surface coverage, allowing to reach a light level of 4500 photoelectrons per MeV and an energy resolution of $\leq 2\%$ at 1 MeV. This volume will be placed inside a cylindrical stainless steel tank with 2.1 m of outer diameter and 2.2 m of height filled with a liquid scintillator buffer. The tank will be cooled to -50° C to mitigate the dark noise of the SiPMs, and will be surrounded by an active 1.2 m thick water



Figure 110. Expected sensitivity of upgraded DANSS detector to sterile neutrino oscillations after 1.5 years of running in comparison with the sensitivity of the DANSS experiment. Reproduced from [110]. © IOP Publishing Ltd. CC BY 3.0.

Cherenkov veto tank, as well as high-density polyethylene in the top and lead in the bottom. Approximately 4000 IBD interactions are expected per day in a central 1 ton fiducial volume with a detection efficiency of about 50%. TAO is scheduled to begin operations around the same time as the rest of the JUNO experiment in 2023.

TAO's primary purpose is to make a precise measurement of the reactor antineutrino spectrum that will serve as a benchmark for JUNO, other experiments, and nuclear databases. However, TAO will also be in an excellent position to search for a distortion in the shape of the reactor antineutrino spectrum caused by sterile neutrino mixing. The exclusion contours expected in JUNO-TAO at 90% and 99.7% C.L. in the absence of a sterile neutrino signal are shown on the right panel of figure 111 [531]. The analysis behind these contours takes into account the physical dimensions of both the detector and the reactor core, and assumes 3 years of exposure with 80% reactor time on. A conservative bin-to-bin uncorrelated uncertainty of 5% (with 50keV bin width) on the predicted reactor spectrum is assumed. The exclusion contours from the Daya Bay [325], PROSPECT phase-I [314], and NEOS [532] experiments are also shown for comparison purposes. TAO's baseline, abundant statistics, and exquisite energy resolution, afford it a leading sensitivity to $\sin^2 2\theta_{14}$ in the $10^{-1} \text{ eV}^2 \leq |\Delta m_{41}^2| \leq 3 \text{ eV}^2$ region.

6.3.3. NEOS II. NEOS-II has taken about 400 live days of reactor-ON data to cover a full fuel cycle and this is to observe time evolution of reactor neutrino spectrum and flux and to investigate whether there is any relation between the '5 MeV excess' and ²³⁵U (or ²³⁹Pu). For background subtraction two periods (total ~125 d) of reactor-OFF data were taken before and after the reactor-ON period. NEOS-II has observed light-yield decrease due to precipitation in the GdLS target from the early stage of the data-taking. The light-yield decrease resulted in



Figure 111. 90.0% and 99.7% C.L. exclusion contours expected at JUNO-TAO with 3 years of data taking. The exclusion contours from the Daya Bay [325], PROSPECT phase-I [314], and NEOS [532] experiments are also shown for comparison purposes. The parameter space to the right of all curves is excluded. Reproduced with permission from [531].

worse energy resolution, from $\sim 5.6\%/\sqrt{1 \text{ MeV}}$ at the beginning to $\sim 7.7\%/\sqrt{1 \text{ MeV}}$ at the end of the data-taking, and increased time between prompt and delayed signals from ~ 28 to $\sim 40 \ \mu s$. According to our study, however, these are found to be minor effects in the NEOS-II data analysis.

Currently two main analyses are underway in NEOS-II. One is on the decomposition of ²³⁵U and ²³⁹Pu spectra to better understand the origin of the '5 MeV excess' and the other is a light sterile neutrino search using rate+shape analysis. Figure 112 shows the 90% C.L. sensitivity (exclusion) curve of NEOS-II (NEOS-I) on a light sterile neutrino search where the Daya Bay spectrum is used as a reference spectrum. However, we plan to use the RENO spectrum as a reference for the final result of NEOS-II. The exclusion curve and sensitivity are drawn using a raster scan. A slight improvement in the sensitivity is expected in the NEOS-II using rate+shape analysis and updated reference from RENO, while a shape-only analysis was used in NEOS-I.

6.3.4. Neutrino-4 Upgrade. The Neutrino-4 experiment is planning for an upgrade of their experiment to increase statistical precision. The upgraded detector will contain 4 identical modules, each consisting of a square grid of 25 segments-amounting to $3 \times$ larger volume. The horizontally positioned segments will have read-out at both the ends, as opposed to vertically placed single read-out segments in Neutrino-4. The target volume will consist of Gd-doped PSD-capable liquid scintillator to reduce background. The combination of the increased volume and reduced backgrounds from higher Gd-loading and addition of the PSD capability is expected to have a three-fold increase in statistical power.

Major Report



Figure 112. NEOS-II 90% C.L. sensitivity (orange) curve to light sterile neutrino oscillations using 400 live days of data. Also overlayed are the NEOS-I final sensitivity (blue-dashed) and exclusion (solid blue, green) curves. Reproduced from [669]. CC BY 4.0.

6.3.5. PROSPECT-II. The PROSPECT experiment has placed some of the most stringent limits on the eV-scale sterile neutrino oscillations [135]. The detector was decommissioned in 2020 after an unexpected HFIR down time. The results are currently statistically limited and could be significantly improved by a longer dataset. PROSPECT-II [533] is an upgrade of the PROSPECT experiment with evolutionary modifications intended to take advantage of the beneficial design aspects of the PROSPECT detector and mitigate the drawbacks in the PROSPECT detector to extend the experiment's statistical reach. The suggested design changes were based on the lessons learned after the decommissioning of the PROSPECT detector. To eliminate the chance of liquid scintillator contacting the PMT electronics, a separate subsystem is being designed that holds the PMTs completely outside the liquid scintillator volume. Additionally, in order to reduce the contact of liquid scintillator with other detector components, a simplified calibration system that runs around the periphery of the detector is being developed. These modifications are expected to mitigate the risk of PMT failure, highly simplify inner detector assembly, significantly improve the choice of detector materials, and allow for increase in the target volume while still maintaining the footprint approved for operations at HFIR.

PROSPECT-II design choices are based on the data collected by PROSPECT and datavalidated Geant4 simulations. Considering all the design modifications, PROSPECT-II is expected to have ~30% increase in exposure and a three-fold increase in S:B from 1.4 to 4.3.



Figure 113. Sensitivity of the PROSPECT-II experiment to sterile neutrinos for 2 years at HFIR (in black), 2 years at HFIR and 2 years at a power reactor (in purple), and 4 years at HFIR (in black) and 2 years at a power reactor (in green). PROSPECT will cover major portion of the suggested sterile neutrino parameter space. A combination of PROSPECT, KATRIN, and Daya Bay experiments will be able to cover significant portion of the parameter space down to the long baseline CP-violation disambuguity limit. Reproduced from [533]. © IOP Publishing Ltd. All rights reserved.

Figure 113 experiment to the sterile neutrino-induced oscillations. To mitigate the reliance on the reactor neutrino models the sensitivities are estimated by comparing at multiple baselines within the detector. As illustrated in figure 113, within two calendar years, PROSPECT-II will be able to fully cover the RAA and Gallium Anomaly suggested regions below 15 eV² While longer baseline and LEU experiments cover the lower oscillation frequencies and the β -decay endpoint experiments like KATRIN and Project 8 cover the higher oscillation frequencies, PROSPECT-II is uniquely situated to complement these results by providing the most stringent limits on $\bar{\nu}_e$ disappearance between 1 and 20 eV².

6.3.6. Other experiments. Several plastic scintillator experiments including CHANDLER [534], and ISMRAN [535], NuLat [536], and SoLID [537] that have the capability to provide modest sensitivity to sterile neutrinos have either been built or planned to be built in the near future. Similar to the experiments described above, these experiments all use segmented detectors for background reduction which would also be useful to perform oscillation search with a single detector. Plastic scintillators provide a convenient and robust way to segment the detectors either in two or three dimensions. The ability to achieve high signal-to-background

Major Report

ratios over large volumes necessary to perform precision oscillation search has yet to be demonstrated for these detector types.

CHANDLER. CHANDLER is a near-field reactor neutrino detector technology consisting of an array of wavelength shifting plastic scintillator cubes, alternating in layers with thin sheets of ⁶Li loaded zinc sulfide (ZnS) scintillator. Positrons from reactor $\bar{\nu}_e$ interactions deposit their energy in the plastic scintillator creating a prompt pulse of light, while the neutrons thermalize and capture on ⁶Li in the ZnS to create a delayed pulse. The ZnS releases its scintillation light 20 times slower than the plastic scintillator, which makes an unmistakable neutron capture tag. Similarly, the high segmentation of the plastic cubes is used to tag the positron annihilation gamma, forming a powerful positron tag. The combination, in conjunction with tight spatial and temporal coincidence requirements, form a strong discriminator against all backgrounds, since correlated, or even random, coincidences of a positron and a neutron are vey rare in nature. Small air gaps between the cubes allow the light to transmitted along the rows and columns of cubes, by total internal reflection (TIR), to the surface of the detector. Light from neutron capture in the ZnS is absorbed by the wavelength shifter in the cube layers on either side of the sheet and re-emitted, so that it too can be transmitted by TIR. The position of the cube where the light originated sits at the intersection of the hit row and column.

The CHANDLER technology was demonstrated in a 2017 deployment to the North Anna Nuclear Generating Station, in Mineral, Virginia [534]. There, the 80 channel, 80 kg MiniCHANDLER prototype was housed in a 14-foot trailer parked 25 meters from Reactor 2, and ran for four and a half months, including one month of reactor off. These data were used to isolate a sample of 2881 events. In this analysis, topological selections for the positron annihilation gammas played a critical role in the observation of reactor IBD events, shifting the signal significance from less than 3σ to 5.5σ . This demonstration used old PMTs, which were only marginally sensitive the Compton edge of the 511 keV gammas. The CHANDLER collaboration is undertaking an upgrade of the MiniCHANDLER prototype with new PMT combined with compound parabolic light guides which will improve the light collection efficiency by a factor of four. This will greatly enhance the efficiency of the positron tag, and achieve an average energy resolution of around 6% at 1 MeV.

CE ν **NS experiments.** The discovery of coherent elastic neutrino-nucleus scattering (CE ν NS) by the COHERENT experiment [474] using a decay-at-rest source, opened up the possibility of using this mechanism to perform several BSM searches. Multiple global efforts [538–545] are underway to leverage the intense source of neutrinos from reactors to perform CE ν NS measurements. CE ν NS has not yet been observed in these experiments owing to a combination of high backgrounds at reactor facilities and a need for extremely low threshold detectors. However, enabling detector technologies are progressing at a fast pace. The observation reactor CE ν NS opens doors to a vibrant BSM physics program, including searches for sterile neutrino oscillations.

6.3.7. Joint analyses. As discussed in section 4.1.3.4, a single experiment cannot have the capability to cover all the sterile neutrino parameter space of interest. Thus it is beneficial to perform joint fits of experimental datasets each of which provide sensitivity to different regions of parameter space. When performed using proper statistical techniques and by including relevant systematics and correlations properly, such joint fits can be extremely powerful in covering wider parameter space as well as in alleviating any unknown systematic effects specific to a single experiment that could mimic oscillations. As highlighted in [334], there are attractive opportunities in combining data from multiple current and upcoming reactor experiments such as Daya Bay, PROSPECT, STEREO, NEOS, and JUNO-TAO.





Figure 114. Left. Nuclear levels for the 65 Zn radioactive source decay. Right. Number of events, *N* as a function of the number of exposures, *m*, for experiments with 51 Cr and 65 Zn sources. Reproduced from [547], with permission from Springer Nature.

Reactor experiments could further be combined with accelerator experiments such as MINOS and MINOS+. It is worth noting that the Daya Bay collaboration plans to publicly release its full data set once all final results have been released [335], allowing such combinations to occur even well after the collaboration has dissolved. Similarly, following the end of data taking in 2016, the MINOS/MINOS+ CL_s surfaces remain available for use in future combinations. Other experiments such as PROSPECT and STEREO are also moving towards releasing data as part of their publications in a format that could be used to perform such joint analyses.

6.4. Radioactive source experiments

The combination of the results obtained by GALLEX, SAGE and BEST leads to a gallium anomaly (section 3.3.3) with a large significance, allowing the possibility that electron neutrinos may disappear through oscillations with the participation of a sterile neutrino at the O(eV) mass scale. The following are experimental proposals aiming to perform additional studies to confirm or refute the gallium anomaly.

6.4.1. BEST-2. The BEST Collaboration has proposed to use their experimental configuration (see section 4.1.5) to run the same type of experiment by measuring the electron neutrinos produced by an artificial source made of 65 Zn [546, 547].

When the isotope ⁶⁵Zn decays (figure 114, left), a neutrino of energy 1.35 MeV is emitted in nearly half of the decay events. The remaining events involve the emission of a neutrino with an energy of 235 keV close to the threshold for capture by gallium (233 keV), and the corresponding cross section is small. On the other hand, the cross section for the capture of 1.35 MeV neutrinos by ⁷¹Ga nuclei is approximately three times as large as that for 0.75 MeV neutrinos from a ⁵¹Cr source [76]. Then, the expected rate of capture of neutrinos from a ⁶⁵Zn source of activity 3 MCi in a single zone at identical dimensions of the sources and target zones is $n_0 = 108 \text{ d}^{-1}$. Since the ⁶⁵Zn lifetime is longer than the ⁵¹Cr lifetime ($T_{1/2}$ = 244.1 and 27.7 d, respectively), BEST-2 could perform measurements with ⁶⁵Zn for a longer time, so that an accumulation of a commensurate data sample may require a substantially lower source activity [547]. To compare against the previously published results, with the data sample accumulated in the BEST experiment, it was estimated that the number of extracted ⁷¹Ge atoms expected in the absence of oscillations was 1657. For the proposed source based on ⁶⁵Zn, the total number of extracted ⁷¹Ge atoms as a function of the number of target exposures, N(m), is given in the right panel of figure 114 for various values of the source activity. The total event number N = 1657 may be attained even with a ⁶⁵Zn source of activity nearly ten times lower than the activity of the ⁵¹Cr source in the BEST experiment [547].

The differences (especially the higher energy of the emitted neutrinos) to be implemented in BEST-2 are expected to reach an extended range of Δm^2 values. In fact, the areas of sensitivity to this parameter in the two experiments (BEST and BEST-2) prove to be shifted in such a way that the highest sensitivity of one experiment corresponds to the lowest sensitivity of the other experiment, and vice versa. According to [548], a global result of the two experiments would provide a better statistical significance of measured oscillations (if there are any in the region of searches), but it will also permit precisely measuring the parameters of these oscillations.

6.4.2. Neutrino oscillometry with Jinping. Another neutrino artificial source proposed to study the possible oscillation of active to sterile neutrinos is the isotope 144 Ce $^{-144}$ Pr. In this case, the idea is to use such a source for the future solar neutrino experiment Jinping [549].

A s a source of $\bar{\nu}_e$, the decay chain of isotopes ¹⁴⁴Ce⁻¹⁴⁴Pr is a suitable option for oscillometry experiment. The antineutrino energy spectrum of ¹⁴⁴Pr is continuous with the end point around 3 MeV and with an overall half-life of 285 d. About 48.5% of the emitted antineutrinos are at energies above the detection threshold of the IBD reaction (the value of the threshold is 1.8 MeV) and thus can be used for the measurements. Based on previous calculations, the maximal source activity can reach 100 kCi [549].

Following [549], the Jinping neutrino detector will be located in the Jinping Mountain, Sichuan Province, China with a maximum overburden around 2400 meters. The Jinping collaboration plans to build a 2 kton detector using slow liquid scintillator (LSc). This delays the scintillation process and thus separates from the Cherenkov light, significantly increasing the background rejection capability using the particle identification method. The inner detector volume will have a spherical shape with a radius around 8.2 meters. The expected energy resolution will be $5\%\sqrt{E[MeV]}$ and the position resolution 10 cm/ $\sqrt{E[MeV]}$.

The proposed experimental setup is shown in figure 115. Two possible locations for the point source are being considered: one at the center of the sphere (position I) and another at a distance 3 m from the edge of the detector (position II). The source at the detector center case gives a higher statistics but a shorter range of baseline. Technically, the more realistic case is when the source is outside of the detector. At both source positions the exposure time is assumed to be 450 d. Initial source activity will be 50 and 100 kCi for position I and position II respectively. The expected non-oscillation event rate is 28.5 K and 73.8 K for position I and position II respectively.

The expected sensitivity to sterile neutrinos, represented as a two-dimensional exclusion plot, compared with another proposed experiments is drawn on figure 116. All currently interesting regions for sterile neutrino searching are covered if position I is implemented. For position II, some part of the RAA and best fit are still not fully covered.

6.4.3. THEIA. The most recently updated sterile neutrino landscape based on the results from BEST and MicroBooNE combined with short-baseline reactor experiments point to the possibility of active to sterile neutrino oscillations in the region with $\Delta m^2 > 1 \text{ eV}^2$ which implies that oscillation pattern is only measurable at distances of less than 10 m.



Figure 115. The schematic layout of the proposed experiment at Jinping. Two positions for the radioactive source are assumed: at the center (position I) and outside of the detector (position II). Two fiducial volume cuts are applied. Inner cut with radius 100 cm and outer cut with width 70 cm. The yellow area is the active volume for position I, the yellow area plus the inner cut region is active volume for position II. Reproduced from [549]. CC BY 4.0.

The three direct independent methods to test sterile neutrino hypothesis include a very short-baseline reactor experiment, accelerator experiment and use of neutrino and antineutrino generators in the vicinity of the large liquid scintillator (LS) or water based liquid scintillator detector (WbLS). The 100 kton (25 kton) WbLS THEIA detector [550, 551] represents a particularly promising venue for a decisive test using neutrino and antineutrino generators, to observe a *distance-dependent* neutrino flux from the source at the distances of the order of oscillation length.

The neutrino oscillation length is given by the following formula:

$$L_{\rm osc}[\mathbf{m}] = 2.48 \frac{E_{\bar{\nu}_e}[\mathrm{MeV}]}{\Delta m_{\rm new}^2 [\mathrm{eV}^2]}.$$
(90)

In the case of sterile neutrino $\Delta m^2 > 1 \text{ eV}^2$, the oscillation distance is of the order of a couple of meters. THEIA spans over tens of meters of distance and the source can be placed within meters of the target, creating a baseline comparable to the sterile neutrino oscillation length. Observation of the oscillation pattern in the distant-dependent measurement in WbLS/LS would represent convincing proof of the existence of sterile neutrinos and their oscillation with the other three flavors.

Sterile neutrino search with antineutrino generators in THEIA. THEIA can detect $\bar{\nu}_e$ via inverse beta decay reaction (IBD) on hydrogen that makes it easily distinguishable from the backgrounds thanks to the double coincidence signature. The most promising antineutrino generator for this search is a pair of beta decaying nuclei ¹⁴⁴ Ce-¹⁴⁴ Pr thanks to high yield of ¹⁴⁴ Ce in the spent nuclear fuel combined with high endpoint at 3 MeV with respect to 1.8 MeV IBD interaction threshold, and relatively long half-life of 285 d. There are two possible configurations: placing the source in the center of the detector for maximum interaction rate or on the side of the detector which is less disruptive and preferred but leads to reduced antineutrino flux. Nevertheless, due to its huge volume of 100 kton THEIA will observe over 100 million antineutrino events in case of no oscillation (a couple of million less in case of sterile neutrino oscillations) with 3.7 PBq source running for 18 months. This measurement is



Figure 116. The exclusion contours on a two-dimensional parameter space. 90% and 95% C.L. are shown for two possible setups. Combined reactor anomaly is also shown [66]. The stars indicate the current best-fit value for the sterile neutrino hypothesis [102], the bes-fit value of the reactor antineutrino anomaly (RAA) [66], and the best-fit result of the Neutrino-4 experiment [670] from left to right, respectively. Reproduced from [549]. CC BY 4.0.

practically background free and statistical error is negligible leading to a very sensitive measurement of large mass sterile neutrino oscillations even if the mixing angle is very small using distance-dependent rate and spectrum measurement. It is worth noting that even with an order of magnitude weaker ¹⁴⁴ Ce-¹⁴⁴ Pr source, THEIA would still make a robust measurement of sterile neutrino oscillations. A 0.37 PBq source has been successfully made in the past and is logistically easier for shielding, handling and transport.

In addition, the IsoDAR [491] collaboration is working on producing a feasible decay at rest source that is based on irradiation of ⁷Li with a strong neutron flux to produce ⁸Li that beta decays within 838 milliseconds. Emitted antineutrinos have a Gaussian energy spectrum in the range from 3 to 13 MeV. This work is progressing steadily, but is not available for production. The advantage of this source is that it produces antineutrinos at higher energy and at a steady rate. IsoDAR would also provide practically background-free high-statistics measurements of sterile neutrino oscillations.

Sterile neutrino search with neutrino generators in THEIA. The most exciting opportunity for THEIA to test the sterile neutrino hypothesis would be to directly cross-check the GA and BEST result by deploying ⁵¹Cr electron neutrino source. As opposed to radiochemical measurements THEIA would measure a Compton-like shoulder due electron neutrino backscatters produced by the ⁵¹Cr source. As opposed to bulk rate measurement of the event rate, THEIA would allow spectrum measurement with electron neutrino detection in real time. ⁵¹Cr emits a monoenergetic 753 keV gamma 90% of the time. This energy translates at the Compton-like shoulder at 0.5 MeV, which is slightly below 0.6 MeV energy threshold of the THEIA pure WbLS phase. However, as part of the Neutrinoless Double Beta Decay search, a 8 m radius balloon filled with pure liquid scintillator (LS) will be deployed.

The LS will have a high scintillation yield, resulting in better energy resolution and lower energy threshold required to detect electron neutrinos from ⁵¹Cr. This would present an excellent opportunity for sterile neutrino search with ⁵¹Cr electron neutrino source, directly testing GA and BEST results with electron neutrino detection in real-time.

Conclusion. In conclusion, a large WbLS detector such as THEIA will provide an exceptional playground for the detailed study of sterile neutrino oscillations, requiring modest energy and vertex resolution. Additional important ingredients include knowledge of the absolute incoming antineutrino (neutrino) flux that can be measured in sources and a few meter distance between the source and the detector. THEIA WbLS detector has a high potential to carry out a powerful search for sterile neutrinos free of statistical limitations that plague current generation of experiments, in the high sterile neutrino mass regime of $\Delta m^2 > 1$ eV ². With its large target mass of up to 100 kton, a tremendous number of IBD interactions from ¹⁴⁴ Ce-¹⁴⁴ Pr source will be collected allowing detailed, high statistics study of the position dependent neutrino flux. THEIA can also directly cross-check GA and BEST high confidence electron neutrino measurement with an independent detection method.

6.5. Atmospheric neutrino experiments

6.5.1. IceCube upgrade. The IceCube Upgrade [552] will begin operation in 2026 and provide both a lower energy threshold and more precise calibration for the full IceCube Neutrino Observatory. Presently dominant sources of systematic uncertainty for IceCube sterile neutrino analyses include absorption and scattering in South Pole glacial ice [553], both expected to be significantly reduced by improved precision of detector calibrations from the Upgrade with its smaller optical module spacing and enhanced calibration systems. Reduced thresholds will enable unprecedented precision for ν_{τ} appearance searches from standard oscillations, and will similarly contribute to improvements of low-energy sterile neutrino searches at IceCube.

6.5.2. DUNE. As noted in section 6.2.2.1 (DUNE LBL), DUNE [554] is primarily designed as an accelerator experiment to accurately measure the value of δ_{CP} using a neutrino beam produced from the LBNF accelerator complex at Fermilab. It will employ a group of near detector (ND) and a far detector (FD) to measure neutrino oscillation parameters. The DUNE FD will be located at the Sanford Underground Research Facility, South Dakota. Benefiting from tpnlow cosmic-muon background, it will also act as an atmospheric neutrino detector, and will make measurements of several oscillation parameters for SM and BSM physics models, which will be complementary to the measurementss made with the accelerator neutrino data.

The FD will be constructed as four separate LArTPC modules, with a total fiducial mass of 40 kt. While DUNE will collect significantly lower event statistics compared to very large volume neutrino telescopes such as IceCube [555], the excellent energy and direction reconstruction will enable competitive atmospheric physics measurements. With LArTPC detectors, very high-resolution spatial tracking (up to millimeter accuracy) of particles produced in neutrino interactions is possible and it can resolve final state morphologies to an extent not possible with other detector technologies. DUNE will also have a finer control of detector and neutrino interaction systematics, enabled by its Near Detectors in the beamline. Between IceCube and DUNE atmospheric neutrino event measurements, there will be a complementarity in terms of large sample size and high resolution. Notably, using atmospheric neutrino events in a few GeV range, DUNE FD will be able to discover the neutrino mass ordering (NMO) with a statistical significance of 3σ

or higher with 10 years of data [554]. DUNE also has potential to measure the value of δ_{CP} using sub-GeV atmospheric neutrino event samples [556]. It will constrain many BSM physics models such as sterile neutrinos, NSI, neutrino decays and Lorentz Invariance Violation (LIV).

Atmospheric neutrinos [557] are produced in the interaction of the cosmic rays with atmospheric nuclei and contain both $\stackrel{(-)}{\nu_e}$ and $\stackrel{(-)}{\nu_{\mu}}$ components. They span a broad energy range from MeV to PeV, with a peak around 1GeV. Propagation of atmospheric neutrinos through the Earth gives accessible L/E from 10 to 10^4 kM/GeV and their measurements can provide sensitivities to $\Delta m^2 \sim (10^{-4} - 1) \text{ eV}^2$. The atmospheric data at DUNE FD will make measurement of u_e and u_μ disappearance as well as $u_ au$ appearance channels. Hence, the DUNE FD can constrain all three new mixing angles θ_{14} , θ_{24} and θ_{34} over the aforementioned broad range of Δm_{41}^2 with the atmospheric data. Sensitivities to these parameters are expected to be enhanced around $10^{-2} - 10^{-3} \text{ eV}^2$ thanks to matter effects, as neutrinos cross the Earth's mantle and core during their propagation. Preliminary sensitivity study suggests that with conservative assumptions on detector properties such as resolutions and PID classification, the DUNE FD can probe sterile parameter space with sensitivities comparable to the current limits from other experiments. Dedicated sensitivity studies to these parameters using DUNE atmospheric neutrino MC simulations is planned in the near future. A joint fit of atmospheric and beam neutrino data is expected to improve parameter constraints over both individual fits. Using DUNE atmospheric data, constraining more complicated models such as decaying sterile neutrinos is also foreseen. Additionally, due to DUNE's capacity to inspect in detail the energy losses of muons, they can infer the energy of uncontained high-energy muons. This provides sensitivity to DUNE to high-energy atmospheric neutrino analysis as shown in [558], which can provide complementary constraints in light sterile neutrino to the ones provided by the contained events.

6.5.3. Hyper-Kamiokande. Hyper-Kamiokande (Hyper-K) is the next-generation large water Cherenkov detector to be deployed in Japan. Beyond their accelerator program discuss previously in this white paper, they also are able to measure atmospheric neutrinos much like its predecessor Super-K.

The expected sensitivity to light sterile neutrinos from Hyper-K has been reported in [559]. As in the case of Super-K they are predominantly sensitive to $|U_{\mu4}|^2$ and $|U_{\tau4}|^2$ mixing elements by looking for distortions in the angular and energy distributions of their events. The expected sensitivity can be seen in figure 117 compared to the previous constraints from Super-K. As can be seen from this figure the improvement on $|U_{\mu4}|^2$ is marginal; however, significant improved sensitivity can is expected in $|U_{\tau4}|^2$. This is interesting since currently ANTARES has reported a small preference for non-zero $|U_{\tau4}|^2$, which would be tested by Hyper-K.

6.5.4. Atmospherics in THEIA. THEIA is a proposed multi-kiloton hybrid optical neutrino detector designed with multiple physics goals in mind [551]. The experiment plans to leverage novel scintillating materials to fully exploit information from scintillation and Cherenkov light production mechanisms using advancements in fast photodetectors and spectral sorting technologies, on a large scale. This 'hybrid' technology will simultaneously enable high light yields for superb position and energy reconstruction, direction reconstruction from Cherenkov light and particle identification (PID) through knowledge of both mechanisms. There is also the potential to employ isotope loading techniques for various physics needs (section 6.4.3).



Figure 117. Hyper-K expected 90% C.L. upper limits on $|U_{\mu4}|^2$ appear as red lines in the left figure. The right figure shows 90% (solid) and 99% (dashed) C.L. limits on $|U_{\mu4}|^2$ versus $|U_{\tau4}|^2$ for a 5.6 Mton-year exposure (red) in comparison with recent limits from Super-K (black). Reproduced with permission from [559].

While physics potentials have been evaluated for long baseline physics, solar neutrinos, supernova neutrinos, DSNB, geoneutrinos, reactor neutrinos, neutrinoless double beta decay and nucleon decay in [551], the potential for leveraging atmospheric neutrinos, especially in regards to the picture of the short-baseline anomalies and possible sterile neutrinos and nonstandard neutrino interactions (NSIs) has not been previously examined. Given the ringimaging capability of a Cherenkov detector, THEIA would be capable of an analysis similar to that performed by Super-Kamiokande [356]. THEIA would have similar sensitivity to Super-Kamiokande, with enhanced samples based on PID techniques -for example, by tagging neutrons more effectively. Yet, it would be many years before THEIA would equal the exposure used in the analysis of [356]. However, as discussed in [560], Hyper-Kamiokande may be able to enhance this picture by simultaneously employing beam and atmospheric neutrinos for both sterile neutrino and NSI searches, which also has complementarity with DUNE. A placement of THEIA in the LBNF beamline could further supplement these experiments, due to utilization of similar detection technology to Hyper-Kamiokande with a location in the same beam as DUNE. This would provide similar detectors in two different locations and beams (Hyper-Kamiokande and THEIA) as well as two different detection technologies in the same location and beam (DUNE and THEIA), which could open significant opportunities in further exploring parameter space and understanding systematic uncertainties across the three experiments. Further studies are planned to examine this possibility.

6.5.5. KM3NeT and ORCA

6.5.5.1. KM3NeT. Neutrino telescopes offer a complementary approach to investigate the anomalies found in short-baseline experiments. In this section we will focus on the capabilities of the KM3NeT detector [561], which is now under construction in the Mediterranean Sea. It should be also mentioned that KM3NeT's predecessor,



Figure 118. Left: median direction resolution as a function of true neutrino energy for showers and tracks. Right: probability distribution of the reconstructed energy as a function of true neutrino energy for upgoing showers and tracks (black lines indicate the quantiles). Reproduced from [562]. CC BY 4.0.

ANTARES, also in the Mediterranean Sea, has also looked for sterile neutrinos with the data gathered in ten years [358] as discussed in prior sections of this work.

The detection principle is as follows. Neutrinos (originated in cosmic sources or in the Earth's atmosphere) can interact in the surroundings of the detector and produce particles which will induce Cherenkov light that can be detected by an array of photomultipliers encapsulated in the so-called Optical Modules (OMs). The KM3NeT neutrino telescope is made of two subdetectors, ORCA and ARCA. ORCA is installed at a depth of 2500 m, off the French coast in Toulon. ARCA is located 3000 m deep, close to Porto Palo, in Sicily. Both use the same technology, being the main difference the distance between the OMs. ORCA is denser and will have 115 lines with 18 OMs each distributed in a volume equivalent to 8 MTon. The energy threshold estimated for this configuration is about 3 GeV. ARCA, with 230 lines, will have a sparser distribution of OMs, which will instrument about one cubic kilometer of water. The difference in size and density drives the physics focus of each configuration. ORCA is more suited for studies on atmospheric neutrinos and low energy astrophysical events, while ARCA's main goals are high energy neutrinos from astrophysical sources (with particular interest in Galactic sources given its geographical location). At the time of this writing, there are 10 lines of ORCA and 8 lines of ARCA already installed and taking data, with plans to install more lines in the following months.

6.5.5.2. ORCA. ORCA will precisely measure various parameters of the three-neutrino standard oscillation framework, namely the neutrino mass ordering, $\sin^2 \theta_{23}$, Δm_{13}^2 , and PMNS unitarity [562]. As a matter of fact, the first oscillation study with a partial configuration and less than one year of exposure finds that oscillations are preferred with a confidence level of 5.9 σ over 'no oscillations' [563]. The main analysis selects events induced by atmospheric neutrinos coming below the horizon and compares the observed energy, direction and topology distributions to predictions under different oscillation hypotheses. In the energy range 3–100 GeV, in which the ORCA detector is most sensitive to neutrinos, events are categorised in two event topologies: tracks, characterised by a long muon track (mostly from muon-neutrino CC interactions), and showers, characterised by events with no distinguishable tracks. ORCA is an excellent instrument to constrain oscillation parameters, primarily due to better energy and angular resolutions than other neutrino telescopes in this energy regime as shown in figure 118.

The presence of light-sterile neutrinos, enhanced by matter effects, would alter the observed rate of events in the GeV regime. For scenarios with large mass splitting and non-zero θ_{24} and θ_{34} , a deficit of high-energy track-like events is expected. This region of the phase space has been probed by neutrino telescopes, such as ANTARES [358] and IceCube/ DeepCore [140]. In addition, ORCA is well suited to test non-zero θ_{14} models, as the rate of showers around 20 GeV depends on this mixing angle.

First sensitivity studies have been performed using a joint binned likelihood for three samples (track, shower, and others), in which systematic uncertainties on the neutrino flux, detector response and standard oscillations were included [564]. Figure 119 shows the ORCA sensitivity to sterile mixing angles in different scenarios after three years of data taking. In the large Δm_{41}^2 limit, ORCA can set competitive constraints in θ_{24} and θ_{34}^{127} . It will be able to improve current limits on $\sin^2 \theta_{24}$, $\sin^2 \theta_{34}$, $\sin^2 \theta_{14}$ and $\sin^2 2\theta_{\mu e}$ for low values of the mass splitting¹²⁸, not yet constrained by cosmology or appearance/disappearance measurements. Furthermore, ORCA will be able to test the majority of the LSND and MiniBoone anomaly region as well as part of the Neutrino-4 allowed region.

Other BSM scenarios can be also tested with ORCA detector. In particular, competitive sensitivities to NSI and invisible neutrino decay can already be achieved with partial configurations of the detector [565, 566]. When completed, the ORCA detector will potentially become a leading tool for probing these models.

6.5.5.3. ARCA. ARCA will be able to observe neutrinos in the high-energy regime [567], exceeding IceCube's capabilities for up-going muon neutrinos as shown in figure 120. Several studies have shown that the presence of sterile neutrinos or NSI result in the disappearance of TeV-scale muon neutrinos passing through the Earth [103, 338, 349]. Recently, IceCube has set world-leading constraints in some of these models characterising atmospheric neutrinos in the few TeV range [151]. Preliminary studies are currently underway to assess ARCA sensitivities.

6.6. Beta and electron capture decay experiments

Beta decay and electron capture experiments are complementary to oscillation experiments in searches for sterile neutrinos. As described in section 4.1.6, the observed spectrum is a superposition of spectra corresponding to each mass state with endpoints shifted by the different neutrino masses. These experiments have the capability to search for sterile neutrinos ranging in mass from the sub-eV up to the MeV.

A new generation of experiments is envisioned to advance sensitivity to sterile neutrinos by orders of magnitude in the $\sin^2 \theta$ (see figure 121). Improvements over former experiments —particularly in source strength, detector threshold, and energy resolution - are enabling gains. Some proposed experiments (KATRIN, Project 8, DUNE) leverage existing detectors targeting other physics with minimal or no modification, while others experiments (HUN-TER, BeEST) require dedicated apparatuses.

6.6.1. KATRIN/TRISTAN. As described in section 4.1.6 KATRIN is sensitive to light sterile neutrinos, providing complementary information to oscillation-based searches. As the analysis interval of KATRIN is currently constrained to 40 eV below E_0 , only sterile neutrinos up to a corresponding mass of about $m_4 < 40$ eV can be searched for.

¹²⁷ In this analysis δ_{24} is kept free in the fit and θ_{14} and δ_{14} are fixed to zero.

¹²⁸ These analyses kept free all relevant sterile neutrino parameters.



Figure 119. The 99% C.L. ORCA sensitivity after three years of data taking for different combination of sterile parameters. Top left: $\theta_{24} - \theta_{34}$, with $\Delta m_{41}^2 = 1eV^2$ (ANTARES [358], IceCube [140], and SK [356]). Top right: $|U_{\mu e}|^2$ (Daya Bay +Bugey-3+MINOS+ [325], LSND [671], and MiniBooNE [33]). Bottom: $\sin^2\theta_{24}$ (SK [356], IceCube [151], MINOS+ [325], and cosmology [451]), $\sin^2\theta_{14}$ (Daya Bay +Bugey-3+MINOS+ [325], STEREO [656], Neutrino-4 [498], global fit [103], reactor fit [110], and cosmology [451]), and $\sin^2\theta_{34}$ (SK [356], IceCube [140], and cosmology [451]). Reproduced from [564]. CC BY 4.0.



Figure 120. Left: ARCA angular resolutions as function of the neutrino energy for ν_{μ} CC events. IceCube resolution was extracted from the point source analyses [672]. Right: ARCA effective area as function of the neutrino energy for the selected event sample. IceCube effective area was extracted from the diffuse analyses [673]. Reproduced with permission from [567]. CC BY-NC-ND 4.0.



Figure 121. Landscape of limits on sterile neutrinos from current and proposed beta decay and electron capture experiments.

To search for heavier sterile neutrinos, a larger range of the tritium β -spectrum must be measured. In particular, keV-scale sterile neutrinos would be of interest, as they are potential dark matter candidates [568]. However, indirect observations and cosmological considerations limit their mixing with active neutrinos to $\sin^2 \theta < 10^{-6}$.

As the count rate increases further away from the endpoint, the statistical uncertainty decreases. Assuming the source strength of KATRIN and a measurement time of about one year, a statistical sensitivity better than $\sin^2 \theta < 10^{-6}$ could be reached [569, 570]. However, a high-statistics measurement of the entire tritium β -decay spectrum poses a new technical challenge: electron rates exceeding 10^8 cps and several new systematic uncertainties become relevant when describing the experimental tritium spectrum far away from the endpoint.

The TRISTAN project is exploring the sensitivity of such a search and is developing a new silicon drift detector focal-plane array for KATRIN with more than 1000 pixels [571, 572]. This technology allows a measurement of a high β -electron flux with an energy resolution of 300 eV (FWHM) for 20 keV electrons, and large (3 mm) pixel footprints. Such a high-resolution detector would enable a differential measurement of the full tritium β -decay spectrum, in order to search for signatures of physics beyond the SM. The new detector will be installed after completion of the neutrino mass measurements.

A demonstration of this procedure has been performed by KATRIN using a low-intensity commissioning dataset from 2018 [365]. With only 0.5% the target tritium activity in this run, the detector could handle scans down to 1.6 keV below the endpoint (as opposed to the 40 eV below the endpoint for the normal operation). This small dataset is sufficient to set leading limits over the mass range $0.1 < m_4 < 1.0$ keV, demonstrating the potential of a full-activity dedicated measurement with the upgraded TRISTAN detector.

Project 8. Project 8 is a β -decay endpoint experiment with an aim to precisely measure the neutrino mass using cyclotron radiation emission spectroscopy (CRES) and atomic tritium as the source [573]. CRES is a frequency-based spectroscopic method that leverages the energy dependence of cyclotron radiation to perform a high resolution differential spectrum measurement over a wide energy range [574]. The use of atomic tritium provides a tight control on systematics by eliminating the uncertainty associated with the final states in the decay of molecular tritium [575].



Figure 122. Sensitivity of the upcoming phases of Project 8 experiment to light sterile neutrinos in the 3 + 1 framework. All curves, including reactor and BEST suggested parameters, are shown at 95 % C.L. The sensitivity is statistically limited and includes the current best knowledge of the systematics from energy resolution arising from thermal Doppler broadening, frequency to energy conversion, and variations in magnetic field. Control of systematic uncertainties leading to sensitivities in $\sin^2(2\theta) > 0.1$ (shown as a solid line) is expected to be straightforward. Further careful systematic control could enable experimental sensitivities down to $\sin^2(2\theta) \sim 0.01$ (shown as dashed lines). In the near future, with Phase-III, Project 8 aims to reach down to $\Delta m_{14}^2 \sim eV^2$ (C.L. 95%) and cover major portions of the reactor [1] and BEST [136] gallium anomaly suggested parameter spaces including BEST best-fit point. Phase-IV of Project 8 aims to completely cover the reactor and BEST gallium anomalies at high significance. Reproduced from [578]. CC BY 4.0.

Project 8 has undertaken a phased approach with each phase demonstrating a critical technological milestone. Phase-I of the experiment demonstrated CRES using internal conversion electrons from ^{83m}Kr [576]. Phase-II performed the first CRES-based molecular tritium endpoint measurement and placed an upper limit on neutrino mass [577]. Critical R&D for Phase-III of the experiment is ongoing with two parallel efforts to demonstrate: (i) CRES in larger volumes and (ii) the use of atomic tritium. Phase-III of the experiment plans to initially use molecular tritium followed by atomic tritium and has a projected ultimate sensitivity to the neutrino mass m_{β} of 0.4eV. Phase-IV is planned to be the ultimate Project 8 mass measurement experiment, with a goal to reach an unprecedented sensitivity to m_{β} of < 0.04 eV (90% C.L.) [578].

As a differential spectroscopy method, CRES allows Project 8 to make simultaneous searches for both active and sterile masses across the entire analysis window. Furthermore, the benefits of the CRES technique for neutrino mass measurement—namely low backgrounds, good resolution, and high event rates—also apply to a search for sterile neutrinos. Hence, a superior sensitivity to the direct neutrino mass also provides superior sensitivity to a sterile neutrino.

The sensitivity of Project 8 to sterile neutrinos is determined using the analytical neutrino mass sensitivity method as suggested in [579]. Figure 122 shows the upcoming phases of Project 8 to be capable of a competitive sterile neutrino search over several orders of magnitude in Δm_{14}^2 . The sensitivity estimates include statistical and systematic uncertainties as noted in the caption. The experiment's sensitivity to higher Δm_{14}^2 is primarily limited by



Figure 123. Projected upper limits (95% C.L.) for $\sin^2 \theta = |U_{e4}|^2$ as a function of heavy neutral lepton mass $m_{heavy} = m_4$ for DUNE. 'DUNE-ALL' refers to a limiting dataset containing every single ³⁹ Ar beta decay occurring during detector operations (2 × 10¹⁶ electrons), while 'DUNE-EXT' refers to a baseline dataset obtained using external triggers and full detector readout without special localized triggering for low-energy activity in the detector (1 × 10¹³ electrons). All upper limits shown are purely statistical (no theoretical/experimental systematic uncertainties included) and assume zero background.

the efficiency of the cyclotron frequency detection method for lower energy electrons since a higher value of m_4 would manifest as a kink at a lower energy in the β -decay spectrum. In generating sensitivity curves in figure 122, the efficiency was assumed to be well-understood for energies tens of eV below the endpoint. This is a fair assumption based on Project 8's ability to quantify the efficiency of the complex Phase-II detector over 2.5 keV below the endpoint [577]. The two different detection methods being investigated for upcoming phases are yet to demonstrate the control of efficiency over the wide energy range, but are expected to have lower complexity in efficiency than in Phase-II. Project 8 through the differential β -decay spectrum measurement using CRES thus provides a promising avenue to search for light sterile neutrinos in the near future.

DUNE. Large liquid argon detectors making use of atmospheric argon, associated with ³⁹Ar beta decay activity of roughly 1 Bq kg⁻¹ [580], will enable the probing of m_4 values below the ³⁹Ar beta decay end point of Q = 565 keV. By utilizing very large detectors with large volumes of liquid argon, a substantial amount of beta decays can be detected as to enable sensitive measurements of $|U_{e4}|^2$ at larger m_4 values than for ³H. Given that liquid argon detectors (such as liquid argon provides both the source and the detector for such measurements, in contrast to measurements using ³H.

DUNE [581–583] will use massive LArTPCs to study accelerator neutrinos (~1 GeV) undergoing flavor oscillations over a long baseline (~1300 km) in order to probe leptonic CP violation. A search for heavy neutral leptons can also be carried out using ionization charge measurements of ³⁹Ar beta decays in the DUNE far detector, enabling sensitivity to $|U_{e4}|^2$ in the 20 keV to 450 keV mass range. Figure 123 shows projected statistical upper limits (95% C.L.) for $|U_{e4}|^2$ at DUNE as a function of m_4 . Recording the full DUNE ³⁹Ar beta decay dataset toward maximum sensitivity to $|U_{e4}|^2$ (better than 10⁻⁶) requires substantial



(magnetic field coils and shielding not shown)

Figure 124. The HUNTER spectrometer. Reproduced from [591]. © IOP Publishing Ltd. All rights reserved.

velopment of the trigger system at DUNE, including selectively recording low-energy activity within "regions of interest"inside of the detector; this capability is currently under development. Reconstruction of ³⁹Ar beta decays has been previously carried out at MicroBooNE [584], a LArTPC neutrino experiment at Fermilab, demonstrating that low thresholds (roughly 100 keV) and good energy resolution from low TPC noise levels (roughly 50 keV) are achievable in large LArTPC detectors [585]; more comprehensive studies at ProtoDUNE-SP [586, 587] are currently in progress. Further studies on the impact of ³⁹Ar beta decay spectrum theoretical uncertainties [588], experimental systematic uncertainties, and radiological backgrounds in the DUNE far detector are also in progress. While these additional considerations may lower the sensitivity to $|U_{e4}|^2$ considerably, given the current global limits on $|U_{e4}|^2$ of 10^{-2} to 10^{-4} in the relevant range of m_4 values [222, 589, 590], significant improvement is expected.

6.6.2. HUNTER. The HUNTER Collaboration brings together techniques from high energy, nuclear, and AMO physics to perform a laboratory sterile neutrino search aimed at the mass range 10–300 keV [591]. It brings together groups from Temple University, UCLA, University of Houston, Racah Institute of Physics, and Princeton University to pursue this physics.

The neutrinos are emitted in the electron capture (EC) decay of ¹³¹Cs atoms captured in a magneto-optical trap and cooled to milliKelvin temperatures where they are essentially at rest. In electron capture, the initial final state is an excited atom recoiling against the emitted neutrino—for a given neutrino mass, the neutrino is mono-energetic. All the observable particles in the EC decay and subsequent de-excitation of the recoiling atom are reconstructed. These particles are a ¹³¹Xe ion, an x-ray, and one or more Auger electrons, each of which requires different measurement techniques. Figure 124 shows a CAD drawing of the HUNTER apparatus. The meter-long horizontal arms of the vessel contain



Figure 125. Left: simulated spectrum (blue histogram) of one year of livetime including a hypothetical 60 keV/ c^2 sterile neutrino with $\sin^2 \theta = 3 \times 10^{-4}$. Right: HUNTER's sensitivity for three phases of the experiment, compared to other laboratory limits. Reproduced from [591]. © IOP Publishing Ltd. All rights reserved.

spectrometers for the ion (left) and electrons (right). In the ion spectrometer, the electrostatic field focuses the ions onto an MCP which allows determination of the ion's vector momentum using the impact location and time of flight. The electron spectrometer is similar, but requires an axial magnetic field to contain the electrons. X-rays are detected via scintillator panels, a signal that triggers the readout and provides the start time for the time of flight measurements.

The left panel of figure 125 is a simulated spectrum including the apparatus resolution and the major backgrounds. A simulated neutrino signal within the reach of the first phase of HUNTER with a year of live-time is shown. The right panel shows the reach of the initial HUNTER configuration and two potential upgrades.

6.6.3. BeEST. The BeEST (Beryllium Electron-capture with Superconducting Tunnel junctions) experiment is a model-independent search for sub-MeV sterile neutrinos using implanted radioactive ⁷Be atoms into high-rate superconducting quantum sensors [592]. The BeEST concept uses momentum reconstruction of the low-energy ⁷Li daughter atom following EC decay of ⁷Be to infer direct mass information of the neutrino. ⁷Be ($T_{1/2} = 53.22$ (6) days [593]) is a pure EC decaying isotope, and the ideal case for neutrino studies via momentum reconstruction due to its large decay energy $Q_{EC} = 861.89(7)$ keV [594], relatively high maximum value for the recoil kinetic energy $T_{D(max)} = 56.826(9)$ eV, and simple atomic and nuclear structure [593]. Due to the explicit neutrino mass dependence on the recoil kinetic energy, the existence of a heavy mass state, m_i , would cause the nuclear recoils to have a lower kinetic energy, and the relative fraction of these shifted events to the total determined by the mixing with the ν_e flavor, $|U_{ei}|^2$.

The BeEST experiment uses superconducting tunnel junction (STJ) sensors to detect the low-energy radiation following EC decay. STJs are high-speed quantum sensors that were originally developed for high-resolution x-ray spectroscopy in astronomy and material science [595]. STJs are a type of Josephson junction that consists of two superconducting electrodes separated by a thin insulating tunneling barrier. The absorption of radiation in one of the electrodes breaks the Cooper pairs of the superconducting ground state and excites free


Figure 126. (left) Image of a 32-pixel Al-based STJ array in preparation for Phase-IV of the BeEST [592] (middle). The low-energy ⁷Li nuclear recoil spectrum for 28 d of acquisition from a single pixel counting at low-rate [371]. The spectrum generated by a hypothetical 300 keV sterile neutrino signal with 1% mixing is shown in red (dashed) (right). Current and projected limits for the BeEST and SuperBeEST experiments compared to existing laboratory decay limits (shaded areas) and the TRISTAN projection. Reprinted (figure) with permission from [371], Copyright (2021) by the American Physical Society.

excess charge carriers above the superconducting energy gap Δ in proportion to the absorbed energy. This results in exceptionally high energy resolution (~1 eV) for low-energy radiation relevant to nuclear recoils. Each STJ detector pixel is able to operate at rates up to 10^4 counts s⁻¹ with exceptionally high predictability on the detector response [596], making them ideal for searches of this type.

The ⁷Be⁺ rare isotopes are implanted into the STJ detectors through Si collimators at TRIUMF's Isotope Separator and ACcelerator (ISAC) facility in Vancouver, Canada at an energy of 25 keV. The ⁷Be⁺ beam is produced using the isotope separation on-line technique via spallation reactions from a 10 μ A, 480 MeV proton beam incident on a stack of thin uranium carbide targets. Following implantation of the rare isotope, The ⁷Li recoil spectrum from the decay of ⁷Be is measured at Lawrence Livermore National Laboratory (LLNL) with the STJ detector at a temperature of ~0.1 K in a two-stage adiabatic demagnetization refrigerator (ADR) and the signals are read out with a specialized current-sensitive preamplifier [597]. For a precision energy calibration, the STJs are simultaneously exposed to 3.49965(15) eV photons from a pulsed Nd:YVO ₄ laser triggered at a rate of 100 Hz [371, 598]. The measured ⁷Li recoil spectrum contains four peaks that result from the two nuclear and two atomic processes following EC decay. These include one for *K*-capture decay to the nuclear ground state (K-GS), one to the excited state of ⁷Li (K-ES), and two for the corresponding *L*-capture decays (L-GS and L-ES, respectively) (figure 126).

Due to the relative simplicity of the ${}^{7}\text{Be} \rightarrow {}^{7}\text{Li}$ EC decay system, the spectral response is able to be precisely evaluated using modern theoretical tools. The atomic de-excitation and auto-ionization (electron shake-up and -off) effects that follow EC decay generate higher-order features in the spectrum and are accessible to *ab initio* atomic theory. Additionally, the sequestration of beryllium in the sensor material generates small (eV-scale) chemical shifts in the EC decay energies. To this end, density functional theory is used to model the electronic structure of beryllium and lithium atoms in different atomic environments of a polycrystalline absorber film of the sensor to provide a further improvement of sensitivity to new physics. This work is already in an advanced stage, and will be employed to provide improved limits to the high-statistics data from the BeEST experiment planned for upcoming phases, and may allow for the extraction of significantly improved limits in the 10–100 keV mass range.

The BeEST experiment has completed its proof-of-concept (Phase-I) in 2020 with the first demonstration of high-resolution nuclear recoil detection with STJs [599]. Following this, the first low-activity (10 Bq) data set using a single tantalum-based STJ detector was

taken for 28 d—including the use of precision *in situ* laser calibration of the sensor response. The resulting Phase-II data allowed for the first extraction of an upper limit for $|U_{ei}|^2$ from the BeEST experiment in the 100–850 keV mass range [371], and improved upon previous decay measurements by up to an order of magnitude (figure 126). Within the next 5 years, the BeEST will complete its four phases of the experiment and achieve limits approaching $|U_{ei}|^2 \approx 10^{-7}$ in the few-hundred keV mass range, as described further in [592]. Following Phase-IV of the BeEST, a dramatic improvement in sensitivity is planned using large arrays (10⁴ pixels) of STJs with new materials. This development is ongoing, with the goal to probe relative couplings to the electron neutrino flavor of $|U_{ei}|^2 \leq 10^{-9}$. The current and projected limits for the BeEST experiment are presented in figure 126 and incorporate known and simulated detector responses under conservative assumptions on the atomic and materials theory work described above.

6.6.4. PTOLEMY. The PTOLEMY experiment is designed to look for relic neutrinos by means of neutrino capture on a tritium target. Searching for spectral distortions near the endpoint with its very intense source, the experiment also projects sensitivity to eV-scale sterile neutrinos [600].

6.6.5. Double beta decay. While nuclear beta-decay has become a valuable tool for searching for MeV-scale sterile neutrinos, the relatively rare double beta-decay process is also sensitive to the presence of sterile neutrinos. Often thought of solely as a background to searches for neutrinoless double-beta decay $(0\nu\beta\beta)$, double beta decay occurs when an atom undergoes a simultaneous emission of two electrons and two antineutrinos. This process leads to a continuous electron energy spectrum that cuts off at the characteristic Q-value for that isotope. The region of this spectrum that is typically studied in detail is the end-point because this is where the signal of neutrinoless double-beta decay will populate. As discussed throughout this section, the presence of heavy sterile neutrinos will lead to deformation of the beta-decay spectra, and this is also true for double beta-decay [601].

In the presence of a heavy sterile neutrino, the kinematics of standard two-neutrino double beta decay is modified in both the energy and angular distribution. The modifications to the summed electron energy is spread across a relatively large range of energies but lead to an overall modulation of the spectrum at low energies, as shown in figure 127. In addition to the impact on the energy of the electrons, the angular distribution is also strongly impacted by the presence of heavy sterile neutrinos.

Constraints from current and planned neutrinoless double beta decay experiments have been explored in [601]. These can be separated into experiments sensitive to just energy that have taken data (CUORE [602], EXO-200 [603], GERDA [604], KamLAND-Zen [605], and the MAJORANA DEMONSTRATOR [606]) and planned to take data (CUPID [607], LEGEND [608], and nEXO [609]) and those sensitive to the angular distributions that have taken data (NEMO-3 [610]) and those planned (SuperNEMO [611]). These can be seen in figure 128, where limits are placed on the sterile neutrino mass and the mixing strength. In figure 128, limits coming from current neutrinoless double beta decay experiments use exposures of O(100) kg-years, which corresponds to $10^4 \ 2\nu\beta\beta$ events, while future next-generation experiments (those targeting the inverted hierarchy region of the neutrinoless double-beta decay phase space) use exposures of O(10^3-10^4) kg-years, with $10^6 \ 2\nu\beta\beta$ events. In the far future, experiments that wish to target the normal hierarchy region of the neutrinoless double beta decay will require exposures of O(10^6) kg-years [612], this increase in the number of events will enable further explorations of this sterile neutrino phase-space.



Figure 127. The summed energy distribution from the standard $2\nu\beta\beta$ spectrum (solid) and the impact from the presence of a 1 MeV sterile neutrino (dashed) for ¹⁰⁰ Mo (purple) and ¹³⁶ Xe (blue). Reproduced from [601]. CC BY 4.0.



Figure 128. The constraint offered by current $0\nu\beta\beta$ experiments sensitive to only the energy (solid blue) and those also sensitive to the angular distributions (solid red). Future (next-generation) experimental sensitivity are shown in dashed lines. Reproduced from [601]. CC BY 4.0.

Leveraging data from $0\nu\beta\beta$ experiments to search for the presence of sterile neutrinos and other BSM processes [613] demonstrate the power diverse experimental data can offer for exploring and searching for anomalies.

6.7. Meson factories

6.7.1. NA62. Rare meson decays are well-known for providing some of the strongest constraints on light dark sectors. Two experimental efforts to measure the flavor-changing NC decay of kaons to neutrinos, $K^+ \rightarrow \pi^+ \nu \overline{\nu}$, are underway: NA62, with a K^+ beam [614, 615], and KOTO, with a K_L beam [616]. In this section we discuss some prospects of NA62 to

The NA62 experiment is a kaon-decay-in-flight experiment located at CERN [617]. To achieve its primary goal, NA62 aims to have a total exposure of $\mathcal{O}(10^{13})$ kaon decays over a few years of run time. The detectors are designed for sub-percent momentum resolution on the kaon and final charged-track momenta. Kaons have an average of 75 GeV momentum and their decays in flight throughout a distance of $\mathcal{O}(65)$ m are detected by a combination of spectrometers arranged in a cylindrical fashion around the beam.

NA62 can also perform several searches for new physics by searching for exotic kaon decays. Light particles have been searched for in $K^+ \rightarrow \ell^+ N_{inv}$ [618–620], as well as in $\pi^0 \rightarrow (\gamma) + inv$ [621, 622]. In the context of short-baseline anomalies, NA62 has a world-leading sensitivity to HNLs invoked in some explanations of the MiniBooNE and LSND anomalies. The current limits on long-lived or invisible HNLs produced in $K_{2\ell}$ decays span the mass region between 100 $\leq m_N < m_K - m_\ell$.

Two channels are of particular interest for HNL searches:

- 1. $K^+ \rightarrow \ell^+$ N, where N is long-lived or invisible,
- 2. $K^+ \to \ell^+ (N \to \nu \ell^+ \ell^-)$, where N decays inside the detector.

The first channel has already been searched for in both flavors, $\ell^+ = e^+ and\mu^+$ [618–620]. The limits reach mixing angles as low as $\mathcal{O}(10^{-9})$, in the region of interest for the Type-I seesaw mass mechanism. The strategy in this case consists in searching for the presence of an invisible resonance in $M_{inv}^2 = (p_K - p_\ell)^2$, therefore it is mostly sensitive to long-lived or invisibly-decaying HNLs. This constraint applies to the models discussed in section 3.2.3.

The second channel with three charged tracks is instead sensitive to short-lived HNLs. This possibility was first discussed in the context of a transition magnetic moment model [623], discussed in section 3.2.1. In that model, the electromagnetic decays of the HNL with an off-shell photon, $N \rightarrow \nu(\gamma^* \rightarrow e^+e^-)$, could dominate the SM rate. The measurement by BNL-E865 [624], $\mathcal{B}(K^+ \rightarrow \mu^+\nu e^+e^-) = (7.06 \pm 0.3) \times 10^{-8}$ allowed to set a constraint on the mixing and branching ratio of the HNL of $|U_{\mu N}|^2 \times \mathcal{B}(N \rightarrow \nu \gamma) < 0.5 \times 10^{-6}$ [623]. This constraint, however, only applies to HNLs with masses $m_N > 145$ MeV due to experimental cuts on $m_{ee} > 145$ MeV, designed to suppress π_D^0 decays. A more precise measurement was performed at NA48, finding $\mathcal{B}(K^+ \rightarrow \mu^+\nu e^+e^-) = (7.81 \pm 0.23) \times 10^{-8}$, applicable to the region $m_{ee} > 140$ MeV.

Three charged track decays are also sensitive to HNLs that decay electromagnetically through a dark photon, Z', such as in the models discussed in section 3.2.2. In this case, the HNLs can decay to either $\nu e^+ e^- or \nu \mu^+ \mu^-$. [208] derived constraints in a specific dark neutrino model with pseudo-Dirac HNLs. The reach of NA48 to the $|U_{\mu4}|^2$ mixing element was around $\mathcal{O}(10^{-8})$ or better. A dedicated search at NA62 can improve upon that reach by one to two orders of magnitude. Backgrounds from π_D^0 decay present the biggest challenge for this search, but can be suppressed by the requirement that the measured $M_{\nu ee}^2 \equiv (p_K - p_\ell)^2 = m_N^2$. In addition, if the mediator is produced on-shell, then a resonance search in the $\ell^+ \ell^-$ invariant mass can be performed. In the case of a signal, NA62 could measure both the dark photon and the HNL mass. The NA62 sensitivity to the dark sector model in [208] is shown in figure 129 under the assumption of no backgrounds. Based on [625], for $K^+ \to \mu^+ \nu e^+ e^-$, a total of $N_K^{\text{Fid}} = 2.14 \times 10^{11}$ fiducial kaon decays were assumed with an acceptance of $A_\beta = 4\%$, while for $K^+ \to \mu^+ \nu \mu^+ \mu^-$, a total of $N_K^{\text{Fid}} = 7.94 \times 10^{11}$ and $A_\beta = 10\%$ were assumed.



Figure 129. Limits on $|U_{\mu}|^2$ as a function of the mass of the HNL *N* (left) and the dark photon *A'* mass (right). The NA62 sensitivity to the prompt decays of HNLs in $K^+ \rightarrow \mu^+ \nu e^+ e^-$ and $K^+ \rightarrow \mu^+ \nu \mu^+ \mu^-$ searches is shown as a solid black line. Figure from and based on the dark sector model of [208]. Reproduced from [208]. CC BY 4.0.

6.8. Collider experiments

6.8.1. FASER ν and FLArE. The flux of broadband neutrinos from the LHC with energies around ~100 GeV to ~1 TeV in the forward direction provides a new opportunity to look for neutrino oscillations. Existing experiments such as FASER ν [626] and SND@LHC [627] as well as proposed experiments which go under the umbrella term of Forward Physics Facility (FPF) will be sensitive to sterile neutrino oscillations. Given a typical baseline of 600m and typical energies in the 100 GeV to 1 TeV range, this corresponds to sensitivities to $\Delta m_{41}^2 \sim 1000 \text{ eV}^2$, or $m_4 \sim 30 \text{ eV}$. While some existing constraints apply at this mass range in the oscillation averaged limit, there are no oscillation probes at this L/E providing a new direct test of oscillations for larger Δm^{2*} s than are usually considered. The forward physics program at the LHC also benefits from the production of all three flavors of neutrinos with hierarchical production rates that differ each by $\gtrsim 1$ order of magnitude. In addition to having all three flavors available at the source, FASER ν and SND@LHC both have flavor discrimination capabilities among the three flavors allowing for, in principle, probes of all 9 oscillation channels, subject to backgrounds and flux uncertainties.

One major challenge is the presence of significant flux uncertainties which affect the normalization and, more importantly, the shape. The relative contribution to the neutrino flux from different particles is rather poorly understood [628] and these shape effects could conceivably mimic a neutrino oscillation signature [629]. Thus significantly more theoretical work to understand these fluxes is essential to use this unique neutrino environment to probe neutrino oscillations.

Nonetheless, it is possible to estimate the sensitivity to sterile neutrino oscillations at FASER ν and FLArE, a proposed LAr detector in the forward direction at the LHC [630]. The most sensitive channel relative to existing constraints is the ν_{μ} disappearance channel which shows sensitivity at the $\Delta m_{41}^2 \sim 100 - 1000 \text{ eV}^2$ range down to mixings $|U_{\mu4}|^2 < 10^{-2}$, better than existing constraints from [102] which are dominated by MINOS/MINOS+ [139] and MiniBooNE [275]. The sensitivity for FASER ν in the upcoming LHC run at 150 fb⁻¹ and FLArE-10 (10 fiducial tons of LAr) with 3 ab⁻¹ in the HL-LHC is shown in figure 130. The primary uncertainty is the flux uncertainty. This is accounted for in a fairly conservative fashion by including an estimate of the impact of shape effects by varying the flux across the range of predictions from different models [628] with an associated 1 σ pull term. This sensitivity is then



Figure 130. The expected sensitivity to sterile neutrino oscillations in the ν_{μ} disappearance channel at the LHC using Asimov and Feldman–Cousins for FASER ν at the upcoming LHC run and a proposed 10-ton LAr detector in a future HL-LHC run. For comparison, the existing oscillation averaged constraints coming mostly from MINOS+ and MiniBooNE are also shown. Reproduced from [630]. CC BY 4.0.

calculated using the Feldman–Cousins procedure [320] at 95% CL including the flux systematic uncertainty. The median sensitivity is estimated with the Asimov method.

It is anticipated that these LHC experiments could have sensitivity to sterile oscillations for the other channels as well, although those will depend on the details of the flux uncertainties which are still quite large and difficult to systematically quantify.

7. The path forward to resolving the anomalies

7.1. Reflecting on the path of the past decade

Sections 2 and 6 serve to demonstrate that a significant portion of the neutrino community's efforts over the past decade has been driven by the goal of establishing or refuting the existence of light sterile neutrinos motivated by these anomalies. To this end, following the requirements identified in [1] has led to a broader understanding of viable interpretations of the anomalies and strengthened experimental efforts—and experimental capabilities—in that direction. Notably, the requirement to probe the anomalies with multiple and orthogonal approaches (accelerator-based short/long-baseline, reactor-based short-baseline, atmospheric neutrinos, and radioactive source) in the same spirit as employed for neutrino oscillations has now been realized through recent, ongoing, or impending experimental programs:

- 1. The development of new radioactive sources and detectors for improved tests of the Gallium Anomaly has been pursued and realized in the form of the BEST experiment.
- The Reactor Antineutrino Anomaly and subsequent reactor-based activities and new results have placed a required emphasis on oscillation-testing short-baseline reactor experiments and on improved understanding of reactor neutrino fluxes.
- 3. The community has just begun a comprehensive multi-channel/multi-baseline accelerator-based short-baseline program to search for 3+N oscillations while directly

addressing the MiniBooNE anomaly both in regards to oscillatory and non-oscillatory solutions.

- 4. A direct test of the LSND Anomaly using an improved decay-at-rest beam facility and experimental arrangement has just begun in the form of the JSNS² experiment.
- 5. Beyond direct anomaly tests, many alternate techniques/facilities, including direct neutrino mass measurements, long-baseline oscillation experiments, and atmospheric and astrophysical neutrino experiments, have been applied to the sterile neutrino explanation of the anomalies.

7.2. Primary focuses of the next decade

As the question of light sterile neutrino oscillations is further explored over the next several years, the community's efforts should be directed toward disentangling the plethora of possibilities that have been identified over the past ten years as viable interpretations of the experimental anomalies in the neutrino sector. The goal of these collective efforts will be to validate and solidify our understanding of the neutrino sector. Regardless of what the ongoing and upcoming experiments observe—be it a deviation from the three-neutrino picture or otherwise —the community should be prepared to address how to put these anomalies to rest or adequately distinguish between different interpretations. We summarize the main experimental, analysis, and theory-driven thrusts that will be essential to achieving this goal as follows:

- Cover all anomaly sectors: given the fundamentally unresolved nature of all four canonical anomalies, it is imperative to support all pillars of a diverse experimental portfolio—source, reactor, decay-at-rest, decay-in-flight, and other methods/sources—to provide complementary probes of and increased precision for new physics explanations.
- 2. **Pursue diverse signatures:** given the diversity of possible experimental signatures associated with allowed anomaly interpretations, it is imperative that experiments make design and analysis choices that maximize sensitivity to as broad an array of these potential signals as possible.
- 3. **Deepen theoretical engagement:** priority in the theory community should be placed on the development of new physics models relevant to all four canonical short-baseline anomalies and the development of tools for enabling efficient tests of these models with existing and future experimental datasets.
- 4. Openly share data: fluid communication between the experimental and theory communities will be required, which implies that both experimental data releases and theoretical calculations should be publicly available. In particular, as it is most likely that a combination of measurements will be needed to resolve the anomalies, global fits should be made public, as well as phenomenological fits and constraints to specific data sets.
- 5. Apply robust analysis techniques: appropriate statistical treatment is crucial to quantify the compatibility of data sets within the context of any given model, and in order to test the absolute viability of a given model. Accurate evaluation of allowed parameter space is also an important input to the design of future experiments.

7.3. Discussion and elaboration

The following section aims to provide further context regarding next decade's points of primary focus above.

Full experimental coverage of all anomaly sectors. To probe as broad a range of potential BSM physics explanations for the short-baseline anomalies as possible, data from diverse final states, energies, and sources will be crucial for disentangling different possible contributing effects, as each effect may or may not manifest itself differently in specific experimental regimes. Many such examples exist: for example, short-baseline accelerator and matter-resonance-affected atmospheric datasets are disparately impacted by sterile NSI effects, reactor and accelerator experiments have differing levels of connection to higher-mass hidden particle sectors, and decay lengths of unstable heavy sterile neutrinos would differ substantially at decay-at-rest and decay-in-flight experiments. These examples clearly emphasize the substantial value added by acquiring data sets from all available experiment types. Beyond this, the lack of a single theoretical framework capable of uniting the experimental anomalies indicates the clear continued need for enhanced, direct tests of each anomaly. Thus, the neutrino community should continue to strive in the coming decade to support the collection of high-precision datasets from all available sources-accelerators, atmospheric, reactors, radioactive sources, and even astrophysical sources-for testing anomaly explanations.

Diverse signatures: enhanced experimental versatility. Recognizing the broad spectrum of possibilities behind the observed anomalies in the neutrino sector, existing and future experiments should rise to the challenge of broadening their physics sensitivity scope: the range of BSM physics models their experiment is originally designed to probe. Some examples of this exist among current experiments: MicroBooNE has targeted potentially oscillation-driven (electron-like) anomalous ν_e signatures as well as anomalous photon-like signatures; PROSPECT performed a separate search for cosmic boosted dark matter; and MiniBooNE itself performed alternative analyses to probe its anomalous result through, for example, adjustments to the BNB beamline (through deployment of a charged pion absorber or 'beam dump' mode running). Diverse new physics testing capabilities should be deemed increasingly important in the coming decade of short-baseline experiments with the disfavoring of the simple 3 + 1 sterile neutrino picture, and the rise of more diverse BSM explanations for the anomalies; see section 3. Enhanced diversity can be achieved on both the experimental and analysis sides in many ways, including optimizing/enhancing experimental designs, implementing new BSM model generators in experiment Monte Carlo simulation frameworks, and designing new selection and analysis techniques to probe less-explored neutrino flavor/interaction channels.

Diverse signatures: specific models versus inclusive categories. In the coming decade, experiments should strive to investigate experimental anomalies both within the context of specific physics models (and thus testing specific model parameters), as well as more inclusively, in the form of model-agnostic and phenomenology-driven searches. The former allows cross-comparisons of results from different experiments—such as comparisons of suggested and excluded 3 + 1 parameter space regions between reactor-based and source-based ν_e disappearance tests—while the latter provides more qualitative and phenomenological information to inform future theoretical developments and experimental searches—such as MicroBooNE's generic approach in searching for an electron-like excess. With the increasing interest of the community in viable explanations for these anomalies (section 3), the ability of experimental collaborations to provide results of BSM searches in as model agnostic and inclusive a manner as possible will allow for broader coverage of an evergrowing theory model landscape. Wherever possible, particle and event reconstruction and identification capabilities should be improved in order to discriminate among different interaction final states, by examining particle types, multiplicities, and kinematics.

Theoretical model and tool development. Over the last decade, the theory community has contributed to resolving the anomalies puzzle on four complementary fronts: performing improved calculations of relevant SM backgrounds, cross sections, and neutrino fluxes, proposing new models that aim to explain the anomaly in light new data, developing and applying tools that aim that are useful for phenomenological and experimental analyses, and placing anomaly explanations in contex with broader questions in particle physics, such as the nature of dark matter and the origin of neutrino masses. It is imperative to continue and deepend pursuit of all of these axes in the next decade. Activities that should be expended to achieve this aim include further support and development of SM neutrino interaction and BSM event generators, and expansion of computational tools for global data fitting (such as Globes [522, 523]), efficient parameter space exploration (such as Markov Chain Monte Carlo techniques [631] and SBNFit), and fast oscillation probability calculation [523, 632–642].

Open data, data sharing, and joint analysis. It is essential that anomaly-relevant results and datasets used to generate them be made publicly available in a format that is easily accessible and versatile. This would allow reproduction of results within a particular physics model but as well as reinterpretation of results within the context of other models, which could be pursued by the theory and phenomenology community to guide more detailed experimental follow-ups in a timely way. This transparency can apply to experimental Monte Carlo simulation as well as experimental data; access to the former could potentially be available earlier than experimental data and would allow the phenomenology community to springboard off the extensive work by the experimental community in driving forward new BSM searches. Parameterizations or covariance matrices that represent systematic uncertainties, as well as χ^2 surfaces, could also be valuable for similar reasons. Public databases (e.g. hepdata, arXiv) would be particularly valuable in disseminating this information.

Applying robust techniques: improved oscillation analyses methodologies. As the statistical precision and breadth/purity of exclusive flavor/interaction channel datasets increase in short-baseline experiments in the coming years, fewer approximations should be employed in testing specific sterile oscillation models. For example, within the context of 3+N oscillations, upcoming accelerator-based searches should consider refraining from performing exclusive oscillation channel measurements (e.g. searching for $\nu_{\mu} \rightarrow \nu_{e}$ appearance while ignoring ν_{e} and/or ν_{μ} and/or NC background disappearance effects), while short-baseline reactor spectral ratio experiments should consider the impacts from neutrino wave packet decoherence on sterile neutrino testing capabilities. While these complexities may in some cases reduce the claimed sensitivity of the involved analysis, it will also provide enhanced clarity as to which BSM models or phase space regions have actually been unambiguously put to rest. Alternate methods of results presentation should also be considered: for example, in the case of 3+N oscillations, final multidimensional parameter spaces should be presented in the form of slices of relevant space, or could be reduced in dimension using profiling or marginalization techniques, albeit at the cost of reduced information.

Applying robust techniques: standardizing model parameterization and presentation. A common language has been established over the years related to the 3 + 1 model and other oscillation models; however, this is not yet the case for several new theory proposals to resolve the anomalies. Different experimental and phenomenological efforts aimed at probing similar families of new models should strive to achieve common notation and language, so that small complicating variations can be reduced and analyses and results can be more easily and directly compared. The use of static benchmarks (single parameter value) in new model studies allows easier comparison between experiments and design of future detectors, while

extensive scans in a multidimensional parameter space provide more complete information on the model.

Applying robust techniques: statistical approaches. As more data is available and the amount of models that need to be tested increases, statistical techniques and procedures must expand and improve. Experiments and theorists should strive when possible to implement frequentist and/or Bayesian statistical methods, providing complementary information and unbiased reported confidence intervals. For experimental studies where asymptotic conditions, such as those of Wilk's theorem, do not hold, test statistic distributions should be explicitly checked. Lacking access to high-throughput computing resources, simple, low-resource methods of unbiased confidence interval setting, such as Gaussian CL_s , are available; intervals set using Wilks' theorem may be acceptable, but should be stated explicitly. For Bayesian analyses, experiments should clearly state chosen priors and rationale; results should be presented with multiple priors when the prior effect is relevant.

8. Conclusions

The identification of a light sterile neutrino program as a high priority through the previous Snowmass process [1] has resulted in the realization of a broad and vibrant experimental program spanning from radioactive source experiments to reactor-based and accelerator-based searches for light sterile neutrino oscillations. The experimental data collected over the past decade are undeniably challenging the simplest theoretical interpretation of experimental anomalies in neutrino physics—namely, that of a light sterile neutrino within the context of a 3 + 1 model—as a single underlying source of the outstanding short-baseline anomalies. This has further motivated theoretical developments in search of an ultimate solution, as well as a diversified experimental program; the vast body of theoretical work produced over the past decade compels the community to keep an open mind on both conventional and BSM possibilities that may lie at the heart of these puzzles.

Despite significant progress in the form of new experimental measurements and theoretical development, the short-baseline experimental neutrino anomalies remain unresolved. While uncertainties in reactor neutrino flux predictions and ν_e cross-section predictions at low energy persist and complicate this picture, experimental anomalies in the radioactive source, reactor, pion decay-at-rest, and pion decay-in-flight neutrino sectors still persist at the level of $1-5\sigma$, and therefore remain as outstanding questions. This suggests that our understanding of the ESM is, at best, incomplete.

Taken at face value, despite spanning a broad energy range of a few MeV to a few GeV, and observed with a variety of different neutrino sources and detector technologies, these anomalies all share the common theme of being associated with electron and muon (anti) neutrino flavors. A big question is whether similar anomalies will be observed with tau neutrinos, as future facilities with higher energy and intensity begin to probe that sector, or through neutral-current inclusive searches.

The neutrino community recognizes the need to advance the study and exploration of an increasing-in-scope and rich phenomenology that will be accessible with current and future-generation experimental facilities. Efforts in this direction will further provide opportunities for synergy with other fields within particle physics, astrophysics, and cosmology.

Acknowledgments

The co-authors, including the editors and contributors to this White Paper, would like to thank the following agencies and foundations for their support: U.S. National Science Foundation; U.S. Department of Energy, Office of Science; Alfred P Sloan Foundation; Gordon and Betty Moore Foundation; UKRI Science and Technology Facilities Council; Spanish Ministerio de Ciencia e Innovación; European Union's Horizon 2020; Ministry of Science and Technology of Taiwan; Korea Institute for Advanced Study; Marie Sklodowska-Curie Foundation; and Swedish Research Council.

Data availability statement

No new data were created or analysed in this study.

ORCID iDs

- M A Acero (1) https://orcid.org/0000-0002-0835-0641
- C A Argüelles https://orcid.org/0000-0003-4186-4182
- M Hostert @ https://orcid.org/0000-0002-9584-8877
- K J Kelly https://orcid.org/0000-0002-4892-2093

B R Littlejohn ⁽¹⁾ https://orcid.org/0000-0002-6912-9684

- P Machado () https://orcid.org/0000-0002-9118-7354
- W Pettus () https://orcid.org/0000-0003-4947-7400
- M Toups (b) https://orcid.org/0000-0001-6584-9011
- M Ross-Lonergan https://orcid.org/0000-0001-7012-8163
- A Sousa https://orcid.org/0000-0002-1862-1409
- P T Surukuchi https://orcid.org/0000-0002-7858-0370

References

- [1] Abazajian K N et al 2012 Light sterile neutrinos: a white paper arXiv:1204.53794
- [2] Fukuda Y et al (Super-Kamiokande Collaboration) 1998 Evidence for oscillation of atmospheric neutrinos Phys. Rev. Lett. 81 1562–7
- [3] Ahmad Q R *et al* (SNO Collaboration) 2001 Measurement of the rate of $\nu_e + d \rightarrow p + p + e^$ interactions produced by ⁸B solar neutrinos at the sudbury neutrino observatory *Phys. Rev. Lett.* **87** 071301
- [4] Ahmad Q R et al (SNO Collaboration) 2002 Direct evidence for neutrino flavor transformation from neutral current interactions in the sudbury neutrino observatory Phys. Rev. Lett. 89 011301
- [5] Zyla P A *et al* (Particle Data Group Collaboration) 2020 Review of particle physics *Progress Theor. Exp. Phys.* **2020** 083C01
- [6] Minkowski P 1977 $\mu \rightarrow e\gamma$ at a rate of one out of 10⁹ muon decays? *Phys. Lett.* B 67 421–8
- [7] Gell-Mann M, Ramond P and Slansky R 1979 Complex spinors and unified theories *Conf. Proc.* C 790927 315–21
- [8] Yanagida T 1979 Horizontal gauge symmetry and masses of neutrinos Conf. Proc. C 7902131 95–9
- [9] Mohapatra R N and Senjanovic G 1980 Neutrino mass and spontaneous parity nonconservation Phys. Rev. Lett. 44 912
- [10] Schechter J and Valle J W F 1980 Neutrino masses in SU(2) x U(1) theories Phys. Rev. D 22 2227
- [11] Magg M and Wetterich C 1980 Neutrino mass problem and gauge hierarchy *Phys. Lett.* B 94 61–4

- [12] Mohapatra R N and Senjanovic G 1981 Neutrino masses and mixings in gauge models with spontaneous parity violation *Phys. Rev.* D 23 165
- [13] Lazarides G, Shafi Q and Wetterich C 1981 Proton lifetime and fermion masses in an SO(10) Model Nucl. Phys. B 181 287–300
- [14] Ma E and Sarkar U 1998 Neutrino masses and leptogenesis with heavy Higgs triplets *Phys. Rev.* Lett. 80 5716–9
- [15] Foot R, Lew H, He X G and Joshi G C 1989 Seesaw neutrino masses induced by a triplet of leptons Z. Phys. C 44 441
- [16] Ma E 1998 Pathways to naturally small neutrino masses *Phys. Rev. Lett.* 81 1171–4
- [17] Schael S et al (ALEPH, DELPHI, L3, OPAL, SLD, LEP Electroweak Working Group, SLD Electroweak Group, SLD Heavy Flavour Group Collaboration) 2006 Precision electroweak measurements on the Z resonance Phys. Rep. 427 257–454
- [18] Parke S and Ross-Lonergan M 2016 Unitarity and the three flavor neutrino mixing matrix *Phys. Rev.* D 93 113009
- [19] Aguilar-Arevalo A *et al* (LSND Collaboration) 2001 Evidence for neutrino oscillations from the observation of $\bar{\nu}_e$ appearance in a $\bar{\nu}_{\mu}$ beam *Phys. Rev.* D 64 112007
- [20] Armbruster B *et al* (KARMEN Collaboration) 2002 Upper limits for neutrino oscillations $\bar{\nu}_{\mu} \rightarrow \bar{\nu}_{e}$ from muon decay at rest *Phys. Rev.* D **65** 112001
- [21] Athanassopoulos C et al (LSND Collaboration) 1997 The liquid scintillator neutrino detector and LAMPF neutrino source Nucl. Instrum. Meth. A 388 149–72
- [22] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2009 The neutrino flux prediction at MiniBooNE Phys. Rev. D 79 072002
- [23] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2009 The MiniBooNE detector Nucl. Instrum. Meth. A 599 28–46
- [24] Aguilar-Arevalo A A *et al* (MiniBooNE Collaboration) 2007 A search for electron neutrino appearance at the $\Delta m^2 \sim 1 \text{ eV}^2$ scale *Phys. Rev. Lett.* **98** 231801
- [25] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2009 Unexplained excess of electron-like events from a 1 GeV neutrino beam Phys. Rev. Lett. 102 101802
- [26] Karagiorgi G, Aguilar-Arevalo A, Conrad J M, Shaevitz M H, Whisnant K, Sorel M and Barger V 2007 Leptonic CP violation studies at MiniBooNE in the (3+2) sterile neutrino oscillation hypothesis *Phys. Rev.* D 75 013011
- [27] Hollenberg S, Micu O and Pas H 2009 Neutrino–antineutrino oscillations as a possible solution for the LSND and MiniBooNE anomalies? *Phys. Rev.* D 80 053010
- [28] Kostelecky V A and Mewes M 2004 Lorentz and CPT violation in neutrinos Phys. Rev. D 69 016005
- [29] Katori T, Kostelecky V A and Tayloe R 2006 Global three-parameter model for neutrino oscillations using Lorentz violation *Phys. Rev.* D 74 105009
- [30] Aguilar-Arevalo A A *et al* (MiniBooNE Collaboration) 2010 Event excess in the MiniBooNE search for $\bar{\nu}_{\mu} \rightarrow \bar{\nu}_{e}$ oscillations *Phys. Rev. Lett.* **105** 181801
- [31] Aguilar-Arevalo A A *et al* (MiniBooNE Collaboration) 2013 Improved search for $\bar{\nu}_{\mu} \rightarrow \bar{\nu}_{e}$ oscillations in the MiniBooNE experiment *Phys. Rev. Lett.* **110** 161801
- [32] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2018 Significant excess of electronlike events in the miniboone short-baseline neutrino experiment *Phys. Rev. Lett.* **121** 221801
- [33] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2021 Updated MiniBooNE neutrino oscillation results with increased data and new background studies Phys. Rev. D 103 052002
- [34] Agafonova N et al (OPERA Collaboration) 2019 Final results on neutrino oscillation parameters from the OPERA experiment in the CNGS beam Phys. Rev. D 100 051301
- [35] Ericson M, Garzelli M V, Giunti C and Martini M 2016 Assessing the role of nuclear effects in the interpretation of the MiniBooNE low-energy anomaly *Phys. Rev.* D 93 073008
- [36] Brdar V and Kopp J 2022 Can standard model and experimental uncertainties resolve the MiniBooNE anomaly? *Phys. Rev.* D 105 115024
- [37] Pas H, Pakvasa S and Weiler T J 2005 Sterile-active neutrino oscillations and shortcuts in the extra dimension *Phys. Rev.* D 72 095017
- [38] Harvey J A, Hill C T and Hill R J 2007 Anomaly mediated neutrino-photon interactions at finite baryon density *Phys. Rev. Lett.* 99 261601
- [39] Nelson A E and Walsh J 2008 Short baseline neutrino oscillations and a new light gauge boson Phys. Rev. D 77 033001

- [40] Gninenko S N 2011 A resolution of puzzles from the LSND, KARMEN, and MiniBooNE experiments Phys. Rev. D 83 015015
- [41] Bemporad C, Gratta G and Vogel P 2002 Reactor based neutrino oscillation experiments Rev. Mod. Phys. 74 297
- [42] Patrick H 2016 Reactor antineutrino fluxes-status and challenges Nucl. Phys. B 908 268-78
- [43] Hayes A C and Vogel P 2016 Reactor neutrino spectra Ann. Rev. Nucl. Part. Sci. 66 219-44
- [44] Estienne M et al 2019 Updated summation model: an improved agreement with the daya bay antineutrino fluxes Phys. Rev. Lett. 123 022502
- [45] Dwyer D A and Langford T J 2015 Spectral structure of electron antineutrinos from nuclear reactors *Phys. Rev. Lett.* **114** 012502
- [46] Sonzogni A A, McCutchan E A, Johnson T D and Dimitriou P 2016 Effects of fission yield data in the calculation of antineutrino spectra for ²³⁵U(n, fission) at thermal and fast neutron energies *Phys. Rev. Lett.* **116** 132502
- [47] Fijałkowska A *et al* 2017 Impact of modular total absorption spectrometer measurements of β decay of fission products on the decay heat and reactor $\overline{\nu}_e$ flux calculation *Phys. Rev. Lett.* **119** 052503
- [48] Guadilla V et al 2019 Total absorption γ-ray spectroscopy of the β-delayed neutron emitters ¹³⁷I and ⁹⁵Rb Phys. Rev. C 100 044305
- [49] Guadilla V et al 2019 Total absorption γ-ray spectroscopy of niobium isomers Phys. Rev. C 100 024311
- [50] Zakari-Issoufou A A *et al* (IGISOL Collaboration) 2015 Total absorption spectroscopy study of ⁹²Rb decay: a major contributor to reactor antineutrino spectrum shape *Phys. Rev. Lett.* 115 102503
- [51] Rasco B C *et al* 2016 Decays of the three top contributors to the reactor $\overline{\nu}_e$ high-energy spectrum, ⁹²Rb, ^{96gs}Y, and ¹⁴²Cs, studied with total absorption spectroscopy *Phys. Rev. Lett.* **117** 092501
- [52] Rice S *et al* 2017 Total absorption spectroscopy study of the β decay of ⁸⁶Br and ⁹¹Rb *Phys. Rev.* C **96** 014320
- [53] Valencia E *et al* 2017 Total absorption γ-ray spectroscopy of β-delayed neutron emitters⁸⁷Br,
 ⁸⁸Br and ⁹⁴Rb *Phys. Rev.* C 95 024320
- [54] Mueller T A et al 2011 Improved predictions of reactor antineutrino spectra Phys. Rev. C 83 054615
- [55] Huber P 2011 On the determination of anti-neutrino spectra from nuclear reactors *Phys. Rev.* C 84 024617
- [56] Feilitzsch F V, Hahn A A and Schreckenbach K 1982 Experimental beta spectra from ²³⁹PU and ²³⁵U thermal neutron fission products and their correlated anti-neutrinos spectra *Phys. Lett.* B 118 162–6
- [57] Schreckenbach K, Colvin G, Gelletly W and Von Feilitzsch F 1985 Determination of the antineutrino spectrum from ²³⁵U thermal neutron fission products up to 9.5 Mev *Phys. Lett.* B 160 325–30
- [58] Hahn A A, Schreckenbach K, Colvin G, Krusche B, Gelletly W and Von Feilitzsch F 1989 Antineutrino spectra from ²⁴¹Pu and ²³⁹Pu thermal neutron fission products *Phys. Lett.* B 218 365–8
- [59] Haag N, Gütlein A, Hofmann M, Oberauer L, Potzel W, Schreckenbach K and Wagner F M 2014 Experimental determination of the antineutrino spectrum of the fission products of ²³⁸U Phys. Rev. Lett. 112 122501
- [60] Kopeikin V, Mikaelyan L and Sinev V 2004 Reactor as a source of antineutrinos: thermal fission energy *Phys. Atom. Nucl.* 67 1892–9
- [61] Balantekin A B et al (PROSPECT Collaboration) 2020 Nonfuel antineutrino contributions in the ORNL high flux isotope reactor (HFIR) Phys. Rev. C 101 054605
- [62] An F-P, Tian X-C, Zhan L and Cao J 2009 Systematic impact of spent nuclear fuel on $\theta(13)$ sensitivity at reactor neutrino experiment *Chin. Phys.* C **33** 711–6
- [63] Zhou B, Ruan X-C, Nie Y-B, Zhou Z-Y, An F-P and Cao J 2012 A study of antineutrino spectra from spent nuclear fuel at Daya Bay *Chin. Phys.* C 36 1–5
- [64] Conant A J 2019 Antineutrino spectrum characterization at the high flux isotope reactor using neutronic simulations *PhD Thesis* Georgia Institute of Technology
- [65] Vogel P, Wen L and Zhang C 2015 Neutrino oscillation studies with reactors Nat. Commun.
 6 6935
- [66] Mention G, Fechner M, Lasserre T, Mueller T A, Lhuillier D, Cribier M and Letourneau A 2011 The reactor antineutrino anomaly *Phys. Rev.* D 83 073006

- [67] Abe Y *et al* (Double Chooz Collaboration) 2012 Indication of reactor $\bar{\nu}_e$ disappearance in the double chooz experiment *Phys. Rev. Lett.* **108** 131801
- [68] An F P et al (Daya Bay Collaboration) 2012 Observation of electron-antineutrino disappearance at Daya Bay Phys. Rev. Lett. 108 171803
- [69] Ahn J K et al (RENO Collaboration) 2012 Observation of reactor electron antineutrino disappearance in the RENO experiment Phys. Rev. Lett. 108 191802
- [70] Davis R 1964 Solar neutrinos: II. Experimental Phys. Rev. Lett. 12 303-5
- [71] Hampel W et al (GALLEX Collaboration) 1998 Final results of the Cr-51 neutrino source experiments in GALLEX Phys. Lett. B 420 114–26
- [72] Abdurashitov J N et al (SAGE Collaboration) 1999 Measurement of the response of the Russian– American gallium experiment to neutrinos from a ⁵¹Cr source Phys. Rev. C 59 2246–63
- [73] Kuzmin V A 1965 Detection of solar neutrinos by means of the ⁷¹Ga(nu, e-)⁷¹Ge reaction Zh. Eksp. Teor. Fiz. 49 1532–4
- [74] Anselmann P et al (GALLEX Collaboration) 1995 First results from the ⁵¹Cr neutrino source experiment with the GALLEX detector Phys. Lett. B 342 440–50
- [75] Abdurashitov J N et al 2006 Measurement of the response of a Ga solar neutrino experiment to neutrinos from an ³⁷Ar source Phys. Rev. C 73 045805
- [76] Bahcall J N 1997 Gallium solar neutrino experiments: absorption cross-sections, neutrino spectra, and predicted event rates *Phys. Rev.* C 56 3391–409
- [77] Giunti C and Laveder M 2011 Significance of the gallium anomaly Phys. Rev. C 83 065504
- [78] Acero M A, Giunti C and Laveder M 2008 Limits on nu(e) and anti-nu(e) disappearance from gallium and reactor experiments *Phys. Rev.* D 78 073009
- [79] Gell-Mann M, Ramond P and Slansky R 1979 Complex spinors and unified theories Conf. Proc. C 790927 315–21
- [80] Cheng T P and Li L-F 1980 Neutrino masses, mixings and oscillations in SU(2) x U(1) models of electroweak interactions *Phys. Rev.* D 22 2860
- [81] He Z 2012 Light sterile neutrino in the minimal extended seesaw *Phys. Lett.* B 714 262–6
- [82] Fukugita M and Yanagida T 1986 Baryogenesis without grand unification Phys. Lett. B 174 45-7
- [83] Dodelson S and Widrow L M 1994 Sterile-neutrinos as dark matter Phys. Rev. Lett. 72 17-20
- [84] Shi X-D and Fuller G M 1999 A New dark matter candidate: nonthermal sterile neutrinos Phys. Rev. Lett. 82 2832–5
- [85] Dolgov A D and Hansen S H 2002 Massive sterile neutrinos as warm dark matter Astropart. Phys. 16 339–44
- [86] Abazajian K, Fuller G M and Patel M 2001 Sterile neutrino hot, warm, and cold dark matter Phys. Rev. D 64 023501
- [87] Boyarsky A, Ruchayskiy O and Shaposhnikov M 2009 The role of sterile neutrinos in cosmology and astrophysics Ann. Rev. Nucl. Part. Sci. 59 191–214
- [88] Asaka T and Shaposhnikov M 2005 The νMSM, dark matter and baryon asymmetry of the universe Phys. Lett. B 620 17–26
- [89] Shaposhnikov M 2008 The ν MSM, leptonic asymmetries, and properties of singlet fermions J. High Energy Phys. JHEP08(2008)08
- [90] de Gouvea A 2005 See-saw energy scale and the LSND anomaly Phys. Rev. D 72 033005
- [91] Mohapatra R N, Nasri S and Yu H-B 2005 Seesaw right handed neutrino as the sterile neutrino for LSND Phys. Rev. D 72 033007
- [92] Harari H and Leurer M 1986 Recommending a standard choice of cabibbo angles and KM phases for any number of generations *Phys. Lett.* B 181 123–8
- [93] Sorel M, Conrad J M and Shaevitz M 2004 A combined analysis of short baseline neutrino experiments in the (3+1) and (3+2) sterile neutrino oscillation hypotheses *Phys. Rev.* D 70 073004
- [94] Karagiorgi G, Djurcic Z, Conrad J M, Shaevitz M H and Sorel M 2009 Viability of $\Delta m^2 \sim 1 \text{ eV}^2$ sterile neutrino mixing models in light of MiniBooNE electron neutrino and antineutrino data from the booster and NuMI beamlines *Phys. Rev.* D **80** 073001
- [95] Kopp J, Maltoni M and Schwetz T 2011 Are there sterile neutrinos at the eV scale? *Phys. Rev. Lett.* 107 091801
- [96] Giunti C and Laveder M 2011 Implications of 3+1 short-baseline neutrino oscillations *Phys. Lett.* B 706 200–7
- [97] Giunti C and Laveder M 2011 3+1 and 3+2 sterile neutrino fits Phys. Rev. D 84 073008

- [98] Conrad J M, Ignarra C M, Karagiorgi G, Shaevitz M H and Spitz J 2013 Sterile neutrino fits to short baseline neutrino oscillation measurements Adv. High Energy Phys. 2013 163897
- [99] Kopp J, Machado P A N, Maltoni M and Schwetz T 2013 Sterile neutrino oscillations: the global picture J. High Energy Phys. JHEP05(2013)050
- [100] Collin G H, Argüelles C A, Conrad J M and Shaevitz M H 2016 First constraints on the complete neutrino mixing matrix with a sterile neutrino *Phys. Rev. Lett.* **117** 221801
- [101] Gariazzo S, Giunti C, Laveder M and Li Y F 2017 Updated global 3+1 analysis of short-baseline neutrino oscillations J. High Energ. Phys. JHEP06(2017)135
- [102] Dentler M, Hernández-Cabezudo A ', Kopp J, Machado P A N, Maltoni M, Martinez-Soler I and Schwetz T 2018 updated global analysis of neutrino oscillations in the presence of eV-scale sterile neutrinos J. High Energy Phys. JHEP08(2018)010
- [103] Diaz A, Argüelles C A, Collin G H, Conrad J M and Shaevitz M H 2020 Where are we with light sterile neutrinos? *Phys. Rep.* 884 1–59
- [104] Giunti C and Lasserre T 2019 eV-scale sterile neutrinos Ann. Rev. Nucl. Part. Sci. 69 163-90
- [105] Böser S, Buck C, Giunti C, Lesgourgues J, Ludhova L, Mertens S, Schukraft A and Wurm M 2020 Status of light sterile neutrino searches *Prog. Part. Nucl. Phys.* 111 103736
- [106] Dasgupta B and Kopp J 2021 Sterile neutrinos Phys. Rep. 928 63
- [107] Maltoni M and Schwetz T 2003 Testing the statistical compatibility of independent data sets *Phys. Rev.* D 68 033020
- [108] Wilks S S 1938 The large-sample distribution of the likelihood ratio for testing composite hypotheses Ann. Math. Stat. 9 60–2
- [109] Coloma P, Huber P and Schwetz T 2021 Statistical interpretation of sterile neutrino oscillation searches at reactors *Eur. Phys. J.* C 81 2
- [110] Berryman J M and Huber P 2021 Sterile neutrinos and the global reactor antineutrino dataset J. High Energy Phys. JHEP01(2021)167
- [111] Argüelles C A, Bertólez-Martínez T and Salvado J 2022 Impact of wave packet separation in low-energy sterile neutrino searches *Phys. Rev.* D 107 036004
- [112] Eliezer S and Swift A R 1976 Experimental consequences of electron neutrino-muon-neutrino mixing in neutrino beams Nucl. Phys. B 105 45–51
- [113] Fritzsch H and Minkowski P 1976 Vector-like weak currents, massive neutrinos, and neutrino beam oscillations *Phys. Lett.* B 62 72–6
- [114] Bilenky S M and Pontecorvo B 1976 Again on neutrino oscillations Lett. Nuovo Cim. 17 569
- [115] Akhmedov E 2019 Quantum mechanics aspects and subtleties of neutrino oscillations Int. Conf. on History of the Neutrino: 1930-2018 p 1
- [116] Giunti C 2004 Coherence and wave packets in neutrino oscillations Found. Phys. Lett. 17 103-24
- [117] Akhmedov E K and Smirnov A Y 2009 Paradoxes of neutrino oscillations Phys. Atom. Nucl. 72 1363–81
- [118] Nussinov S 1976 Solar neutrinos and neutrino mixing Phys. Lett. B 63 201-3
- [119] Kayser B 1981 On the quantum mechanics of neutrino oscillation Phys. Rev. D 24 110
- [120] Kiers K, Nussinov S and Weiss N 1996 Coherence effects in neutrino oscillations *Phys. Rev.* D 53 537–47
- [121] Beuthe M 2003 Oscillations of neutrinos and mesons in quantum field theory Phys. Rep. 375 105–218
- [122] Akhmedov E, Hernandez D and Smirnov A 2012 Neutrino production coherence and oscillation experiments J. High Energy Phys. JHEP04(2012)052
- [123] Akhmedov E, Kopp J and Lindner M 2017 Collective neutrino oscillations and neutrino wave packets J.Cosmol. Astropart. Phys. JCAP09(2017)017
- [124] Giunti C and Kim C W 2007 Fundamentals of Neutrino Physics and Astrophysics (Oxford University Press)
- [125] Bernardini A E and Leo S D 2004 An analytic approach to the wave packet formalism in oscillation phenomena *Phys. Rev.* D 70 053010
- [126] Naumov D V and Naumov V A 2010 A Diagrammatic treatment of neutrino oscillations J. Phys. G 37 105014
- [127] Giunti C and Kim C W 1998 Coherence of neutrino oscillations in the wave packet approach Phys. Rev. D 58 017301
- [128] Stodolsky L 1998 The Unnecessary wave packet Phys. Rev. D 58 036006
- [129] Gouvêa A d, Romeri V D and Ternes C A 2021 Combined analysis of neutrino decoherence at reactor experiments J. High Energy Phys. JHEP06(2021)042

- [130] An F P et al (Daya Bay Collaboration) 2017 Study of the wave packet treatment of neutrino oscillation at Daya Bay Eur. Phys. J. C 77 606
- [131] An F P et al (Daya Bay Collaboration) 2017 Measurement of electron antineutrino oscillation based on 1230 days of operation of the Daya Bay experiment Phys. Rev. D 95 072006
- [132] An F P et al (Daya Bay Collaboration) 2016 Improved search for a light sterile neutrino with the full configuration of the Daya Bay experiment Phys. Rev. Lett. 117 151802
- [133] Ko Y J et al (NEOS Collaboration) 2017 Sterile neutrino search at the NEOS experiment Phys. Rev. Lett. 118 121802
- [134] Ashenfelter J et al (PROSPECT Collaboration) 2018 First search for short-baseline neutrino oscillations at HFIR with PROSPECT Phys. Rev. Lett. 121 251802
- [135] Andriamirado M et al (PROSPECT Collaboration) 2021 Improved short-baseline neutrino oscillation search and energy spectrum measurement with the PROSPECT experiment at HFIR Phys. Rev. D 103 032001
- [136] Barinov V V et al 2022 Results from the baksan experiment on sterile transitions (BEST) Phys. Rev. Lett. 128 232501
- [137] Fong C S, Minakata H and Nunokawa H 2017 A framework for testing leptonic unitarity by neutrino oscillation experiments J. High Energy Phys. JHEP02(2017)114
- [138] Fong C S, Minakata H and Nunokawa H 2019 Non-unitary evolution of neutrinos in matter and the leptonic unitarity test J. High Energy Phys. JHEP02(2019)015
- [139] Adamson P et al (MINOS+ Collaboration) 2019 Search for sterile neutrinos in MINOS and MINOS+ using a two-detector fit Phys. Rev. Lett. 122 091803
- [140] Aartsen M G et al (IceCube Collaboration) 2017 Search for sterile neutrino mixing using three years of IceCube DeepCore data Phys. Rev. D 95 112002
- [141] Wolfenstein L 1978 Neutrino oscillations in matter Phys. Rev. D 17 2369-74
- [142] Karagiorgi G, Shaevitz M H and Conrad J M 2012 Confronting the short-baseline oscillation anomalies with a single sterile neutrino and non-standard matter effects arXiv:1202.1024
- [143] Liao J and Marfatia D 2016 Impact of nonstandard interactions on sterile neutrino searches at IceCube Phys. Rev. Lett. 117 071802
- [144] Liao J, Marfatia D and Whisnant K 2019 MiniBooNE, MINOS+ and IceCube data imply a baroque neutrino sector *Phys. Rev.* D 99 015016
- [145] Denton P B, Farzan Y and Shoemaker I M 2019 Activating the fourth neutrino of the 3+1 scheme Phys. Rev. D 99 035003
- [146] Asaadi J, Church E, Guenette R, Jones B J P and Szelc A M 2018 New light Higgs boson and short-baseline neutrino anomalies *Phys. Rev.* D 97 075021
- [147] Alves D S M, Louis W C and deNiverville P G 2022 Quasi-sterile neutrinos from dark sectors I. BSM matter effects in neutrino oscillations and the short-baseline anomalies J. High Energy Phys. JHEP08(2022)034
- [148] Pal P B and Wolfenstein L 1982 Radiative decays of massive neutrinos Phys. Rev. D 25 766
- [149] Nieves J F 1983 Two photon decays of heavy neutrinos Phys. Rev. D 28 1664
- [150] Moss Z, Moulai M H, Argüelles C A and Conrad J M 2018 Exploring a nonminimal sterile neutrino model involving decay at IceCube *Phys. Rev.* D 97 055017
- [151] Aartsen M G et al (IceCube Collaboration) 2020 eV-scale sterile neutrino search using eight years of atmospheric muon neutrino data from the icecube neutrino observatory Phys. Rev. Lett. 125 141801
- [152] Aartsen M G et al (IceCube Collaboration) 2020 Searching for eV-scale sterile neutrinos with eight years of atmospheric neutrinos at the IceCube Neutrino Telescope Phys. Rev. D 102 052009
- [153] Aartsen M G et al (IceCube Collaboration) 2016 Searches for Sterile Neutrinos with the IceCube Detector Phys. Rev. Lett. 117 071801
- [154] Moulai M H, Argüelles C A, Collin G H, Conrad J M, Diaz A and Shaevitz M H 2020 Combining sterile neutrino fits to short baseline data with IceCube data *Phys. Rev.* D 101 055020
- [155] Moulai M H 2021 Light, unstable sterile neutrinos: phenomenology, a search in the IceCube experiment, and a global picture PhD Thesis MIT
- [156] Machado P A, Palamara O and Schmitz D W 2019 The short-baseline neutrino program at Fermilab Ann. Rev. Nucl. Part. Sci. 69 363–87
- [157] Palomares-Ruiz S, Pascoli S and Schwetz T 2005 Explaining LSND by a decaying sterile neutrino J. High Energy Phys. JHEP09(2005)048

- [158] Bai Y, Lu R, Lu S, Salvado J and Stefanek B A 2016 Three twin neutrinos: evidence from LSND and MiniBooNE Phys. Rev. D 93 073004
- [159] Gouvêa A D, Prakash S and Stenico G V 2020 On The decaying-sterile neutrino solution to the electron (anti)neutrino appearance anomalies J. High Energy Phys. 07 141
- [160] Dentler M, Esteban I, Kopp J and Machado P 2020 Decaying sterile neutrinos and the short baseline oscillation anomalies *Phys. Rev.* D 101 115013
- [161] Hostert M and Pospelov M 2021 Constraints on decaying sterile neutrinos from solar antineutrinos Phys. Rev. D 104 055031
- [162] Dasgupta B and Kopp J 2014 Cosmologically safe eV-scale sterile neutrinos and improved dark matter structure *Phys. Rev. Lett.* **112** 031803
- [163] Hannestad S, Hansen R S and Tram T 2014 How self-interactions can reconcile sterile neutrinos with cosmology *Phys. Rev. Lett.* **112** 031802
- [164] Chu X, Dasgupta B, Dentler M, Kopp J and Saviano N 2018 Sterile neutrinos with secret interactions—cosmological discord? J. Cosmol. Astropart. Phys. JCAP11(2018)049
- [165] Yaguna C E 2007 Sterile neutrino production in models with low reheating temperatures J. High Energy Phys. JHEP06(2007)002
- [166] Saviano N, Mirizzi A, Pisanti O, Serpico P D, Mangano G and Miele G 2013 Multi-momentum and multi-flavour active-sterile neutrino oscillations in the early universe: role of neutrino asymmetries and effects on nucleosynthesis *Phys. Rev.* D 87 073006
- [167] Giovannini M, Kurki-Suonio H and Sihvola E 2002 Big bang nucleosynthesis, matter antimatter regions, extra relativistic species, and relic gravitational waves *Phys. Rev.* D 66 043504
- [168] Bezrukov F, Chudaykin A and Gorbunov D 2017 Hiding an elephant: heavy sterile neutrino with large mixing angle does not contradict cosmology J. Cosmol. Astropart. Phys. JCAP06 (2017)051
- [169] Farzan Y 2019 Ultra-light scalar saving the 3 + 1 neutrino scheme from the cosmological bounds Phys. Lett. B 797 134911
- [170] Cline J M 2020 Viable secret neutrino interactions with ultralight dark matter Phys. Lett. B 802 135182
- [171] Archidiacono M, Gariazzo S, Giunti C, Hannestad S and Tram T 2020 Sterile neutrino selfinteractions: H₀ tension and short-baseline anomalies J. Cosmol. Astropart. Phys. JCAP12 (2020)029
- [172] Valentino E D, Gariazzo S, Giunti C, Mena O, Pan S and Yang W 2021 Minimal dark energy: key to sterile neutrino and Hubble constant tensions? *Phys. Rev.* D 105 103511
- [173] Bergmann S and Grossman Y 1999 Can lepton flavor violating interactions explain the LSND results? Phys. Rev. D 59 093005
- [174] Babu K S, McKay D W, Mocioiu I and Pakvasa S 2016 Light sterile neutrinos, lepton number violating interactions, and the LSND neutrino anomaly *Phys. Rev.* D 93 113019
- [175] Carena M, Li Y-Y, Machado C S, Machado P A N and Wagner C E M 2017 Neutrinos in large extra dimensions and short-baseline v_e appearance Phys. Rev. D 96 095014
- [176] Döring D and Päs H 2019 Sterile neutrino shortcuts in asymmetrically warped extra dimensions Eur. Phys. J. C 79 604
- [177] Döring D, Päs H, Sicking P and Weiler T J 2020 Sterile neutrinos with altered dispersion relations as an explanation for neutrino anomalies *Eur. Phys. J.* C 80 1202
- [178] Barenboim G A, Martínez-Miravé P, Ternes C A and Tórtola M A 2020 Sterile neutrinos with altered dispersion relations revisited J. High Energy Phys. JHEP03(2020)070
- [179] Argüelles C A, Esteban I, Hostert M, Kelly K J, Kopp J, Machado P A N, Martinez-Soler I and Perez-Gonzalez Y F 2022 MicroBooNE and the ν_e interpretation of the MiniBooNE low-energy excess *Phys. Rev. Lett.* **128** 241802
- [180] Kostelecky V A and Mewes M 2004 Lorentz and CPT violation in the neutrino sector *Phys. Rev.* D 70 031902
- [181] Kostelecky A and Mewes M 2012 Neutrinos with Lorentz-violating operators of arbitrary dimension Phys. Rev. D 85 096005
- [182] Diaz J S and Alan Kostelecky V 2011 Three-parameter Lorentz-violating texture for neutrino mixing Phys. Lett. B 700 25–8
- [183] Diaz J S and Kostelecky A 2012 Lorentz- and CPT-violating models for neutrino oscillations Phys. Rev. D 85 016013
- [184] Masip M, Masjuan P and Meloni D 2013 Heavy neutrino decays at MiniBooNE J. High Energy Phys. JHEP01(2013)106

- [185] Saúl-Sala E 2021 Open problems in the physics of neutrino interactions with nucleons and nuclei PhD Thesis Universidad de Valencia https://ific.uv.es/nucth/Thesis_Saul_Sala.pdf.
- [186] Alvarez-Ruso L and Saul-Sala E 2021 Neutrino interactions with matter and the MiniBooNE anomaly Eur. Phys. J. Special Top. 230 4373–89
- [187] Gninenko S N 2009 The MiniBooNE anomaly and heavy neutrino decay Phys. Rev. Lett. 103 241802
- [188] Gninenko S N and Gorbunov D S 2010 The MiniBooNE anomaly, the decay $D_s^+ \rightarrow \mu^+ \nu_{\mu}$ and heavy sterile neutrino *Phys. Rev.* D **81** 075013
- [189] Magill G, Plestid R, Pospelov M and Tsai Y-D 2018 Dipole portal to heavy neutral leptons *Phys. Rev.* D 98 115015
- [190] Alvarez-Ruso L and Saul-Sala E 2017 Radiative decay of heavy neutrinos at MiniBooNE and MicroBooNE Prospects in Neutrino Physics p 4 arXiv:1705.00353
- [191] Vergani S, Kamp N W, Diaz A, Argüelles C A, Conrad J M, Shaevitz M H and Uchida M A 2021 Explaining the MiniBooNE excess through a mixed model of neutrino oscillation and decay *Phys. Rev.* D 104 095005
- [192] David M and Pospelov M 2010 Muon capture constraints on sterile neutrino properties *Phys. Rev.* D 82 113018
- [193] Coloma P, Machado P A N, Martinez-Soler I and Shoemaker I M 2017 Double-Cascade Events from New Physics in Icecube Phys. Rev. Lett. 119 201804
- [194] Auerbach L B et al (LSND Collaboration) 2001 Measurement of electron–neutrino–electron elastic scattering Phys. Rev. D 63 112001
- [195] Agostini M et al (Borexino Collaboration) 2017 Limiting neutrino magnetic moments with Borexino Phase-II solar neutrino data Phys. Rev. D 96 091103
- [196] Schwienhorst R et al (DONUT Collaboration) 2001 A new upper limit for the tau-neutrino magnetic moment Phys. Lett. B 513 23–9
- [197] Geiregat D et al (CHARM-II Collaboration) 1991 An Improved determination of the electroweak mixing angle from muon-neutrino electron scattering Phys. Lett. B 259 499–507
- [198] Gninenko S N and Krasnikov N V 1999 Limits on the magnetic moment of sterile neutrino and two photon neutrino decay *Phys. Lett.* B 450 165–72
- [199] Vannucci F 2014 The NOMAD experiment at CERN Adv. High Energy Phys. 2014 129694
- [200] Anelli M et al (SHiP Collaboration) 2015 A facility to search for hidden particles (SHiP) at the CERN SPS arXiv:1504.04956
- [201] Atkinson M, Coloma P, Martinez-Soler I, Rocco N and Shoemaker I M 2021 Heavy neutrino searches through double-bang events at Super-Kamiokande J. High Energy Phys. JHEP04 (2022)174
- [202] Schwetz T, Zhou A and Zhu J-Y 2020 Constraining active-sterile neutrino transition magnetic moments at DUNE near and far detectors J. High Energy Phys. JHEP21(2020)200
- [203] Bertuzzo E, Jana S, Machado P A N and Funchal R Z 2018 Dark neutrino portal to explain MiniBooNE excess Phys. Rev. Lett. 121 241801
- [204] Ballett P, Pascoli S and Ross-Lonergan M 2019 U(1)' mediated decays of heavy sterile neutrinos in MiniBooNE Phys. Rev. D 99 071701
- [205] Fischer O, Hernández-Cabezudo Á and Schwetz T 2020 Explaining the MiniBooNE excess by a decaying sterile neutrino with mass in the 250 MeV range *Phys. Rev.* D 101 075045
- [206] Chang C H V, Chen C-R, Ho S-Y and Tseng S-Y 2021 Explaining the MiniBooNE anomalous excess via a leptophilic ALP-sterile neutrino coupling *Phys. Rev.* D 104 015030
- [207] Bertuzzo E, Jana S, Machado P A N and Funchal R Z 2019 Neutrino masses and mixings dynamically generated by a light dark sector *Phys. Lett.* B 791 210–4
- [208] Ballett P, Hostert M and Pascoli S 2020 Dark neutrinos and a three portal connection to the standard model Phys. Rev. D 101 115025
- [209] Abdullahi A, Hostert M and Pascoli S 2021 A dark seesaw solution to low energy anomalies: MiniBooNE, the muon (g - 2), and BaBar Phys. Lett. B 820 136531
- [210] Datta A, Kamali S and Marfatia D 2020 Dark sector origin of the KOTO and MiniBooNE anomalies Phys. Lett. B 807 135579
- [211] Dutta B, Ghosh S and Li T 2020 Explaining $(g 2)_{\mu,e}$, the KOTO anomaly and the MiniBooNE excess in an extended Higgs model with sterile neutrinos *Phys. Rev.* D **102** 055017
- [212] Abdallah W, Gandhi R and Roy S 2020 Understanding the MiniBooNE and the muon and electron g 2 anomalies with a light Z' and a second Higgs doublet J. High Energy Phys. JHEP12(2020)188

- [213] Abdallah W, Gandhi R and Roy S 2021 Two-Higgs doublet solution to the LSND, MiniBooNE and muon g – 2 anomalies *Phys. Rev.* D 104 055028
- [214] Hammad A, Rashed A and Moretti S 2022 The dark Z' and sterile neutrinos behind current anomalies Phys. Lett. B 827 136945
- [215] Dutta B, Kim D, Thompson A, Thornton R T and Van de Water R G 2022 Solutions to the MiniBooNE anomaly from new physics in charged meson decays *Phys. Rev. Lett.* **129** 111803
- [216] Ballett P, Hostert M and Pascoli S 2019 Neutrino masses from a dark neutrino sector below the electroweak scale *Phys. Rev.* D 99 091701
- [217] Argüelles C A, Hostert M and Tsai Y-D 2019 Testing new physics explanations of the MiniBooNE anomaly at neutrino scattering experiments *Phys. Rev. Lett.* **123** 261801
- [218] Bansal S, Paz G, Petrov A, Tammaro M and Zupan J 2023 Enhanced neutrino polarizability J. High Energy Phys. 2023 142
- [219] Atre A et al 2009 The search for heavy majorana neutrinos J. High Energy Phys. JHEP05 (2009)030
- [220] Duk V A *et al* (ISTRA+ Collaboration) 2012 Search for heavy neutrino in $K^- \rightarrow \mu^- \nu_h (\nu_h \rightarrow \nu \gamma)$ decay at ISTRA+ setup *Phys. Lett.* B **710** 307–17
- [221] Drewes M and Garbrecht B 2017 Combining experimental and cosmological constraints on heavy neutrinos Nucl. Phys. B 921 250–315
- [222] de Gouvêa A and Kobach A 2016 Global constraints on a heavy neutrino *Phys. Rev.* D93 033005
- [223] Aguilar-Arevalo A A et al (MiniBooNE DM Collaboration) 2018 Dark matter search in nucleon, pion, and electron channels from a proton beam dump with MiniBooNE Phys. Rev. D 98 112004
- [224] Jordan J R, Kahn Y, Krnjaic G, Moschella M and Spitz J 2019 Severe constraints on new physics explanations of the MiniBooNE excess *Phys. Rev. Lett.* **122** 081801
- [225] Bryman D A and Shrock R 2019 Improved constraints on sterile neutrinos in the MeV to GeV mass range Phys. Rev. D 100 053006
- [226] Coloma P 2019 Icecube/DeepCore tests for novel explanations of the MiniBooNE anomaly Eur. Phys. J. C 79 748
- [227] Bryman D A and Shrock R 2019 Constraints on sterile neutrinos in the MeV to GeV mass range Phys. Rev. D 100 073011
- [228] Brdar V, Fischer O and Smirnov A Y 2021 Model-independent bounds on the nonoscillatory explanations of the MiniBooNE excess *Phys. Rev.* D 103 075008
- [229] Córsico A H, Althaus L G, Miller Bertolami M M, Kepler S O and García-Berro E 2014 Constraining the neutrino magnetic dipole moment from white dwarf pulsations J. Cosmol. Astropart. Phys. JCAP08(2014)054
- [230] Díaz S A, Schröder K-P, Zuber K, Jack D and Barrios E E B 2019 Constraint on the axionelectron coupling constant and the neutrino magnetic dipole moment by using the tip-RGB luminosity of fifty globular clusters arXiv:1910.10568
- [231] Studenikin A 2021 Electromagnetic neutrino: the basic processes and astrophysical probes J. Phys. Conf. Ser. 2156 012120
- [232] Radionov A 2013 Constraints on electromagnetic properties of sterile neutrinos from MiniBooNE results Phys. Rev. D 88 015016
- [233] Abdallah W, Gandhi R and Roy S 2022 Requirements on common solutions to the LSND and MiniBooNE excesses: a post-MicroBooNE study J. High Energy Phys. JHEP 06(160)160
- [234] Vilain P et al (CHARM-II Collaboration) 1994 Precision measurement of electroweak parameters from the scattering of muon-neutrinos on electrons Phys. Lett. B 335 246–52
- [235] Valencia E et al (MINERvA Collaboration) 2019 Constraint of the MINER νA medium energy neutrino flux using neutrino–electron elastic scattering Phys. Rev. D 100 092001
- [236] Perevalov D, Tayloe R and (MiniBooNE Collaboration) 2009 Measurement of neutrino–nucleon neutral current elastic scattering in MiniBooNE AIP Conf. Proc. 1189 175
- [237] Alves A, Dias A G and Lopes D D 2020 Probing ALP-sterile neutrino couplings at the LHC J. High Energy Phys. JHEP08(2020)074
- [238] Hill R J 2010 Low energy analysis of $\nu N \rightarrow \nu N \gamma$ gamma in the Standard Model *Phys. Rev.* D 81 013008
- [239] Serot B D and Zhang X 2012 Neutrinoproduction of photons and pions from nucleons in a chiral effective field theory for nuclei *Phys. Rev.* C 86 015501

- [240] Wang E, Alvarez-Ruso L and Nieves J 2014 Photon emission in neutral current interactions at intermediate energies Phys. Rev. C 89 015503
- [241] Hernández E and Nieves J 2017 Neutrino-induced one-pion production revisited: the $\nu_{\mu}n \rightarrow$ $\mu^- n\pi^+$ channel *Phys. Rev.* D **95** 053007
- [242] Rosner J L 2015 Low-energy photon production in neutrino neutral-current interactions Phys. Rev. D 91 093001
- [243] Zhang X and Serot B D 2012 Incoherent neutrinoproduction of photons and pions in a chiral effective field theory for nuclei Phys. Rev. C 86 035502
- [244] Zhang X and Serot B D 2013 Can neutrino-induced photon production explain the low energy excess in MiniBooNE? Phys. Lett. B 719 409-14
- [245] Wang E, Alvarez-Ruso L and Nieves J 2015 Single photon events from neutral current interactions at MiniBooNE Phys. Lett. B 740 16-22
- [246] Chanfray G and Ericson M 2021 γ production in neutrino interactions with nuclei *Phys. Rev.* C 104 015203
- [247] Giunti C, Ioannisian A and Ranucci G 2020 A new analysis of the MiniBooNE low-energy excess J. High Energy Phys. JHEP11(2020)146
- [248] Abratenko P et al (MicroBooNE Collaboration) 2022 Search for neutrino-induced neutral-current Δ radiative decay in MicroBooNE and a first test of the MiniBooNE low energy excess under a single-photon hypothesis Phys. Rev. Lett. 128 111801
- [249] Alvarez-Ruso L, Nieves J, Saúl Sala E and Wang E 2018 Neutrino interactions in the few-GeV region and the MiniBooNE anomaly J. Phys. Conf. Ser. 1056 012001
- [250] Hayes A C, Friar J L, Garvey G T, Jungman G and Jonkmans G 2014 Systematic uncertainties in the analysis of the reactor neutrino anomaly Phys. Rev. Lett. 112 202501
- [251] 2019 Antineutrino spectra and their applications IAEA Report INDC(NDS)-0786 https://wwwnds.iaea.org/publications/indc/indc-nds-0786.pdf
- [252] Hayen L, Kostensalo J, Severijns N and Suhonen J 2019 First-forbidden transitions in the reactor anomaly Phys. Rev. C 100 054323
- [253] Fallot M et al 2012 New antineutrino energy spectra predictions from the summation of beta decay branches of the fission products Phys. Rev. Lett. 109 202504
- [254] Hayes A C, Jungman G, McCutchan E A, Sonzogni A A, Garvey G T and Wang X 2018 Analysis of the daya bay reactor antineutrino flux changes with fuel burnup Phys. Rev. Lett. 120 022503
- [255] An F P et al (Daya Bay Collaboration) 2017 Evolution of the reactor antineutrino flux and spectrum at Daya Bay Phys. Rev. Lett. 118 251801
- [256] Bak G et al (RENO Collaboration) 2019 Fuel-composition dependent reactor antineutrino yield at RENO Phys. Rev. Lett. 122 232501
- [257] Giunti C, Li Y F, Ternes C A and Xin Z 2022 Reactor antineutrino anomaly in light of recent flux model refinements Phys. Lett. B 829 137054
- [258] Matthews E F, Bernstein L A and Younes W 2021 Stochastically estimated covariance matrices for independent and cumulative fission yields in the ENDF/B-VIII.0 and JEFF-3.3 evaluations At. Data Nucl. Data Tables 140 101441
- [259] Goldhagen K, Maltoni M, Reichard S E and Schwetz T 2022 Testing sterile neutrino mixing with present and future solar neutrino data Eur. Phys. J. C 82 116
- [260] Kaether F, Hampel W, Heusser G, Kiko J and Kirsten T 2010 Reanalysis of the GALLEX solar neutrino flux and source experiments Phys. Lett. B 685 47-54
- [261] Frekers D et al 2011 The Ga-71(He-3, t) reaction and the low-energy neutrino response Phys. Lett. B 706 134-8
- [262] Frekers D et al 2013 Penning-trap Q-value determination of the ${}^{71}Ga(\nu, e^{-}){}^{71}Ge$ reaction using
- threshold charge breeding of on-line produced isotopes *Phys. Lett.* B **722** 233–7 [263] Frekers D *et al* 2015 Precision evaluation of the ⁷¹Ga(ν_e , *e*) solar neutrino capture rate from the (³He,t) charge-exchange reaction *Phys. Rev.* C **91** 034608
- [264] Giunti C, Laveder M, Li Y F, Liu Q Y and Long H W 2012 Update of short-baseline electron neutrino and antineutrino disappearance Phys. Rev. D 86 113014
- [265] Haxton W C 1998 Cross-section uncertainties in the gallium neutrino source experiments Phys. Lett. B 431 110-8
- [266] Krofcheck D et al 1985 Gamow-Teller strength function in ⁷¹GE via the (p, n) reaction at medium-energies Phys. Rev. Lett. 55 1051-4

- [267] Honma M, Otsuka T, Mizusaki T and Hjorth-Jensen M 2009 New effective interaction for f₅pg₉shell nuclei Phys. Rev. C 80 064323
- [268] Kostensalo J, Suhonen J, Giunti C and Srivastava P C 2019 The gallium anomaly revisited Phys. Lett. B 795 542–7
- [269] Gemmeke H et al 1990 The High resolution neutrino calorimeter KARMEN Nucl. Instrum. Meth. A 289 490–5
- [270] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2008 Measurement of muon neutrino quasi-elastic scattering on carbon Phys. Rev. Lett. 100 032301
- [271] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2009 A Search for muon neutrino and antineutrino disappearance in MiniBooNE Phys. Rev. Lett. 103 061802
- [272] Naples D *et al* (CCFR/NuTeV Collaboration) 1999 A high statistics search for $\nu_e (\nu_e) \rightarrow \nu_\tau (\nu_\tau)$ oscillations *Phys. Rev.* D **59** 031101
- [273] Stockdale I E *et al* 1984 Search for ν_{μ} and $\bar{\nu}$ oscillations in the mass range 15 eV²/ $c^4 < \delta M^2 < 1000 \text{ eV}^2/c^4$ Conf. Proc. C 841031 258
- [274] Mahn K B M *et al* (SciBooNE, MiniBooNE Collaboration) 2012 Dual baseline search for muon neutrino disappearance at 0.5 eV² $< \Delta m^2 < 40 \text{ eV}^2$ Phys. Rev. D **85** 032007
- [275] Cheng G *et al* (MiniBooNE, SciBooNE Collaboration) 2012 Dual baseline search for muon antineutrino disappearance at 0.1 eV² < Δm^2 < 100 eV² Phys. Rev. D **86** 052009
- [276] Adamson P et al (MINOS Collaboration) 2011 Active to sterile neutrino mixing limits from neutral-current interactions in MINOS Phys. Rev. Lett. 107 011802
- [277] Adamson P et al (MiniBooNE, MINOS Collaboration) 2009 First measurement of ν_{μ} and ν_{e} events in an off-axis horn-focused neutrino beam *Phys. Rev. Lett.* **102** 211801
- [278] Abratenko P et al (MicroBooNE Collaboration) 2023 First constraints on light sterile neutrino oscillations from combined appearance and disappearance searches with the MicroBooNE detector Phys. Rev. Lett. 130 011801
- [279] Abratenko P et al (MicroBooNE Collaboration) 2022 Search for an anomalous excess of inclusive charged-current ve interactions in the MicroBooNE experiment using Wire-Cell reconstruction Phys. Rev. D 105 112005
- [280] Abratenko P et al (MicroBooNE Collaboration) 2022 Search for an excess of electron neutrino interactions in MicroBooNE using multiple final-state topologies Phys. Rev. Lett. 128 241801
- [281] Abratenko P et al (MicroBooNE Collaboration) 2022 Search for an anomalous excess of charged-current quasi-elastic ve interactions with the MicroBooNE experiment using deeplearning-based reconstruction Phys. Rev. D 105 112003
- [282] Abratenko P et al (MicroBooNE Collaboration) 2022 Search for an anomalous excess of charged-current ve interactions without pions in the final state with the MicroBooNE experiment Phys. Rev. D 105 112004
- [283] Peter B Denton 2021 Sterile neutrino searches with MicroBooNE: electron neutrino disappearance Phys. Rev. Lett. 129 061801
- [284] Aguilar-Arevalo A A et al (MiniBooNE Collaboration) 2022 MiniBooNE and MicroBooNE combined fit to a 3+1 sterile neutrino scenario Phys. Rev. Lett. 129 201801
- [285] Astier P *et al* (NOMAD Collaboration) 2003 Search for $\nu_{\mu} \rightarrow \nu_{e}$ oscillations in the NOMAD experiment *Phys. Lett.* B **570** 19–31
- [286] Stockdale I E et al 1984 Limits on muon neutrino oscillations in the mass range 55-eV**2
 Delta m**2 < 800-eV**2 Phys. Rev. Lett. 52 1384</p>
- [287] Dydak F *et al* 1984 A search for muon-neutrino oscillations in the Δm^2 range 0.3–eV² to 90 eV² *Phys. Lett.* V **134** 281
- [288] Adamson P et al (NOvA Collaboration) 2017 Search for active-sterile neutrino mixing using neutral-current interactions in NOvA Phys. Rev. D 96 072006
- [289] Acero M A et al (NOvA Collaboration) 2021 Search for active-sterile antineutrino mixing using neutral-current interactions with the NOvA experiment *Phys. Rev. Lett.* **127** 201801
- [290] Abe K et al (T2K Collaboration) 2011 The T2K experiment Nucl. Instrum. Meth. A 659 106-35
- [291] Fukuda Y et al (Super-Kamiokande Collaboration) 2003 The Super-Kamiokande detector Nucl. Instrum. Meth. A 501 418
- [292] Abe K et al (T2K Collaboration) 2013 T2K neutrino flux prediction Phys. Rev. D 87 012001
- [293] Abe K et al 2012 Measurements of the T2K neutrino beam properties using the INGRID on-axis near detector Nucl. Instrum. Meth. A 694 211–23
- [294] Koga T et al 2015 Water/CH neutrino cross section measurement at J-PARC (WAGASCI experiment) JPS Conf. Proc. 8 023003

- [295] Hallsjö S-P and (Baby MIND Collaboration) 2018 Baby MIND: a magnetised spectrometer for the WAGASCI experiment PoS NuFact2017 078
- [296] Abe K et al (T2K Collaboration) 2019 T2K ND280 upgrade—technical design report arXiv:1901.03750
- [297] Abdurashitov J N et al (SAGE Collaboration) 2009 Measurement of the solar neutrino capture rate with gallium metal: III. Results for the 2002–2007 data-taking period Phys. Rev. C 80 015807
- [298] Abe K *et al* (T2K Collaboration) 2015 Search for short baseline ν_e disappearance with the T2K near detector *Phys. Rev.* D **91** 051102
- [299] Abe K et al (T2K Collaboration) 2019 Search for light sterile neutrinos with the T2K far detector Super-Kamiokande at a baseline of 295 km Phys. Rev. D 99 071103
- [300] Abe K *et al* (T2K Collaboration) 2021 Improved constraints on neutrino mixing from the T2K experiment with 3.13×10^{21} protons on target *Phys. Rev.* D 103 112008
- [301] Bordoni S, Dewhurst D, Stowell P and (T2K Collaboration) 2017 Charged-current muon neutrino shape and rate analysis at the T2K off-axis near detector J. Phys. Conf. Ser. 888 012188
- [302] Bodek A, Sarica U, Naples D and Ren L 2012 Methods to determine neutrino flux at low energies:investigation of the low ν method *Eur. Phys. J.* C **72** 1973
- [303] Acquafredda R et al 2009 The OPERA experiment in the CERN to Gran Sasso neutrino beam J. Instrum. 4 P04018
- [304] Ade P A R et al (Planck Collaboration) 2016 Planck 2015 results. XIII. Cosmological parameters Astron. Astrophys. 594 A13
- [305] Antonello M et al 2013 Experimental search for the "LSND anomaly" with the ICARUS detector in the CNGS neutrino beam Eur. Phys. J. C 73 2345
- [306] Palazzo A 2015 Consistent analysis of the $\nu_{\mu} \rightarrow \nu_{e}$ sterile neutrinos searches of ICARUS and OPERA *Phys. Rev.* D **91** 091301
- [307] Danilov M 2020 New results from the DANSS experiment arXiv:2211.01208
- [308] Atif Z et al (RENO, NEOS Collaboration) 2020 Search for sterile neutrino oscillation using RENO and NEOS data Phys. Rev. D 105 L111101
- [309] Danilov M V and Skrobova N A 2020 Comment on "Analysis of the results of the Neutrino-4 experiment on the search for the sterile neutrino and comparison with results of other experiments" (JETP Letters 112, 199 (2020)) JETP Lett. 112 452–4
- [310] Giunti C 2020 Statistical significance of reactor antineutrino active-sterile oscillations *Phys. Rev.* D 101 095025
- [311] Andriamirado M et al (PROSPECT, STEREO Collaboration) 2020 Preparation of the Neutrino-4 experiment on search for sterile neutrino and the obtained results of measurements arXiv:2005. 05301
- [312] Ashenfelter J et al (PROSPECT Collaboration) 2018 Performance of a segmented ⁶Li-loaded liquid scintillator detector for the PROSPECT experiment J. Instrum. 13 P06023
- [313] Almazán H et al (STEREO Collaboration) 2023 Interpreting reactor antineutrino anomalies with STEREO data Nature 613 257
- [314] Ashenfelter J et al (PROSPECT Collaboration) 2016 The PROSPECT physics program J. Phys. G G 43 113001
- [315] Ashenfelter J et al (PROSPECT Collaboration) 2019 The PROSPECT reactor antineutrino experiment Nucl. Instrum. Meth. A 922 287–309
- [316] Ashenfelter J et al (PROSPECT Collaboration) 2019 Lithium-loaded Liquid Scintillator Production for the PROSPECT experiment J. Instrum. 14 P03026
- [317] Ashenfelter J et al (PROSPECT Collaboration) 2015 Light collection and pulse-shape discrimination in elongated scintillator cells for the PROSPECT reactor antineutrino experiment J. Instrum. 10 P11004
- [318] Ashenfelter J et al (PROSPECT Collaboration) 2019 A low mass optical grid for the PROSPECT reactor antineutrino detector J. Instrum. 14 P04014
- [319] Ashenfelter J et al (PROSPECT Collaboration) 2016 Background radiation measurements at high power research reactors Nucl. Instrum. Meth. A 806 401–19
- [320] Feldman G J and Cousins R D 1998 A unified approach to the classical statistical analysis of small signals *Phys. Rev.* D 57 3873–89
- [321] Qian X, Tan A, Ling J J, Nakajima Y and Zhang C 2016 The Gaussian CL_s method for searches of new physics *Nucl. Instrum. Meth.* A 827 63–78

- [322] Almazán H et al (STEREO Collaboration) 2018 Sterile neutrino constraints from the STEREO experiment with 66 days of reactor-on data Phys. Rev. Lett. 121 161801
- [323] An F P et al (Daya Bay Collaboration) 2014 Search for a light sterile neutrino at Daya Bay Phys. Rev. Lett. 113 141802
- [324] Adamson P et al (Daya Bay, MINOS Collaboration) 2016 Limits on active to sterile neutrino oscillations from disappearance searches in the MINOS, Daya Bay, and Bugey-3 experiments *Phys. Rev. Lett.* **117** 151801
- [325] Adamson P et al (MINOS+, Daya Bay Collaboration) 2020 Improved constraints on sterile neutrino mixing from disappearance searches in the MINOS, MINOS+, Daya Bay, and Bugey-3 experiments Phys. Rev. Lett. 125 071801
- [326] AbrahÃo T et al 2021 Search for signatures of sterile neutrinos with double chooz Eur. Phys. J. C 81
- [327] Choi J H et al (RENO Collaboration) 2016 Observation of energy and baseline dependent reactor antineutrino disappearance in the RENO experiment Phys. Rev. Lett. 116 211801
- [328] Park K S et al 2012 Construction and properties of acrylic vessels in the RENO detector Nucl. Instrum. Meth. A 686 91–9
- [329] Park J S et al 2013 Production and optical properties of Gd-loaded liquid scintillator for the RENO neutrino detector Nucl. Instrum. Meth. A 707 45–53
- [330] Ma K J et al 2011 Time and amplitude of afterpulse measured with a large size photomultiplier tube Nucl. Instrum. Meth. A 629 93–100
- [331] Bak G et al (RENO Collaboration) 2018 Measurement of reactor antineutrino oscillation amplitude and frequency at RENO Phys. Rev. Lett. 121 201801
- [332] Read A L 2002 Presentation of search results: the CL(s) technique J. Phys. G 28 2693-704
- [333] Berryman J M, Coloma P, Huber P, Schwetz T and Zhou A 2022 Statistical significance of the sterile-neutrino hypothesis in the context of reactor and gallium data *J. High Energy Phys.* JHEP02(2022)11
- [334] An F P et al Joint experimental oscillation analyses in search of sterile neutrinos. Snowmass 2021 letter of interest, 2020. https://www.snowmass21.org/docs/files/summaries/NF/ SNOWMASS21-NF2_NF1_Joint_Oscillation_Analyses_at_Reactors-115.pdf
- [335] An F P et al Legacy of the daya bay reactor neutrino experiment. Snowmass 2021 letter of interest, 2020. https://www.snowmass21.org/docs/files/summaries/NF/SNOWMASS21-NF1_NF2_Daya_Bay-086.pdf
- [336] Aartsen M G et al (IceCube Collaboration) 2017 The IceCube neutrino observatory contributions to ICRC 2017 part IV: searches for beyond the standard model physics arXiv:1710.01197
- [337] Esmaili A and Smirnov A Y 2013 Restricting the LSND and MiniBooNE sterile neutrinos with the IceCube atmospheric neutrino data J. High Energy Phys. JHEP12(2013)014
- [338] Nunokawa H, Peres O L G and Funchal R Z 2003 Probing the LSND mass scale and four neutrino scenarios with a neutrino telescope *Phys. Lett. B* 562 279–90
- [339] Petcov S T 2017 On the IceCube result on *v

 μ

 μ* → *v

 s* oscillations *Int. J. Mod. Phys.* A 32 1750018
 [340] Barger V, Gao Y and Marfatia D 2012 Is there evidence for sterile neutrinos in IceCube data? *Phys. Rev.* D 85 011302
- [341] Coloma P, Lopez-Pavon J, Martinez-Soler I and Nunokawa H 2018 Decoherence in neutrino propagation through matter, and bounds from IceCube/DeepCore Eur. Phys. J. C 78 614
- [342] Stuttard T and Jensen M 2020 Neutrino decoherence from quantum gravitational stochastic perturbations *Phys. Rev.* D 102 115003
- [343] Stuttard T 2021 Neutrino signals of lightcone fluctuations resulting from fluctuating spacetime Phys. Rev. D 104 056007
- [344] Abbasi R et al (IceCube Collaboration) 2010 Search for a Lorentz-violating sidereal signal with atmospheric neutrinos in IceCube Phys. Rev. D 82 112003
- [345] Aartsen M G et al (IceCube Collaboration) 2018 Neutrino interferometry for high-precision tests of Lorentz symmetry with IceCube Nat. Phys. 14 961–6
- [346] Katori T, Argüelles C A, Farrag K, Mandalia S and (IceCube Collaboration) 2020 Test of lorentz violation with astrophysical neutrino flavor at IceCube 8th Meeting on CPT and Lorentz Symmetry pp 166–9 arXiv:1906.09240
- [347] Smithers B R, Jones B J P, Argüelles C A, Conrad J M and Diaz A 2021 Cascade appearance signatures of sterile neutrinos at 1–100 TeV Phys. Rev. D 105 11

- [348] Abbasi R *et al* (IceCube Collaboration) 2022 Search for quantum gravity using astrophysical neutrino flavour with IceCube *Nature Phys.* **18** 1287
- [349] Salvado J, Mena O, Palomares-Ruiz S and Rius N 2017 Non-standard interactions with highenergy atmospheric neutrinos at IceCube J. High Energy Phys. JHEP01(2017)141
- [350] Abbasi R et al (IceCube Collaboration) 2021 All-flavor constraints on nonstandard neutrino interactions and generalized matter potential with three years of IceCube DeepCore data Phys. Rev. D 104 072006
- [351] Abbasi R *et al* (IceCube Collaboration) 2022 Strong constraints on neutrino nonstandard interactions from TeV-scale ν_{μ} disappearance at IceCube *Phys. Rev. Lett.* **129** 011804
- [352] Razzaque S and Smirnov A Y 2012 Searches for sterile neutrinos with IceCube DeepCore Phys. Rev. D 85 093010
- [353] Trettin A and (IceCube Collaboration) 2021 Sensitivity of a search for eV-scale sterile neutrinos with 8 years of IceCube DeepCore data J. Instrum. 16 C09005
- [354] Lindner M, Rodejohann W and Xu X-J 2016 Sterile neutrinos in the light of IceCube J. High Energy Phys. JHEP01(2016)124
- [355] Gavrin V N, Gorbachev V V, Veretenkin E P and Cleveland B T 2010 Gallium experiments with artificial neutrino sources as a tool for investigation of transition to sterile states arXiv:1006.2103
- [356] Abe K et al (Super-Kamiokande Collaboration) 2015 Limits on sterile neutrino mixing using atmospheric neutrinos in Super-Kamiokande Phys. Rev. D 91 052019
- [357] Ageron M et al (ANTARES Collaboration) 2011 ANTARES: the first undersea neutrino telescope Nucl. Instrum. Meth. A 656 11–38
- [358] Albert A et al (ANTARES Collaboration) 2019 Measuring the atmospheric neutrino oscillation parameters and constraining the 3 + 1 neutrino model with ten years of ANTARES data J. High Energy Phys. JHEP06(2019)113
- [359] Abe K et al (Super-Kamiokande Collaboration) 2015 Limits on sterile neutrino mixing using atmospheric neutrinos in Super-Kamiokande Phys. Rev. D 91 052019
- [360] Barinov V, Gavrin V, Gorbunov D and Ibragimova T 2016 BEST sensitivity to O(1) eV sterile neutrino Phys. Rev. D 93 073002
- [361] Barinov V V et al 2022 Search for electron-neutrino transitions to sterile states in the BEST experiment Phys. Rev. C 105 065502
- [362] Belesev A I, Berlev A I, Geraskin E V, Golubev A A, Likhovid N A, Nozik A A, Pantuev V S, Parfenov V I and Skasyrskaya A K 2014 The search for an additional neutrino mass eigenstate in the 2–100 eV region from 'Troitsk nu-mass' data: a detailed analysis J. Phys. G 41 015001
- [363] Abdurashitov J N et al 2017 First measeurements in search for keV-sterile neutrino in tritium beta-decay by Troitsk nu-mass experiment Pisma Zh. Eksp. Teor. Fiz. 105 723–4
- [364] Aker M et al (KATRIN Collaboration) 2022 Improved eV-scale sterile-neutrino constraints from the second KATRIN measurement campaign Phys. Rev. D 105 072004
- [365] Aker M et al (KATRIN Collaboration) 2023 Search for keV-scale sterile neutrinos with first KATRIN data Eur. Phys. J. C 83 763
- [366] Galeazzi M, Fontanelli F, Gatti F and Vitale S 2001 Limits on the existence of heavy neutrinos in the range 50–1000 eV from the study of the ¹⁸⁷Re beta decay *Phys. Rev. Lett.* 86 1978–81
- [367] Holzschuh E, Kundig W, Palermo L, Stussi H and Wenk P 1999 Search for heavy neutrinos in the beta spectrum of ⁶³Ni Phys. Lett. B 451 247–55
- [368] Holzschuh E, Palermo L, Stussi H and Wenk P 2000 The beta-spectrum of ³⁵S and search for the admixture of heavy neutrinos *Phys. Lett.* B 482 1–9
- [369] Schreckenbach K, Colvin G and Von Feilitzsch F 1983 Search for mixing of heavy neutrinos in the β^+ and β^- spectra of the ⁶⁴Cu decay *Phys. Lett.* B **129** 265–8
- [370] Derbin A V, Drachnev I S, Lomskaya I S, Muratova V N, Pilipenko N V, Semenov D A, Tukkhonen L M, Unzhakov E V and Khusainov A K 2018 Search for a neutrino with a mass of 0.01–1.0 MeV in beta decays of ¹⁴⁴Ce–¹⁴⁴Pr nuclei *JETP Lett.* **108** 499–503
- [371] Friedrich S et al 2021 Limits on the existence of sub-MeV sterile neutrinos from the decay of ⁷Be in superconducting quantum sensors Phys. Rev. Lett. **126** 021803
- [372] Deutsch J, Lebrun M and Prieels R 1990 Searches for admixture of massive neutrinos into the electron flavor Nucl. Phys. A 518 149–55
- [373] Osipowicz A et al (KATRIN Collaboration) 2001 KATRIN: a next generation tritium beta decay experiment with sub-eV sensitivity for the electron neutrino mass. Letter of intent arXiv:hepex/0109033

- [374] Angrik J et al (KATRIN Collaboration) 2005 KATRIN Design Report 2004 FZKA-7090
- [375] Arenz M et al (KATRIN Collaboration) 2018 First transmission of electrons and ions through the KATRIN beamline J. Instrum. 13 P04020
- [376] Aker M et al (KATRIN Collaboration) 2019 Improved upper limit on the neutrino mass from a direct kinematic method by KATRIN Phys. Rev. Lett. 123 221802
- [377] Aker M et al (KATRIN Collaboration) 2022 Direct neutrino-mass measurement with subelectronvolt sensitivity Nat. Phys. 18 160–6
- [378] Aker M et al (KATRIN Collaboration) 2021 Bound on 3+1 active-sterile neutrino mixing from the first four-week science run of KATRIN Phys. Rev. Lett. 126 091803
- [379] Aker M et al (KATRIN Collaboration) 2021 Analysis methods for the first KATRIN neutrinomass measurement Phys. Rev. D 104 012005
- [380] Batell B, Pospelov M and Ritz A 2009 Exploring portals to a hidden sector through fixed targets *Phys. Rev.* D 80 095024
- [381] Bjorken J D, Essig R, Schuster P and Toro N 2009 New fixed-target experiments to search for dark gauge forces *Phys. Rev.* D 80 075018
- [382] deNiverville P, Pospelov M and Ritz A 2011 Observing a light dark matter beam with neutrino experiments *Phys. Rev.* D 84 075020
- [383] deNiverville P, McKeen D and Ritz A 2012 Signatures of sub-GeV dark matter beams at neutrino experiments *Phys. Rev.* D 86 035022
- [384] Abe K et al (T2K Collaboration) 2019 Search for heavy neutrinos with the T2K near detector ND280 Phys. Rev. D 100 052006
- [385] Abratenko P et al (MicroBooNE Collaboration) 2020 Search for heavy neutral leptons decaying into muon-pion pairs in the MicroBooNE detector Phys. Rev. D 101 052001
- [386] Acciarri R *et al* (ArgoNeuT Collaboration) 2021 New constraints on tau-coupled heavy neutral leptons with masses $m_N = 280-970$ MeV *Phys. Rev. Lett.* **127** 121801
- [387] Bernardi G et al 1988 Further limits on heavy neutrino couplings Phys. Lett. B 203 332-4
- [388] Argüelles C A, Foppiani N and Hostert M 2022 Heavy neutral leptons below the kaon mass at hodoscopic neutrino detectors *Phys. Rev.* D 105 095006
- [389] Gorbunov D, Krasnov I and Suvorov S 2022 Revisiting PS191 limits on sterile neutrinos Phys. Lett. B 830 137173
- [390] Bergsma F et al (CHARM Collaboration) 1986 A search for decays of heavy neutrinos in the mass range 0.5–2.8 GeV Phys. Lett. B 166 473–8
- [391] Vilain P et al (CHARM II Collaboration) 1995 Search for heavy isosinglet neutrinos Phys. Lett. B 343 453–8
- [392] Vaitaitis A et al (NuTeV, E815 Collaboration) 1999 Search for neutral heavy leptons in a highenergy neutrino beam Phys. Rev. Lett. 83 4943–6
- [393] Boyarsky A, Ovchynnikov M, Ruchayskiy O and Syvolap V 2021 Improved big bang nucleosynthesis constraints on heavy neutral leptons *Phys. Rev.* D 104 023517
- [394] Abe K et al (T2K Collaboration) 2019 Search for heavy neutrinos with the T2K near detector ND280 Phys. Rev. D 100 052006
- [395] Argüelles C A, Foppiani N and Hostert M 2023 Efficiently exploring multi-dimensional parameter spaces beyond the standard model *Phys. Rev.* D 107 035127
- [396] Abe K et al 2016 Proposal for an extended run of T2K to 20×1021 POT arXiv:1609.04111
- [397] Blanpied G *et al* 1997 $N \to \Delta$ transition from simultaneous measurements of $p(\gamma, \pi)$ and $p(\gamma, \gamma)$ *Phys. Rev. Lett.* **79** 4337–40
- [398] Sparveris N F *et al* 2008 Virtual compton scattering measurements in the $\gamma * N \rightarrow \Delta$ transition *Phys. Rev.* C **78** 018201
- [399] Gebre Y, Littlejohn B R and Surukuchi P T 2018 Prospects for improved understanding of isotopic reactor antineutrino fluxes *Phys. Rev.* D 97 013003
- [400] Almazán H et al (STEREO Collaboration) 2020 Accurate measurement of the electron antineutrino yield of ²³⁵U fssions from the STEREO experiment with 119 days of reactor-on data Phys. Rev. Lett. 125 201801
- [401] Kopeikin V, Skorokhvatov M and Titov O 2021 Reevaluating reactor antineutrino spectra with new measurements of the ratio between ²³⁵U and ²³⁹Pu β spectra *Phys. Rev.* D **104** L071301
- [402] Bergevin M et al 2019 Applied antineutrino physics 2018 proceedings arXiv:1911.06834
- [403] Giunti C, Ji X P, Laveder M, Li Y F and Littlejohn B R 2017 Reactor fuel fraction information on the antineutrino anomaly J. High Energy Phys. JHEP10(2017)143

- [404] Giunti C, Li Y F, Littlejohn B R and Surukuchi P T 2019 Diagnosing the reactor antineutrino anomaly with global antineutrino flux data *Phys. Rev.* D 99 073005
- [405] Raffelt G G 1996 Stars as Laboratories for Fundamental Physics: The Astrophysics of Neutrinos, Axions, and Other Weakly Interacting Particles (The University of Chicago Press) vol 5
- [406] Froustey J, Pitrou C and Volpe M C 2020 Neutrino decoupling including flavour oscillations and primordial nucleosynthesis J. Cosmol. Astropart. Phys. JCAP12(2020)015
- [407] Bennett J J, Buldgen G, Salas P F D, Drewes M, Gariazzo S, Pastor S and Wong Y Y 2021 Towards a precision calculation of N_{eff} in the standard model: II. Neutrino decoupling in the presence of flavour oscillations and finite-temperature QEDJCAP04(2021)073
- [408] Fixsen D J, Cheng E S, Gales J M, Mather J C, Shafer R A and Wright E L 1996 The cosmic microwave background spectrum from the full COBE FIRAS data set Astrophys. J. 473 576
- [409] Fixsen D J 2009 The Temperature of the cosmic microwave background Astrophys. J. 707 916–20
- [410] Hasenkamp J 2012 Dark radiation from the axino solution of the gravitino problem Phys. Lett. B 707 121–8
- [411] Hasenkamp J 2014 Daughters mimic sterile neutrinos (almost!) perfectly J. Cosmol. Astropart. Phys. JCAP09(2014)048
- [412] Chen M-C, Ratz M and Trautner A 2015 Nonthermal cosmic neutrino background *Phys. Rev.* D 92 123006
- [413] Roland S B and Shakya B 2017 Cosmological imprints of frozen-in light sterile neutrinos J. Cosmol. Astropart. Phys. JCAP05(2017)027
- [414] Shakya B 2016 Sterile neutrino dark matter from freeze-in *Mod. Phys. Lett.* A **31** 1630005
- [415] Shakya B and Wells J D 2017 Sterile neutrino dark matter with supersymmetry Phys. Rev. D 96 031702
- [416] Shakya B and Wells J D 2019 Exotic sterile neutrinos and pseudo-goldstone phenomenology J. High Energy Phys. JHEP02(2019)174
- [417] Davidson S et al 2000 Updated bounds on millicharged particles J. High Energy Phys. JHEP05 (2000)003
- [418] Diacoumis J A D and Wong Y Y 2019 Trading kinetic energy: how late kinetic decoupling of dark matter changes N_{eff} J. Cosmol. Astropart. Phys. JCAP01(2019)001
- [419] Kainulainen K 1990 Light singlet neutrinos and the primordial nucleosynthesis Phys. Lett. B 244 191–5
- [420] Enqvist K, Kainulainen K and Thomson M J 1992 Stringent cosmological bounds on inert neutrino mixing Nucl. Phys. B 373 498–528
- [421] Cline J M 1992 Constraints on almost Dirac neutrinos from neutrino-anti-neutrino oscillations Phys. Rev. Lett. 68 3137–40
- [422] Shi X, Schramm D N and Fields B D 1993 Constraints on neutrino oscillations from big bang nucleosynthesis *Phys. Rev.* D 48 2563–72
- [423] Sigl G and Raffelt G 1993 General kinetic description of relativistic mixed neutrinos Nucl. Phys. D 406 423–51
- [424] McKellar B H J and Thomson M J 1994 Oscillating doublet neutrinos in the early universe Phys. Rev. D 49 2710–28
- [425] Oldengott I M and Schwarz D J 2017 Improved constraints on lepton asymmetry from the cosmic microwave background *Europhys. Lett.* 119 29001
- [426] Gariazzo S, de Salas P F and Pastor S 2019 Thermalisation of sterile neutrinos in the early Universe in the 3+1 scheme with full mixing matrix J. Cosmol. Astropart. Phys. JCAP07 (2019)014
- [427] Notzold D and Raffelt G 1988 Neutrino dispersion at finite temperature and density Nucl. Phys. B 307 924–36
- [428] Hannestad S, Tamborra I and Tram T 2012 Thermalisation of light sterile neutrinos in the early universe J. Cosmol. Astropart. Phys. JCAP07(2012)025
- [429] Bell N F, Volkas R R and Wong Y Y Y 1999 Relic neutrino asymmetry evolution from first principles *Phys. Rev.* D 59 113001
- [430] Kainulainen K and Sorri A 2002 Oscillation induced neutrino asymmetry growth in the early universe J. High Energy Phys. JHEP02(2002)020
- [431] Chu Y-Z and Cirelli M 2006 Sterile neutrinos, lepton asymmetries, primordial elements: how much of each? *Phys. Rev.* D 74 085015

- [432] Hannestad S, Hansen R S, Tram T and Wong Y Y 2015 Active-sterile neutrino oscillations in the early Universe with full collision terms J. Cosmol. Astropart. Phys. JCAP08(2015)019
- [433] Hannestad S, Hansen R S and Tram T 2013 Can active-sterile neutrino oscillations lead to chaotic behavior of the cosmological lepton asymmetry? J. Cosmol. Astropart. Phys. JCAP04 (2013)032
- [434] Melchiorri A, Mena O, Palomares-Ruiz S, Pascoli S, Slosar A and Sorel M 2009 Sterile neutrinos in light of recent cosmological and oscillation data: a multi-flavor scheme approach J. Cosmol. Astropart. Phys. JCAP01(2009)036
- [435] Pitrou C, Coc A, Uzan J-P and Vangioni E 2018 Precision big bang nucleosynthesis with improved Helium-4 predictions *Phys. Rep.* 754 1–66
- [436] Arbey A 2012 AlterBBN: a program for calculating the BBN abundances of the elements in alternative cosmologies *Comput. Phys. Commun.* 183 1822–31
- [437] Arbey A, Auffinger J, Hickerson K P and Jenssen E S 2020 AlterBBN v2: a public code for calculating Big-Bang nucleosynthesis constraints in alternative cosmologies *Comput. Phys. Commun.* 248 106982
- [438] Pisanti O, Cirillo A, Esposito S, Iocco F, Mangano G, Miele G and Serpico P D 2008 PArthENoPE: public algorithm evaluating the nucleosynthesis of primordial elements *Comput. Phys. Commun.* **178** 956–71
- [439] Consiglio R, de Salas P F, Mangano G, Miele G, Pastor S and Pisanti O 2018 PArthENoPE reloaded *Comput. Phys. Commun.* 233 237–42
- [440] Aver E, Olive K A and Skillman E D 2015 The effects of He I λ10830 on helium abundance determinationsJCAP07(2015)011
- [441] Cooke R J, Pettini M and Steidel C C 2018 One percent determination of the primordial deuterium abundance Astrophys. J. 855 102
- [442] Sbordone L et al 2010 The metal-poor end of the Spite plateau: I. Stellar parameters, metallicities and lithium abundances Astron. Astrophys. 522 A26
- [443] Bania T M, Rood R T and Balser D S 2002 The cosmological density of baryons from observations of 3He+ in the Milky Way Nature 415 54–7
- [444] Hou Z, Keisler R, Knox L, Millea M and Reichardt C 2013 How massless neutrinos affect the cosmic microwave background damping tail *Phys. Rev.* D 87 083008
- [445] Aghanim N et al (Planck Collaboration) 2020 Planck 2018 results. VI. Cosmological parameters Astron. Astrophys. 641 A6
- [446] Dunkley J et al 2011 The atacama cosmology telescope: cosmological parameters from the 2008 power spectra Astrophys. J. 739 52
- [447] Keisler R et al 2011 A measurement of the damping tail of the cosmic microwave background power spectrum with the south pole telescope Astrophys. J. 743 28
- [448] Beutler F, Blake C, Colless M, Jones D H, Staveley-Smith L, Campbell L, Parker Q, Saunders W and Watson F 2011 The 6dF Galaxy survey: baryon acoustic oscillations and the local hubble constant *Mon. Not. R. Astron. Soc.* **416** 3017–32
- [449] Ross A J, Samushia L, Howlett C, Percival W J, Burden A and Manera M 2015 The clustering of the SDSS DR7 main Galaxy sample –I. A 4 percent distance measure at z = 0.15 Mon. Not. R. Astron. Soc. 449 835–47
- [450] Alam S et al (BOSS Collaboration) 2017 The clustering of galaxies in the completed SDSS-III baryon oscillation spectroscopic survey: cosmological analysis of the dR12 galaxy sample Mon. Not. R. Astron. Soc. 470 2617–52
- [451] Hagstotz S, de Salas P F, Gariazzo S, Gerbino M, Lattanzi M, Vagnozzi S, Freese K and Pastor S 2021 Bounds on light sterile neutrino mass and mixing from cosmology and laboratory searches *Phys. Rev.* D 104 123524
- [452] Riess A G et al 2018 New parallaxes of galactic cepheids from spatially scanning the hubble space telescope: implications for the hubble constant Astrophys. J. 855 136
- [453] Scolnic D M et al (Pan-STARRS1 Collaboration) 2018 The complete light-curve sample of spectroscopically confirmed SNe Ia from Pan-STARRS1 and cosmological constraints from the combined pantheon sample Astrophys. J. 859 101
- [454] Hamann J, Hannestad S, Raffelt G G and Wong Y Y Y 2011 Sterile neutrinos with eV masses in cosmology: how disfavoured exactly? JCAP 09 034
- [455] Foot R and Volkas R R 1995 Reconciling sterile neutrinos with big bang nucleosynthesis Phys. Rev. Lett. 75 4350

- [456] Abazajian K, Bell N F, Fuller G M and Wong Y Y Y 2005 Cosmological lepton asymmetry, primordial nucleosynthesis, and sterile neutrinos *Phys. Rev.* D 72 063004
- [457] Mirizzi A, Saviano N, Miele G and Serpico P D 2012 Light sterile neutrino production in the early universe with dynamical neutrino asymmetries *Phys. Rev.* D 86 053009
- [458] Saviano N, Pisanti O, Mangano G and Mirizzi A 2014 Unveiling secret interactions among sterile neutrinos with big-bang nucleosynthesis *Phys. Rev.* D 90 113009
- [459] Archidiacono M, Hannestad S, Hansen R S and Tram T 2015 Cosmology with self-interacting sterile neutrinos and dark matter—a pseudoscalar model *Phys. Rev.* D 91 065021
- [460] Archidiacono M, Gariazzo S, Giunti C, Hannestad S, Hansen R, Laveder M and Tram T 2016 Pseudoscalar—sterile neutrino interactions: reconciling the cosmos with neutrino oscillationsJ-CAP08(2019)067
- [461] Chu X, Dasgupta B and Kopp J 2015 Sterile neutrinos with secret interactions–lasting friendship with cosmologyJCAP10(2015)011
- [462] Forastieri F, Lattanzi M, Mangano G, Mirizzi A, Natoli P and Saviano N 2017 Cosmic microwave background constraints on secret interactions among sterile neutrinosJCAP07 (2017)038
- [463] Cherry J F, Friedland A and Shoemaker I M 2016 Short-baseline neutrino oscillations, Planck, and IceCube5 arXiv:1605.06506
- [464] Gelmini G, Palomares-Ruiz S and Pascoli S 2004 Low reheating temperature and the visible sterile neutrino Phys. Rev. Lett. 93 081302
- [465] Gelmini G B, Lu P and Takhistov V 2020 Visible sterile neutrinos as the earliest relic probes of cosmology Phys. Lett. B 800 135113
- [466] Gelmini G B, Lu P and Takhistov V 2019 Cosmological dependence of non-resonantly produced sterile neutrinos J. Cosmol. Astropart. Phys. JCAP12(2019)047
- [467] Hasegawa T, Hiroshima N, Kohri K, Hansen R S L, Tram T and Hannestad S 2020 MeV-scale reheating temperature and cosmological production of light sterile neutrinos J. Cosmol. Astropart. Phys. JCAP08(2020)015
- [468] Gelmini G B, Lu P and Takhistov V 2020 Cosmological dependence of resonantly produced sterile neutrinosJCAP06(2020)008
- [469] Harada M et al (JSNS2 Collaboration) 2013 Proposal: a search for sterile neutrino at J-PARC materials and life science experimental facility arXiv:1310.1437
- [470] Ajimura S et al 2017 Technical design report (TDR): searching for a sterile neutrino at J-PARC MLF (E56, JSNS2) arXiv:1705.08629
- [471] Ajimura S et al (JSNS2 Collaboration) 2021 The JSNS2 detector Nucl. Instrum. Meth. A 1014 165742
- [472] Ajimura S et al 2020 Proposal: JSNS2-II arXiv:2012.10807
- [473] Hino Y et al 2022 Characterization of the correlated background for a sterile neutrino search using the first dataset of the JSNS2 experiment Eur. Phys. J. C 82 331
- [474] Akimov D et al (COHERENT Collaboration) 2017 Observation of coherent elastic neutrinonucleus scattering Science 357 1123-6
- [475] Akimov D et al (COHERENT Collaboration) 2021 First measurement of coherent elastic neutrino-nucleus scattering on argon Phys. Rev. Lett. 126 012002
- [476] Chernyak D, Pershey D, Liu J, Ding K, Saunders N and Oli T 2020 Prospect of undoped inorganic crystals at 77 Kelvin for low-mass dark matter search at Spallation Neutron Source *Eur. Phys. J.* C 80 547
- [477] Aguilar-Arevalo A A et al 2022 First leptophobic dark matter search from coherent CAPTAIN-Mills Phys. Rev. Lett. 129 021801
- [478] Aguilar-Arevalo A A et al (CCM Collaboration) 2022 First dark matter search results from coherent CAPTAIN-Mills Phys. Rev. D 106 012001
- [479] Aguilar-Arevalo A A et al (CCM Collaboration) 2023 Axion-like particles at coherent CAPTAIN-Mills Phys. Rev. D 107 095036
- [480] Rott C and (JSNS2 Collaboration) 2020 Status of JSNS² J-PARC sterile neutrino search at J-PARC spallation neutron source J. Phys. Conf. Ser. 1468 012176
- [481] Toups M et al 2022 PIP2-BD: GeV proton beam dump at Fermilab's PIP-II Linac3 2022 Snowmass Summer Study arXiv:2203.08079
- [482] Ainsworth R et al 2021 An upgrade path for the fermilab accelerator complex arXiv:2106.02133

- [483] Anderson A J, Conrad J M, Figueroa-Feliciano E, Ignarra C, Karagiorgi G, Scholberg K, Shaevitz M H and Spitz J 2012 Measuring active-to-sterile neutrino oscillations with neutral current coherent neutrino-nucleus scattering *Phys. Rev.* D 86 013004
- [484] Allison J et al 2006 Geant4 developments and applications IEEE Trans. Nucl. Sci. 53 270
- [485] Axani S, Collin G, Conrad J M, Shaevitz M H, Spitz J and Wongjirad T 2015 Decisive disappearance search at high Δm^2 with monoenergetic muon neutrinos *Phys. Rev.* D 92 092010
- [486] Cost estimates for the KPIPE experiment. https://dspace.mit.edu/bitstream/handle/1721.1/ 98388/Kpipecosts.pdf
- [487] Mokhov N, Aarnio P, Eidelman Y, Gudima K, Konobeev A, Pronskikh V, Rakhno I, Striganov S and Tropin I 2014 MARS15 Code Developments Driven By The Intensity Frontier Needs Prog. Nucl. Sci. Tech. 4 496–501
- [488] Ball M et al 2017 The pip-ii conceptual design report, https://www.osti.gov/biblio/1346823
- [489] Eldred J, Lebedev V and Valishev A 2019 Rapid-cycling synchrotron for multi-megawatt proton facility at Fermilab J. Instrum. 14 P07021
- [490] Toups M et al 2022 SBN-BD: O(10 GeV) proton beam dump At Fermilab's PIP-II Linac 2022 Snowmass Summer Study 3 arXiv:2203.08102
- [491] Alonso J R et al 2022 IsoDAR@Yemilab: a report on the technology, capabilities, and deployment J. Instrum. 17 P09042
- [492] Winklehner D, Adelmann A, Conrad J M, Mayani S, Muralikrishnan S, Schoen D and Yampolskaya M 2021 Order-of-magnitude beam current improvement in compact cyclotrons *New J. Phys.* 24 023038
- [493] Winklehner D, Conrad J M, Smolsky J and Waites L 2021 High-current H2+ beams from a filament-driven multicusp ion source *Rev. Sci. Instrum.* 92 123301
- [494] Koser D, Waites L, Winklehner D, Frey M, Adelmann A and Conrad J 2022 Input beam matching and beam dynamics design optimization of the isodar rfq using statistical and machine learning techniques *Front.in Phys.* 10 875889
- [495] Bungau A, Alonso J, Bartoszek L, Conrad J, Shaevitz M and Spitz J 2019 Optimizing the ⁸Li yield for the IsoDAR neutrino experiment J. Instrum. 14 P03001
- [496] Bungau A, Alonso J, Bartoszek L, Conrad J M, Dunton E and Shaevitz M H 2020 The shielding design concept for the ISODAR neutrino target J. Instrum. 15 T07002
- [497] Alonso J et al 2022 Neutrino physics opportunities with the IsoDAR source at Yemilab Phys. Rev. D 105 052009
- [498] Serebrov A P et al 2021 Search for sterile neutrinos with the Neutrino-4 experiment and measurement results Phys. Rev. D 104 032003
- [499] Hostert M, McKeen D, Pospelov M and Raj N 2022 Dark sectors in neutron-shining-through-awall and nuclear absorption signals *Phys. Rev.* D 107 075034
- [500] Antonello M et al (MicroBooNE, LAr1-ND, ICARUS-WA104 Collaboration) 2015 A proposal for a three detector short-baseline neutrino oscillation program in the fermilab booster neutrino beam arXiv:1503.01520
- [501] Furmanski A P and Hilgenberg C 2021 Neutrino energy estimation in neutral current interactions and prospects for sterile neutrino searches *Phys. Rev.* D 103 112011
- [502] Cianci D, Furmanski A, Karagiorgi G and Ross-Lonergan M 2017 Prospects of light sterile neutrino oscillation and CP violation searches at the fermilab short baseline neutrino facility *Phys. Rev.* D 96 055001
- [503] Abratenko P et al (MicroBooNE Collaboration) 2021 Search for a Higgs portal scalar decaying to electron-positron pairs in the MicroBooNE detector Phys. Rev. Lett. 127 151803
- [504] The European Strategy Group (ESG) 2020 Update of the European Strategy for Particle Physics Available at https://cds.cern.ch/record/2721370/files/CERN-ESU-015-2020%20Update% 20European%20Strategy.pdf
- [505] ENUBET collaboration Enhanced NeUtrino BEams from kaon Tagging, https://www.pd.infn. it/eng/enubet/
- [506] ENUBET collaboration Enhanced NeUtrino BEams from kaon Tagging. https://cordis.europa. eu/project/id/681647
- [507] Torti M et al 2020 A high precision narrow-band neutrino beam: the ENUBET project Int. J. Mod. Phys. A 35 2044017
- [508] nuSTORM collaboration nuSTORM, Neutrinos from Stored Muons, https://www.nustorm. org/trac/

- [509] Ahdida C C et al 2020 nuSTORM at CERN: feasibility studyCERN-PBC-REPORT-2019-00310
- [510] Physics Beyond Colliders study group, the Physics Beyond Colliders Study Group, https://pbc. web.cern.ch
- [511] Muon Collider collaboration website, https://muoncollider.web.cern.ch
- [512] Adey D et al (nuSTORM Collaboration) 2014 Light sterile neutrino sensitivity at the nuSTORM facility Phys. Rev. D 89 071301
- [513] Kopp J 2010 Sterile neutrinos and non-standard neutrino interactions in GLoBES https://www. mpi-hd.mpg.de/personalhomes/globes/tools/snu-1.0.pdf
- [514] Abi B et al (DUNE Collaboration) 2021 Prospects for beyond the standard model physics searches at the deep underground neutrino experiment Eur. Phys. J. C 81 322
- [515] Gouvêa A D, Kelly K J, Stenico G V and Pasquini P 2019 Physics with beam tau-neutrino appearance at DUNE Phys. Rev. D 100 016004
- [516] Abe K et al (Hyper-Kamiokande Collaboration) 2018 Hyper-Kamiokande design report arXiv:1805.04163
- [517] Abe K et al (T2K Collaboration) 2017 Search for Lorentz and CPT violation using sidereal time dependence of neutrino flavor transitions over a short baseline *Phys. Rev.* D 95 111101
- [518] Mitsuka G et al (Super-Kamiokande Collaboration) 2011 Study of non-standard neutrino interactions with atmospheric neutrino data in Super-Kamiokande I and II Phys. Rev. D 84 113008
- [519] Bhadra S et al (nuPRISM Collaboration) 2014 Letter of intent to construct a nuPRISM detector in the J-PARC neutrino beamline arXiv:1412.3086
- [520] Baussan E *et al* (ESSnuSB Collaboration) 2014 A very intense neutrino super beam experiment for leptonic CP violation discovery based on the European spallation source linac *Nucl. Phys. B* 885 127–49
- [521] Wildner E et al (ESSnuSB Collaboration) 2016 The opportunity offered By the ESSnuSB project to exploit the larger leptonic CP violation signal at the second oscillation maximum and the requirements of this project on the ESS accelerator complex Adv. High Energy Phys. 2016 8640493
- [522] Huber P, Lindner M and Winter W 2005 Simulation of long-baseline neutrino oscillation experiments with GLoBES (General Long Baseline Experiment Simulator) Comput. Phys. Commun. 167 195
- [523] Huber P, Kopp J, Lindner M, Rolinec M and Winter W 2007 New features in the simulation of neutrino oscillation experiments with GLoBES 3.0: General Long Baseline Experiment Simulator Comput. Phys. Commun. 177 432–8
- [524] Agostino L et al (MEMPHYS Collaboration) 2013 Study of the performance of a large scale water-Cherenkov detector (MEMPHYS) JCAP 01 024
- [525] Coloma P, Huber P, Kopp J and Winter W 2013 Systematic uncertainties in long-baseline neutrino oscillations for large₁₃ Phys. Rev. D 87 033004
- [526] Esteban I, Gonzalez-Garcia M C, Hernandez-Cabezudo A, Maltoni M and Schwetz T 2019 Global analysis of three-flavour neutrino oscillations: synergies and tensions in the determination of θ_{23} , δ_{CP} , and the mass ordering *JHEP* **01** 106
- [527] Abe K et al (Hyper-Kamiokande Collaboration) 2018 Physics potentials with the second Hyper-Kamiokande detector in Korea Progress Theor. Exp. Phys. 2018 063C01
- [528] Acciarri R et al (DUNE Collaboration) 2015 Long-Baseline Neutrino Facility (LBNF) and Deep Underground Neutrino Experiment (DUNE): Conceptual Design Report, Volume 2: The Physics Program for DUNE at LBNF 12 arXiv:1512.06148
- [529] Gandhi R, Kayser B, Masud M and Prakash S 2015 The impact of sterile neutrinos on CP measurements at long baselines JHEP 11 039
- [530] Svirida D and (DANSS Collaboration) 2020 DANSS experiment: current status and future plans J. Phys. Conf. Ser. 1690 012179
- [531] Abusleme A et al (JUNO Collaboration) 2020 TAO Conceptual Design Report: A Precision Measurement of the Reactor Antineutrino Spectrum with Sub-percent Energy Resolution 5 arXiv:2005.08745
- [532] Ko Y J et al (NEOS Collaboration) 2017 Sterile Neutrino Search at the NEOS Experiment Phys. Rev. Lett. 118 121802
- [533] Andriamirado M et al (PROSPECT Collaboration) 2022 PROSPECT-II physics opportunities J. Phys. G 49 070501

- [534] Haghighat A, Huber P, Li S, Link J M, Mariani C, Park J and Subedi T 2020 Observation of Reactor Antineutrinos with a Rapidly-Deployable Surface-Level Detector *Phys. Rev. Applied* 13 034028
- [535] Behera S P, Mishra D K and Pant L M 2020 Active-sterile neutrino mixing constraints using reactor antineutrinos with the ISMRAN setup Phys. Rev. D 102 013002
- [536] Lane C et al (NuLat Collaboration) 2015 A new type of Neutrino Detector for Sterile Neutrino Search at Nuclear Reactors and Nuclear Nonproliferation Applications 1 arXiv:1501.06935
- [537] Abreu Y et al (SoLid Collaboration) 2021 SoLid: a short baseline reactor neutrino experiment JINST 16 P02025
- [538] Billard J et al 2017 Coherent Neutrino Scattering with Low Temperature Bolometers at Chooz Reactor Complex J. Phys. G 44 105101
- [539] Choi J J 2020 Neutrino Elastic-scattering Observation with NaI[Tl](NEON) PoS NuFact2019 047
- [540] Aguilar-Arevalo A et al (CONNIE Collaboration) 2021 Search for coherent elastic neutrinonucleus scattering at a nuclear reactor with CONNIE 2019 data 10 arXiv:2110.13033
- [541] Bonet H et al (CONUS Collaboration) 2021 Constraints on Elastic Neutrino Nucleus Scattering in the Fully Coherent Regime from the CONUS Experiment Phys. Rev. Lett. 126 041804
- [542] Agnolet G et al (MINER Collaboration) 2017 Background Studies for the MINER Coherent Neutrino Scattering Reactor Experiment Nucl. Instrum. Meth. A 853 53–60
- [543] Rothe J et al (NUCLEUS Collaboration) 2019 NUCLEUS: Exploring Coherent Neutrino-Nucleus Scattering with Cryogenic Detectors J. Low Temp. Phys. 199 433–40
- [544] Akimov D Y et al (RED-100 Collaboration) 2020 First ground-level laboratory test of the twophase xenon emission detector RED-100 JINST 15 P02020
- [545] Wong H T-K 2015 Taiwan EXperiment On NeutrinO History and Prospects The Universe 3 22–37
- [546] Gavrin V N et al 2018 On the gallium experiment BEST-2 with a ⁶⁵Zn source to search for neutrino oscillations on a short baseline 7 arXiv:1807.02977
- [547] Gavrin V N et al 2019 Neutrino-Oscillation Searches in the Short-Baseline Gallium Experiment BEST-2 with a⁶⁵Zn Source Phys. Atom. Nucl. 82 70–6
- [548] Barinov V, Cleveland B, Gavrin V, Gorbunov D and Ibragimova T 2018 Revised neutrinogallium cross section and prospects of BEST in resolving the Gallium anomaly *Phys. Rev.* D 97 073001
- [549] Smirnov M V, Hu Z, Ling J J, Novikov Y, Wang Z and Yang G 2020 Sterile neutrino oscillometry with Jinping Eur. Phys. J. C 80 609
- [550] Orebi Gann G D and (THEIA Interest Group Collaboration) 2015 Physics Potential of an Advanced Scintillation Detector: Introducing THEIA 4 arXiv:1504.08284
- [551] Askins M et al (Theia Collaboration) 2020 THEIA: an advanced optical neutrino detector Eur. Phys. J. C 80 416
- [552] Aartsen M G et al (IceCube Collaboration) 2019 Neutrino astronomy with the next generation IceCube Neutrino Observatory 11 arXiv:1911.02561
- [553] Aartsen M G *et al* (IceCube Collaboration) 2019 Efficient propagation of systematic uncertainties from calibration to analysis with the SnowStorm method in IceCube *JCAP* **10** 048
- [554] Abi B et al (DUNE Collaboration) 2020 Deep Underground Neutrino Experiment (DUNE), Far Detector Technical Design Report, Volume II: DUNE Physics 2 arXiv:2002.03005
- [555] Aartsen M G et al (IceCube Collaboration) 2014 IceCube-Gen2: A Vision for the Future of Neutrino Astronomy in Antarctica 12 arXiv:1412.5106
- [556] Kelly K J, Machado P A, Soler I M, Parke S J and Perez Gonzalez Y F 2019 Sub-GeV Atmospheric Neutrinos and CP-Violation in DUNE Phys. Rev. Lett. 123 081801
- [557] Honda M, Kajita T, Kasahara K and Midorikawa S 2011 Improvement of low energy atmospheric neutrino flux calculation using the JAM nuclear interaction model *Phys. Rev.* D 83 123001
- [558] Schneider A, Skrzypek B, Argüelles C A and Conrad J M 2021 Closing the neutrino BSM gap: Physics potential of atmospheric through-going muons at DUNE Phys. Rev. D 104 092015
- [559] Abe K et al (Hyper-Kamiokande Collaboration) 2018 Hyper-Kamiokande Design Report 5 arXiv:1805.04163
- [560] Kelly K J 2017 Searches for new physics at the Hyper-Kamiokande experiment Phys. Rev. D 95 115009

- [561] Adrian-Martinez S et al (KM3Net Collaboration) 2016 Letter of intent for KM3NeT 2.0 J. Phys. G 43 084001
- [562] Aiello S et al (KM3Net Collaboration) 2022 Determining the neutrino mass ordering and oscillation parameters with KM3NeT/ORCA Eur. Phys. J. C 82 26
- [563] Nauta L et al (KM3NeT Collaboration) 2021 First neutrino oscillation measurement in KM3NeT/ORCA PoS ICRC2021 1123
- [564] Aiello S et al (KM3NeT Collaboration) 2021 Sensitivity to light sterile neutrino mixing parameters with KM3NeT/ORCA JHEP 10 180
- [565] Manczak J et al (KM3NeT Collaboration) 2021 Neutrino non-standard interactions with the KM3NeT/ORCA detector PoS ICRC2021 1165
- [566] de Salas P F, Pastor S, Ternes C A, Thakore T and Tórtola M 2019 Constraining the invisible neutrino decay with KM3NeT-ORCA *Phys. Lett.* B 789 472–9
- [567] Muller R et al (KM3NeT Collaboration) 2021 Sensitivity estimates for diffuse, point-like and extended neutrino sources with KM3NeT/ARCA PoS ICRC2021 1077
- [568] Drewes M et al 2017 A White Paper on keV Sterile Neutrino Dark Matter JCAP 01 025
- [569] Mertens S et al 2015 Sensitivity of Next-Generation Tritium Beta-Decay Experiments for keV-Scale Sterile Neutrinos JCAP 02 020
- [570] Mertens S, Dolde K, Korzeczek M, Glueck F, Groh S, Martin R D, Poon A W P and Steidl M 2015 Wavelet approach to search for sterile neutrinos in tritium β -decay spectra *Phys. Rev.* D **91** 042005
- [571] Mertens S et al (KATRIN Collaboration) 2019 A novel detector system for KATRIN to search for keV-scale sterile neutrinos J. Phys. G 46 065203
- [572] Mertens S et al 2020 Characterization of Silicon Drift Detectors with Electrons for the TRISTAN Project J. Phys. G 48 015008
- [573] Esfahani A A et al (Project 8 Collaboration) 2017 Determining the neutrino mass with cyclotron radiation emission spectroscopy—Project 8 J. Phys. G 44 054004
- [574] Monreal B and Formaggio J A 2009 Relativistic Cyclotron Radiation Detection of Tritium Decay Electrons as a New Technique for Measuring the Neutrino Mass Phys. Rev. D 80 051301
- [575] Robertson R G H and Knapp D A 1988 Direct Measurements of Neutrino Mass Ann. Rev. Nucl. Part. Sci. 38 185–215
- [576] Asner D M et al (Project 8 Collaboration) 2015 Single electron detection and spectroscopy via relativistic cyclotron radiation Phys. Rev. Lett. 114 162501
- [577] Esfahani A A et al (Project 8 Collaboration) 2022 Tritium Beta Spectrum and Neutrino Mass Limit from Cyclotron Radiation Emission Spectroscopy 12 arXiv:2212.05048
- [578] Esfahani A A et al (Project 8 Collaboration) 2022 The Project 8 Neutrino Mass Experiment 2022 Snowmass Summer Study 3 arXiv:2203.07349
- [579] Formaggio J A, de Gouvêa A L C and Hamish Robertson R G 2021 Direct Measurements of Neutrino Mass Phys. Rept. 914 1–54
- [580] Benetti P et al (WARP Collaboration) 2007 Measurement of the specific activity of ar-39 in natural argon Nucl. Instrum. Meth. A 574 83–8
- [581] Abi B et al (DUNE Collaboration) 2020 Deep Underground Neutrino Experiment (DUNE), Far Detector Technical Design Report, Volume I Introduction to DUNE JINST 15 T08008
- [582] Abi B et al (DUNE Collaboration) 2020 Deep Underground Neutrino Experiment (DUNE), Far Detector Technical Design Report, Volume III: DUNE Far Detector Technical Coordination JINST 15 T08009
- [583] Abi B et al (DUNE Collaboration) 2020 Deep Underground Neutrino Experiment (DUNE), Far Detector Technical Design Report, Volume IV: Far Detector Single-phase Technology JINST 15 T08010
- [584] Acciarri R et al (MicroBooNE Collaboration) 2017 Design and Construction of the MicroBooNE Detector JINST 12 P02017
- [585] MicroBooNE Collaboration. http://microboone.fnal.gov/wp-content/uploads/MICROBOONE-NOTE-1050-PUB.pdf.
- [586] Abi B et al (DUNE Collaboration) 2017 The Single-Phase ProtoDUNE Technical Design Report 6 arXiv:1706.07081
- [587] Abi B et al (DUNE Collaboration) 2020 First results on ProtoDUNE-SP liquid argon time projection chamber performance from a beam test at the CERN Neutrino Platform JINST 15 P12004

- [588] Kostensalo J, Suhonen J and Zuber K 2018 Spectral shapes of forbidden argon β decays as background component for rare-event searches J. Phys. G 45 025202
- [589] Dragoun O and Vénos D 2016 Constraints on the Active and Sterile Neutrino Masses from Beta-Ray Spectra: Past, Present and Future J. Phys. 3 77-113
- [590] Bolton P D, Deppisch F F and Bhupal Dev P S 2020 Neutrinoless double beta decay versus other probes of heavy sterile neutrinos JHEP 03 170
- [591] Martoff C J et al 2021 HUNTER: precision massive-neutrino search based on a laser cooled atomic source Quantum Sci. Technol. 6 024008
- [592] Leach K G, Friedrich S and (BeEST Collaboration) 2021 19th International Workshop on Low Temperature Detectors 12 arXiv:2112.02029
- [593] Tilley D R, Cheves C M, Godwin J L, Hale G M, Hofmann H M, Kelley J H, Sheu C G and Weller H R 2002 Energy levels of light nuclei A=5, A=6, A=7 Nucl. Phys. A 708 3-163
- [594] Huang W J, Wang M, Kondev F G, Audi G and Naimi S 2021 The AME 2020 atomic mass evaluation (I). Evaluation of input data, and adjustment procedures Chin. Phys. C 45 030002
- [595] Kurakado M 1982 Possibility of high resolution detectors using superconducting tunnel junctions Nucl. Instrum. Methods Phys. Res. 196 275-7
- [596] Friedrich S, Ponce F, Hall J A and Cantor R 2020 Non-linearities in superconducting tunnel junction radiation detectors and their mca readout J. Low Temp. Phys. 200 200-5
- [597] Warburton W K, Harris J T and Friedrich S 2015 High density processing electronics for superconducting tunnel junction x-ray detector arrays Nucl. Instrum. Methods Phys. Res., Sect. A 784 236-41
- [598] Ponce F, Swanberg E, Burke J, Henderson R and Friedrich S 2018 Accurate measurement of the first excited nuclear state in U235 Phys. Rev. C 97 054310
- [599] Fretwell S et al 2020 Direct Measurement of the ⁷Be L/K Capture Ratio in Ta-Based Superconducting Tunnel Junctions Phys. Rev. Lett. 125 032701
- [600] Betti M G et al 2019 Neutrino physics with the PTOLEMY project: active neutrino properties and the light sterile case JCAP 07 047
- [601] Bolton P D, Deppisch F F, Gráf L and Šimkovic F 2021 Two-Neutrino Double Beta Decay with Sterile Neutrinos Phys. Rev. D 103 055019
- [602] Adams D Q et al 2022 Search for Majorana neutrinos exploiting millikelvin cryogenics with CUORE Nature 604 53-8
- [603] Albert J B et al 2018 Search for Neutrinoless Double-Beta Decay with the Upgraded EXO-200 Detector Phys. Rev. Lett. 120 072701
- [604] Agostini M et al 2020 Final Results of GERDA on the Search for Neutrinoless Double-β Decay Phys. Rev. Lett. 125 252502
- [605] Abe S et al 2022 First Search for the Majorana Nature of Neutrinos in the Inverted Mass Ordering Region with KamLAND-Zen 3 arXiv:2203.02139
- [606] Arnquist I J et al 2022 Final Result of the MAJORANA DEMONSTRATOR's Search for Neutrinoless Double- β Decay in ⁷⁶Ge 7 arXiv:2207.07638
- [607] Armstrong W R et al 2019 CUPID pre-CDR 7 arXiv:1907.09376
- [608] Abgrall N et al 2021 The Large Enriched Germanium Experiment for Neutrinoless $\beta\beta$ Decay: LEGEND-1000 Preconceptual Design Report 7 arXiv:2107.11462
- [609] Adhikari G et al 2022 nEXO: neutrinoless double beta decay search beyond 10^{28} year half-life sensitivity J. Phys. G 49 015104
- [610] Arnold R et al 2016 Measurement of the $2\nu\beta\beta$ decay half-life of ¹⁵⁰Nd and a search for $0\nu\beta\beta$ decay processes with the full exposure from the NEMO-3 detector Phys. Rev. D 94 072003
- [611] Arnold R et al 2010 Probing New Physics Models of Neutrinoless Double Beta Decay with SuperNEMO Eur. Phys. J. C 70 927-43
- [612] Biller S D 2013 Probing Majorana neutrinos in the regime of the normal mass hierarchy Phys. Rev. D 87 071301
- [613] Cepedello R, Deppisch F F, González L, Hati C and Hirsch M 2019 Neutrinoless Double-β Decay with Nonstandard Majoron Emission Phys. Rev. Lett. 122 181801
- [614] Gil E C et al 2019 First search for $K^+ \to \pi^+ \nu \bar{\nu}$ using the decay-in-flight technique Phys. Lett. B **791** 156–66
- [615] Gil E C *et al* 2020 An investigation of the very rare $K^+ \to \pi^+ \nu \bar{\nu}$ decay *JHEP* **11** 042 [616] Ahn J K *et al* 2019 Search for the $K_L \to \pi^0 \nu \bar{\nu}$ and $K_L \to \pi^0 X^0$ decays at the J-PARC KOTO experiment Phys. Rev. Lett. 122 021802
- [617] Gil E C et al 2017 The Beam and detector of the NA62 experiment at CERN JINST 12 P05025

- [618] Gil E C *et al* 2018 Search for heavy neutral lepton production in K^+ decays *Phys. Lett.* B 778 137–45
- [619] Lazzeroni C *et al* 2017 Search for heavy neutrinos in $K^+ \rightarrow \mu^+ \nu_\mu$ decays *Phys. Lett.* B 772 712–8
- [620] Gil E C *et al* 2020 Search for heavy neutral lepton production in K^+ decays to positrons *Phys.* Lett. B **807** 135599
- [621] Gil E C et al 2019 Search for production of an invisible dark photon in π^0 decays JHEP 05 182
- [622] Gil E C et al 2021 Search for π^0 decays to invisible particles JHEP 02 201
- [623] Dib C, Helo J C, Kovalenko S and Schmidt I 2011 Sterile neutrino decay explanation of LSND and MiniBooNE anomalies *Phys. Rev.* D 84 071301
- [624] Poblaguev A A et al 2002 Experimental study of the radiative decays K+ -> mu+ nu e+ e- and K+ -> e+ nu e+ e- Phys. Rev. Lett. 89 061803
- [625] Gil E C et al 2019 Searches for lepton number violating K^+ decays Phys. Lett. B 797 134794
- [626] Abreu H et al 2020 Detecting and Studying High-Energy Collider Neutrinos with FASER at the LHC Eur. Phys. J. C 80 61
- [627] Ahdida C et al 2020 SND@LHC 2 arXiv:2002.08722
- [628] Kling F and Nevay L J 2021 Forward neutrino fluxes at the LHC *Phys. Rev.* D 104 113008
- [629] Bai W, Diwan M, Garzelli M V, Jeong Y S and Reno M H 2020 Far-forward neutrinos at the Large Hadron Collider JHEP 06 032
- [630] Anchordoqui L A et al 2022 The Forward Physics Facility: Sites, experiments, and physics potential Phys. Rept. 968 1–50
- [631] Foreman-Mackey D, Hogg D W, Lang D and Goodman J 2013 emcee: The MCMC Hammer Publ. Astron. Soc. Pac. 125 306–12
- [632] Freund M 2001 Analytic approximations for three neutrino oscillation parameters and probabilities in matter Phys. Rev. D 64 053003
- [633] Akhmedov E K, Johansson R, Lindner M, Ohlsson T and Schwetz T 2004 Series expansions for three flavor neutrino oscillation probabilities in matter JHEP 04 078
- [634] Minakata H and Parke S J 2016 Simple and Compact Expressions for Neutrino Oscillation Probabilities in Matter JHEP01(2016)180
- [635] Denton P B, Minakata H and Parke S J 2016 Compact Perturbative Expressions For Neutrino Oscillations in Matter JHEP06(2016)051
- [636] Barenboim G, Denton P B, Parke S J and Ternes C A 2019 Neutrino Oscillation Probabilities through the Looking Glass *Phys. Lett.* B 791 351–60
- [637] Denton P B, Parke S J and Zhang X 2020 Fibonacci Fast Convergence for Neutrino Oscillations in Matter Phys. Lett. B 807 135592
- [638] Denton P B and Parke S J 2019 Simple and Precise Factorization of the Jarlskog Invariant for Neutrino Oscillations in Matter *Phys. Rev.* D 100 053004
- [639] Calland R G, Kaboth A C and Payne D 2014 Accelerated Event-by-Event Neutrino Oscillation Reweighting with Matter Effects on a GPU JINST 9 P04016
- [640] Wallraff M and Wiebusch C 2014 Calculation of oscillation probabilities of atmospheric neutrinos using nucraft 09 http://arxiv.org/abs/1409.1387
- [641] Argüelles C A and Jones B J P 2019 Neutrino Oscillations in a Quantum Processor Phys. Rev. Research. 1 033176
- [642] Argüelles C A, Salvado J and Weaver C N 2021 nuSQuIDS: A toolbox for neutrino propagation 12 arXiv:2112.13804
- [643] Jones B J P 2015 Dynamical pion collapse and the coherence of conventional neutrino beams Phys. Rev. D 91 053002
- [644] Sousa A B and (MINOS, MINOS+ Collaboration) 2015 First MINOS+ Data and New Results from MINOS AIP Conf. Proc. 1666 110004
- [645] Adamson P et al (MINOS Collaboration) 2016 Search for Sterile Neutrinos Mixing with Muon Neutrinos in MINOS Phys. Rev. Lett. 117 151803
- [646] Gariazzo S, Giunti C, Laveder M, Li Y F and Zavanin E M 2016 Light sterile neutrinos J. Phys. G 43 033001
- [647] Sousa A B et al (NOvA Collaboration) 2011 Status of NOvA 14th International Workshop on Neutrino Telescopes: Un altro modo di guardare il cielo 3 193–202
- [648] Hartnell J and (NOvA Collaboration) 2022 Accelerator neutrino i_nova https://doi.org/10. 5281/zenodo.6683827

- [649] Astier P et al (NOMAD Collaboration) 2001 Final NOMAD results on muon-neutrino —> tauneutrino and electron-neutrino —> tau-neutrino oscillations including a new search for tauneutrino appearance using hadronic tau decays Nucl. Phys. B 611 3–39
- [650] Eskut E et al (CHORUS Collaboration) 2008 Final results on nu(mu) —> nu(tau) oscillation from the CHORUS experiment Nucl. Phys. B 793 326–43
- [651] Danilov M 2022 Review of sterile neutrino searches at very short-baseline reactor experiments Phys. Scripta 97 094001
- [652] Declais Y et al 1995 Search for neutrino oscillations at 15-meters, 40-meters, and 95-meters from a nuclear power reactor at Bugey Nucl. Phys. B 434 503–34
- [653] Choi J H et al (RENO Collaboration) 2020 Search for Sub-eV Sterile Neutrinos at RENO Phys. Rev. Lett. 125 191801
- [654] Gariazzo S, Giunti C, Laveder M and Li Y F 2018 Model-independent $\bar{\nu}_e$ short-baseline oscillations from reactor spectral ratios *Phys. Lett.* B **782** 13–21
- [655] Danilov M and (DANSS Collaboration) 2020 Recent results of the DANSS experiment PoS EPS-HEP2019 401
- [656] Almazán H et al (STEREO Collaboration) 2020 Improved sterile neutrino constraints from the STEREO experiment with 179 days of reactor-on data Phys. Rev. D 102 052002
- [657] Kraus C, Singer A, Valerius K and Weinheimer C 2013 Limit on sterile neutrino contribution from the Mainz Neutrino Mass Experiment *Eur. Phys. J.* C 73 2323
- [658] Belesev A I, Berlev A I, Geraskin E V, Golubev A A, Likhovid N A, Nozik A A, Pantuev V S, Parfenov V I and Skasyrskaya A K 2013 An upper limit on additional neutrino mass eigenstate in 2 to 100 eV region from 'Troitsk nu-mass' data JETP Lett. 97 67–9
- [659] Giunti C, Laveder M, Li Y F and Long H W 2013 Short-baseline electron neutrino oscillation length after troitsk *Phys. Rev.* D 87 013004
- [660] Kelly K J and Machado P A N 2021 MicroBooNE experiment, NuMI absorber, and heavy neutral leptons Phys. Rev. D 104 055015
- [661] Haag N, von Feilitzsch F, Oberauer L, Potzel W, Schreckenbach K and Sonzogni A A 2014 Republication of the data from the BILL magnetic spectrometer: The cumulative β spectra of the fission products of ²³⁵U, ²³⁹Pu, and ²⁴¹Pu 5 arXiv:1405.3501
- [662] Kopeikin V I, Panin Y and Sabelnikov A A 2021 Measurement of the Ratio of Cumulative Spectra of Beta Particles from ²³⁵U and ²³⁹Pu Fission Products for Solving Problems of Reactor-Antineutrino Physics *Phys. Atom. Nucl.* 84 1–10
- [663] Maruyama T and (JSNS2, JSNS2 -II, Collaboration) 2022 The status of JSNS² and JSNS²-II PoS NuFact2021 159
- [664] Asaadi J et al 2022 Physics Opportunities in the ORNL Spallation Neutron Source Second Target Station Era 2022 Snowmass Summer Study 9 arXiv:2209.02883
- [665] Abi B et al (DUNE Collaboration) 2020 Long-baseline neutrino oscillation physics potential of the DUNE experiment Eur. Phys. J. C 80 978
- [666] Todd J R 2018 Search for Sterile Neutrinos with MINOS and MINOS+ *PhD thesis* (Cincinnati U)
- [667] Ghosh M and Ohlsson T 2020 A comparative study between ESSnuSB and T2HK in determining the leptonic CP phase Mod. Phys. Lett. A 35 2050058
- [668] Berryman J M, Delgadillo L A and Huber P 2022 Future searches for light sterile neutrinos at nuclear reactors Phys. Rev. D 105 035002
- [669] Jinyu K 2022 Neos-ii new results https://doi.org/10.5281/zenodo.6680618
- [670] Serebrov A P et al (NEUTRINO-4 Collaboration) 2019 First Observation of the Oscillation Effect in the Neutrino-4 Experiment on the Search for the Sterile Neutrino Pisma Zh. Eksp. Teor. Fiz. 109 209–18
- [671] Athanassopoulos C et al (LSND Collaboration) 1995 Candidate events in a search for anti-muonneutrino —> anti-electron-neutrino oscillations Phys. Rev. Lett. 75 2650–3
- [672] Aartsen M G et al (IceCube Collaboration) 2020 Time-Integrated Neutrino Source Searches with 10 Years of IceCube Data Phys. Rev. Lett. 124 051103
- [673] Stettner J and (IceCube Collaboration) 2020 Measurement of the Diffuse Astrophysical Muon-Neutrino Spectrum with Ten Years of IceCube Data PoS ICRC2019(2020)1017
- [674] Semenov S V 2020 Cross Section of Neutrino Absorption by the Gallium-71 Nucleus Phys. Atom. Nucl. 83 1549–52
- [675] Palomares-Ruiz S, Pascoli S and Schwetz T 2005 Explaining LSND by a decaying sterile neutrino JHEP09(2005)048

- [676] Akhmedov E K and Schwetz T 2011 New MiniBooNE results and non-standard neutrino interactions Nucl. Phys. B Proc. Suppl. 217 217–9
- [677] Bramante J 2013 Sterile Neutrino Production Through a Matter Effect Enhancement at Long Baselines Int. J. Mod. Phys. A 28 1350067
- [678] Smirnov A Y and Valera V B 2021 Resonance refraction and neutrino oscillations JHEP 09 177
- [679] Jones B J P and Spitz J 2019 Neutrino Flavor Transformations from New Short-Range Forces 11 arXiv:1911.06342
- [680] Berryman J M, Gouvêa A D, Kelly K J and Zhang Y 2018 Lepton-Number-Charged Scalars and Neutrino Beamstrahlung Phys. Rev. D 97 075030
- [681] Gninenko S N 2012 New limits on radiative sterile neutrino decays from a search for single photons in neutrino interactions *Phys. Lett.* B710 86–90
- [682] Alekseev I et al (DANSS Collaboration) 2018 Search for sterile neutrinos at the DANSS experiment Phys. Lett. B 787 56-63
- [683] Bennett J J, Buldgen G, Drewes M and Wong Y Y Y 2020 Towards a precision calculation of the effective number of neutrinos N_{eff} in the Standard Model I: the QED equation of stateJCAP03 (2020)003